

NOVEL COMBINATION TREATMENT WITH TRAIL FOR OSTEOSARCOMA THERAPY: THE USE OF INHIBITORS OF THE MEVALONATE PATHWAY.

AGATHA LABRINIDIS B.Sc. (Hons)

Submitted in total fulfilment of the requirements for the degree of Doctor of Philosophy

May, 2005

Department of Orthopaedics and Trauma Faculty of Health Science The University of Adelaide Australia

ABSTRACT

Osteosarcoma (OS) is the most common primary malignant tumour of the bone, occurring most frequently in children and young adults. In addition to surgical intervention, the current choice of treatment for OS is chemotherapy however despite significant improvements in patient survival, some patients continue to have a poor prognosis due to problems including tumour resistance to chemotherapy and metastatic spread. Therefore, there is a pressing need to develop better and safe alternative approaches for the treatment of OS. Investigating combinations of agents that alone have little toxicity, but which have additive anti-tumour activity is appealing because of the possibility of using lower drug doses with reduced toxic side effects.

This thesis investigates potential new agents for the treatment of OS. These include TRAIL and various inhibitors of the mevalonate pathway (IMP), including the nitrogen-containing bisphosphonate, zoledronic acid (ZOL), the statins, lovastatin (LOV) and mevastatin (MEV) and the prenyl transferase inhibitors (PTIs), geranylgeranyl transferase inhibitor (GGTI) and farnesyl transferase inhibitor (FTI).

By exploring new combinations of agents that can kill bone tumour cells *in vitro*, the surprising finding was made that ZOL and TRAIL act synergistically to kill bone cancer cells. More importantly, ZOL strongly sensitised TRAIL-resistant cells to the cytotoxic effects of the ligand to an extent comparable with chemotherapeutic agents. The statins and PTIs also resulted in a significantly greater induction of cell death, when used in combination with TRAIL, than either agent alone. Further investigations revealed the importance of prenylation for cell survival. Analysis of cell surface receptors revealed an upregulation of TRAIL death receptors, but not decoy receptors,

i

in response to ZOL treatment, suggesting one possible mechanism for the observed synergy.

An *in vivo* model of OS was used to examine the anti-tumour and anti-metastatic effects of ZOL and TRAIL. Mice inoculated with K-HOS cells, and left untreated, developed large lesions that invaded the marrow cavity and began to erode the cortical bone, as assessed by radiography, micro-computer tomography and histology. In contrast, animals treated with ZOL, showed remarkable conservation of the tibiae with no evidence of bone destruction and significant increases in cortical thickness and trabecular density. This was reflected in the bone volume, which was significantly increased in ZOL treated animals. As both the ZOL treated and combination ZOL and TRAIL treated animals showed no osteolysis, it was not possible to establish whether the combination treatment was any more beneficial than ZOL treatment alone. The area of tumour growth was lowest in TRAIL treated mice suggesting that TRAIL may have an effect on the extra-osseous tumour. ZOL alone or ZOL and TRAIL in combination failed to inhibit metastases to lungs. Although no definitive relationship between ZOL and TRAIL could be determined *in vivo*, the use of ZOL alone exhibited powerful anticancer properties.

The use of statins and PTIs may prove very beneficial as alternative agents to ZOL, as they do not have a high affinity for bone, thus their widespread systemic distribution may be used for the treatment of other non-osseous malignancies including lung metastases, often associated with increased mortality in OS. Most importantly, the lack of effect on normal bone cells by the IMP make them very desirable anti-cancer agents however further investigation is required to fully understand the mechanism of augmentation with TRAIL. By understanding the pathways involved and exploiting

ii

their use, there is exciting potential for a targeted therapeutic approach in the management of OS clinically.

TABLE OF CONTENTS

Abstract	ii
Declaration	xiv
Acknowledgments	XV
Publications arising during PhD candidature	xvi
Conference presentations	xvi
Awards received	xvii
List of abbreviations	xviii
List of figures	xxiii
List of tables	xxvi

CHAPTER 1: Introduction

1.1	Osteosarcoma (OS)	2
1.1.1	Definition, classification, epidemiology and localisation	2
1.1.2	Diagnosis of osteosarcoma	3
1.1.3	Treatment of osteosarcoma	5
1.2	The apoptotic pathway	7
1.2.1	Introduction	7
1.2.2	The history of apoptosis	8
1.2.3	Morphological and biochemical changes of apoptosis	9
1.2.4	Anoikis	11
1.3	The caspase cascade	12
1.3.1	Caspase structure and function	12
1.3.2	Caspase activation	13
1.3.3	Inhibition of caspase activation	14
1.3.4	Importance of caspase activity	16

1.4	The intrinsic pathway of apoptosis		17
1.4.1	The Bcl-2 family		17
1.4.2	The role of the Bcl-2 family in cancer		19
1.4.3	The mitochondrial intrinsic pathway		20
1.4.4	Regulation of the mitochondrial intrinsic pathway		22
1.4.5	Caspase-independent cell death		24
1.5	The extrinsic pathway of apoptosis		26
1.5.1	Introduction		26
1.5.2	The TNF ligand family		27
1.5.3	The TNF receptor family		27
1.5.4	The death receptor/extrinsic pathway of apoptosis		28
1.6	TRAIL and its receptors		29
1.6.1	TRAIL		29
1.6.2	Molecular mechanism of TRAIL-induced apoptosis		34
1.6.3	TRAIL sensitivity and resistance		35
1.6.4	Physiological role of the TRAIL system		37
1.6.5	Potential for cancer therapy		39
1.6.6	Combined treatment with TRAIL and other agents		41
1.7	Bisphosphonates		44
1.7.1	General properties		44
1.7.2	The effect of bisphosphonates on bone cells		46
1.7.3	Molecular mechanism of nitrogen containing bisphosphonates		48
1.7.4	Anti-tumour effects of bisphosphonates		50
1.7.5	Clinical use of bisphosphonates as cancer therapy		53
1.7.6	Combined treatment with bisphosphonates and other agent	ů.	55
1.8	Other inhibitors of the mevalonate pathway		57
1.8.1	Statins		57

v

1.8.2	Prenyl transferase inhibitors (PTIs)	61
1.9	Aims and significance of project	66
СНА	PTER 2: Materials and methods	
2.1	Cell lines and cell culture	70
2.1.1	Osteosarcoma cell lines	70
2.1.2	Normal human bone donor cells	70
2.1.3	Cell culture	71
2.2	Chemicals and reagents used	71
2.2.1	General chemicals and reagents	71
2.2.2	Novel anticancer agents	74
2.2.3	Enzymes and inhibitors of caspase activity	75
2.2.4	Antibodies	75
2.2.5	Solutions, buffers and kits	76
2.2.6	Antibiotics	77
2.3	Specialised equipment used	78
2.3.1	Flow cytometer	78
2.3.2	Phospho-imager	78
2.3.3	Plate reader	78
2.3.4	Radiography	78
2.3.5	Micro-computer tomography analysis	79
2.4	Methods	79
2.4.1	Measuring cell viability using crystal violet	79
2.4.2	Measuring cell viability using WST-1 assay	79
2.4.3	Measuring caspase activity	80

V1

		0.1
2.4.4	DAPI staining	81
2.4.5	Protein estimation	81
2.4.6	DNA fragmentation	82
2.4.7	Anoikis assay	82
2.4.8	Western blot analysis	82
2.4.9	Flow cytometric analysis of receptor expression	83
2.4.10	Flow cytometric analysis of the cell cycle	84
2.4.11	Immunofluorescence assay	85
2.4.12	Statistical analysis	85
2 5	Le vive enclusis of onti tumour activity	86
2.5	In vivo analysis of anti-tumour activity	00
2.5 2.5.1	Cell culture	86
2.52.5.12.5.2	Cell culture Cell lines	86 86
2.52.5.12.5.22.5.3	Cell culture Cell lines Animals	86 86 86
 2.5 2.5.1 2.5.2 2.5.3 2.5.4 	Cell culture Cell lines Animals Tibial implantation	86 86 86 86
 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.5 	Cell culture Cell lines Animals Tibial implantation Treatment	86 86 86 86 86 87
 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.5 2.5.6 	Cell culture Cell lines Animals Tibial implantation Treatment Histology	86 86 86 86 87 88
 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.5 2.5.6 2.5.7 	Cell culture Cell lines Animals Tibial implantation Treatment Histology Radiography	86 86 86 86 87 88 88
2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.5 2.5.6 2.5.7 2.5.8	Cell culture Cell lines Animals Tibial implantation Treatment Histology Radiography Micro-computer tomography analysis	86 86 86 86 87 88 88 88

CHAPTER 3: Zoledronic acid induces apoptosis in human osteosarcoma cells

3.1	Introduction	91
3.2	Materials and methods	93
3.2.1	Cells and cell culture	93
3.2.2	Measuring cell viability	93
3.2.3	Measuring caspase activity	93
3.2.4	DAPI staining of nuclei	93
3.2.5	Cell surface receptor analysis after ZOL treatment	93

vii

3.2.6	Cell cycle analysis after ZOL treatment	93
3.2.7	Statistical analysis	94

3.3	Results	95
3.3.1	ZOL induces dose-dependent apoptosis	95
3.3.2	ZOL induces time-dependent apoptosis	96
3.3.3	ZOL treatment leads to increased caspase activity	96
3.3.4	Caspase inhibitors do not protect against ZOL-induced apoptosis	97
3.3.5	Geranylgeraniol protects against ZOL-induced apoptosis	98
3.3.6	ZOL treatment leads to anoikis	99
3.3.7	ZOL induces S-phase arrest of the cell cycle	100

3.4 Discussion

102

CHAPTER 4: Zoledronic acid sensitises human osteosarcoma cell lines to TRAIL-induced apoptosis: are dual treatments more effective?

4.1	Introduction	107
4.2	Materials and methods	109
4.2.1	Cells and cell culture	109
4.2.2	Measuring cell viability of ZOL and TRAIL treated cells	109
4.2.3	Measuring caspase activity	109
4.2.4	Cell attachment dependent survival using polyHema	109
4.2.5	Effect of geranylgeraniol on ZOL and TRAIL co-treatment in OS	
	cells	109
4.2.6	Western blot analysis	110
4.2.7	Cell surface receptor analysis after treatment	110
4.2.8	Cell cycle analysis after treatment	110
4.2.9	Statistical analysis	110

4.3	Results	111
4.3.1	ZOL augments TRAIL-induced apoptosis in human OS cells	111
4.3.2	The augmented apoptotic effect of ZOL and TRAIL is time-	
	dependent	112
4.3.3	ZOL-induced apoptosis in OS cells is associated with caspase	
	activation	113
4.3.4	Geranylgeraniol prevents sensitisation to TRAIL by ZOL	113
4.3.5	Loss of cell anchorage sensitises cells to TRAIL-induced apoptosis	114
4.3.6	Cell cycle analysis of ZOL treated OS cells leads to S-phase arrest	
	whereas combination with TRAIL leads to apoptosis	115
4.3.7	Normal human bone donor cells are not sensitised to TRAIL	
	induced apoptosis by ZOL	116
4.3.8	ZOL upregulates TRAIL cell surface death receptors	116

4.4 Discussion

117

CHAPTER 5: Other inhibitors of the mevalonate pathway: statins and prenyl transferase inhibitors induce apoptosis in osteosarcoma cells- preliminary studies. are there effective alternatives to zoledronic acid?

5.1	Introduction	121
5.2	Materials and methods	123
5.2.1	Cells and cell culture	123
5.2.2	Measuring cell viability	123
5.2.3	Measuring caspase activity	123
5.2.4	Western blot analysis	123
5.2.5	Statistical analysis	123

ix

5.3	Results	124
5.3.1	Statins induce dose-dependent apoptosis	124
5.3.2	Statins augment TRAIL-induced apoptosis in a dose-dependent	
	manner	125
5.3.3	Prenyl transferase inhibitors induces dose-dependent apoptosis	126
5.3.4	Prenyl transferase inhibitors augments TRAIL-induced apoptosis in	
	a dose-dependent manner	126
5.3.5	Normal human bone donor cells are not sensitised to TRAIL	
	induced apoptosis by statins and prenyl transferase inhibitors	127
5.3.6	Caspase inhibitors do not protect against apoptosis induced by	
	inhibitors of the mevalonate pathway	127
5.3.7	Geranylgeraniol is able to rescue cells from the effects of statins but	
	not prenyl transferase inhibitors	128
	* -	

5.4 Discussion

CHAPTER 6: Establishment of *in vivo* osteosarcoma mouse model

6.1	Introduction	135
6.2	Materials and methods	136
6.2.1	Cells and cell culture	136
6.2.2	Mouse specifications and maintenance	136
6.2.3	Intra-tibial implantation: delivery and schedule	
6.2.4	Histology and image analysis	136
		105
6.3	Results	137
6.3.1	Lytic lesions identified using radiograph analysis	137
6.3.2	Improved imaging of lytic lesions using micro-CT analysis	138

6.3.3 Histological analysis

х

139

CHAPTER 7: *In vivo* investigation into the effect of zoledronic acid and TRAIL treatment in nude mice: will it work in a living model?

7.1	Introduction	143
7.2	Materials and methods	144
7.2.1	Cells and cell culture	144
7.2.2	Mouse specifications and maintenance	144
7.2.3	In vivo treatment: delivery and schedule	144
7.2.4	Histology and image analysis	144
7.2.5	Statistical analysis	144
7.3	Results	146
7.3.1	ZOL protects the skeleton from tumour-induced osteolysis	146
7.3.2	ZOL increases bone volume	149
7.3.3	ZOL has no effect on tumour burden in tibia but TRAIL reduces	
	tumour burden	150
7.3.4	ZOL and TRAIL have no effect on lung metastases	151

7.4 Discussion

153

CHAPTER 8: General discussion

8.1 General discussion

BIBLIOGRAPHY

167

DECLARATION

This thesis contains no material which has been accepted for the award of any other degree or diploma in a University or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text.

I give consent for this copy of my thesis, when deposited in the University Library, to be available for loan and photocopying.

Agatha Labrinidis

ACKNOWLEDGMENTS

My biggest thankyou is to my mentor, Dr Andreas Evdokiou whom I greatly admire and respect and whose enquiring mind and passion for research makes me want to be a better scientist. Thankyou for giving me the opportunity to become part of your lab. I have learnt so much and really appreciate your time and effort over the years.

Professor David Findlay, your endless knowledge has been invaluable, your encouragement has been inspirational and your excitement regarding results contagious. Your support throughout the years has been greatly appreciated. God bless you.

A special thanks to Shelley Hay, for her expertise in the lab and for putting up with my never-ending questions without losing the smile on her face...you are a gem! Thankyou to Gerald Atkins, Katie Welldon, Steve Bouralexis, Vasilios Liapis, Thai Le Minh and Tina Vincent for making the lab a great place to work in!! To all the wonderful friends I have made at the IMVS, especially Peter, Amanda, Angela, Fiona, Penny, Rosa, Sally, Andrea, Chris, Debbie and Anne-Marie. Thanks for the morning teas, coffees, lunches and good times that I will never forget.....friends for life!!

Thankyou to Dr Peter Choong, for providing the KRIB cells and for his expertise in intra-tibial injections; Dr Brendan Coventry for his collaboration with our lab regarding the mouse work. Thankyou to Dr Peter Self for his wonderful explanations and help during micro-CT analysis and to Dr Rob Moore for use of the Faxitron for radiograph analysis. A special thankyou to Justin Lokhorst from Health Sector consulting, Faculty of Health Sciences, University of Adelaide for his invaluable statistical support. I extend my appreciation to the Anti-Cancer foundation of SA and the Australian Research Council for supporting this work.

My biggest thankyou is to my wonderful family, for their support and encouragement over the years!! Mum and dad I love you more than life itself...you give me energy!! Stav, Nick and Paul, you guys are the best and I'm proud to be your sister!!! Thanks for putting up with me for so long! Finally, a huge thankyou to all my friends (old and new) and relatives (especially my precious thitsa), who believed in me and supported me through this time. You are all in my heart!

xiv

PUBLICATIONS ARISING DURING CANDIDATURE

- Evdokiou A, Labrinidis A, Bouralexis S, Hay S, Findlay DM. (2003). Induction of cell death of human osteogenic sarcoma cells by zoledronic acid resembles anoikis. *Bone* **33**: 216-228
- Bouralexis S, Clayer M, Atkins GJ, Labrinidis A, Hay S, Graves S, Findlay DM and Evdokiou A. (2004). Sensitivity of fresh isolates of soft tissue sarcoma, osteosarcoma and Giant Cell Tumour cells to TRAIL and doxorubicin. *Int J Oncol*, **24**: 1263-70
- Bouralexis S, Findlay DM, Atkins GJ, Labrinidis A, Hay S and Evdokiou A. (2003). Progressive resistance of BTK-143 osteosarcoma cells to TRAIL-induced apoptosis is mediated by acquisition of DcR2: Re-sensitisation with chemotherapy. *British J Cancer* **89**:206-14
- Atkins GJ, Bouralexis S, Evdokiou A, Hay S, Labrinidis A and Findlay DM. (2002) Human osteoblasts are resistant to TRAIL-mediated apoptosis. *Bone* **31**:(4): 448-56

CONFERENCE PRESENTATIONS:

- Labrinidis A., Nguyen O., Bouralexis S., Hay S., Findlay DM. and Evdokiou A. 2003 Inhibitors of the mevalonate pathway sensitise osteosarcoma cells to TRAIL induced cell death. International Conference on Metabolic Bone Disease, Coolum, Old. Oral presentation.
- Labrinidis A., Nguyen O., Bouralexis S., Hay S., Findlay DM. and Evdokiou A. 2003 Inhibitors of the mevalonate pathway sensitise osteosarcoma cells to TRAIL induced cell death. Australian Society for Medical Research (ASMR) SA scientific meeting, Adelaide, SA. Oral Presentation.
- Labrinidis A. Bouralexis S., Hay S., Findlay DM. and Evdokiou A. 2002 The nitrogencontaining bisphosphonate, zoledronic acid sensitises osteogenic sarcoma cells to

TRAIL-induced cell death. Australian Society for Medical Research (ASMR) SA scientific meeting, Adelaide, SA. Oral Presentation.

- Evdokiou A., Labrinidis A., Bouralexis S., Hay S., Atkins GJ. and Findlay DM. 2002 Zoledronic acid enhances TRAIL cytotoxicity of human osteosarcoma cells: A novel non-toxic approach to bone cancer therapy. The 9th International Congress on TNF-related cytokines, San Diego, California, USA. Oral presentation.
- Bouralexis S., Findlay D.M., Atkins G.J., Labrinidis A., Hay S, and Evdokiou A. 2002 Chemotherapy Sensitises Resistant BTK-143 Osteogenic Sarcoma cells to TRAIL-Induced Apoptosis. Australian Society for Medical Research (ASMR) SA scientific meeting, Adelaide, SA. Oral Presentation.
- Labrinidis A, Bouralexis S., Hay S., and Evdokiou A. 2002 The nitrogen-containing bisphosphonate, zoledronate sensitises bone cancer cells to TRAIL-induced cell death. 14th Lorne Cancer Conference, Lorne, Vic. Poster Presentation.
- Bouralexis S., Findlay D.M., Atkins G.J., Labrinidis A., Hay S., and Evdokiou A. 2002 Progressive resistance of BTK-143 Osteogenic Sarcoma cells to TRAIL-Induced Apoptosis is Mediated by Acquisition DcR2. 14th Lorne Cancer Conference, Lorne, Vic. Poster Presentation.
- Labrinidis A, Bouralexis S., Hay S., and Evdokiou A. 2002 The nitrogen-containing bisphosphonate, zoledronate sensitises bone cancer cells to TRAIL-induced cell death. Australian and New Zealand Bone and Mineral Society 12th Annual Scientific Meeting, Adelaide, SA.. Poster Presentation.

AWARDS RECEIVED:

1. Roger Melick Young Investigator award, received at the International Conference on Metabolic Bone disease, Qld, Australia in June 2003.

2. BioRad New Investigator Award for best oral presentation received at the Australian Society for Medical Research Adelaide Meeting in May 2003.

3. Research abroad scholarship awarded by the University of Adelaide to attend the 4th International Conference on Cancer-Induced Bone Diseases in San Antonio, Texas, USA, December 2003.

4. Travel grant awarded to attend the International Conference on Metabolic Bone disease, Qld, Australia, June 2003.

5. Travel grant awarded to attend the 13th Lorne Cancer Conference, Vic, Australia, February 2002.

LIST OF ABBREVIATIONS

αΜΕΜ	α -Minimum Essential Media
ANCOVA	Analysis of Covariance
ANOVA	Analysis of Variance
Apo2L	apoptosis-2 ligand
aa	amino acid
AIF	apoptosis inducing factor
Apaf-1	apoptosis protease-activating factor-1
ATCC	American Type Culture Collection
BH	Bcl-2 homology domain
BIR	Baculovirus IAP repeat
BMD	bone mineral density
BMP	bone morphogenetic protein
bp	base pairs
BPs	bisphosphonates
BSA	bovine serum albumin
CAD	caspase-activated deoxyribonuclease
CARD	caspase-activating recruitment domain
CDK	cyclin dependent kinase
CRD	cysteine-rich domains
cDNA	complementary deoxyribonucleic acid
c-FLIP	cellular FLICE inhibitory protein
CMV	cytomegalovirus
CPM	cyclophosphamide
Crm A	Cytokine response modifier A

DAPI	4',6-diamidine-2'-phenylindole dihydrochloride
DcR1	decoy receptor 1
DcR2	decoy receptor 2
DD	death domain
DED	death effector domain
DNA	deoxyribonucleic acid
DIABLO	direct IAP binding protein with low pI
DISC	death-inducing signalling complex
DMEM	Dulbecco's modified Eagles medium
DMSO	dimethyl sulphoxide
DR3	death receptor 3
DR4	death receptor 4
DR5	death receptor 5
DTT	Dithiothreitol
EDTA	Ethylenediaminetetraacetic acid
ER	endoplasmic reticulum
Erk	extracellular signal-regulating kinase
FADD	Fas-associated death domain
FAK	focal adhesion kinase
FCS	foetal calf serum
FITC	fluorescein
FLICE	FADD-like ICE
FLIP	FLICE inhibitory protein
FTI	farnesyl transferase inhibitor

GAPDH	glyceraldehyde-3-phosphate dehydrogenase
GGTI	geranylgeranyl transferase inhibitor
GPI	glycosyl-phosphatidylinositol
HMG-CoA	3-hydroxy-3-methylglutaryl coenzyme A
Hsp	heat shock protein
Htr2A	High temperature requirement-A2
IAP	inhibitor of apoptosis protein
IBM	IAP-binding motif
ICAD	inhibitor of caspase-activated deoxyribonuclease
ICE	interleukin-1-beta-converting enzyme
IFN	interferon
Ig	immunoglobulin
ІкВ	inhibitor of ĸB
IL	interleukin
IMP	inhibitors of the mevalonate pathway
IMVS	Institute of Medical and Veterinary Science
iv	intravenous
JNK	c-Jun N-terminal kinase
kD	kilodalton
LDL	low density lipoprotein
LOV	lovostatin
LT-a	lymphotoxin-α
Mab	monoclonal antibody
МАРК	mitogen activated protein kinase
MEV	mevastatin

XIX

MLS	mitochondrial localisation sequence
MMP	mitochondrial membrane permeabilization
MMTV	mouse mammary tumour virus
MMR	mismatch-repair
mRNA	messenger ribonucleic acid
N-BPs	nitrogen containing bisphosphonates
NF-ĸB	nuclear factor kappa beta
NH ₂	amino group
NK	natural killer
NSAIDs	non-steroidal anti-inflammatory drugs
OH	hydroxyl group
OPG	osteoprotegerin
OS	osteogenic sarcoma (osteosarcoma)
PARP	poly(ADP-ribose) polymerase
PBS	phosphate buffered saline
PCD	programmed cell death
PE	phycoerythrin
PI	propidium iodide
PI-3-kinase	phosphatidylinositol 3-kinase
РКС	protein kinase C
PMSF	phenylmethylsulphanyl fluoride
PTI	prenyl transferase inhibitors
RANKL	receptor activator NF-kB ligand
Rb	retinoblastoma protein
RIP	receptor-interaction protein

XХ

rpm	revolutions per minute
SC	subcutaneous
SD	standard deviation
SDS	sodium dodecyl sulphate
Smac	second mitochondria-derived activator of caspase
SRE	skeletal-related event
tBID	truncated Bid
TBS	Tris buffered saline
THD	TNF homology domain
TNF	tumour necrosis factor
TNFR	tumour necrosis factor receptor
TRAIL	TNF related apoptosis inducing ligand
TRAIL-R1	TRAIL receptor 1 (also known as DR4)
TRAIL-R2	TRAIL receptor 2 (also known as DR5)
TRAIL-R3	TRAIL receptor 3 (also known as DcR1)
TRAIL-R4	TRAIL receptor 4 (also known as DcR2)
TRADD	TNFR-associated death domain
TRAF	TNFR-associated factor
TRAIL	TNF-related apoptosis-inducing ligand
Tween 20	Polyxyethylene Sorbitan Monolaerate
UV	ultraviolet
v-FLIP	viral FLICE inhibitory proteins
XIAP	X-linked inhibitor of apoptosis protein
ZOL	zoledronic acid

xx1

LIST OF FIGURES

- Figure 1.1 Radiologic images of an OS
- Figure 1.2 Magnetic resonance imaging of an OS
- Figure 1.3 The human caspase gene family
- Figure 1.4 Structural representation of the IAP family
- Figure 1.5 Formation of the apoptosome
- Figure 1.6 The mitochondrial intrinsic pathway
- Figure 1.7 The intrinsic and extrinsic pathway
- Figure 1.8 The five TRAIL receptors
- Figure 1.9 The general structures of BPs compared to pyrophosphate
- Figure 1.10 The chemical structure of ZOL
- Figure 1.11 The mevalonate pathway and inhibition by N-BPs
- Figure 1.12 The mevalonate pathway and inhibition by Statins
- Figure 1.13 The mevalonate pathway and inhibition by PTIs
- Figure 2.1 Intra-tibial injection of a mouse
- Figure 3.1 ZOL induces dose-dependent apoptosis in OS cells
- Figure 3.2 Combined ZOL dose graph
- Figure 3.3 ZOL leads to detachment and apoptosis
- Figure 3.4 ZOL induces time-dependent apoptosis
- Figure 3.5 Increasing doses of ZOL leads to increased caspase activity
- Figure 3.6 ZOL treatment leads to increased caspase activity over time
- Figure 3.7 Caspase inhibitors do not protect against ZOL induced apoptosis
- Figure 3.8 DAPI images of Hos cells treated with ZOL and/or caspase inhibitors
- Figure 3.9 GGO dose-dependently abrogates the effect of ZOL in OS cells
- Figure 3.10 GGO protects against ZOL-induced apoptosis in all OS cell lines
- Figure 3.11 GGO inhibits an increase in caspase activation
- Figure 3.12 ZOL inhibits protein prenylation in OS cells
- Figure 3.13 ZOL treatment leads to anoikis
- Figure 3.14 ZOL induces S-phase arrest of the cell cycle
- Figure 3.15 Inhibition of bone resorption by N-BPs

Figure 3.16 ZOL inhibits the mevalonate pathway and inhibits prenylation

Figure 3.17 GGO acts as an intermediate of the mevalonate pathway

- Figure 4.1ZOL augments TRAIL-induced apoptosis in OS cells
- **Figure 4.2** OS cells are differentially sensitised to combination ZOL and TRAIL treatment
- Figure 4.3 The augmented apoptotic effect of ZOL and TRAIL is time-dependent
- Figure 4.4 Morphological characteristics of apoptosis in HOS cells
- Figure 4.5 ZOL treatment leads to DNA fragmentation and induces PARP cleavage
- Figure 4.6 Caspase activation in OS cells treated with ZOL and TRAIL is dose and time dependent
- Figure 4.7 Caspase inhibitor prevents augmented apoptosis in OS cells treated with ZOL and TRAIL
- Figure 4.8 GGO prevents sensitisation to TRAIL in combination with ZOL
- **Figure 4.9** Western blot analysis of HOS cells treated with ZOL and TRAIL, either alone or in combination.
- Figure 4.10 Loss of cell anchorage sensitises HOS cells to TRAIL-induced apoptosis
- Figure 4.11 ZOL leads to S-phase cell cycle arrest whereas combination treatment with TRAIL leads to apoptosis
- **Figure 4.12** Normal human osteoblast-like cells did not exhibit augmented apoptotic effects with co-treatment of ZOL and TRAIL
- Figure 4.13ZOL upregulates TRAIL death receptors on the cell surface of OS cellsbut not normal human osteoblast-like cells
- Figure 5.1 LOV induces dose-dependent apoptosis in OS cells
- Figure 5.2 MEV induces dose-dependent apoptosis in OS cells
- Figure 5.3 LOV augments TRAIL-induced apoptosis in OS cells
- Figure 5.4 MEV augments TRAIL-induced apoptosis in OS cells
- Figure 5.5 FTI induces dose-dependent apoptosis in OS cells
- Figure 5.6 GGTI induces dose-dependent apoptosis in OS cells
- Figure 5.7 FTI augments TRAIL-induced apoptosis in OS cells
- Figure 5.8 GGTI augments TRAIL-induced apoptosis in OS cells

- **Figure 5.9** The inhibitors of the mevalonate pathway do not affect the viability of normal human osteoblast-like cells
- **Figure 5.10** Caspase inhibitors do not protect against apoptosis induced by inhibitors of the mevalonate pathway
- Figure 5.11 GGO is able to rescue OS cells from the effects of statins but not PTIs
- **Figure 5.12** Western blots of from ZOL and statin single and combination treatments with TRAIL.
- Figure 5.13 GGO acts as an intermediate of the mevalonate pathway
- Figure 6.1 Intra-tibial injections of human OS cells in nude mice results in the development of tumours
- Figure 6.2 Radiographic analysis of mice injected with K-HOS cells
- Figure 6.3 Radiographic analysis of mice injected with KRIB cells
- **Figure 6.4** Micro-CT analysis of the tibiae from a nude mouse injected with K-HOS cells
- **Figure 6.5** Micro-CT analysis of the tibiae from a nude mouse injected with KRIB cells
- **Figure 6.6** Histological examination of an intra-tibial OS
- Figure 6.7 Histological examination of lung tissue from nude mice injected with human OS cells
- Figure 7.1 Dosing regimen for *in vivo* OS experiment
- Figure 7.2 Radiograph images of control untreated group
- Figure 7.3 Radiograph images of TRAIL treated group
- Figure 7.4 Radiograph images of ZOL treated group
- Figure 7.5 Radiograph images of ZOL and TRAIL treated group
- Figure 7.6 Micro-CT images of control untreated group
- Figure 7.7 Micro-CT images of tibiae sliced in half from control untreated group
- Figure 7.8 Micro-CT images of TRAIL treated group
- Figure 7.9 Micro-CT images of tibiae sliced in half from TRAIL treated group
- Figure 7.10 Micro-CT images of ZOL treated group
- Figure 7.11 Micro-CT images of tibiae sliced in half from ZOL treated group
- Figure 7.12 Micro-CT images of ZOL and TRAIL treated group

- Figure 7.13 Micro-CT images of tibiae sliced in half from ZOL and TRAIL treated group
- **Figure 7.14** Comparison images of the osteoblastic phenotype taken at different points through out from micro-CT process
- **Figure 7.15** Histological examination of the bone/periosteum border from an inoculated tibiae showing the osteoblastic phenotype
- Figure 7.16 Comparison radiograph images of inoculated tibiae at two-week intervals
- **Figure 7.17** Comparison micro-CT images of tibiae sliced in half from control untreated group and ZOL treated group
- Figure 7.18 Bone volume comparisons between the treatment groups
- Figure 7.19 Histological examination of stained sections of mouse tibiae 35 days post-inoculation with K-HOS cells
- Figure 7.20 Average tumour area per treatment group

LIST OF TABLES

Table 1.1	The categories of OS
Table 1.2	The evolution of BPs
Table 2.1	List of antibodies used
Table 5.1	IC_{50} values for each cell line with statin and TRAIL treatment
Table 5.2	IC_{50} values for each cell line with IMP and TRAIL treatment
Table 7.1	Area of tumour in the inoculated tibiae of each animal
Table 7.2	Animals with lesions and lung metastases

CHAPTER ONE

INTRODUCTION

1.1 OSTEOGENIC SARCOMA

1.1.1 Definition, classification, epidemiology and localisation

Osteogenic sarcoma commonly referred to as osteosarcoma (OS), is the most common non-haematopoietic primary malignancy of the skeleton [Pringle 1999]. Osteosarcoma is defined as a spindle cell neoplasm with proliferating malignant cells that produce osteoid (unmineralised bone) or bone [Campanacci 1999]. The production of osteoid is central to the diagnosis of OS but it may be focal or minimal and therefore not obvious [Pringle 1999; Unni 1996]. A diagnosis of OS may be reasonable provided the neoplasm exhibits other typical features even though the tumour has other predominant elements such as chondroid or fibromatoid differentiation [Unni 1996]. There are consequently several varieties of OS that are not distinct enough to warrant a separate classification, as well as several that possess special features (summarised in **Table 1.1**).

Between 1100 and 1500 new cases of OS are reported in the United States yearly [Whyte 1999]. Males are 1.5 times more likely to develop OS than females. The majority of cases occur before the age of 30 with peak incidence in the second decade of life. The cause of most OS is unknown, however several cases seen in later life, particularly the fifth decade, are usually secondary to Paget's disease, radiation, chondrosarcoma, fibrous dysplasia, bone infarct or chronic osteomyelitis [Campanacci 1999; Unni 1996]. Rare cases of siblings presenting with OS have also been reported [Harmon and Norton 1966].

Three quarters of all cases fall into the "classic" OS category, which is considered a high-grade malignant bone tumour and includes osteoblastic, fibroblastic and chondroblastic OS. The remaining 25 percent are classified as "variants" on the basis of clinical, morphological characteristics and location [Pringle 1999]. A representative radiograph image of an OS is shown in **Figure 1.1** [Campanacci 1999].

2



Figure 1.1 Radiological images of an osteosarcoma. The image on the left represents an osteosarcoma located in the distal femur and the image on the right shows the location at the proximal tibia.

High Grade Malignancy

+ Classic osteosarcoma (osteoblastic, fibroblastic, chondroblastic)

- + Telangiectatic osteosarcoma
- + Osteosarcoma of the jaws
- + Small cell osteosarcoma
- + Osteosarcoma of the bone surface (Peripheral conventional osteosarcoma)
- + Intracortical osteosarcoma
- + Multifocal osteosarcoma
- + Secondary osteosarcoma to disease
- + Secondary osteosarcoma to radiation

Low Grade Malignancy

- + Parosteal (juxtacortical) osteosarcoma
- + Periosteal osteosarcoma
- + Low-grade central osteosarcoma

Table 1.1 Osteosarcoma are divided into two major categories with various types within each category. These are based on histology and location of the tumour.



Figure 1.2 Visualisation of an osteosarcoma within the proximal tibia using magnetic resonance imaging. Lytic lesions in the bone are indicated by white arrows and soft tissue tumour is indicated by a black arrow.

Osteosarcoma is predominantly an intramedullary tumour but in rare cases may originate at the bone surface in the case of peripheral conventional OS and parosteal OS or intracortically (intracortical OS) [Campanacci 1999]. Localisation is often found in the metaphyseal part of the long bones with common sites involving the distal femur, proximal tibia and proximal humerus [Campanacci 1999]. Almost 70 percent of all OSs are located in close proximity to the knee or the shoulder with symptoms including pain and swelling of the affected area [Unni 1996]. Tumours are evaluated further using numerous visual methods such as radiography, angiography, bone scanning, computer tomography and magnetic resonance imaging [Whyte 1999]. Histopathology studies also help determine tumour type and staging, critical for choosing the most beneficial treatment and determining prognosis [Whyte 1999]. Metastatic spread to the lungs is common to 20% of presenting patients, with lymph nodes and other sites of the skeleton less common sites of metastases [Saeter *et al.* 1997; Saeter *et al.* 1997].

1.1.2 Diagnosis of osteosarcoma

The diagnosis of OS depends on a combination of clinical, imaging and histological data thus requiring multiple methods of investigation. To base diagnosis exclusively on one method may be misleading as sampling size of a biopsy alone may not identify all cell types, or imaging alone may not be characteristic for that tumour type [Huvos 1991; Campanacci 1999].

Clinical tools or important information include the age of a patient, which will help identify possible tumour types and the growth rate of the lesion, which can differentiate between high grade (rapid growth), or low grade (slow growth) malignancy. Symptoms such as pain and fever also give clues to diagnosing the neoplasm. The most important information regarding the type of tumour is the site and

localisation and this can be investigated in detail using a range of imaging methods [Huvos 1991; Campanacci 1999]. These include:

1. Conventional radiography- is an essential element in the diagnosis of bone tumours. It provides a negative image of the tumour and reveals the extent of damage to the host bone. Radiographs can often reveal a combination of radiolucency due to osteolysis and radiodensity due to a predominantly osteoblastic phenotype. Substantial osteolysis with undefined borders usually indicates a rapidly aggressive tumour whereas a defined osteogenic border indicates a slow growing tumour [Campanacci 1999]. Radiographs also aid in localising biopsy site for further investigations. Occasionally the presence of an OS is not apparent on a radiograph and further analysis is required [Doney and Vilke 2004].

2. Radioactive isotope bone scans, which identify sites of increased bone turnover by isotope uptake, are useful in exploring the entire skeleton from the skull to the toes, revealing all tumours localised to the bone as well as adjacent soft tissue tumours that may affect the bone.

3. Magnetic Resonance Imaging (MRI)- can distinguish between different tissue types and therefore give superior analysis of soft tissues and bone marrow [Exner *et al.* 1990; Dousset and Buthiau 1998]. The advantages include lack of ionising radiation so it is safe for pregnant women. As OSs are hypervascular, the use of MR angiography is useful in identifying vessels within and around the tumour [Dousset and Buthiau 1998]. A representative MRI image of an OS is shown in **Figure 1.2** [Campanacci 1999].

4. Computerised Tomography (CT) is by far the most sensitive method of tumour detection of the bone. In bone tumours it reveals: the extent of tumour in the bone, bone marrow spaces, soft tissue and extra-compartmental spaces; the involvement with the bone cortex if any; tumour involvement of the joint; and relationship with nearby neurovascular structures [Murphey *et al.* 2003; Dousset and Buthiau 1998]. CT is also

4

used to reveal pulmonary metastases, as it is more sensitive than standard radiography. The presence of metal such as plates and screws inhibits the generation of good quality images but this is improving with new technological advances such as new generation CT instruments and metal artefact reduction [Dousset and Buthiau 1998; Looney *et al.* 2002; Puri *et al.* 2002; Schwarz *et al.* 2003; Stamenkov *et al.* 2003].

After visualisation, a small biopsy, to minimise spreading the tumour, is carefully taken from the site and sent for histological testing [Mankin *et al.* 1982]. The tumours are staged using the histological data, in combination with extension of the tumour into other compartments, as well as metastases. The reason for staging tumours is to try and classify them into groups with a similar behaviour and prognosis and as a guide for treatment [Huvos 1991]. Prognosis is related to the stage, with a higher survival rate observed in early stage patients that those diagnosed at a later stage.

1.1.3 Treatment of osteosarcoma

The treatment of OS has undergone considerable changes over the past 20 years with the efficacy of chemotherapy significantly improving long-term survival to between 55-70%. This is compared to less than 20% survival after 2 years with surgery alone [Link *et al.* 1986; Provisor *et al.* 1997].

Current treatment consists of preoperative chemotherapy (neoadjuvant), to improve the stage of the tumour, followed by surgery (conducted in over 80 percent of OS patients), followed by postoperative (adjuvant) chemotherapy, to prevent local recurrences and metastases [Provisor *et al.* 1997; Kumta *et al.* 2002]. Response to chemotherapy depends on the type and combination of drugs used, the doses given and sensitivity/resistance of the tumour cells. Chemotherapeutic agents used include doxorubicin, etoposide, vincristine, methotrexate and many more, all which act to arrest the growth and induce necrosis in tumour cells.

The surgical procedure in practice today is termed limb-sparing surgery and has replaced the former preference for amputation [DiCaprio and Friedlaender 2003]. This has been made possible with the use of chemotherapy, which improves tumour burden preoperatively therefore making tumour removal more successful [Rougraff *et al.* 1994; DiCaprio and Friedlaender 2003]. Comparison studies of limb salvage versus amputation in OS patients found no significant difference in the rate of survival between the two groups. There was a slightly higher local recurrence and an additional operation on the limb was necessary more often for patients who had limb-sparing surgery [Rougraff *et al.* 1994]. Comparisons also revealed little difference in the quality of life between these two groups [Refaat *et al.* 2002].

The most important factor to prognosis for long-term survival is response to chemotherapy. The development of drug resistance to chemotherapeutic agents is evident in the clinic and remains a problem [Chan *et al.* 1997]. Drug resistance can vary significantly and recent studies suggest the use of gene expression profiles for the prediction of tumour response to adjuvant chemotherapy [Ochi *et al.* 2004; Desai *et al.* 2003]. This prediction would prove very beneficial in identifying tumours with poor responsiveness to chemotherapy so a more intense treatment regime can be administered. Pharmacogenomic studies are also proving to be important by investigating the inherited basis of variable drug response in patients, based on variations in drug metabolising enzymes and drug transporters [Desai *et al.* 2003]. This knowledge will advance treatment outcome, however has yet to become standard procedure prior to the commencement of chemotherapy. Additional improvements to survival might be obtained with new agents, which have differing mechanisms of action, and these may also be used in combination to increase their effectiveness in the clinic.

6

1.2 THE APOPTOTIC PATHWAY

1.2.1 Introduction

While the phenomenon of programmed cell death had been accepted for many years within the scientific community, it was not until 1972 that the term 'apoptosis', was first proposed, by Kerr and colleagues. Apoptosis is commonly identified in single cells usually surrounded by healthy-looking neighbours and defined as a series of morphological events that are distinct from necrosis. Apoptosis has been observed in a diverse range of cells and tissues undergoing cell death. Programmed cell death is essential in controlling cell turnover and together with cell proliferation controls the homeostasis of tissues [Jacobson et al. 1997]. Increased levels of apoptosis have been implicated in a number of diseases, including autoimmune disease and degenerative conditions. In contrast a failure to initiate apoptosis can be detrimental leading to tumour development and growth [Johnstone et al. 2002; Degli-Esposti 1999]. Apoptosis also has an integral role in the programmed destruction of cells in many physiological and pathological events. This includes: the destruction of cells during embryogenesis and metamorphosis, hormone-dependent atrophy, cell deletion in proliferating cell populations, cell death in tumours and in immune cells, cell death induced by cytotoxic T cells and cell death due to injurious stimuli including virally infected and damaged cells [Kerr and Harmon 1991; Degli-Esposti 1999; Nagata 1997]. Therefore, understanding the mechanism of apoptosis may prove to be beneficial in identifying the causes and developing treatment strategies for countless diseases.

Another type of cell death was also identified known as necrosis, usually occurring due to physical or chemical trauma. Necrosis is phenotypically characterised by cytoplasmic swelling and eventual bursting of the cell, releasing its contents into the extra-cellular space and causing an inflammatory response [Trump *et al.* 1965; Nagata

7
1997]. Since 1972, apoptosis and necrosis have been the two general processes used to describe cell death.

1.2.2 The history of apoptosis

Naturally occurring cell death was first reported in 1842 by Carl Vogt, who noticed this phenomenon in the notochord and adjacent cartilage of metamorphic toads [cited by Vaux 2002]. This led to other landmark discoveries, including the description of cell death by Weismann in 1864, chondrocyte death during endochondral ossification by Steida in 1872 and phagocytosis associated cell death in the muscles of metamorphic toads by Metschnikoff in 1883 [cited by Clarke and Clarke 1996]. In 1889, Beard discovered that an entire population of neurons is lost in fish embryos, and Felix reported scattered cell death in developing mammalian muscle. These reports triggered many papers on the subject but it was Flemming in 1885, who gave the first morphological description of apoptosis, in which he termed "chromatolysis", in a study of naturally regressing ovarian follicles [cited by Clarke and Clarke 1996]. In the early 20th century the popularity of the concept of naturally occurring cell death declined, largely due to the decreasing number of German speaking scientists unable to read early reports. It was not until 1951, when Glucksmann's review re-discovered this phenomena, that a fresh surge of interest was generated. The term 'programmed cell death' was first used in 1965 by Lockshin and Williams to describe developmental cell deaths in insect systems [Lockshin and Williams 1965].

In 1972, Kerr and colleagues used the term 'apoptosis' to describe programmed cell death after realising that the morphology of dying cells was similar to that described by Glucksmann, noting a common morphological pattern that was likely not accidental. Following Kerr's description of apoptosis came two exciting discoveries that carried apoptotic research through to the 21st century. Firstly came the observation that DNA

was degraded into nucleosomal-sized fragments during cell death, which was later reported in 1997 to be due to an endogenous nuclease activated in response to apoptotic stimuli [Liu and Zou 1997; Wyllie 1980]. Secondly, the use of nematode worm *Caenorhabditis elegans* was invaluable in studying apoptosis during development, with the observation that exactly 131 of the 1090 somatic cells generated during development underwent apoptosis [Sulston and Horvitz 1977]. Mutants of worm *C. elegans* were used to identify 12 genes involved in programmed cell death [Hedgecock *et al.* 1983; Hengartner *et al.* 1992; Ellis and Horvitz 1986].

It was not until the late 1980's that the explosion of interest came with an insight into the identification of the biochemical and genetic processes involved in apoptosis, beginning with Bcl-2 and its anti-apoptotic capabilities [Vaux *et al.* 1988].

1.2.3 Morphological and biochemical changes of apoptosis

In contrast to necrosis, the controlled process of apoptosis is characterised by the cell and its nucleus condensing itself for engulfment by macrophages or neighbouring cells without any leakage of cytosolic components that will initiate an inflammatory response [Kerr *et al.* 1972]. Using electron microscopy, the morphological changes associated with apoptosis were identified. These begin with cell shrinkage including a dense cytoplasm and tightly packed organelles. This is followed by chromatin condensation, the most characteristic feature of apoptosis, where the chromatin clumps peripherally under the nuclear membrane with occasional fragmentation of the nucleus. This leads to cell surface blebbing and fragmentation and the formation of membrane bound apoptotic cells or bodies. The apoptotic bodies are phagocytosed by adjacent healthy cells or macrophages and degraded within lysosomes [Searle *et al.* 1975; Kerr *et al.* 1972; Wyllie *et al.* 1981].

Distinct biochemical changes, specifically the activation of a group of cysteine proteases, cause most of the morphological changes observed during apoptosis. This group of cysteine proteases are homologous to each other and are known as caspases [Alnemri *et al.* 1996]. Caspases are highly conserved across different species and over a dozen have been identified in humans [Earnshaw *et al.* 1999]. All caspases possess an active-site cysteine, which cleaves substrates C-terminal to aspartate residues, with the distinct substrate specificity determined by four residues amino terminal to the cleavage site [Thornberry *et al.* 1997]. Activation of caspases leads to the selective cleavage of target proteins resulting in either their subsequent activation or inactivation.

Markers of apoptosis are often used to identify apoptosis. This includes the hallmark 'DNA ladder' due to cleavage of nuclear DNA by endonucleases, which are activated by caspase-3-mediated cleavage [Sakahira *et al.* 1998], generating fragments of 180-200 base pair lengths representing nicks at the linker regions between nucleosomes [Walker *et al.* 1997; Wyllie 1987]. This differs from the random fragmentation of DNA seen in necrosis. It has been shown that apoptotic cell death can occur in the absence of DNA fragmentation [Cohen *et al.* 1992].

The cleavage of the enzyme poly(ADP-ribose) polymerase (PARP-1) is another marker of apoptosis [Kaufmann *et al.* 1993]. PARP-1 is involved in DNA repair, replication and transcription, therefore cleavage of the protein may be necessary for cell death to occur [Jeggo 1998; D'Amours *et al.* 1999]. The enzymes caspase-3 and caspase-7, are able to cleave PARP-1, producing an 89 kDa C-terminal fragment and a 24 kDa fragment, thus preventing DNA repair, vital for survival [Kaufmann *et al.* 1993]. It has been reported that liver apoptotic cells do not show PARP-1 cleavage and PARP-1 gene knockout studies in mice have not shown any apoptotic defects [Jones *et al.* 1999; Wang *et al.* 1995]. Alternatively, possible PARP-1 cleavage has also been

detected in non-apoptotic cells, suggesting that there may be exceptions to the rule [Budihardjo et al. 1998].

Over 100 caspase substrates have been identified, including cleavage of the nuclear lamins during apoptosis, which results in nuclear shrinking and budding, and the cleavage of fodrin and gelsolin, disrupting cytoskeletal integrity [Nicholson 1999; Rao *et al.* 1996; Kothakota *et al.* 1997].

1.2.4 Anoikis

Cell death can be triggered by numerous factors including the loss of cell anchorage. In 1994, Frisch and Francis first used the term anoikis, derived from the Greek language meaning state of homelessness, to describe this form of apoptosis. Since then, the association of cell anchorage and cell survival has been studied in numerous adherent cell types and found to play an important role in tissue homeostasis *in vivo*. The role of anoikis in the pathogenesis of disease is of particular importance in trying to understand how cells are rendered resistant to loss of cell anchorage such as tumour metastasis [Ruoslahti and Reed 1994]. Overexpression of oncogenes such as ras, raf and rac has been shown to render cells resistant to anoikis and these oncogenes are implicated in neoplasms that are highly metastatic [McFall *et al.* 2001; Rytomaa *et al.* 2000; Coniglio *et al.* 2001].

Cell anchorage is an extremely complex process involving many adhesion molecules such as integrins and cadherins, that are capable of interacting with varying structural components, including the extracellular matrix (cell-matrix anchorage) and other cells (cell-cell anchorage) for review, [Giancotti and Ruoslahti 1999; Aberle *et al.* 1996]. The process of loss of anchorage alters anchorage-dependent survival signals ultimately leading to the cell's demise [Khwaja and Downward 1997; Cardone *et al.* 1997; Barberis *et al.* 2000].

1.3 THE CASPASE CASCADE

1.3.1 Caspase structure and function

Caspases are fundamental mediators of apoptosis that are highly conserved across phyla and essential for the normal development of higher organisms. Thornberry and Molinequx identified the first caspase, ICE (interleukin-1 β converting enzyme; caspase-1), in 1992, by using sequence homology studies based on the *C. elegans* death gene, *ced-3*, essential for all 131 programmed cell deaths that occur during hermaphrodite development. The *ced-3* gene product was found to be homologous to mammalian ICE, implicating a similar cell death pathway in mammals [Yuan *et al.* 1993]. More importantly, this information led to the realisation that specific proteolysis is an essential role in apoptotic cell death. Caspase substrates were also identified, shedding light on the biochemical complexity involved following caspase cleavage.

The caspase family now consists of at least 14 members, of which 11 human enzymes are known, with similarities in amino acid sequence, structure and substrate specificity [Nicholson and Thornberry 1997]. All members are cysteine proteases capable of cleaving proteins C-terminal to an aspartic acid residue [Thornberry *et al.* 1997]. They are produced as inactive zymogens containing an amino terminal prodomain, a large subunit domain (~20kD) and a small subunit domain (~10kD) [Thornberry and Lazebnik 1998]. Phylogenetic analysis reveals two sub-families related to either ICE (caspase-1), important for inflammation, or the mammalian equivalent of CED-3, important for apoptosis (**Figure 1.3**). Further subdivisions can be made depending on the size of the prodomain [Nicholson 1999]. The first group, such as caspases-1, -2, -4, -5, -9, -11 and -12, are the initiator caspases and contain a long prodomain (>10kDa) and a N-terminal caspase-activating recruitment domain (CARD). Caspase-8 and -10 have a N-terminal death effector domain (DED). Caspase -1, -4, -5, -11 and -12 are mainly responsible for processes other than apoptosis. The other group



Figure 1.3 The human caspase gene family. Two major subfamilies exist in the caspase family related to either ICE or CED-3. Further subdivisions can be made based on their proteolytic specificities (not shown). Each subfamily has a role in either inflammation or apoptosis.

consists of the effector caspases, such as caspases-3, -6, -7, which contain a short prodomain (<30 aa) and lack the N-terminal enzymatic domain. These differences lead to different modes of activation between the two groups.

A pro-apoptotic signal can lead to either death receptor- or mitochondrialmediated activation of initiator caspases, capable of triggering the proteolytic caspase cascade during apoptosis [Medema *et al.* 1997; Green 1998].

1.3.2 Caspase activation

At least three mechanisms of caspase activation exist in mammalian cells; (1) recruitment-activation, (2) trans-activation and (3) auto-activation. In the first case, recruitment of the proenzyme to an oligomeric activation complex using an adaptor molecule leads to activation, such as, recruitment of pro-caspase-8 by FADD, after ligand binding to CD95/Apo-1/Fas/TNF death receptors. This leads to the formation of a death inducing signalling complex (DISC) and subsequent caspase activation (will be discussed in detail in section 1.5.4), [Bodmer *et al.* 2000; Medema *et al.* 1997]. Another example of recruitment activation involves the cytosolic protein apoptosis protease activating factor-1 (Apaf-1), which remains latent until the release of cytochrome c from the mitochondria during apoptosis [Li *et al.* 1997; Green and Reed 1998]. Cytochrome c forms a complex with Apaf-1 that is able to enhance autoactivation of caspase-9 in the presence of dATP. Activated caspase-9 acts as the initiator caspase activating pro-caspase-3, resulting in activation of the caspase cascade. Initiator caspases -8 and -9 represent the starting points for caspase activation either *via* death receptor binding or mitochondrial participation, respectively.

Secondly, trans-activation occurs when an active caspase is able to induce proteolytic activation by coming into very close proximity to the pro-caspase, with the help of Hsp60 activity [Buckley *et al.* 1999]. The pro-domain is removed by

autocatalysis and the large and small subunit domains then associate to form heterodimers, which then associate with other heterodimers to form tetramers, containing two independent catalytic sites. Both subunits contribute to form the catalytic sites that are highly specific, only cleaving after an aspartic acid with a minimum of four NH₂-terminal amino acids to the cleavage site for effective proteolysis [Walker *et al.* 1998]. This is a common mechanism, by which initiator caspases activate effector caspases in sequential order and recognition of this sequential activation has led to the common term 'caspase cascade' [Walker *et al.* 1998].

Finally, caspases can also undergo autocatalytic-activation to ensure a rapid response to apoptotic stimuli [Buckley *et al.* 1999].

1.3.3 Inhibition of caspase activation

The endogenous inhibitors of caspases in humans are members of the inhibitor of apoptosis (IAP) family. They act to counteract cell death by directly inhibiting active caspases and blocking caspase activation [Holcik and Korneluk 2001; Deveraux and Reed 1999]. All IAP family members contain up to three baculovirus IAP repeat (BIR) domains of ~70 amino acids, essential for their inhibitory function. Proteins containing BIR domains have been identified in a wide range of species including yeast, insects, birds and mammals [Deveraux and Reed 1999]. The apoptotic program is unclear in yeast, suggesting that proteins containing BIR domains may be involved in other processes [Fraser and James 1998]. Eight members of the IAP family have been identified in humans including, XIAP/hILP, c-IAP/HIAP-2, c-IAP/HIAP-1, NAIP, Survivin, BRUCE, MLIAP and ILP-2 as shown in **Figure 1.4** [LeBlanc 2003]. The first member of the IAP family identified in humans, X-linked IAP (XIAP), is a potent inhibitor of caspases-3, -7 and -9 [Deveraux *et al.* 1997; Deveraux and Reed 1999]. More specifically, the second BIR domain (BIR2) targets caspases 3 and 7 and the third



Figure 1.4 Structural representation of the IAP family. Eight human IAPs have been identified and are represented above. The amino acid length is indicated to the right of the IAP and the location of the Baculovirus IAP repeat (BIR), caspase recruitment domain (CARD), really interesting new gene (RING) and ubiquitin conjugating (UBC) domains are indicated.

BIR domain (BIR3) targets caspase 9, indicating that a single BIR domain can be sufficient to inhibit caspases [Chai *et al.* 2001; Huang *et al.* 2001; Riedl *et al.* 2001]. Other family members have also been shown to act as caspase inhibitors including c-IAP/HIAP-2, c-IAP/HIAP-1, ML-IAP and ILP-2 [Salvesen and Duckett 2002]. In addition to regulation of apoptosis, IAP members have also been implicated in other processes for example the involvement of the protein survivin during mitosis [Reed and Bischoff 2000].

IAPs do not bind or inhibit caspase-8, the initiator caspase in CD95/Apo-1/Fas/TNF death receptor-induced apoptosis, instead they protect the cell by binding to its substrate, caspase-3 and inhibit further caspase activation [Deveraux *et al.* 1997]. Conversely, activation of the mitochondrially dependent pro-caspase-9 is directly inhibited by IAPs, preventing activation by the cytochrome c/Apaf-1 complex. IAPs also inhibit the next substrate in the cascade, caspase-3, preventing activation of procaspase-9 *via* a feedback loop mechanism [Srinivasula *et al.* 1998]. Over-expression studies have confirmed that IAPs are able to protect against anti-apoptotic proteins of the Bcl-2-family such as Bax, which are able to induce cytochrome c release from the mitochondria [Deveraux *et al.* 1997; Jurgensmeier *et al.* 1998]. Over-expression studies of pro-caspases 3, 7, or 9 leads to apoptosis but co-transfection of IAPs prevents activation of caspases suggesting that the ratio of caspase to IAPs is critical in determining cell life or death [Deveraux *et al.* 1997].

The activity of IAPs is controlled at various levels. At the transcriptional level, NF κ B is capable of stimulating expression of c-IAP/HIAP-1, c-IAP/HIAP-2 and XIAP [Deveraux and Reed 1999]. In contrast, caspase-mediated cleavage is regulated by the release of the proteins Smac/Diablo and HtrA2/Omi from the mitochondria, which bind and inhibit IAP, hence promoting apoptosis [Holcik and Korneluk 2001].

1.3.4 Importance of caspase activity

The caspase connection to apoptosis was first revealed in the nematode *C. elegans* when a *ced-3* gene mutation led to the survival of cells normally destined to die [Ellis and Horvitz 1986]. Genetic studies in mammals have also revealed the importance of caspases in normal development. Gene deletion experiments have provided much information regarding the importance of caspases in development, however the significance of these deleted genes may be underestimated, as alternative caspases may compensate for the loss of a particular gene, therefore double knockouts may be more informative in defining their roles. Mammalian caspase mutants often display tissue-specific defects in apoptosis probably due to the large number of caspases that exist.

The first knockout studies to show developmental defects due to insufficient apoptosis were observed in caspase-3^{-/-} knockout mice, which led to excess brain tissue and eventual postnatal death [Kuida *et al.* 1998]. No abnormalities were seen in any other organs. Caspase-9^{-/-} knockout mice exhibited similar but less extensive abnormalities, suggesting that caspase-3 plays a critical role in the development of the mammalian brain where it depends on caspase-9 for activation [Hakem *et al.* 1998]. Caspase-6^{-/-} mice develop normally, with minor defects in B cell development [Zheng and Flavell 2000]. Conversely, knockout of caspase-7 leads to embryonic death, indicating that compensation by other caspases exist for caspase-6 but not caspase-7. Premature animal death seen in caspase-7^{-/-} knockouts, as opposed to other caspase knockout mice, might be due to differences in the localisation and onset of expression [Sadowski-Debbing *et al.* 2002].

Caspase-1 and -11 knockout mice revealed normal development but immune deficiencies. Caspase-1 deficient animals had problems with the production of IL-1 α and IL-1 β , TNF- α , IL-6 and interferon γ production indicating a vital role in the immune response [Kuida *et al.* 1995]. Mice deficient in caspase-11 were also found to

have problems with the production of IL-1 α and IL-1 β and have fewer apoptotic cells due to a defect in caspase-3 activation, suggesting that caspase-11 may act as an initiator caspase under certain conditions [Kang *et al.* 2000]. Both caspase-1 and -11 play an important role in inflammation but not necessarily apoptosis. In humans, mutations in caspase-10, leads to defective apoptosis of lymphocytes and is associated with type II auto-immune lymphoproliferative syndrome [Wang *et al.* 1999].

Caspase-2 and -12 knockout mice exhibit no marked abnormalities except excess numbers of female germ cells in the caspase-2 deficient mice [Bergeron *et al.* 1998].

Caspase-8 is an important caspase in death receptor signal-mediated apoptosis. Cell lines deficient in caspase-8 are completely resistant to death receptor induced apoptosis and partially resistant to apoptosis induced by UV-irradiation, adriamycin and etoposide [Juo *et al.* 1998]. In mice, caspase-8 deficiency leads to death *in utero*, due to abnormal heart development, abdominal haemorrhage and erythrocytosis in many organs, demonstrating a role in haematopoeisis and a non-redundant role in development [Varfolomeev *et al.* 1998].

Knockout studies have unveiled the role of caspases in apoptosis and normal development as well as providing clues into the specific participation of caspases in other systems such as cytokine maturation, immunity and haematopoeisis.

1.4 THE INTRINSIC PATHWAY OF APOPTOSIS

1.4.1 The Bcl-2 family

The Bcl-2 family of proteins are key regulators of apoptosis capable of responding to death and survival signals generated both inside and outside the cell [Strasser *et al.* 2000]. The Bcl-2 family of proteins are involved in apoptosis during embryogenesis and continue to play an important role in haematopoiesis and tissue homeostasis after birth

[Jacobson *et al.* 1997]. More importantly, the Bcl-2 protein family regulates apoptosis induced by cytokine withdrawal or cytotoxic stress conditions, such as those elicited by anti-cancer drugs [Adams and Cory 1998].

C. elegans encodes only two members of the Bcl-2 family, CED-9 survival factor and EGL-1 death factor. The human Bcl-2 family includes more than 30 anti- and proapoptotic molecules that bind to each other to form homo- and heterodimers [Tsujimoto 2003]. The Bcl-2 family is subdivided into 3 categories: (i) (CED-9-like) anti-apoptotic *Bcl-2-like survival factors* such as Bcl-2, Bcl-x_L, Mcl-1, Al/Bfl-1, NR-13, Boo/Diva/Bcl2-L-10 and Bcl-B; (ii) pro-apoptotic, *Bax-like death factors* such as Bax, Bak, Bok/Mtd, Bcl-xs; and (iii) EGL-1-like pro-apoptotic, *BH3-only death factors* such as Bik/Nbk, Blk, Hrk/DP5, BNIP3, Bim₁/Bod Bad, Bid, Noxa, PUMA/Bbc3 and Bmf [Huang and Strasser 2000]. Interactions between a pro-apoptotic and anti-apoptotic factor affects the threshold, which determines the fate of the cell.

The *Bcl-2-like survival factors* contain three to four Bcl-2 homology domains (BH1-BH4), capable of mediating interactions with other proteins and essential to their survival function [Farrow and Brown 1996; Hanada *et al.* 1995; Kelekar and Thompson 1998]. These proteins are commonly bound to intracellular membranes such as the mitochondria and nuclear/ER membranes, and act as scavenger molecules for *BH3-only* and *Bax-like death factors* thereby inhibiting their apoptotic function. They have been shown to localise to other regions within the cell and still act as survival factors, but to a lesser extent [Borner *et al.* 1994; Borner 2003].

As the name suggests, the *BH3-only death factors* share only the short BH3 domain [Huang and Strasser 2000]. The levels of *BH3-only death factors* in mammalian cells are controlled at the transcriptional and post-translational level, and are activated in response to an apoptotic signal *via* several mechanisms [Puthalakath and Strasser 2002]. These include (i) transcriptional induction- PUMA/Bbc3 and NOXA are induced

by a p53-dependent apoptotic signal, (ii) post-translational dephosphorylation- BAD is normally phosphorylated and sequestered in the cytoplasm by binding to scaffold proteins until an apoptotic stimulus leads to its dephosphorylation and dissociation so that it can bind to Bcl-2 survival proteins (iii) proteolysis- the inactive cytosolic form of BID is cleaved by caspase-8 into a truncated form that translocates to the mitochondria, (vi) cytoskeletal sequestration- Bmf interacts with the actin cytoskeleton and is released in response to anoikis stress signals. Thus *BH3-only death factors* act as sensors and are activated in response to an apoptotic signal.

Bax-like death factors are converted into pro-apoptotic proteins after removal of the N-terminal BH4 domain by proteolysis [Cheng *et al.* 1997]. Modified *Bax-like death factors* contain three BH domains and undergo a conformational change that favours apoptosis [Adams and Cory 1998]. This cannot entirely explain the difference between *Bcl-2-like survival* and *Bax-like* death *factors* as some *Bcl-2-like* lack a BH4 domain and the addition of BH4 to Bax is insufficient to convert it then to survival factors, suggesting that other protein domains may be involved [Adams and Cory 1998; Borner 2003]. Activation of the *Bax-like death factors* leads to their stable insertion into the mitochondrial membrane to form a pore, thus increasing permeability of the outer membrane [Antonsson 2001]. How activation occurs at the mitochondrial membrane in response to apoptotic stimuli remains unknown.

1.4.1 The role of the Bcl-2 family in cancer

Bcl-2 and its homologues have been associated with numerous types of cancer including human follicular centre lymphoma in which up to 85% of cases contain a translocation of the Bcl-2 gene leading to overexpression of the protein and increased survival of the tumour cell [Tsujimoto *et al.* 1988]. Overexpression of Bcl-2 has also been reported in prostate, breast and colon carcinomas and glioblastomas [Bettaieb *et al.*

2003]. Another *Bcl-2-like survival factor*, Mcl-1, is often overexpressed in relapsed AML resulting in poorer prognosis. Similarly loss of function mutations in pro-apoptotic proteins such as Bax have also been implicated in certain cancers leading to changes in the pro- and anti-apoptotic protein ratio [Bettaieb *et al.* 2003].

As part of its role in the cell, the tumour suppressor gene p53 can induce cell cycle arrest or apoptosis in response to DNA damage. Mutations in p53 are found in almost half of all human cancers and mutations in other genes involved in the p53 signalling pathway, such as arf and mdm2, have also been identified despite the presence of a wild-type p53 [Levine 1997]. Defective p53 signalling also affects the induction of the *BH3-only death factors*, PUMA/Bbc3 and NOXA effectively tipping the balance of anti- and pro-apoptotic proteins towards tumour cell survival [Oda *et al.* 2000; Nakano and Vousden 2001].

Since overexpression of anti-apoptotic proteins promotes tumourigenesis, then pro-apoptotic proteins may serve as tumour suppressors such as the induction of the *BH-3-only death factor* Bik/Nbk in response to an overexpressing adenovirus protein [Mathai *et al.* 2002].

It still remains controversial as to whether the outcome of cancer treatment is determined by apoptosis sensitivity or a result of the irreparable damage sustained by the cell [Brown and Wouters 1999].

1.4.3 The mitochondrial intrinsic pathway

The first critical step in the intrinsic apoptotic pathway is the release of proteins from the mitochondrial membrane, which ultimately leads to caspase activation. This is regulated by numerous factors which leads to mitochondrial membrane permeabilization (MMP), and can be induced by pro-apoptotic second messengers including Ca^{2+} , reactive oxygen species, certain lipids such as ceramide and stress

•

kinases [Zamzami and Kroemer 2003]. As mentioned earlier, pro-apoptotic Bcl-2 proteins also promote MMP, whereas anti-apoptotic Bcl-2 proteins act to inhibit MMP.

An apoptotic signal can lead to cell death *via* activation of either initiator caspase -8 or -9, which are capable of mediating two distinct sets of signals. Activation of caspase-8, *via* death receptor-mediated apoptosis (discussed in detail in section 1.5.4), leads to the cleavage of the Bcl-2 protein Bid [Luo *et al.* 1998]. Evidence suggests that truncated Bid (tBid) translocates from the cytosol to the mitochondrial membranes, where it induces MMP together with either Bax or Bak (Bcl-2 anti-apoptotic proteins), which are able to insert into the membrane and cause permeabilization. Translocation of Bid specifically to the mitochondria is due to the membrane lipid cardiolipin, which is not present in any other organelles [Lutter *et al.* 2000]. Anti-apoptotic Bcl-2 proteins bound to the mitochondrial membrane inhibit apoptosis by antagonizing Bax or Bak. Thus the ratio of pro- and anti-apoptotic proteins will determine MMP and thus the fate of the cell [Letai *et al.* 2002].

Similarly apoptotic signals without the involvement of the death receptors and caspase-8, still leads to cell death *via* Bcl-2 proteins. Activation of pro-apoptotic proteins disrupts the Bcl-2 protein ratio and as a consequence mitochondrial membrane integrity is lost, leading to permeabilization and the release of toxic proteins from the mitochondria involved in both caspase-dependent and -independent apoptotic pathways [Saelens *et al.* 2004].

MMP leads to the release of cytochrome c into the cytosol where it binds to the Cterminal region of Apaf-1. Apaf-1 is a cytosolic protein with an N-terminal caspaserecruitment domain (CARD), a nucleotide-binding domain with homology to C. elegans CED-4, and a C-terminal domain containing 12-13 WD-40 repeats [Zou *et al.* 1997]. The WD-40 repeats facilitate binding to cytochrome c and through association of dATP with Apaf-1 leads to the exposure and subsequent oligomerization of the N-terminal

CARD, allowing the recruitment of pro-caspase-9, and activation through a CARD-CARD interaction [Adrain *et al.* 1999; Jiang and Wang 2000]. The cytochrome *c*/Apaf-1/procaspase-9 complex is termed the apoptosome and activated caspase-9 in turn cleaves and activates caspase-3, initiating the caspase cascade as shown in **Figure 1.5** [Li *et al.* 1997; Wang 2001].

1.4.4 Regulation of the mitochondrial intrinsic pathway

Procaspase activation is the critical mechanism within the apoptotic pathway and is regulated by complex processes at various levels including caspase recruitment and procaspase oligomerisation of initiator caspases. Inhibitors of apoptosis proteins (IAP) directly inhibit caspase activation (as mentioned previously in 1.3.3). Regulation also occurs prior to caspase/adaptor interactions with the involvement of Bcl-2 anti-apoptotic proteins such as Bcl-X_L and Boo, which bind to Apaf-1 and procaspase-9 to form a complex that inhibits caspase-9 activation as shown in Figure 1.6 [Pan et al. 1998; Song et al. 1999]. Other anti-apoptotic proteins include Bcl-2, which functions to block the release of cytochrome c from the mitochondria, thereby preventing Apaf-1 mediated activation of caspase-9 [Green and Reed 1998]. Conversely, Bid acts to promote cytochrome c release [Luo et al. 1998; Jiang and Wang 2000]. Other anti-apoptosis proteins include, catalytically inactive molecules, which mimic caspases and possess an intact CARD or DED. A variant of caspase-9 has been identified, which interferes with the formation of a functional Apaf-1/caspase-9 activating complex [Seol and Billiar 1999]. Caspase-8-like molecules such as FLIP, contain DED but lack catalytic function and act by blocking caspase-8 recruitment to FADD [Nicholson and Thornberry 1997].

To add complexity to the regulation process, IAP antagonists also exist to overcome IAP inhibition of caspase activation. Numerous proteins exist in *Drosophila* including Hid, Grim, Reaper, Sickle and Jafrac2, and at least two mitochondrial proteins



Figure 1.5 Formation of the apoptosome adapted from Cain *et al.*, 2002.



Figure 1.6 The mitochondrial intrinsic pathway.

in mammals, Smac/DIABLO and Omi/HtrA2 [Srinivasula *et al.* 2002; Vaux and Silke 2003]. All these proteins share a conserved four residue IAP-binding motif (IBM) at their N-terminus, which allows them to bind IAPs.

The Smac/DIABLO protein was independently identified by two groups. Smac was found to stimulate caspase-3 activation in humans and DIABLO was found in mice [Verhagen *et al.* 2000; Du *et al.* 2000]. The nuclear encoded Smac/DIABLO protein is expressed in most human tissues and is synthesized as a precursor protein with a mitochondrial localisation sequence (MLS) required for import into the mitochondria. Once there it is modified by proteolysis to remove the MLS and to expose the IBM of the mature protein, which exists as a dimer [Srinivasula *et al.* 2001]. The IBM of the Smac/DIABLO is able to bind XIAP, cIAP1, cIAP2 and surviving (**Figure 1.6**). Each Smac/DIABLO within the dimer is able to bind to BIR2 and BIR3 (baculovirus IAP repeat), which is essential for IAP activity. BIR3 normally binds to caspase-9 and prevents activation, however the binding of IBM of Smac/DIABLO displaces caspase-9 from XIAP and promotes caspase activity [Srinivasula *et al.* 2001].

High temperature requirement-A2 (Htr2A/Omi) is so named due to its upregulation of both expression and proteolytic activity, following heat shock [Gray *et al.* 2000]. Like Smac/DIABLO, Htr2A/Omi protein is imported into the mitochondria with the help of the MLS, where it is also modified to expose the IBM of the mature 35 kD protein, that exist as a trimer [Suzuki *et al.* 2001; Hegde *et al.* 2002]. Apoptotic signals lead to the release of Htr2A/Omi from the mitochondria into the cytosol, where it contributes to caspase-dependent apoptosis by binding to and inhibiting IAP proteins and caspase-independent cell death by signalling pathways *via* its protease activity [Hegde *et al.* 2002; Blink *et al.* 2004]. Htr2A/Omi function resembles Smac/DIABLO by competing with caspase-3, -7 and -9 for XIAP, cIAP1 and cIAP2 binding, thus promoting caspase activity [Suzuki *et al.* 2001; Verhagen *et al.* 2002]. Unlike

Smac/DIABLO, Htr2A/Omi preferentially binds to the BIR2 domain of XIAP and not BIR3 [Verhagen *et al.* 2002]. Following binding to its IBM, Htr2A/Omi then cleaves the IAPs, rendering them irreversibly inactive [Srinivasula *et al.* 2003; Yang *et al.* 2003].

It has been reported by several groups that procaspases are localised within the mitochondria and may promote apoptosis after their release and subsequent activation in the cytosol [Mancini *et al.* 1998; Susin *et al.* 1999; Zhivotovsky *et al.* 1999]. Exactly how the procaspases are imported into the mitochondria remains to be identified, as they all lack a MLS except for procaspase-2. Van Loo and colleagues were unable to identify the presence of caspases within the mitochondria therefore caspase localisation within the cell remains controversial [van Loo *et al.* 2002].

1.4.5 Caspase-independent cell death

Bax-like death factors are directly involved in mitochondrial permeabilisation and therefore provoke the release of pro-apoptotic proteins, including cytochrome c and Smac/DIABLO, which have their own specific functions. Up to 30 different proteins were released into the cytoplasm after mitochondrial perforation in Bid- treated mitochondria. These proteins were found to be involved in both caspase-dependent and caspase-independent pathways [van Loo *et al.* 2002].

Endonuclease G is a 33 kD protein, which contains a mitochondrial targeting sequence required for import into the organelle. Endonuclease G is able to induce caspase-independent cell death, by translocating to the nucleus after mitochondrial release, where it degrades DNA [Van Loo *et al.* 2001]. DNA fragmentation occurs in a sequential manner with the cleavage of genomic DNA into high molecular weight sizes followed by nucleosome-sized fragments and is independent of caspases [Widlak *et al.* 2001; Li *et al.* 2001].

Apoptosis-inducing factor (AIF) is a 57 kD flavoprotein, with NADH oxidase activity required for protection against oxidative stress [Miramar *et al.* 2001]. This function contrasts with AIFs ability to also contribute to apoptosis. AIF is released into the cytosol where it translocates to the nucleus following an apoptotic stimulus. In the nucleus AIF contributes to large-scale (50 kb) DNA fragmentation and condensation [Lorenzo *et al.* 1999]. Studies have revealed that AIF itself cannot cleave DNA and may recruit or activate an endonuclease to facilitate DNA fragmentation and condensation [Susin *et al.* 1999; Ye *et al.* 2002]. Specific inhibition of AIF translocation was shown to inhibit nuclear apoptosis in fibroblasts, confirming the importance of AIF nuclear translocation in the execution of cell death [Susin *et al.* 1999]. Mice deficient in AIF show developmental defects including lack of blastocysts, suggesting that caspase-independent cell death is a crucial part of normal development [Joza *et al.* 2001].

Although the activity of Endonuclease G and AIF is caspase-independent, their release into the cytosol may be both caspase dependent and independent, subject to the apoptotic stimulus.

Other non-caspase mediators of programmed cell death are the calpains. The calpains are cysteine proteases and are localised in the cytosol in an inactive form. Activation requires an increase in calcium levels and leads to proteolytic cleavage of the zymogen followed by association with membrane phospholipids. They participate in apoptotic signalling both upstream and downstream of caspases [Nakagawa and Yuan 2000; Waterhouse *et al.* 1998].

The cathepsin protease family have also been implicated in caspase-independent cell death, more specifically cysteine cathepsins B and L and aspartate cathepsin D [Johnson 2000]. Cathepsins are activated in the endosomal/lysosomal compartment by autoproteolysis under low pH, or proteolysis by other cathepsins, and translocate to the

nucleus [Mathiasen and Jaatela 2002]. Cathepsin participation occurs in both caspase dependent and independent cell death and can be triggered by various stimuli.

Studies on the serine protease proteins granzyme A and B, demonstrated that granzyme B is able to cleave and directly activate caspases in caspase-mediated apoptosis only, whereas granzyme A is able to induce death by cleaving DNA even in the presence of caspase inhibitors [Talanian *et al.* 1997; Beresford *et al.* 2001].

Interestingly, calpains, cathepsins and granzymes can also trigger MMP and may be able to activate pro-apoptotic Bcl-2 proteins [Roberg *et al.* 1999; Trapani *et al.* 2000; Waterhouse *et al.* 1998].

1.5 THE EXTRINSIC PATHWAY OF APOPTOSIS

1.5.1 Introduction

The TNF superfamily consists of the TNF receptor (TNFR) family and the corresponding TNF ligand family and is involved in proliferation, survival, cell differentiation and apoptosis [Mallett and Barclay 1991; Smith *et al.* 1994]. To date, 19 different ligand members that signal through 29 receptors of the TNF superfamily have been identified. Many of them have been found to associate with a wide variety of diseases including cancer, septic shock, viral replication, bone resorption, rheumatoid arthritis, diabetes, allergy, atherosclerosis, myocardial infarction, and acquired immune deficiency disease. All the cytokines of the TNF superfamily mediate their effects through the activation of the transcription factors NF-kappaB and c-Jun N-terminal kinase, apoptosis, and proliferation [Gaur and Aggarwal 2003].

1.5.2 The TNF ligand family

The TNF ligands are expressed as type II transmembrane proteins that can have both a membrane embedded as well as cleaved, soluble forms [Idriss and Naismith 2000]. Both forms are processed into biologically active trimers. The TNF ligands share 25-30% sequence homology and are characterised by a conserved C-terminal domain of 150 amino acids, termed the TNF homology domain (THD), which is responsible for trimer assembly but have little sequence similarity in their external surface, which accounts for receptor specificity [Bodmer *et al.* 2002; Locksley *et al.* 2001]. The TNF ligand superfamily consists of TNF α , LT- α , LT- β , CD271, CD30L, CD40L, CD95L, 4-1BBL, OX40L, TRAIL, RANKL, TWEAK and Apo-3 Ligand [Smith *et al.* 1994]. Each member of the TNF ligand family binds at least one receptor from the TNF receptor family, with some ligands such as APO2L/TRAIL, capable of binding several receptors. In addition, many ligands bind the same receptors [Browning *et al.* 1993]. Generally, the TNF ligand family members cause multimerization of the receptors after binding [Banner *et al.* 1993].

1.5.3 The TNF receptor family

The TNFR family members are type 1 transmembrane glycoproteins containing highly conserved cysteine-rich domains (CRDs) involved in ligand binding [Gruss and Dower 1995]. Up to 65% protein sequence homology within the extracellular, amino-terminal CRD region is shared between TNFR family members. The number of CDRs can range from one to six, but is commonly three or four and is responsible for receptor folding resulting in a signalling unit of three receptors often bound by a trimeric ligand molecule [Idriss and Naismith 2000].

The interactions between ligands and their corresponding receptors induce cellular responses such as activation, proliferation, differentiation and apoptosis [French and Tschopp 1999]. The type of response depends on the receptor and the cell type and the final signal received by the cell. Some TNFR family members have the ability to transmit an intracellular death signal capable of mediating apoptosis, these are called death receptors [Locksley et al. 2001]. Other receptors do not signal, instead it has been suggested they act as decoys that compete with signalling receptors for the corresponding ligand [Idriss and Naismith 2000]. Signalling members can be divided into two groups based on their cytoplasmic region. One class of receptors contain a cytoplasmic death domain (DD) and are called the death receptors, the other class lack this DD [Ashkenazi and Dixit 1999]. Among the 8 homologous death receptors, at least 6 can stimulate apoptosis through caspase activation [Screaton and Xu 2000]. These include DR3 (WSL-1/Apo-3/TRAMP/LARD), DR6, TNF receptor 1 (TNF-R1/p55-R), TRAIL-R2 TRAIL-R1 (Apo2/DR4) and (CD95/APO-1/APT1), Fas (DR5/TRICK2A/TRICKB/KILLER), [Locksley et al. 2001]. The involvement of this receptor class in apoptosis will be discussed in section 1.5.4., below.

1.5.4 The death receptor/extrinsic pathway of apoptosis

The extrinsic pathway is death receptor dependent and requires ligand binding to initiate apoptosis. Stimulation of receptors results in receptor aggregation and recruitment of DD-containing adaptor proteins. These include FADD (Fas-associated death domain/MORT1/MACH) and TRADD (TNFR-associated death domain), which bind to death receptors *via* the DD [Boldin *et al.* 1996; Chinnaiyan *et al.* 1995; Hsu *et al.* 1995]. Caspase-8 is recruited by the adaptor protein to form the death inducing signalling complex (DISC), which results in the subsequent activation of caspase-8 and

initiation of the caspase cascade *via* direct cleavage of downstream effectors [Schulze-Osthoff *et al.* 1998; Walczak and Krammer 2000].

The extrinsic and intrinsic pathway can be interconnected at various levels [Roy and Nicholson 2000]. Activation of caspase-8 *via* death receptor stimulation can result in the cleavage of Bid, a Bcl-2 family member. Initially it was thought that the truncated Bid (tBid) translocates to the mitochondria and triggers cytochrome c release [Luo *et al.* 1998; Li *et al.* 1998]. Recently, Deng and colleagues reported that tBid acts on Bax and Bak to induce mitochondrial release of Smac/DIABLO into the cytosol where it can bind inhibitor of apoptosis proteins, as shown in **Figure 1.7** [Deng *et al.* 2002]. In addition, caspase-6 activation by mitochondria may feed back into the extrinsic pathway by cleaving caspase-8 [Slee *et al.* 1999].

1.6 TRAIL AND ITS RECEPTORS

1.6.1 **TRAIL**

TNF-related apoptosis-inducing ligand (APO-2 Ligand/TRAIL) was identified independently by two groups, using a conserved TNF member sequence to search the expressed sequence tag (EST) database [Pitti *et al.* 1996; Wiley *et al.* 1995]. For the purpose of this thesis APO2L/TRAIL will be referred to as TRAIL. TRAIL is a type II transmembrane protein with a predicted molecular weight of 32 kD. Human TRAIL is 281 amino acids long, divided into a 243 amino acid residue extracellular region, 21 amino acid transmembrane region and a 17 amino acid cytoplasmic tail [Wiley *et al.* 1995; Pitti *et al.* 1996]. The carboxy-terminal extracellular region can be proteolytically cleaved from the cell surface into a soluble form [Mariani and Krammer 1998]. The carboxy-terminal extracellular roms homotrimers that bind three receptor molecules [Hymowitz *et al.* 1999]. Unlike other TNF ligand family members,



Figure 1.7 The intrinsic and extrinsic pathway

TRAIL contains a zinc ion at the trimer interface held in place by the single unpaired cysteine residue (Cys 230) of each monomer [Mariani and Krammer 1998]. This zinc ion is vital for trimer stability of TRAIL and substituting it leads to reduced apoptotic capability [Bodmer *et al.* 2000; Mariani and Krammer 1998]. Studies have shown that TRAIL has the ability to induce apoptosis in a variety of tumour cells, but not of most normal cells, potentially making it an ideal anticancer agent [El-Deiry 2001; Ashkenazi and Dixit 1999; Jo *et al.* 2000; Walczak *et al.* 1999; Ashkenazi *et al.* 1999; Atkins *et al.* 2002; Evdokiou *et al.* 2002].

TRAIL mRNA expression is widespread in normal tissues and has been reported for peripheral blood leukocytes, bone, brain, colon, kidney, liver, lung, ovary, prostate, spleen, testis and thymus [Wiley *et al.* 1995]. The broad range of tissues expressing TRAIL suggests that TRAIL-induced death may be regulated at the level of the TRAIL receptors, unlike the expression of other TNF family members, which is tightly regulated.

TRAIL induces apoptosis by interacting with its receptors [Wang and El-Deiry 2003]. To date five TRAIL receptors have been identified (see Figure 1.8). DR4 and DR5 are known as the death receptors and encode the typical type I transmembrane protein structure, shared with other members of the TNF receptor family. Like several other TNF receptor family members, DR4 and DR5 also possess a sequence within their cytoplasmic domain that is homologous to the death domain [Pan *et al.* 1997; Pan *et al.* 1997; Degli-Esposti *et al.* 1997; Degli-Esposti *et al.* 1997; Degli-Esposti *et al.* 1997; Degli-Esposti *et al.* 1997; Difficult receptors increased the complexity of the TRAIL ligand-receptor system, particularly in regard to the difference in sensitivity of normal and tumour cells to TRAIL-induced apoptosis. However, the receptors known as DcR1 (TRAIL-R3/TRID), DcR2 (TRAIL-R4) and OPG were found to lack apoptotic capability, potentially providing a logical explanation for the protection of normal cells



Figure 1.8 The five receptors for TRAIL. Adapted from Degli-Esposti *et al.*, 1999. Numbers indicate amino acid length.

to TRAIL-induced apoptosis. It appears that these receptors have the ability to act as 'decoys', capable of binding TRAIL and therefore competing with the death -domain containing receptors DR4 and DR5.

The four membrane-bound TRAIL receptors are tightly clustered on human chromosome 8p21-22, suggesting that recent evolution occurred *via* gene duplication [Walczak *et al.* 1997; Degli-Esposti *et al.* 1997]. OPG also maps to chromosome 8 at position 23-24 of the q arm [Simonet *et al.* 1997].

Pan and colleagues identified the first of the TRAIL death receptors, DR4, using the EST data base system [Pan *et al.* 1997]. DR4 is a 468 amino acid type I transmembrane protein that contains a 23 amino acid signal sequence, a 226 amino acid extracellular region, a 19 amino acid transmembrane region and a 220 amino acid cytoplasmic region. Within the extracellular domain there are two cysteine-rich motifs homologous to those within the ligand-binding domain of other TNF receptor family members [Pan *et al.* 1997]. DR4 binds TRAIL with a very high affinity and is able to induce an apoptotic signal due to a functional death domain within the cytoplasmic region [Degli-Esposti 1999; Pan *et al.* 1997; Pan *et al.* 1997; Truneh *et al.* 2000]. DR4 mRNA is expressed in most of the same tissues as TRAIL mRNA [Pan *et al.* 1997; Pan *et al.* 1997].

Soon after the discovery of the first TRAIL receptor, several groups reported a second receptor, DR5, also known as TRAIL-R2/TRICK2A/TRICKB/KILLER [Sheridan *et al.* 1997; Screaton and Xu 2000; Walczak *et al.* 1997; Wu *et al.* 1997]. DR5 is a 411 amino acid type I transmembrane protein that contains a 51 amino acid signal sequence, a 132 amino acid extracellular region, a 22 amino acid transmembrane region and a 206 amino acid cytoplasmic region [Pan *et al.* 1997]. There are also two cysteine-rich motifs in the extracellular region of DR5 also present in DR4. Northern

blot analysis revealed that most tissues expressing DR4 mRNA also expressed DR5 mRNA [Pan et al. 1997].

DcR1, also known as TRID, TRAIL-R3 and LIT, was identified using the sequence of the ligand binding domain of TNFR1 [Pan *et al.* 1997; Sheridan *et al.* 1997; Walczak *et al.* 1997]. The 299 amino acid DcR1, contains a 23 amino acid signal sequence, a 217 amino acid extracellular region and a 19 amino acid transmembrane region [Pan *et al.* 1997]. The two cysteine-rich motifs present in the extracellular region have 58% and 54% homology with DR4 and DR5 respectively. Due to the absence of a cytoplasmic region, the receptor is bound to the cell surface *via* a glycosylphosphatidylinositol (GPI) anchor and its ability to transmit an intracellular signal has yet to be determined. It is, however, able to bind TRAIL at the cell surface, possibly acting as a 'decoy' and inhibiting a response by the death domain-containing receptors DR4 and DR5. DcR1 mRNA is not as widely expressed as DR4 and DR5 mRNA [Pan *et al.* 1997; Sheridan *et al.* 1997; Degli-Esposti *et al.* 1997]. An absence of DcR1 alone does not appear to correlate with sensitivity to TRAIL-induced apoptosis [Bretz, *et al.* 1999; Degli-Esposti 1999].

The fourth receptor, DcR2, was identified from a cDNA library using the sequence of the DcR1 ligand-binding domain as a probe [Degli-Esposti *et al.* 1997; Marsters *et al.* 1997]. DcR2 is a classical type 1 transmembrane protein that shares 58% and 57% homology with DR4 and DR5 and 70% homology with DcR1. All four receptors possess two cysteine motifs within the extracellular region, necessary for ligand binding. DcR2 binds to TRAIL with an extremely high affinity and soluble DcR2 protein is able to block TRAIL-mediated apoptosis in TRAIL-sensitive cell lines [Marsters *et al.* 1997; Truneh *et al.* 2000]. Despite the similarities in ligand binding between TRAIL receptors, DcR2 contains a truncated death domain within the

cytoplasmic region and does not mediate apoptosis [Degli-Esposti *et al.* 1997; Marsters *et al.* 1997]. DcR2 gene expression in normal tissues is as widespread as DR4 and DR5 gene expression. As with DcR1, the presence or absence of DcR2 mRNA does not correlate with sensitivity or resistance of cells to TRAIL-mediated apoptosis [Degli-Esposti 1999; Bretz *et al.* 1999]. Moreover, Zhang and colleagues demonstrated that mRNA expression of DcR2 within the cell did not reflect receptor protein expression at the cell surface [Zhang *et al.* 1999]. Importantly, it was also demonstrated that mRNA transcripts located within the cytoplasm were redistributed to the cell surface upon activation.

The fifth TRAIL receptor, osteoprotegerin (OPG), exists as a secreted protein and has been identified as an antagonist of RANKL-mediated bone resorption [Truneh *et al.* 2000; Emery *et al.* 1998; Degli-Esposti 1999]. Therefore, unlike any of the other TRAIL receptors, OPG lacks a transmembrane domain [Simonet *et al.* 1997]. Reports have also identified that OPG has an affinity for TRAIL and is able to inhibit TRAIL-mediated apoptosis as seen with DcR1 and DcR2 [Truneh *et al.* 2000; Emery *et al.* 1998]. Not only is OPG able to bind to TRAIL but it also binds to an osteoblast differentiation factor RANKL also known as ODF, TRANCE and OPGL, which is involved in regulating bone resorption [Emery *et al.* 1998]. Simonet *et al.* 1997]. It has been shown that TRAIL blocks the anti-osteoclastogenic activity of OPG suggesting a potential cross-regulatory mechanism [Emery *et al.* 1998]. The physiological role of OPG in TRAIL signalling is unclear although a recent report has suggested that cancerderived OPG may be an important survival factor in hormone-resistant prostate cancer cells. This is supported by a reduction of TRAIL-induced apoptosis in cancer cells producing higher than normal levels of OPG [Holen *et al.* 2002].

1.6.2 Molecular mechanisms of TRAIL-induced apoptosis

Apoptosis induced by TRAIL occurs via the extrinsic pathway, as for all members of the TNF superfamily. Ligation of TRAIL to either DR4 or DR5 death receptors results in trimerisation of the receptor and clustering of the receptor's intracellular DD [Ozoren and El-Deiry 2003]. This leads to the recruitment of the adaptor protein Fasassociated death domain (FADD/Mort1), through its own death domain, to the receptorligand complex followed by binding of procaspase-8 to form the DISC [French and Tschopp 1999; Degli-Esposti 1999; Walczak and Krammer 2000; Bodmer et al. 2000; Wang and El-Deiry 2003; Wang et al. 2001; Kischkel et al. 2001]. The formation of the DISC leads to the autocleavage of procaspase-8 and release of active caspase into the cytosol [Medema et al. 1997; Kischkel et al. 2001; Boldin et al. 1996; Juin and Evan 2000; Zhang et al. 1999]. Procaspase-10 has been shown to be recruited to the DISC and become activated, however it remains controversial as to whether caspase-10 can initiate apoptosis in the absence of caspase-8 [Kischkel et al. 2001; Sprick et al. 2002]. Sufficient levels of caspase-8 are required to activate caspase-3 and initiate the caspase cascade [Kim et al. 2000]. Amplification of the apoptotic signal occurs when the extrinsic and intrinsic pathway are linked via caspase-8 cleavage of Bid, leading to its translocation to the mitochondria to promote cytochrome c and Smac/DIABLO release through interaction with Bax and Bak [Deng et al. 2002; Green 2000].

Two different cell types have been identified with respect to the degree of involvement of the intrinsic and/or extrinsic apoptotic pathway, following death receptor signalling. In Type I cells, death receptor ligation is able to recruit and activate sufficient levels of caspase-8 to induce apoptosis, by initiating the caspase cascade, without the need of the mitochondrial intrinsic pathway. On the contrary, type II cells produce insufficient levels of activated caspase-8 and additionally require Bcl-2 and the mitochondrial intrinsic pathway to amplify the apoptotic signal after ligand binding

[Ozoren and El-Deiry 2002]. The difference between the two cell types is based on the levels of caspase-8 recruited by FADD to the DISC [Scaffidi *et al.* 1999; Ozoren and El-Deiry 2002]. The two different cell types have been identified in response to FasL and TRAIL [Burns and El-Deiry 2001; El-Deiry 2001; Ozoren *et al.* 2000].

The DISC can be inhibited by FLICE inhibitory protein (also known as FLIP) [Wang *et al.* 2000; Irmler *et al.* 1997; Kataoka *et al.* 2000]. Inhibition of the DISC by FLIP leads to the activation of signalling pathways such as NF-κB and Erk [Kataoka *et al.* 2000].

1.6.3 TRAIL sensitivity and resistance

By comparison with normal cells, many cancer cells (because some are as resistant as normal cells) are more sensitive to TRAIL-induced apoptosis after TRAIL treatment. A proposed explanation for the observed difference in TRAIL sensitivity is a greater expression of TRAIL death receptors in cancer cells [Zhang *et al.* 1999]. This, however, is not supported by mRNA expression studies of the death receptors, which show them to be widely distributed in both normal and malignant tissues [Chaudhary *et al.* 1997]. The discovery of the decoy receptors offered a possible explanation, however, it was assumed that only normal cells would express surface decoy receptors [Gura 1997]. This assumption proved incorrect and the link between TRAIL receptor expression and TRAIL sensitivity remains unclear [Hersey and Zhang 2001; Ozoren and El-Deiry 2003].

The cellular FLICE-inhibitory proteins (c-FLIPS) have been identified, which are able to negatively regulate death receptor signalling. These proteins have sequence homology to caspase-8 and -10 and can be recruited to the DISC, preventing the interaction between FADD and the procaspases [French and Tschopp 1999; Tschopp *et al.* 1998; Jiang *et al.* 1999; Igney and Krammer 2002; Igney and Krammer 2002].

Multiple splice variants of c-FLIP have been reported, designated c-FLIP_S and c-FLIP_L, representing the short and long forms respectively. The short form of c-FLIP has been identified as conferring TRAIL resistance in a sensitive colon carcinoma cell line [Burns and El-Deiry 2001]. A correlation of high FLIP expression and TRAIL resistance has been found in many cancer cells [French and Tschopp 1999]. In addition, high FLIP expression has also been implicated in enhanced tumour progression [French and Tschopp 1999]. Recent reports suggest that FLIP may promote caspase-8 activation, leaving FLIPs role in apoptosis unclear [Chang *et al.* 2002].

The role of caspases is central to the process of apoptosis and may be a factor in resistance. Individual caspase expression levels may have an impact on overall activity, with frameshift mutations and alternative splicing identified in cancer cell lines [Srinivasula *et al.* 1998]. Caspase mutations are a plausible cause for TRAIL resistance, however detection of such in human cancer is rare [Mandruzzato *et al.* 1997]. It is more probable that other factors hinder caspase function and confer resistance in cancer cells [Baylin and Bestor 2002]. Hypermethylation of regulatory sequences of the caspase-8 gene was found to inhibit caspase-8 expression in numerous tumour cell types both *in vitro* and *in vivo* [Teitz *et al.* 2000; Fulda *et al.* 2001; Baylin and Bestor 2002]. Demethylation treatment or gene transfer experiments to restore caspase-8 expression led to sensitisation of these cells to apoptotic signals.

Mutations in proteins of the Bcl-2 family can occur, such as Bax mutations in mismatch-repair (MMR)-deficient human tumours. Resistance to TRAIL-induced apoptosis can occur due to Bax inactivation, probably by preventing the release of Smac/DIABLO, needed to antagonise the IAP protein family [Deng *et al.* 2002; Burns and El-Deiry 2001]. Recent studies have shown that combination treatment of chemotherapy and TRAIL is able to reverse the resistance to TRAIL [Bouralexis *et al.* 2004; LeBlanc *et al.* 2002; Wu *et al.* 2004; Van Valen *et al.* 2003; Zhu *et al.* 2004].

Evidence suggests that chemotherapy sensitises the cells to the effects of TRAIL by mechanisms such as increased pro-apoptotic protein expression and increased death receptor expression.

Sensitivity to death receptor mediated apoptosis can be regulated by several factors and it is the sum of these factors, which decides between life and death of a cell.

1.6.4 Physiological role of the TRAIL system

The physiological role the TRAIL system is not well understood. Initially it was assumed that the functionally distinct receptors could explain the role of the TRAIL system and the apparent differences in sensitivity to TRAIL between normal and transformed cells. The competing 'decoy' and death receptors provided a simple explanation to the observed phenomenon. However the levels of TRAIL receptor expression do not always correlate with resistance or sensitivity to TRAIL-induced apoptosis suggesting the influence of other factors [Degli-Esposti 1999]. The cell surface expression of TRAIL has been discovered for cell types that had previously been known to induce apoptosis in target cells via an unknown mechanism. These include type II interferon (IFNy)-stimulated monocytes [Griffith et al. 1999], type I interferon (INF α and IFN β) [Walczak and Krammer 2000], TCR stimulated T cells [Kayagaki et al. 1999; Kayagaki et al. 1999], cytomegalovirus (CMV)-infected fibroblasts [Sedger et al. 1999], non-stimulated CD4+ T cells [Thomas and Hersey 1998; Mariani and Krammer 1998; Kayagaki et al. 1999], lipopolysaccharide activated monocytes and macrophages [Halaas et al. 2000], INF α and IFN γ - stimulated as well as measles virus-infected dendritic cells [Vidalain et al. 2000; Fanger et al. 1999] and natural killer (NK) cells [Kashii et al. 1999; Johnsen et al. 1999; Zamai et al. 1998; Kayagaki et al. 1999]. Interferon stimulation has been associated with surface

37
expression of TRAIL such as in human lymphoid cells, therefore, IFNs may play a role in mediating death in TRAIL-sensitive tumour cells [Kayagaki et al. 1999; Griffith et al. 1999]. The antiviral activity of IFN has been reported to induce TRAIL-mediated apoptosis in virally infected cells but not normal cells [Sedger et al. 1999]. Therefore the TRAIL apoptotic system may be important in controlling virally transformed cells. The expression of TRAIL, together with other apoptosis-related cytokines such as FasL, TNF and lymphotoxin- α (LT- α) by NK cells, has led to the speculation that TRAIL is one apoptotic component of many, aiding NK cells in their pursuit of defective cells for elimination [Kashii et al. 1999]. Other than its involvement in apoptosis, TRAIL has also been identified as an inhibitory protein in auto-immune inflammation by blocking cell cycle progression in rheumatoid arthritis [Song et al. 2000]. The absence of TRAIL led to hyper-proliferation of cells and increased expression of cytokines and autoantibodies whereas treatment with TRAIL prevented cell cycle progression [Song et al. 2000]. A role for TRAIL in the immune privilege of uteri was also proposed, with the discovery of TRAIL expression in human placentas [Phillips et al. 1999]. The constitutive expression of TRAIL on murine NK cells in liver was recently reported and may play a role in prevention against tumour metastasis [Hayakawa et al. 2001]. These studies suggest that the TRAIL ligand-receptor system is critical in the destruction of virally infected cells, inhibition of auto-immune inflammation and surveillance of malignant transformations.

The differing activities of TRAIL further emphasise its complexity and continued studies will provide a better understanding of the TRAIL ligand-receptor system and hopefully identify the mechanisms that regulate tissue homeostasis.

1.6.5 Potential for cancer therapy

The preferential activity of TRAIL against cancer cells makes it potentially ideal as a cancer therapeutic alone or in combination with chemotherapy and radiation therapy. Normal cells appear to be resistant to TRAIL, which makes the use of TRAIL therapy more appealing [Ashkenazi 2002; Atkins *et al.* 2002; Evdokiou *et al.* 2002]. Most importantly, TRAIL-induced apoptosis is independent of p53 status, which makes it potentially effective against tumours that are resistant to conventional chemotherapy [El-Deiry 2001].

TRAIL, in its soluble recombinant form, induces apoptosis in a number of transformed cell lines derived from leukaemia, multiple myeloma, neuroblastoma, and cancers of the colon, lung, breast, prostate, pancreas, kidney, central nervous system and thyroid [Yu *et al.* 2000; Rieger *et al.* 1998; Keane *et al.* 1999; Mitsiades *et al.* 2001; Mizutani *et al.* 2001; Gazitt 1999; Walczak *et al.* 1999]. More importantly, the use of TRAIL *in vivo* has also been successful. Mice bearing solid tumours showed increased tumour cell apoptosis, suppressed tumour progression and improved survival, without any normal tissue toxicity, after the administration of soluble TRAIL [Ashkenazi *et al.* 1999; Walczak *et al.* 1999]. Significant anti-tumour activity without systemic toxicity has been reported for many human xenograft models in mice derived from multiple myeloma, malignant glioma, and cancers of the breast and colon [Pollack *et al.* 2001; Roth *et al.* 1999; Mitsiades *et al.* 2001; Kelley *et al.* 2001]. Additionally, TRAIL was also shown to inhibit growth of tumours immediately after implantation, as well as inducing apoptosis in established tumours.

The use of an alternative delivery system has been explored by means of adenoviral gene therapy. This method has been used in both *in vitro* and *in vivo* experiments with promising results [Kagawa *et al.* 2001; Griffith and Broghammer 2001; Griffith *et al.* 2000]. Anti-tumour activity of TRAIL gene therapy has been

verified in cancers such as colon, breast, prostate, glioblastoma and hepatocellular carcinoma [Yamashita *et al.* 2002; Lee *et al.* 2002; Norris *et al.* 2001].

To date, four recombinant versions of human TRAIL have been generated. One of the first versions generated contains TRAIL amino acids 114-281 fused to an aminoterminal polyhistidine tag [Pitti *et al.* 1996]. To promote and stabilise the formation of ligand trimers, the second version generated contains an amino-terminal modified yeast Gal4 leucine zipper fused to residues 95-281 [Walczak *et al.* 1999]. The third variant contains the TRAIL amino acids 95-281 fused to an amino-terminal 'Flag' epitope tag. In some cell lines the use of anti-flag antibodies cross-link with this version of TRAIL and improve its activity [Bodmer *et al.* 2000]. The fourth version generated has potential to be the most clinically favoured because it contains amino acids 114-281 of human TRAIL but lacks any exogenous sequence and therefore is the least probable to be immunogenic in humans [Ashkenazi and Dixit 1999]. Non-tagged, recombinant soluble TRAIL is produced by bacteria at a neutral pH, as a stable homotrimer, optimised by the addition of zinc and reducing agent to the culture media [Kelley *et al.* 2001; Lawrence *et al.* 2001].

The use of the non-tagged, zinc-bound version of TRAIL is reported not harmful to most normal human cells, with some cells including astrocytes, hepatocytes and keratinocytes sensitive to the tagged, non-optimised versions of TRAIL [Ashkenazi *et al.* 1999; Walczak *et al.* 1999; Lawrence *et al.* 2001; Jo *et al.* 2000; Qin *et al.* 2001]. The observed difference may be determined by the amount of multimerisation of the death receptors DR4 and DR5. The tagged versions of TRAIL, which lack a zinc atom, have low solubility and aggregate or precipitate at high concentrations. This is also seen with the antibody cross-linked version of TRAIL. This may result in over multimerisation of the death receptors resulting in an overwhelming death signal and eventual apoptosis. Conversely, the non tagged, zinc-bound version of TRAIL is highly

stable and soluble as a trimer, forming only trimeric death receptor multimerisation complexes, which are well below the apoptotic threshold in normal cells, but sufficient to induce apoptosis in cancer cells due to a lowered threshold [Evan and Littlewood 1998; Jo *et al.* 2000; Lawrence *et al.* 2001; Ashkenazi *et al.* 1999]. This hypothesis is also supported by the observation that unlike the poly-histidine tagged version of TRAIL, which binds irreversibly to the death receptors, the untagged, zinc-bound version of TRAIL has reversible binding [Jo *et al.* 2000; Lawrence *et al.* 2001].

Studies in mice show no obvious toxicity to various versions of recombinant TRAIL, however mice are more resistant to toxicity than primates [Walczak *et al.* 1999]. Non-human primates, were used to determine safety of the untagged zinc- bound TRAIL. Cynomolgus monkeys and chimpanzees were intravenously injected with TRAIL with varying doses over 7 days, with no toxicity observed [Ashkenazi *et al.* 1999; Lawrence *et al.* 2001; Kelley *et al.* 2001]. More importantly, there was no evidence of hepatotoxicity, suggesting that this version of the recombinant molecule may be the best candidate for clinical assessment.

1.6.6 Combined treatment with TRAIL and other agents

Response to cancer therapy is determined by the induction of apoptosis, which is limited by lack of selectivity in the currently available apoptosis-inducing agents. In addition, most cancers are resistant to growth arrest and apoptosis. Current standard cancer therapy is restricted by its toxicity to normal cells and the consequent side effects observed. It is therefore central to the application of therapeutics that the apoptosisinducing agents given to patients are effective and selective at killing cancer cells, with normal cells spared this fate. The approaches for therapy can range from selective inhibition of anti-apoptotic pathways, anti-angiogenic therapy, tissue-specific therapy such as immunotherapy, and the exploitation of cancer specific features such as

unrestricted cell cycles and hypoxicity. To ensure both effective and selective treatment, combinations of drugs are proving useful by providing multiple apoptotic signals *via* different pathways. Examples of this include the use of anti-apoptotic kinases, such as inhibitors of Raf-1, MEK and Akt, to increase sensitivity to chemotherapy [Dancey and Sausville 2003; Solit *et al.* 2003; Yu *et al.* 2001; MacKeigan *et al.* 2000]. This combines activators of both the extrinsic and intrinsic pathway, leading to caspase activation [Milner *et al.* 2002] and combination therapy of anticancer agents that potentiate the apoptotic signal.

In an effort to overcome resistance of cancer cells to TRAIL, combination therapy is proving to be valuable experimentally. Combination therapy can result in either a synergistic, additive or antagonistic apoptotic effect. A combination of agents that is more effective than the sum of both agents is said to show synergy. Additivism is no more and no less effective than expected and antagonism is a combination that is less effective than expected [Berenbaum 1977].

Synergistic combinations usually consist of one agent that sensitises the cell to the cytotoxic effects of the second agent. This may be by blocking anti-apoptotic pathways, such as down regulating expression of FLIP [Sayers *et al.* 2003], or the use of DNA damaging drugs, which induce the mitochondrial release of inhibitors of IAP to reactivate caspase-3 [Leverkus *et al.* 2003]. The use of synergistic combinations may also cause unwanted synergy in normal cells, which may be translated into side effects in the clinic. It is therefore crucial to find combinations that are selective, to minimise the effects on normal cells.

The use of recombinant soluble TRAIL in combination with other agents is being actively investigated in the field of cancer therapy [Nagane *et al.* 2001]. Studies in OS, Ewing's sarcoma, leukaemia, colorectal, multiple myeloma, neuroblastoma, melanoma, breast, bladder, glioma, liver, lung, ovarian, pancreatic, renal and thyroid cancer have

shown that combination treatment between TRAIL and chemotherapeutic drugs or radiation, results in greater tumour cell death than standard chemotherapy alone [Ahmad and Shi 2000; Bouralexis *et al.* 2004; Cuello *et al.* 2001; Chinnaiyan *et al.* 2000; Dejosez *et al.* 2000; Eggert *et al.* 2001; Evdokiou *et al.* 2002; Gliniak and Le 1999; Gong and Almasan 2000; Griffith *et al.* 1998; Jazirehi *et al.* 2001; Keane *et al.* 1999; Lacour *et al.* 2001; Mizutani *et al.* 2001; Nagane *et al.* 2000; Ng *et al.* 2002; Rohn *et al.* 2001; Sun *et al.* 2000; Van Valen *et al.* 2000; Wu *et al.* 2002; Yamanaka *et al.* 2000]. Chemotherapeutic drugs or radiation may activate an apoptotic pathway, which is different from the apoptotic pathway initiated by TRAIL, therefore generating a more potent apoptotic signal.

It has been observed that DNA damage and radiation causes upregulation of DR5 transcription and therefore may enhance the cytotoxicity of TRAIL through upregulation of its death receptor DR5. [Chinnaiyan *et al.* 2000; Gong and Almasan 2000; Sheikh *et al.* 1998; Wu *et al.* 1997]. Cotreatment of TRAIL with chemotherapeutic agents cisplatin and etoposide in malignant gliomas, resulted in an augmented apoptotic effect due to increased expression of DR5 [Nagane *et al.* 2000]. In vitro experiments using sublethal doses, which are commonly used for the treatment of malignant glioma in humans, caused upregulation of DR5 transcripts but did not induce apoptosis in most gliomas tested. Treatment with both TRAIL in combination with DNA damaging drugs caused synergistic apoptosis *in vitro* as well as growth suppression and even regression in established human glioblastoma xenografts in nude mice [Nagane *et al.* 2000]. Conversely, single treatments were ineffective, supporting the usefulness of combination therapy for gliomas and even TRAIL resistant therapy. Upregulation of DR4 has also been reported in response to DNA damaging agents [Guan *et al.* 2001]. There has been a recent suggestion that the synergy between TRAIL

and chemotherapeutic agents may be independent of DR4 and DR5 upregulation and that other mechanisms may be involved [Rohn *et al.* 2001].

Although most studies have been performed with cell lines, the effectiveness of TRAIL and chemotherapeutic drug combination treatments *in vitro* has also been shown to reflect sensitivity *in vivo*. Chemotherapeutic drugs such as doxorubicin, 5-flouracil and camptothecin augment TRAIL cytotoxicity both *in vitro* and *in vivo* for colon, breast and bladder cancer [Keane *et al.* 1999; Gliniak and Le 1999; Bonavida *et al.* 1999].

Therefore although the mechanisms are not clearly defined, combination treatment of TRAIL with standard chemotherapeutic agents is able to synergistically augment cell death *in vivo*, providing an alternative strategy for the treatment of human cancers. Ideally, a combination of TRAIL and another less toxic agent, in lieu of chemotherapy, may prove far superior in cancer treatment by reducing the well-known side effects associated with current chemotherapeutic agents. Scientists are in pursuit of such a new agent with many possible candidates currently being explored for future management strategies.

1.7 BISPHOSPHONATES

1.7.1 General properties

Bisphosphonates (BPs) are synthetic analogues of inorganic pyrophosphate, commonly used for the prevention and treatment of various bone diseases associated with enhanced bone resorption, such as Paget's disease and osteoporosis. First synthesised in 1865, BPs were found to inhibit precipitation of calcium carbonate and were mainly used as anti-corrosive and anti-scaling agents for industrial purposes. It was not until 1968 that the biological properties of BPs were first reported by Fleisch

44

and his group [Fleisch *et al.* 1968]. BPs are all characterised by a phosphorous-carbonphosphorous group [Fleisch *et al.* 1968; Fleisch 2000]. They are similar to pyrophosphates but have a carbon atom replacing the central oxygen atom, thus allowing the binding of two extra side chains, R1 and R2 (see **Figure 1.9**). Structural changes of BPs occur at these side chains of the central carbon atom and even minor alterations can result in significant differences in the physicochemical and biological properties of the compound [Fleisch 2000; Shinoda *et al.* 1983]. The three-dimensional structure of BPs allows them to bind to divalent metal ions such as Ca^{2+} , Mg^{2+} and Fe^{2+} *via* the oxygen atom from each phosphate group, forming insoluble complexes [Ebetino *et al.* 1998]. If the R1 side chain is a hydroxyl (OH) or an amino group (NH₂) then the affinity for Ca^{2+} increases due to changes in structure and improved binding [Jung *et al.* 1973]. The highly selective localisation and retention of BPs in bone is the basis for its use in skeletal disorders. The degree of inhibition of bone resorption varies greatly depending on the structure of the BP.

BPs are divided into three main groups, as shown in **Table 1.2**: (i) The first generation group of compounds, such as etidronate and clodronate, is the non-nitrogen containing group of BPs and these are the weakest in their resorptive inhibitory function. They are structurally similar to ATP and are incorporated into metabolic pathways as non-hydrolyzable analogues of ATP, thereby inhibiting ATP-dependent function. An important example of this is the vacuolar H+-ATPase proton extrusion pump, which is essential for osteoclast function [Blair *et al.* 1989]; (ii) the second group, which includes pamidronate and alendronate, contain a nitrogen atom (N-BPs) on the side chain, which strongly increases BP activity [Rogers *et al.* 2000]; (iii) the third and most active group of N-BPs contain a cyclic substituent and either one nitrogen atom in the pyridyl ring, such as risedronate, or two nitrogen atoms in an imidazole ring, such as zoledronic acid (**Figure 1.10**) [Rogers *et al.* 2000]. This group



BISPHOSPHONATES





Figure 1.9 The general structures of bisphosphonates compared to pyrophosphate.

BISPHOSPHONATE CLASS	MECHANISM OF ACTION	STRUCTURE	POTENCY	
Non-nitrogen-containing * First generation BPs -etidronate, clodronate	Cytotoxic, non-hydrolyzable analogues of ATP	Side chains la Aliphatic side single N atom	cking N atom chain with	+ ++
Nitrogen-containing * Second generation BPs - pamidronate, alendronate - risedronate	Inhibit protein prenylation	Heterocyclic r single N atom	ing with	+++
* Third generation BPs -zoledronic acid		Imidazole ring N atoms	, with two	++++

Table 1.2 The evolution of Bisphosphonates. Bisphosphonates are divided into three groups based on their mechanism of action and their structure.

.

8 3 2 3



Figure 1.10 Chemical structure of zoledronic acid (2-[imidazol-1-yl]-1-hydroxyethylidene-1,1-bisphosphonic acid).

of N-BPs are up to 10,000 fold more potent than the non-nitrogen containing BPs, with respect to their ability to inhibit bone resorption *in vitro* [Green *et al.* 1994; Fleisch 2000]. Zoledronic is the most potent inhibitor of bone resorption of the currently available BPs [Green *et al.* 1994]. Studies have shown that ZOL preserves bone mass, structure and strength in eostrogen-deficient animals without affecting bone mineralisation [Binkley *et al.* 1998; Bare *et al.* 1997]. Already used in the treatment of bone disease such as Paget's disease, ZOL also shows exciting potential as an anticancer agent, as will be discussed throughout the thesis. Significantly, prolonged treatment with zoledronic acid seems to be safe and well tolerated [Ali *et al.* 2001; Rosen *et al.* 2004].

1.7.2 The effect of bisphosphonates on bone cells

Bone remodelling by osteoblasts and osteoclasts is an ongoing process in adult bones. This allows the maintenance of the mechanical integrity of the skeleton and is also involved in calcium homeostasis [Rodan 1997]. The normal remodelling rate in a human adult is between 2 and 10 % of the skeletal mass per year. The process of remodelling starts with the recruitment of osteoclasts to a particular site, where they erode the bone to form a resorption cavity. This is tightly coupled with the formation of new bone by the osteoblasts, over the resorption cavity [Manolagas and Jilka 1995; Rodan 1996].

Bone destruction is a common manifestation associated with skeletal malignancy and is mediated primarily by the osteoclasts and their bone resorbing activity [Goltzman 2001; Taube *et al.* 1994]. The initial appearance of malignant cells within the bone microenvironment leads to the release of numerous agents produced directly by the tumour cells including hormones, growth factors and cytokines, which act on osteoblastic stromal cells to enhance the production of osteoclastic stimulating factors.

This leads to increased osteoclast number and increased bone resorption. Osteoclasts resorb bone by secreting proteases that dissolve the bone extracellular matrix and release bone mineral and growth factors [Chirgwin and Guise 2000]. Bone resorption occurs at the extracellular space/bone interface, under the ruffled border of the osteoclast's plasma membrane [Blair *et al.* 1989]. The factors released from the bone are able to stimulate tumour growth resulting in the establishment of a mutually beneficial relationship, often termed "the vicious cycle" due to its progressively destructive nature.

Due to their safety and efficacy, BPs have been used extensively over the years for the treatment of many diseases characterised by increased bone resorption [Body 2000]. BPs preferentially concentrate to bone surfaces that are actively remodelling, at the bone/osteoclast interface, where bone mineral is most exposed [Masarachia *et al.* 1996]. During acidification of bone mineral by osteoclasts, BP is released and is internalised by osteoclasts [Dunford *et al.* 2001]. Following internalisation, the primary action of BPs is to inhibit osteoclast activity.

The action of BPs on the osteoclasts involves various processes; reduction in osteoclast activity and a reduction in osteoclast number either by apoptosis or *via* decreased recruitment [Hughes *et al.* 1995]. Although several studies have shown that BPs to induce osteoclast apoptosis, it appears that the first effect is to reduce osteoclast activity, resulting in reduced bone resorption [Hughes *et al.* 1995; Coxon *et al.* 2000; Rogers *et al.* 2000]. The effect of BPs on osteoclast function is due to loss of ruffled border, altered actin cytoskeleton, decreased acid production, decreased enzyme activity, decreased protein prenylation, disruption of integrin-induced intracellular signalling and altered trafficking of membranes and intracellular molecules [Hughes *et al.* 1995; Coxon *et al.* 2000; Rogers *et al.* 2000]. Indirect effects *via* the osteoblasts also affect osteoclast activity.

Osteoblasts are key regulators in bone cell differentiation and function. BPs have been shown to directly influence osteoblast function [Giuliani et al. 1998; Garcia-Moreno et al. 1998; Khokher and Dandona 1989]. Recent investigations have reported BPs to cause a stimulatory effect on osteoblast proliferation [Giuliani et al. 1998; Fromigue and Body 2002; Viereck et al. 2002; Tsuchimoto et al. 1994]. The anabolic effects on osteoblasts are dose dependent with higher doses causing inhibition and even cell death. Maturation of osteoblasts has been reported after BP treatment, with an earlier commitment towards the osteoblastic phenotype [Im et al. 2004; Fromigue and Body 2002]. Evidence from in vitro and in vivo studies confirm that BPs inhibit osteoclastogenesis and the recruitment of osteoclast progenitors into bone [Sahni et al. 1993; Clohisy et al. 2001; Van Beek et al. 2002]. BPs can also influence changes in osteoblastic gene expression and the secretion of soluble factors involved in regulating osteoclast differentiation and activity [Giuliani et al. 1998; Reinholz et al. 2000; Tokuda et al. 1998; Vitte et al. 1996]. One example is increased osteoprotegerin (OPG) production in osteoblasts treated with BPs [Pan et al. 2004; Viereck et al. 2002]. Osteoclast development is regulated by the receptor activator NF- κ B ligand (RANKL), produced by bone marrow stromal cells and osteoblasts [Lacey et al. 1998]. OPG is an antagonist soluble decoy receptor that binds RANKL [Grimaud et al. 2001]. The ratio between RANKL and OPG regulates osteoclastogenesis, therefore disruption of this balance by BPs may lead to reduced bone resorption. Overall, BPs act via multiple mechanisms to reduce osteoclast number and activity and thus inhibit bone resorption.

1.7.3 Molecular mechanism of nitrogen-containing bisphosphonates

The newer N-BPs inhibit farnesyl pyrophosphate synthase, a key enzyme of the mevalonate pathway, by blocking the pyrophosphate Mg^{2+} binding site of the enzyme [Luckman *et al.* 1998; Coxon *et al.* 2000; Dunford *et al.* 2001; Fisher *et al.* 2000; Rodan

1998; Rogers *et al.* 2000]. The mevalonate pathway produces various end products that are important for many different cellular functions. These include the production of isoprene units incorporated into sterols such as cholesterol, essential in membrane integrity; steroid production; ubiquinone, for electron transport and cell respiration; farnesyl and geranylgeranyl isoprenoids, involved in the post-translational modification of proteins; dolichol, for glycoprotein synthesis; and isopentenyladenine, required for tRNA function and protein synthesis [Goldstein and Brown 1990; Kabakoff *et al.* 1990; Edwards and Ericsson 1998; Wong *et al.* 2002].

The post-translational lipid modification process that occurs as part of the mevalonate pathway is called prenylation. More specifically, prenylation is divided into geranylgeranylation and farnesylation, which involves the covalent attachment of hydrophobic geranylgeranyl (20 carbon) and farnesyl (15 carbon) prenyl lipids, to a conserved C-terminal cysteine residue contained within prenylation motifs on the target proteins [Casey 1992; Zhang and Casey 1996]. These reactions are specifically catalysed by geranylgeranyl transferase and farnesyl transferase, respectively (shown in **Figure 1.11**). The GTPase family consists of over 100 proteins such as Ras, Rho, Rac and Rab, which are major substrates for prenylation and are involved in a large variety of cellular processes such as cytoskeletal organisation, nucleocytoplasmic transport, cell attachment, integrin signalling and microtubule organisation [Edwards and Ericsson 1998; Ridley *et al.* 1992; Ridley and Hall 1992; Takai *et al.* 2001; Zhang and Casey 1996]. N- BPs prevent prenylation by inhibiting the enzyme farnesyl pyrophosphate synthase, which results in depletion of farnesyl pyrophosphate and geranylgeranyl pyrophosphate, utilised for prenylation [Dunford *et al.* 2001].

The role of GTPases in osteoclasts has not been widely investigated and although many GTPases are involved in general cell maintenance, cell type specific GTPases have also been identified [Lutcke *et al.* 1993]. It is possible that specific GTPases,



Figure 1.11 The mevalonate pathway. Nitrogen-containing bisphosphonates inhibit the enzyme as indicated above, preventing downstream processes including farnesylation and geranylgeranylation.

1.128

linked with the actin ring and ruffled border, unique to osteoclasts, may exist. Many GTPases have been identified in playing a role in the regulation of osteoclast activity, such as Rab7 involvement in bone resorption. It is therefore not surprising that inhibition of prenylation by N-BPs induces multiple negative effects on osteoclast activity [Fisher *et al.* 1999; Coxon and Rogers 2003; Zhao *et al.* 2001].

1.7.4 Anti-tumour effects of bisphosphonates

The use of BPs as inhibitors of bone resorption has been long known, however it was only until fairly recently that the anti-tumour potential of BPs was discovered. Shipman and his colleagues were the first to demonstrate the anti-cancer effects of N-BPs in myeloma cells [Shipman et al. 1997; Shipman et al. 1998; Shipman et al. 1998]. The mechanism of apoptosis in the malignant cells was confirmed to be the same as that seen in the osteoclasts, that is, inhibition of the mevalonate pathway leading to loss of important signalling proteins [Shipman et al. 1998; Shipman et al. 2000]. N-BPs clearly decreased myeloma cell proliferation and induced apoptosis, whereas the non-nitrogen containing BPs had little or no effect [Shipman et al. 1998]. Consistent with the proposed mechanism of N-BP action, the addition of geranylgeraniol and farnesol, both intermediates of the mevalonate pathway, prevented N-BP-induced apoptosis and partially reversed cytostasis. A wide variety of tumour cell lines have also been used to demonstrate the anti-tumour activity of N-BPs, including Ewings sarcoma [Sonnemann et al. 2003], leukaemia [Kimura et al. 2004], melanoma [Riebeling et al. 2002], breast cancer [Senaratne and Colston 2002], ovarian cancer [Sawada et al. 2002], prostate cancer [Lee et al. 2001], pancreatic cancer [Tassone et al. 2003] and OS [Evdokiou et al. 2003; Farese et al. 2004]. These in vitro studies have shown N-BPs to dosedependently inhibit proliferation and induce apoptosis in tumour cell lines. The concentrations of N-BPs used to produce these cytostatic and apoptotic effects in vitro

were in the range of 5-2000 μ M. This concentration range is within that achieved in bone at an active resorption site [Sato *et al.* 1991]. The potency of the N-BP and the cell line used, will determine the concentration of N-BP required to induce apoptosis [Fromigue *et al.* 2000].

Tumour cell adhesion and invasion are necessary for effective spreading of tumour cells. Van der Pluijm showed that N-BPs reduce tumour burden in bone by inhibiting breast cancer cell adhesion [van der Pluijm *et al.* 1996]. This suggested a preventative role for N-BPs in deterring tumour cells away from the bone environment, by preventing attachment. Reduced migration has been reported in other cancer cell types pretreated with low levels of N-BPs that did not induce apoptosis [Boissier *et al.* 2000]. BPs inhibit several enzymes required for tumour cell invasion *in vitro*, such as the matrix metalloproteinase enzymes (MMP-1, -2, -3, -7, -8, -9, -12, -13, -14) [Boissier *et al.* 2000; Heikkila *et al.* 2002; Teronen *et al.* 1999]. Investigation of MMP activity by several groups have produced conflicting results, which may be due to the varying N-BPs and cell lines used.

Angiogenesis is essential for the growth and survival of metastatic tumours, to ensure vascularisation and access to nutrient rich blood supply. *In vitro* experiments using zoledronic acid have shown clear inhibition of cell proliferation induced by growth factors bFGF and VEGF or serum [Wood *et al.* 2002]. Subsequent animal studies revealed that when zoledronic acid was administered systemically to mice, it potently inhibited angiogenesis induced by subcutaneous implants impregnated with bFGF [Wood *et al.* 2002]. The anti-angiogenic properties of N-BPs, specifically zoledronic acid, have also been reported in other *in vivo* models. 5T2 myeloma-bearing mice treated with zoledronic acid showed decreased tumour burden and reduced angiogenesis, evident by a marked decrease in micro-vessel density within the myeloma bone lesions [Croucher *et al.* 2003]. The additional ability of N-BPs to inhibit

angiogenesis makes them very appealing for the preventative treatment of metastatic cancers.

In vivo studies have clearly demonstrated a reduction in tumour-induced osteolysis, using BP treatment in numerous animal models of metastatic cancer including breast [Hall and Stoica 1994; Sasaki *et al.* 1995; Yoneda *et al.* 2000; Lipton 2000; Hiraga *et al.* 2001; Peyruchaud *et al.* 2001], prostate [Pollard and Luckert 1985; Pollard and Luckert 1986; Nemoto *et al.* 1988; Shevrin *et al.* 1991; Corey *et al.* 2003], bladder [Nemoto *et al.* 1991; Nemoto *et al.* 1992; Nemoto *et al.* 1993] and multiple myeloma [Croucher *et al.* 2003; Cruz *et al.* 2001; Dallas *et al.* 1999]. Within the bone microenvironment BPs work to inhibit bone resorption therefore decreasing the release of tumour-promoting growth factors from bone and thus hindering the progression of bone metastases. Not only is tumour-induced osteolysis decreased, but also the number and size of tumours within bone.

There is ongoing debate as to whether the inhibition of osteolysis seen with BP treatment is due to a direct anti-tumour effect *in vivo* or whether the BP induces a change in the bone environment that makes it less conducive to tumour cell growth. Supportive, but inconclusive evidence it plentiful. This includes evidence showing increased apoptosis of tumour cells within osteolytic lesions of mice [Yaccoby *et al.* 2002; Hiraga *et al.* 2001]; lowered serum levels of tumour markers in tumour-bearing animals treated with BPs [Yaccoby *et al.* 2002; Corey *et al.* 2003]; and modest increases in survival time [Sasaki *et al.* 1995; Cruz *et al.* 2001]. The effect of BPs on cancer cells in non-bone sites is unlikely, due to the chemical properties of the compound. BP concentrations can reach an adequate level in bone that may cause apoptosis of cancer cells, however such concentrations are not achievable in visceral organs *in vivo* [Fleisch 2000]. Although BPs cause few adverse effects in non-bone organs, changes in visceral metastases after BP treatment in tumour bearing animals have been reported. Results

have been contradictory, with trends towards either a decrease [Alvarez et al. 2003; Nobuyuki et al. 2001; Sasaki et al. 1998] or an increase in visceral metastases [Stearns and Wang 1996; Sasaki et al. 1998; Sasaki et al. 1995; Cruz et al. 2001]. This was despite a marked decrease in new bone metastases [Yoneda et al. 2000; Michigami et al. 2002]. One report noted an increase in bone metastases after BP treatment [Kostenuik et al. 1993]. These inconsistencies may be due to unusual characteristics of some animal tumour models and cannot be reproduced consistently. One study, using the MDA-MB-231 breast cancer animal model, resulted in marked inhibition of bone metastases with ibandronate treatment, but an increase in adrenal metastases [Michigami et al. 2002]. No increase in adrenal metastases was observed with coadministration of the chemotherapeutic agent doxorubicin with ibandronate [Michigami et al. 2002]. Cancer patients who receive BPs are also treated with anticancer agents, however it still remains possible that preventative BP therapy may have adverse effects on visceral metastases. It is imperative that the risk of visceral metastases due to preventative BP therapy be thoroughly investigated. An increased risk of non-osseous metastases may far outweigh the prophylactic benefit currently observed.

1.7.5 Clinical use of bisphosphonates as cancer therapy

The unique properties of the nitrogen containing BPs include the selective accumulation in bone, the suppression of osteoclastogenesis and the promotion of osteoclast apoptosis, which all lead to inhibition of bone resorption. For this reason, BPs are used for the treatment of cancer-related bone disease particularly for bone metastases, bone pain and hypercalcemia. Cancers frequently associated with metastatic bone disease include advanced, breast cancer [Rosen *et al.* 2003; Lipton 2000], prostate cancer [Saad *et al.* 2002; Adami 1997], lung cancer [Rosen *et al.* 2003] and multiple myeloma [Rosen *et al.* 2003; Sezer *et al.* 2003; Bruno *et al.* 2004; Berenson *et al.*

2002]. From the diagnosis of bone metastases, patients experience severe bone pain often associated with increased skeletal morbidity, rendering them very susceptible to fractures that often require surgery [Martin and Moseley 2000]. BP treatment is the current standard practice for palliative treatment of bone metastases. Clodronate and the more potent N-BPs, including pamidronate and zoledronic acid, are commonly used [Body 2003]. A large number of clinical trials have assessed the clinical benefit of BPs in patients with bone metastases. These studies have measured the occurrence of skeletal-related events (SREs), which are defined as pathologic fracture, spinal cord compression, bone pain that required palliative radiation therapy, bone surgery, or hypercalcemia of malignancy.

The N-BPs pamidronate and zoledronic acid have proven to be more effective in large non-randomised clinical trials than the non-N-BP, clodronate. However only zoledronic acid has demonstrated significant benefits in patients with prostate cancer [Rogers *et al.* 2000; Coleman and Seaman 2001]. Trials using breast cancer patients with at least one bone lesion indicate that treatment with 4 mg of zoledronic acid was more effective than 90 mg of pamidronate in reducing skeletal complications [Rosen *et al.* 2004]. The use of zoledronic acid in four randomised, controlled multicentre, phase III trials involving over 3,000 patients, demonstrated significant improvement in patients with bone lesions secondary to breast cancer, multiple myeloma, prostate cancer, lung cancer and other solid tumours [Rosen *et al.* 2003; Berenson *et al.* 2001; Rosen *et al.* 2003]. Zoledronic acid is the first BP to show significant clinical benefit in patients with bone metastases from various primary tumours and therefore seems far superior to other BPs.

1.7.6 Combined treatment with bisphosphonates and other agents

Until recently, most *in vitro* and *in vivo* experiments were designed to assess to the anti-cancer activity of BPs as a single agent. This is unlike the clinical scenario, where BPs are given to patients with metastatic bone lesions as adjuvant therapy with conventional antineoplastic agents [Body *et al.* 1998; Diel *et al.* 1998]. In 1996, Sterns and Wang found that the combination of alendronate and paclitaxel showed enhanced anti-cancer activity in tumour bearing mice models of human prostate cancer. Alendronate alone was found to suppress metastases to the bone with no effect on visceral metastases. However, combination of alendronate with paclitaxel not only prevented metastases to the bone, but also prevented non-osseous metastases, resulting in increased survival in the combination treated mice [Stearns and Wang 1996].

In two murine models of bone metastases using breast cancer cells, mice treated with either ibandronate or zoledronic acid and UFT (a pro-drug of flourouracil) resulted in a reduction in bone metastases, and lung and liver metastases and increased survival in the combination treated mice [Yoneda *et al.* 1999]. A reduction in both bone and visceral metastases has also been observed with minodronate and etoposide dual treatment in mice bearing small-cell lung tumours [Yano *et al.* 2003]. More importantly, the combination treatment prolonged the survival of the cancer-bearing mice. Zoledronic acid and Glivec also showed a significant improvement in survival in a mouse model of leukaemia [Kuroda *et al.* 2003].

Enhanced anti-tumour behaviour *in vitro* was also demonstrated using a combination of ibandronate and taxoid. Invasion and migration assays of breast cancer cells treated with this combination resulted in a dose dependent additive effect on inhibition of cell adhesion and cell migration [Magnetto *et al.* 1999].

In addition, experiments have revealed that the combination of zoledronic acid, with the anti-cancer agents dexamethasone and paclitaxel, leads to clear synergistic

apoptotic effects in myeloma and breast cancer cell lines [Tassone et al. 2000; Jagdev et al. 2001].

Farnesyl transferase inhibitors (FTIs) act by inhibiting farnesyl transferase, the enzyme required for the covalent attachment of a farnesyl lipid to proteins. Some FTIs are also able to inhibit geranylgeranyl transferase but their potency varies depending on the FTI. The use of multiple inhibitors of the mevalonate pathway, specifically FTIs with N-BPs, have led to synergistic anti-tumour activity on *in vitro* murine carcinoma cell invasiveness and *in vivo* metastasis in a murine model of spontaneous lung metastases, supporting the use of this combination for the prevention of metastatic progression [Andela *et al.* 2002]. The limitations of N-BPs for the prevention of metastases, is that it is limited to the skeleton. Use of combination treatment including FTIs, potentially extends its usefulness in the clinical setting to include prevention of visceral metastases.

Specific resistance to N-BPs has been reported in myeloma cell lines after continuous exposure to alendronate for 6 weeks [Salomo *et al.* 2003]. This was due to increased activity of farnesyl pyrophosphate synthase, the target enzyme inhibited by N-BPs, without upregulation of gene transcription suggesting post-transcriptional regulation. The development of resistance is unlikely to occur *in vivo*, as the levels of N-BPs achievable would be the highest at the osteoclast/resorption site interface. More importantly, myeloma cells remained sensitive to conventional anti-neoplastic agents used for the treatment of myeloma thus not compromising N-BP treatment [Salomo *et al.* 2003].

It is apparent from the numerous independent reports that the synergistic and additive behaviour of BPs in combination with other agents is far more beneficial for the treatment of cancer than either agent alone. The interaction with other agents potentially reduces the dose of N-BPs required for the treatment of cancers that

previously were only susceptible to N-BP-induced apoptosis at unacceptably higher doses. These data provide solid evidence for future preclinical and clinical investigations, to identify suitable combinations of N-BPs with other anti-cancer and novel agents, for synergistic anti-tumour activity in the treatment of cancer patients.

1.7 OTHER INHIBITORS OF THE MEVALONATE PATHWAY

1.7.1 Statins

Developed in the late 1970's, the statins, also known as HMG-CoA (3-hydroxy-3methylglutaryl coenzyme A) reductase inhibitors, are a class of drugs that inhibit the rate-limiting step of the mevalonate pathway required for cholesterol biosynthesis (Figure 1.12) [Goldstein and Brown 1990]. Statins are characterised by an open-ring side chain, which is able to bind to the active site of HMG-CoA reductase, inhibiting the enzyme [Istvan et al. 2000]. Compared to the natural substrate, statins are able to bind approximately 1,000 times more effectively to HMG-CoA reductase [Istvan and Deisenhofer 2001]. Inhibition of cholesterol biosynthesis leads to increases in hepatic LDL receptor, which promotes hepatic LDL-cholesterol uptake and clearance of cholesterol from the bloodstream, therefore lowering endogenous cholesterol levels [Ness et al. 1996]. For this reason, statins are widely used for the treatment of hypercholesterolemia. Like the N-BPs, the statins also inhibit protein prenylation, preventing the post-translational lipid attachment modifications of many proteins essential for their function, including the GTPases. Other beneficial uses of statins include, anti-inflammatory, anti-thrombotic and immunosuppressive effects [Rosenson et al. 1999; Rosenson and Tangney 1998; Bellosta et al. 2000]. Recent reports of statin



Figure 1.12 The mevalonate pathway. Statins inhibit the enzyme as indicated above, preventing downstream processes including cholesterol biosynthesis, farnesylation and geranylgeranylation.

activity have included the stimulation of bone formation and anti-cancer activity [Bellosta *et al.* 2000].

In 1999, Mundy reported that statins were potent stimulators of bone formation in vitro due to associated increases in BMP-2 gene expression in bone cells [Mundy et al. 1999]. Subsequent reports supported statin-induced BMP-2 expression in human OS cells, which was inhibited by the addition of downstream metabolites of the mevalonate pathway, indicating that gene activation was directly due to enzyme inhibition [Sugiyama et al. 2000]. It has been suggested that the degree of water solubility of statins, determines their ability to induce increases in BMP-2 expression. Lipophilic statins, such as lovastatin are far better at increasing BMP-2 expression than watersoluble statins, such as pravastatin [Sugiyama et al. 2000]. Lipophilic statins are osteogenic and promote osteoblastic calcification independently of BMP-2 [Izumo et al. 2001]. In vivo studies have confirmed increases in cancellous bone mass and bone strength [Oxlund et al. 2001]. In humans, several clinical studies have described the skeletal effects of statins. In a study of 6,110 subjects, a 71% reduction of hip fractures was reported among statin users [Wang et al. 2000]. These results have been confirmed by other population-based case-control studies, with significant reductions of fractures and associated increased BMD in statin users [Chan et al. 2000; Meier et al. 2000; Pasco et al. 2002]. This was compared with users of other lipid lowering agents to rule out that hyperlipidemia itself is not associated with fewer fractures [Bauer 2003].

It is evident from the escalating use of statin therapy in the past 15 years that longterm use is safe and well tolerated. Initial concerns regarding associated increases in cancer incidence were raised, however long term follow up trials have demonstrated no association between the use of statins and increased risk of cancer [Bjerre and LeLorier 2001; Hebert *et al.* 1997; Pedersen *et al.* 2000].

Analogous to the N-BPs, statins also exhibit anti-proliferative, anti-invasive and apoptotic effects in transformed cells [Wong et al. 2002; Chan et al. 2003]. The mechanism of the effects of statins has been shown to be the inhibition of protein prenvlation [Keyomarsi, 1991; [Denoyelle et al. 2001; Xia et al. 2001]. All effects were reversed upon the addition of an intermediate of the mevalonate pathway, such as importance of prenylation, particularly confirming the geranylgeraniol, geranylgeranylation [Crick et al. 1998; Kusama et al. 2001; Agarwal et al. 1999]. Tumour cell types sensitive to statin-induced apoptosis include acute myeloid leukaemia (AML) [Dimitroulakos et al. 1999; Newman et al. 1994]; juvenile myelomonocytic leukaemia, squamous carcinoma of the head and neck and cervix, rhabdomyosarcoma, medulloblastoma, neuroblastoma [Dimitroulakos et al. 2001]; mesothelioma [Rubins et al. 1998]; astrocytoma [Jones et al. 1994]; and pancreatic cancer [Muller et al. 1998]. Statins, like BPs, trigger cell death in a tumour specific manner. In AML, statin induced apoptosis occurs in both primary tumours and tumour cell lines but not myeloid progenitor cells from normal bone marrow [Dimitroulakos et al. 1999; Newman et al. 1997].

In addition to their *in vitro* efficacy, statins have been used in many different animal models and shown to have anti-tumour effects. Anti-tumour effects were observed *in vivo* in leukaemia, glioma, neuroblastoma and liver, pancreatic and lung cancer [Kikuchi *et al.* 1997; Clutterbuck *et al.* 1998; Maltese *et al.* 1985; Kawata *et al.* 1992; Hawk *et al.* 1996]. A significant reduction in metastases was also observed for lymphoma, fibrosarcoma, melanoma, breast and colon cancer [Matar *et al.* 1998; Matar *et al.* 1999; Alonso *et al.* 1998; Broitman *et al.* 1996; Jani *et al.* 1993]. The use of statins in chemoprevention has been demonstrated in radiation-induced mammary tumourigenesis and chemical-induced colon tumourigenesis [Inano *et al.* 1997; Narisawa *et al.* 1996; Narisawa *et al.* 1996].

Clinical safety and efficacy of statins as anticancer agents has been evaluated in numerous clinical trials. Phase 1 clinical studies evaluated maximum tolerable dose and toxicity of lovastatin in advanced malignancies with doses reaching up to 30-45 mg/kg/day [Thibault *et al.* 1996; Larner *et al.* 1998]. Clinical efficacy of lovastatin has been reported with conflicting results. A phase II study in gastric adenocarcinoma generated disappointing results, however a randomised trial in patients with advanced liver cancer resulted in median survival of 18 months in the statin treated group compared with only 9 months in the control group [Kim *et al.* 2001; Kawata *et al.* 2001]. This suggests that statins may be of value in disease control when combined with standard chemotherapeutic therapy [Miller *et al.* 1993]. There is increased interest in the use of statin therapy for the treatment of cancer and clinical trials are ongoing. In advanced or recurrent cancers, statins are more likely to be effective when given in combination with other agents.

Combination therapy of statins with other agents has been investigated and found to act both additively and synergistically. These include the agents 5-flouracil, N,N'bis(2-chloroethyl)-N-nitrosurea, cisplatin, doxorubicin, cytosine arabinoside and non-steroidal anti-inflammatory drugs (NSAIDS) [Agarwal *et al.* 1999; Agarwal *et al.* 1999; Soma *et al.* 1992; Feleszko *et al.* 1998; Feleszko *et al.* 2000; Holstein and Hohl 2001; Wong *et al.* 2001]. Combination therapy has also been successful in animal models, with synergism observed in three tumour models in mice using lovastatin and doxorubicin [Feleszko *et al.* 2000]. Additionally, lovastatin has been used to potentiate anti-tumour activity of TNF- α and carmustine, in mouse tumour models of melanoma and astrocytoma [Feleszko *et al.* 1999; Feleszko *et al.* 1995; Soma *et al.* 1992]. One conflicting report by Bardeleben, suggests that the statin lovastatin, protects cells from doxorubicin and etoposide-induced cell death *in vitro* [Bardeleben *et al.* 2002]. These experiments have been repeated in an independent laboratory, where no evidence for

protection against doxorubicin and etoposide was found (unpublished data, Labrinidis *et al.*). Therefore, there is uncertainty regarding the validity of Bardeleben's data.

In a study of the relationship between cell cycle progression and TRAIL-induced apoptosis, it was found that a combination of simvastatin or lovastatin and TRAIL significantly augmented apoptosis in tumour cells but not normal cells [Jin *et al.* 2002]. This has been the first documented report of statin and TRAIL combination treatment *in vitro* and signifies the therapeutic potential for combining inhibitors of the mevalonate pathway with death receptor-mediated apoptosis.

The identification and improvement of new and known combinations of agents with statin therapy, needs to be investigated further. Additional optimisation and evaluation of statin therapy, alone and in combination, will improve efficacy in the clinical setting.

1.8.2 Prenyl transferase inhibitors (PTIs)

Post-translational modification or prenylation refers to the covalent addition of a 15-carbon farnesyl or a 20-carbon geranylgeranyl isoprenoid onto acceptor proteins. Many small GTPase proteins require prenylation for proper biological function [Zhang and Casey 1996]. This requires the enzyme farnesyl transferase or geranylgeranyl transferase [Casey 1992]. The need to develop inhibitors of these enzymes was quickly recognised when it was discovered that the Ras GTPase protein was a major contributor to human cancer with expression of the Ras oncoprotein identified in over 30% of cancers [Sebti and Hamilton 1997]. Four different forms of the Ras protein, H-, N-, K_A-, and K_B-Ras require farnesylation to induce malignant transformation [Casey 1992; Zhang and Casey 1996]. Alternatively, R-Ras and TC-21 require geranylgeranylation to mediate Ras transformation and malignancy [Zohn *et al.* 1998]. Studies with the mutated K-Ras demonstrated that the protein is able to be cross-prenylated, that is either

farnesylated or geranylgeranylated [James *et al.* 1995; Rowell *et al.* 1997]. Other GTPases, such as RhoA, Rac1 and cdc42 also require geranylgeranylation to promote tumourigenesis [Zohn *et al.* 1998].

This led to the development of farnesyl transferase inhibitors (FTIs) and geranylgeranyl transferase inhibitors (GGTIs), (or in general terms referred to as PTIs), specifically block the farnesylation anticancer agents, to and as novel geranylgeranylation process, as shown in Figure 1.13 [Sebti and Hamilton 2000]. This new approach to cancer therapy mimics the mechanism of action of N-BPs and statins. That is, inhibition of prenylation, the gradual depletion of prenylated proteins, an increase in unprenylated proteins leading to the inability of the cell to function and ultimately cell death. As is seen with the other inhibitors of the mevalonate pathway, variations in structure greatly affect the bioavailability and potency of the PTIs.

It is not surprising that PTIs induce much the same effect in tumour cells as other inhibitors of the mevalonate pathway, such as anti-proliferation and apoptosis in human cancer cells [Ashar *et al.* 2001; Bolick *et al.* 2003; Morgan *et al.* 2003; van de Donk *et al.* 2003; Vogt *et al.* 1997]. To elucidate the processing required for protein function, most *in vitro* studies to date have focused on the ability of FTIs and or GGTIs to potently and selectively inhibit prenylation of specific Ras mutants [Sebti and Hamilton 2000; Lobell *et al.* 2001]. It has been shown that the ability of FTIs to inhibit growth of human cancer cell lines does not correlate with Ras mutation status [Sepp-Lorenzino *et al.* 1995; Nagasu *et al.* 1995].

The efficacy of PTIs in animal models has been investigated with encouraging results. Although K-Ras requires both FTIs and GGTIs to inhibit prenylation *in vitro*, each alone is able to inhibit tumour growth in nude mouse xenograft models [Sun *et al.* 1998]. This verifies the concept that additional proteins requiring farnesylation, other than K-Ras, are critical for tumour growth and survival. Similarly, geranylgeranylated



Figure 1.13 The mevalonate pathway. PTIs inhibit the enzymes as indicated above, preventing farnesylation and geranylgeranylation.

proteins are also vital for tumour progression, as GGTI treatment led to tumour growth inhibition, supporting previous *in vitro* studies that RhoA and Rac1 are involved in tumourigenesis [Zohn *et al.* 1998]. Numerous laboratories have shown human tumour growth inhibition in nude mice xenografts treated with FTIs [Sebti and Hamilton 2000; Cox and Der 1997; Gibbs and Oliff 1997; Sun *et al.* 1999]. Lantry demonstrated the ability of FTIs to inhibit the growth of chemically induced lung tumours in nude mice [Lantry *et al.* 2000]. These results illustrate the anti-tumour potential of FTIs, by not only inhibiting tumour growth in xenografts but also in the tumours own microenvironment. FTIs do not simply inhibit tumour growth, but actually cause tumour regression. In a transgenic mouse model of mammary cancer, Ha-*ras* oncogenes expressed under the control of mouse mammary tumour virus (MMTV) promoter, cause the development of aggressive mammary and salivary carcinomas [Gibbs and Oliff 1997]. Upon FTI treatment of animals bearing tumours, regression was observed in 100% of the mice. Variants of the Ras oncogene produced varying results *in vivo*, suggesting that genetic differences may influence prognosis.

Combination therapy using both FTIs and GGTIs was found to significantly enhance the chemotherapeutic effect of human colon cancer cells both *in vitro* and *in vivo* [Di Paolo *et al.* 2001]. It seems that inhibition of both geranylgeranylation and farnesylation triggers a magnified apoptotic signal that enhances cell death.

The use of standard cytotoxic agents in combination with PTIs, has been reported by several independent groups [Adjei *et al.* 2001; Shi *et al.* 2000; Sun *et al.* 1999]. Sun demonstrated that the co-addition of cisplatin, taxol or gemcitabine with either FTI or GGTI, resulted in greater anti-tumour efficacy than monotherapy [Sun *et al.* 1999]. Similarly, taxol and desoxypothilone in combination with FTI produced synergistic killing effects in breast cancer cell lines [Moasser *et al.* 1998]. Theoretically, cytotoxic agents that do not induce cell cycle arrest should be more effective agents in

combination with FTIs. This holds true for cisplatin, which is not cell cycle specific and has a synergistic apoptotic effect with FTIs [Sun *et al.* 1999].

Combining FTIs and radiation therapy has been one approach to overcoming radiation resistance. Ras activation has been identified as the link to increased radiation resistance in transformed cells [FitzGerald *et al.* 1985]. Inhibition of farnesylation by the addition of FTIs to H-Ras transformed fibroblasts, led to the sensitisation of the cells to radiation-induced apoptosis [Bernhard *et al.* 1996]. K-Ras transformed cells however required both FTI and GTI to become sensitised to radiation [Bernhard *et al.* 1998]. Other tumour cell lines tested using a variety of FTIs *in vitro* and *in vivo* include bladder, breast, colon and lung [Sebti and Hamilton 2000; Kohl *et al.* 1995; Britten *et al.* 2001; End *et al.* 2001].

The most important feature of FTIs is their very low toxicity to normal cells at concentrations that affect transformed cells [Sebti and Hamilton 2000]. Long term use of FTIs in mice, with over two months worth of daily injections, showed no toxicity or weight loss in the animals treated [Sun *et al.* 1999; Sun *et al.* 1998]. Many reports have also observed the lack of toxicity with FTI use [Cox and Der 1997; Gibbs and Oliff 1997; Sebti and Hamilton 1997; Sebti and Hamilton 2000]. The reason for lack of toxicity is unknown. Lower FTI concentrations may be adequate to inhibit Ras processing in transformed cells compared to other proteins, such as nuclear lamins, in normal cells [Garcia *et al.* 1993]. Another explanation may be that Ras signalling occurs through some but not all pathways so that FTI inhibition of oncogenic Ras does not affect the function of normal R-Ras2/TC21, which is expressed ubiquitously in many cells [Carboni *et al.* 1995].

On the other hand, GGTIs have been reported as showing a higher level of toxicity *in vivo* than FTIs [Lobell *et al.* 2001]. This has not been consistent, with encouraging *in vivo* results reported by Di Paolo using the GGTI, BAL9611 [Di Paolo

et al. 2001]. GGTIs have biological activities such as inhibition of human tumour growth *in vitro* and *in vivo* that are highly desirable for novel anticancer agents. There is therefore a pressing need to evaluate current forms of GGTIs and to identify new, structurally different forms, that exhibit enhanced anti-cancer activity with low toxicity. The recent development of inhibitors capable of inhibiting both farnesylation and geranylgeranylation, known as dual inhibitors, looks very promising, with increased efficacy and reduced toxicity in both *in vitro* and xenograft models of cancer [Vigushin *et al.* 2004; Dinsmore *et al.* 1999; Dinsmore *et al.* 2004].

FTIs have displayed impressive efficacy in preclinical models of cancer. Phase I trials have evaluated the maximum tolerable dose and have demonstrated that they are well tolerated and suitable for prolonged administration [Brunner et al. 2003]. Karp et al. reported encouraging results in patients with leukaemia, with a 30% response rate [Karp et al. 2001]. Phase II trials in haematological malignancies have confirmed improved response rates of 60% in chronic myelogenous leukaemia [Keating 2002], 30% in myelodysplastic syndromes [Kurzrock et al. 2001; Kurzrock et al. 2001], 29% in myeloproliferative disorders [Gotlib et al. 2002] and 75% in multiple myeloma [Alsina et al. 2002]. Solid tumour malignancies on the other hand have been disappointing in clinical trials using single-agent FTIs in bladder cancer [Winquist et al. 2001], prostate cancer [Haas et al. 2002], colorectal cancer [Sharma et al. 2002], and pancreatic cancer [Cohen et al. 2002]. Breast cancer, small cell lung cancer and glioma cancers however have provided positive results in clinical trials suggesting that tumours with specific molecular characteristics are more susceptible to FTI therapy [Johnston et al. 2002; Heymach et al. 2004; Cloughesy et al. 2002]. Preliminary results of phase III trials have been reported for colorectal cancer and pancreatic cancer. Both trials showed no statistical difference between the placebo group and the group for median overall survival [Cunningham et al. 2002; Van Cutsem et al. 2002]. Other phase III trials are still under way. Although this has dampened enthusiasm for the use of FTIs, many of the initial studies were based on the assumption that K-Ras mutations are sensitive to FTIs. Subsequent studies suggest that sensitivity to inhibition of tumours by FTIs is independent of Ras mutation status and tumours with these mutations may actually be more resistant to FTIs. Clinical trials are often carried out on patients in the late stages of their disease with often highly metastatic cancers that are not responsive to FTI monotherapy. It may be more beneficial to treat patients in the early stages of their disease where the anti-angiogenic and pro-apoptotic properties of FTIs may prove more favourable. Preclinical studies investigating synergistic interactions using combinations of FTIs with other classes of anticancer agents, proved to be highly effective in inducing cancer cell death and may well be the therapy of choice in treating patients in the clinic.

1.9 AIMS AND SIGNIFICANCE OF PROJECT

Osteosarcoma is the most common primary malignancy of the bone in children and adolescents. The treatment of OS has undergone considerable changes over the past 20 years with the efficacy of chemotherapy significantly improving long-term survival compared to surgery alone. Despite this, patients continue to have a poor prognosis with the frequent acquisition of drug-resistant phenotypes and the occurrence of "second malignancies" associated with chemotherapy. It is therefore vital to find new alternative approaches for the treatment of OS.

Bisphosphonates (BPs), statins and prenyl transferase inhibitors (PTIs) all act by inhibiting the mevalonate pathway, a biosynthetic pathway responsible for the production of many intracellular products. Preclinical data clearly demonstrates the cytostatic, anti-proliferative and pro-apoptotic properties of these agents. More importantly, N-BPs, statins and PTIs exhibit anti-cancer properties that are amplified in combination with chemotherapeutic agents. The identification of novel combinations that induce significant cell death but lack the associated toxicity observed with current therapy, is highly desirable. BPs high affinity for bone, make it ideal for the treatment of OS as bioavailability is optimal, whereas the use of statins and PTIs are useful in OS metastases and other soft tissue sarcomas.

TRAIL, a member of the TNF family, is capable of inducing apoptotic cell death in a wide variety of tumour cell lines, but does not seem to be cytotoxic to many normal cell types *in vitro*, or *in vivo*. It was found that TRAIL-resistant tumour cells can frequently be sensitised to TRAIL by treatment with cytostatic or cytotoxic agents. In contrast normal human osteoblasts are not sensitive to TRAIL alone and are not sensitised by conventional chemotherapeutic agents. These observations suggest that TRAIL may be a useful anti-tumour agent for OS and that it compares favourably with its TNF-ligand superfamily counterparts, Fas and TNF- α , which are cytotoxic to normal cells.

The aim of the current study was to investigate the effect of N-BPs, statins and PTIs in combination with TRAIL, in delivering an apoptotic stimulus to OS cells. Such an outcome would bring the hope of achieving substantial apoptotic response in this tumour type without the toxic side effects of current therapies.

The specific aims of this study are:

- 1. To investigate the apoptotic effect induced by the bisphosphonate, zoledronic acid, on OS cell lines.
- 2. To investigate the additive/synergistic interactions observed between zoledronic acid and TRAIL and to elucidate the molecular mechanisms involved.
- 3. To assess whether other inhibitors of the mevalonate pathway, specifically statins and prenyl transferase inhibitors, exhibit additive/synergistic activity in combination with TRAIL.
- 4. To establish an *in vivo* OS model and to assess the effects of zoledronic acid and TRAIL therapy as single agents and in combination.

CHAPTER TWO

MATERIALS AND METHODS

2.1 CELL LINES AND TISSUE CULTURE

2.1.1 Osteosarcoma cell lines

The cell lines listed below were purchased from the American Type Culture Collection (ATCC), Manassus, VA, USA.

HOS BTK-143 G-292

MG-63

SJSA-1

Saos2

K-HOS

The cell line listed below was provided by Dr Berlin, Department of Orthopaedics, Sahlgrenska Hospital, Goteborg, Sweden.

KRIB

2.1.2 Normal human bone donor cells

Human osteoblastic cells were obtained from trabecular bone of osteoarthritic patients at joint replacement surgery, as described previously [Gronthos *et al.* 1999] or were grown from needle aspirates from the iliac crest of normal healthy young (25-35 years) donors. Briefly, specimens of bone were dissected into 0.3-0.5 mm pieces and washed extensively in phosphate buffered saline (PBS), pH 7.4. The fragments were seeded as explants into 75 cm² culture dishes (Corning, Costar Corp. Cambridge, MA, USA) and cultured at 37°C in a humidified atmosphere of 5% CO₂ in α MEM containing 10% heat-inactivated fetal calf serum and L-ascorbic acid 2-phosphate. The medium

was replaced at 4-day intervals. Cells typically reached confluence after 4-6 weeks. Cells were then subcultured by treatment with a 0.1% (w/v) mixture of collagenase and dispase. Cells from first passage were used for all experiments.

2.1.3 Cell culture

All tissue culture was performed under sterile conditions within laminar flow hoods. Cell cultures were maintained at 37°C in a humidified atmosphere containing 5% CO_2 . A haemocytometer and trypan blue dye exclusion was used to determined cell density. Cells were incubated in 0.4%(w/v) trypan blue in phosphate buffered saline (PBS) exclusion and counted using a light microscope. The osteosarcoma cell lines were cultured in Dulbecco's modified Eagle's medium (DMEM), (supplied by the Media Production Unit, IMVS, SA). This was supplemented with glutamine (2 mM), penicillin (100 IU/ml), streptomycin (100 μ g/ml), gentamicin (160 μ g/ml) and 10% fetal bovine serum. The medium was filter sterilised and stored at 4°C.

Cell lines were maintained by subculturing every 3 to 4 days using 0.05% (w/v) Trypsin digestion (supplied by the Media Production Unit, IMVS, SA).

2.2 CHEMICALS AND REAGENTS USED

2.2.1 General chemicals and reagents

The following chemicals and reagents were purchased from:

Sigma Chemical Co. Ltd., St Louis, MO, USA.

Ampicillin

β-mercaptoethanol

Bovine Serum Albumin

Bromophenol Blue

Dithiothreitol (DTT)

Ethylenediaminetetraacetic acid (EDTA)

Ethidium Bromide

Formamide

Hepes

Puromycin

Coomasie Stain

MOPS

Polyxyethylene Sorbitan Monolaerate (Tween 20)

Phenylmethylsulphanyl Fluoride (PMSF)

Saponin

Sephadex G₅₀

Sodium dodecyl sulphate (SDS)

Sodium Bicarbonate

Sodium Chloride

Sodium Vanadate

Sucrose

Tris-HCl

TRIS[hydroxymethyl]aminomethane

Triton-X-100

BDH Laboratory Supplies, Poole ,U.K.

Acetic Acid

Chloroform

Crystal violet Dimethyl Sulphoxide (DMSO) Ethanol Formaldehyde Formalin Glycerol Methanol Sodium Chloride Sodium Hydroxide

Roche Molecular Biochemicals, Castle Hill, NSW, Australia.

Cell Proliferation Reagent (WST-1)

4',6-Diamidine-2'-phenylindole dihydrochloride (DAPI)

Glycogen (molecular grade)

Magnesium chloride

Invitrogen, BV CH, Groningen, Netherlands.

4-12% Bis-Tris Pre-Cast Gels
NuPage Antioxidant
NuPage LDS Sample buffer 4X PVDF membrane
NuPage MES SDS Running Buffer 20X
NuPage Sample reducing agent NuPage Bis-Tris Gels
NuPage Transfer Buffer 20X Superscript II
SeeBlue plus 2 protein markers

Sources of other routinely used reagents:

DNA grade Agarose	Progen Industries, QLD, Australia
Low Melting point Agarose	Progen Industries, QLD, Australia
CHAPS	Pierce, Rockford, IL, USA
Dulbeccos Mod. Eagles. Media	ICN Biomedicals, Aurora, OH, USA
α -Minimum Essential Media	ICN Biomedicals, Aurora, OH, USA
Diploma instant skim milk	Bonlac Foods Ltd, VIC, Australia
Dextran	Progen Industries, QLD, Australia
Foetal Calf Serum	TRACE Laboratories, VIC, Australia
Reagent A	Pierce, Rockford, IL, USA
Reagent B	Pierce, Rockford, IL, USA
DNA molecular weight markers	GeneWorks Pty Ltd, SA, Australia
SYBR-Gold Nucleic acid stain	Molecular Probes, Eugene, OR, USA
Recombinant human TRAIL	Peprotech, Rocky Hill, NJ, USA
Trypsin-EDTA	Media Production Unit, IMVS, SA, Australia
L-Glutamine	Media Production Unit, IMVS, SA, Australia
Hybond H^+ nylon membrane	Amersham Pharmacia Biotech, UK
Univert mountant	Merck, Kilsyth, VIC, Australia

2.2.2 Novel anticancer agents

Recombinant human TRAIL	Genentech Inc. San Francisco, CA, USA
Zoledronic Acid (Zometa)	Novartis Pharmaceuticals, Australia Pty Ltd.
	(Sydney, NSW, Australia)
Lovostatin	Sigma-Aldrich Co.
Mevastatin	Sigma-Aldrich Co.
FTI-277	Calbiochem, Alexandria, NSW, Australia

2.2.3 Enzymes and inhibitors of caspase activity

The following enzymes and inhibitors were used to detect or inhibit caspase

activation.

Caspase-3 inhibitor II (Z-DEVD-fmk)	Calbiochem, Alexandria, NSW, AUS
Caspase-3/CPP32 Fluorogenic-	Calbiochem, Alexandria, NSW, AUS
Substrate (Ac-DEVD-AFC)	
Caspase Inhibitor III (Boc-D-fmk)	Calbiochem, Alexandria, NSW, AUS
Caspase Inhibitor I (Z-VAD-fmk)	Calbiochem, Alexandria, NSW, AUS

2.2.4 Antibodies and recombinant proteins

A list of Antibodies used in this thesis is listed below.

Table 2.1

Specificity	Antibody	Isotype	Source
TRAIL-R1	M271	Mouse IgG2a	Immunex, Seattle, WA, USA
TRAIL-R2	M413	Mouse IgG1	Immunex, Seattle, WA, USA
TRAIL-R3	M430	Mouse IgG1	Immunex, Seattle, WA, USA
TRAIL-R4	M444	Mouse IgG1	Immunex, Seattle, WA, USA
OPG	805	Mouse IgG1	R&D systems, Minneapolis, MN, USA
OPG	8051	Mouse IgG2A	R&D systems, Minneapolis, MN, USA
Poly-(ADP-Rib	ose)-	Rabbit IgG	Roche Molecular Biochemicals, Castle Hill, NSW, Australia

Unprenylated	sc-1482	Rabbit IgG	Santa Cruz Biotech. California, USA
RAP1A (C-17)			
Total RAP 1A	sc-65	Goat IgG	Santa Cruz Biotech. California, USA
BAX (N-20)	sc-493	Rabbit IgG	Santa Cruz Biotech. California, USA
Bcl-2 (N-20)	sc-492	Rabbit IgG	Santa Cruz Biotech. California, USA
Actin (I-19)	sc-1616	Mouse IgG1	Santa Cruz Biotech. California, USA
1D4.5		Mouse IgG2a	Dr Leonie Ashman,
			University of Newcastle, NSW, AUS
1B5		Mouse IgG1	Dr Leonie Ashman,
			University of Newcastle, NSW, AUS
Sheep IgG -FIT	ЪС	Mouse Ig	Southern Biotechnology,
			Birmingham, AL, USA
Goat IgG-PE		Mouse IgG	Southern Biotechnology,
			Birmingham, AL, USA

2.2.5 Solutions, buffers and kits

DNA loading buffer (6X):	0.25% Bromophenol Blue, 40% (w/v) sucrose in water.
RNA loading buffer (2X):	50% Formamide, 6.2% Formaldehyde, 10% glycerol, 30μg/ml Ethidium bromide.
MOPS buffer:	0.2M MOPS, 50mM Na acetate, 10mM EDTA (pH 8.0).
Protein Lysis Solution:	25mM MOPS (pH 7.2), 60mM β-glycerolphosphate, 15mM MgCl ₂ , 5mM EGTA, 1mM DTT, 0.1mM Na Vanadate, 1% Triton X-100, 1mM PMSF.
Caspase-3 Assay Buffer:	50mM HEPES, 0.1% CHAPS, 10% Sucrose, 10mM DTT, pH 7.4.

Caspase-3 Lysis Solution:	5mM EDTA, 5mM TRIS-HCl, 0.5% NP40 solution, pH 7.4.
Crystal Violet Solution:	1% Crystal Violet, in a 2% ethanol solution (total volume of 100mls).
TAE 50X:	40mM TRIS-acetate, 0.001M EDTA (pH 8.0).
EDTA (0.5M):	93.06gms EDTA in 400mls of water, pH to 8.0, make up to 500mls with water.
Anti-Fade (propylgallate):	2gms n-propylgallate, 80% glycerol, 20% 1x PBS, rotate at room temperature overnight until dissolved and store at 4° C.
Propidium Iodide(PI)/ Triton X-100	0.1% w/v Triton X-100 in 10mls PBS, 2 mg Dnase-free Rnase A, 30µl of 0.1% w/v Triton X-100 in 10mls PBS, 2 mg Dnase-free Rnase A, 30µl of 10mg/ml PI (PI stock- dissolve 1mg PI in 1 ml H ₂ 0 and store at 4oC)
WST-1 cell proliferation reagent assay kit	Roche Molecular Biochemicals, Manheim, Germany
BCA Protein Assay kit	Pierce Biotechnology Inc, Rockford, IL, USA
NuPAGE® System	Invitrogen Corporation, Carlsbad, CA, USA
2.2.6 Antibiotics	
GENETICIN (G-418)	Invitrogen, BV CH, Groningen, Netherlands
Penicillin (100 IU/ml)	Media Production Unit IMVS, S.A., Australia

Streptomycin (100 µg/ml) Media Production Unit IMVS, S.A., Australia

Gentamicin (160 µg/ml)Media Production Unit IMVS, S.A., AustraliaPuromycin (2 µg/ml)Sigma Chemical Co. Ltd., St Louis, MO, USA.

2.3 SPECIALISED EQUIPMENT USED

2.3.1 Flow cytometer

Analysis was carried out using the Becton and Dickinson "FACscan" flow cytometer coupled with LYSIS II or CellQuest software. Graphs were aligned using the WinMDI 2.8 software package. See section 2.4.8 and 2.4.9 for a detailed description of receptor analysis and cell cycle analysis methods using flow cytometry.

2.3.2 Phospho-imager

Imaging for western blot analysis (refer to 2.4.9) and DNA electrophoresis imaging (2.4.7) was performed using the Typhoon 9410 Variable Mode Imager (Amersham Biosciences, Buckinghamshire, UK).

2.3.3 Plate reader

The Perkin Elmer LS50 spectrofluorometer was used to measure optical density of crystal violet assays, WST-1 assays and protein estimation assays as described in sections 2.4.2, 2.4.3 and 2.4.6, respectively.

2.3.4 Radiography

Radiographs were taken using the HP cabinet X-Ray System-Faxitron Series, model no: 43855A. Exposure time and intensity was optimised with final settings set for 18 sec at 60 kVp. Kodak min R-2000 film was used supplied by Kodak (Australasia) Pty Ltd, Melbourne, Australia.

2.3.5 Micro-computer tomography analysis

High-resolution computer micro-tomography analysis was performed using SkyScan-1072 micro-CT (Skyscan, Aartselaar, Begium). The SkyScan software CONE-REC, 3D creator and CTan were used for file conversion, 3D reconstruction of images and bone volume analysis, respectively.

2.4 METHODS

2.4.1 Measuring cell viability using crystal violet

For determination of cell viability, 1×10^4 cells per well were seeded in 96 well microtiter plates and allowed to adhere to the plate overnight. The cell density varied between different experiments, based upon the size of well and experimental design; however, the level of confluency for experiments did not exceed 70%. Cells were then treated for 72 hours with various combinations of novel anti-cancer agents. Cell viability was determined by staining with crystal violet and measuring OD₅₇₀ of cell lysates.

2.4.2 Measuring cell viability using WST-1 assay

For determination of cell viability, 1x10⁴ cells per well were seeded in 96 well microtiter plates and allowed to adhere to the plate overnight. Cells were then treated for 72 hours with various combinations of novel anti-cancer agents. Cell viability was determined using the WST-1 cell proliferation reagent assay kit, a colorimetric assay based on the cleavage of the tetrazolium salt WST-1 by mitochondrial dehydrogenase in viable cells. Media was removed from the wells at termination of the experiment and 200µl of WST-1 solution made up in DMEM media (100µl of WST-1 for every 1000µl

of medium) was added and left to incubate between 30 mins to 3 hours at 37° C. The absorbance was measured using an ELISA reader at 420-480 nm at 30 min intervals. The reference wavelength used was 600 nm.

2.4.3 Measuring caspase activity

DEVD-caspase was assayed by cleavage of zDEVD-AFC (z-asp-glu-val-asp-7amino-4-trifluoro-methyl-coumarin), a fluorogenic substrate based on the peptide sequence at the caspase-3 cleavage site of poly (ADP-ribose) polymerase [Medina *et al.* 1997]. Cells were seeded at a density of $5x10^5$ in 24 well plates were treated as indicated, washed once with HBSS and resuspended in 200 µl of NP-40 lysis buffer containing 5 mM Tris-HCl, 5 mM EDTA and 0.5% NP40, pH 7.5. After 15 min in lysis buffer at 4°C, insoluble material was pelleted at 15,000g and an aliquot of the lysate was tested for protease activity. To each assay tube containing 8 µM of substrate in 1 ml of protease buffer (50 mM Hepes, 10% sucrose, 10 mM DTT, 0.1% CHAPS, pH 7.4) was added to 20 µl of cell lysate. Reactions were allowed to proceed for 4 hour at room temperature in total darkness, whereupon fluorescence was quantified (Exc 400 nm, Emis 505 nm). Optimal amounts of added lysate and duration of assay were determined in preliminary experiments. One unit of caspase activity was taken as one fluorescence unit (at slit widths of 12.5 nm) per 4 hour incubation with the fluorogenic substrate.

In experiments where caspase inhibition was used, the tetrapeptide pan-caspase inhibitors Z-VAD-fmk, BOC-D-fmk and Z-DEVD-fmk were resuspended in DMSO, and were added to cells at the indicated concentrations as the same time as all other agents. Control cells were incubated with the equivalent concentration of DMSO.

2.4.4 DAPI staining

Cells were seeded on plastic chamber slides (Nunc, Inc, Naperville, IL, USA) allowed to attach overnight and treated as indicated. After two washes with PBS, cells were fixed in 200 µl of an ethanol/acetic acid (ratio of 6:1) for 10 min at room temperature. The cells were washed in Milli Q water and allowed to air dry overnight. The fixed cells were then incubated with 0.8 mg/ml of 4', 6-diamidine-2'-phenylindole dihydrochloride (DAPI) for 15 min at room temperature. The cells were then rinsed once with 1X PBS and once with MilliQ water, and allowed to dry in total darkness. The coverslips were mounted on propylgallate (anti-fade) and kept in the dark until ready to visualise. DAPI staining was visualised by fluorescence microscopy using an Olympus BH2-RFCA fluorescence microscope and imaged using a Photometrics Coolsnap Fx digital camera (Roper Scientific, NJ, USA).

2.4.5 Protein estimation

The amount of protein per sample was estimated using the BCA Protein Assay kit (Pierce Biotechnology Inc). A series of 11 standards from 0 to 40µg were added to a 96 well plate using BSA 0.0004g/ml in caspase lysis buffer or protein lysis buffer. Frozen samples obtained for either caspase assay or western blot analysis were thawed and 5µl were aliquoted into the plate. Fifty parts of Reagent A was added to 1 part Reagent B and 200µl was added to each sample. The plate was incubated at 37°C for 30 mins and scanned at 570 nm in a plate reader. The values obtained were compared against the standards and the amount of protein in µg was calculated.

2.4.6 DNA fragmentation

Cells were harvested by washing twice with PBS, after treatments as indicated, and incubated overnight at 37°C in lysis buffer. DNA was extracted twice with equal volumes of phenol-chloroform-isoamylalcohol (25:24:1) and then precipitated in ethanol. Samples were electrophoresed in a 1.2% agarose gel, stained with ethidium bromide and visualised under UV light.

2.4.7 Anoikis assay

coated with polyHEMA (poly(2 culture plates were Tissue hydroxyethylmethacrylate); (Sigma-Aldrich, Castle Hill, Australia) as follows: 1 ml of 10 mg/ml of polyHEMA in 95% ethanol was applied to each well and allowed to dry. After drying, the wells were washed in PBS and stored dry until use. Cells were plated onto polyHEMA-coated or uncoated control wells at 1×10^5 cells per well in a six well plate. At the indicated times, cells were harvested from polyHEMA-coated wells by centrifugation, followed by trypsin/EDTA to yield single cell suspension. The number of viable cells was determined by trypan blue exclusion and counting on a haemocytometer.

2.4.8 Western blot analysis

Cells were treated as indicated and lysed in buffer containing 10 mM Tris HCl, pH 7.6, 150 mM NaCl, 1% Triton X-100, 0.1% sodium dodecyl sulphate (SDS), 2 mM sodium vanadate and a cocktail of protease inhibitors. Lysates were centrifuged for 5 min at 12,000 rpm, the supernatant fraction of the lysate was then removed and stored at -70°C until ready to use. Cell extracts containing between 35-50µg of protein were mixed with an equal volume of sample buffer containing 12 mM Tris HCl pH 6.8, 6%

SDS, 10% β - mercaptoethanol, 20% glycerol, and 0.03% bromophenol blue. Protein samples were boiled for 5 min and electrophoresed under reducing conditions in 4-12% gradient polyacrylamide gels. Separated proteins were electrophoretically transferred to PVDF transfer membrane and blocked in PBS containing 5% blocking reagent, for 1 h at room temperature. Immunodetection was performed overnight at 4°C in PBS/blocking reagent, using antibodies described in Table 2.1, and diluted according to manufacturers instructions. Filters were rinsed several times with PBS containing 0.1% Tween 20 and incubated with 1:5000 dilution of anti-mouse or anti-rabbit alkaline phosphatase-conjugate for 1 h. Bound proteins were detected and quantitated using the Vistra ECF substrate reagent kit using a Fluorimager (Molecular Dynamics Inc).

2.4.9 Flow cytometric analysis of receptor expression

Analysis was carried out using the Becton and Dickinson "FACscan" flow cytometer coupled with LYSIS II or CellQuest software. Graphs were aligned using the WinMDI 2.8 software package. Cells were reseeded into fresh culture flasks one day prior to the assay, rinsed twice with PBS and detached using 2 mM EDTA in PBS at 37°C for 5 min. For flow cytometric analysis, all subsequent incubation steps were performed on ice and centrifugation steps were performed at 4°C. Cells were washed twice in PBS by centrifugation at 200g for 5min. For analysis of internal antigen expression, cells were pelleted by centrifugation and then resuspended at 10⁷ cells/ml in 2% paraformaldehyde in PBS and incubated for a further 5 min. The volume was increased to 10 ml with PBS, and the cells were then pelleted and washed twice by centrifugation at 200g for 5 min in wash buffer (PBS + 0.1% Azide). The cells were then resuspended in permeabilisation buffer, incubated for 10 min, washed twice in wash buffer by centrifugation at 200g. Both intact and permeabilised cells were

resuspended at 2 x 10^6 cells/ml in blocking buffer (10% BSA/PBS + 0.1% Azide). 50µl aliquots of the cell suspensions were added to polypropylene FACS tubes. To each of these was added 50µl of monoclonal antibody solution (Mabs), as indicated, or isotype-matched non-binding control Mabs, each diluted to 10μ g/ml in blocking buffer and incubated for 45 min. The cells were washed twice in 2ml/tube of wash buffer by centrifugation at 300g. To the resuspended cell pellets was added 50µl of FITC-labelled F(ab')₂ sheep anti-mouse Ig or goat anti-mouse IgG-PE, both diluted 1/50 in blocking buffer. The cells were incubated for a further 45 min in the dark, washed twice as above, then resuspended and fixed in 0.5ml of cold 1% w/v paraformaldehyde for analysis by flow cytometry.

2.4.10 Flow cytometric analysis of the cell cycle

Analysis was carried out using the Becton and Dickinson "FACscan" flow cytometer coupled with LYSIS II or CellQuest software. Graphs were aligned using the WinMDI 2.8 software package. Cells were detached using 2mM EDTA in PBS, transferred to centrifuge tube and centrifuged for 5 min at 1100 rpm. Supernatant was removed, cells were resuspended well into 0.5 ml PBS and transferred into tubes containing 70% ethanol for storage at -20 °C until required. Preparation of the cells for cell cycle analysis required centrifugation of the ethanol suspended cells for 5 mins at 1100 rpm. The ethanol was decanted and the pellet was resuspended in 5ml PBS and centrifuged again for 5 min at 1100 rpm. The pellet was resuspended in 1ml PI/Triton X-100 staining solution with Rnase A and incubated at 37° C for 15 min.

2.4.11 Immunofluorescence assay

Cells were seeded into 8-chamber slides (Nunc, Inc, Naperville, IL, USA) at 2 x 10⁴ cells per well, and cultured for 48 hours. In situ immunofluorescence was performed on ice. Cells were rinsed once in PBS, fixed in 2% paraformaldehyde in PBS for 5 min, and permeabilised with 0.1% saponin in PBS containing 10% heat inactivated pooled normal human serum (permeabilization buffer), for 10 min. The cells were then washed thrice with PBS containing 0.1% BSA and 0.1% NaN3 (wash buffer), and blocked in 5% normal goat serum containing 0.1% w/v NaN₃ (blocking buffer) for 30 min. The blocking buffer was removed and monoclonal antibodies (Mab) specific for human TRAIL receptors TR1, TR2, TR3, TR4, TRAIL, OPG (Mab 8051) or isotypematched non-binding control Mabs, each diluted to 10 µg/ml in blocking buffer as above, were added for 45 min. The slides were washed thrice in PBS containing 0.05% (v/v) TritonX-100. To reveal primary antibody reactivity, cells were incubated with a 1/50 dilution in PBS of FITC-conjugated goat anti-mouse $F(ab)_2$ antisera for 30 min. The cells were then washed as above and stained for 1 min with 1 µg/ml DAPI solution, rinsed in PBS and mounted in Univert mountant. The labelled specimens were examined using an Olympus BH2-RFCA fluorescence microscope or an Olympus Bx 51 fluorescence microscope and imaged using a Photometrics Coolsnap Fx digital camera (Roper Scientific, NJ, USA).

2.4.12 Statistical analysis

All graphs were generated using Microsoft Excel 2000. This program was used to determine standard deviation (SD) based on experiments that were performed in triplicate and repeated at least three times.

2.5 IN VIVO ANALYSIS OF ANTI-TUMOUR ACTIVITY

2.5.1 Cell culture

See section 2.1.3 for a detailed description of cell culture protocol. For the purpose of *in vivo* transplantation, cells were removed from flasks by gentle 2mM EDTA digestion, followed by dilution of cells in sterile PBS.

2.5.2 Cell lines

The human osteosarcoma cell line K-HOS was used in this study because it has been demonstrated to produce an osteolytic lesion when implanted into bone. *In vitro* studies have also demonstrated the highly synergistic apoptotic effect of combination ZOL and TRAIL treatment in K-HOS cells.

2.5.3 Animals

Four to six week old Balb/c Nu/Nu female mice were housed under pathogen free conditions, at a constant temperature of 23°C, under controlled 12 hour light-dark cycles, in accordance with the protocol approved by the IMVS Animal Research Committee. Animals were acclimatised for one week prior to commencement of procedures. IMVS Animal Ethics approval was obtained for all procedures and animal care was in accordance with guidelines. During the experiment all animals were monitored closely and weighed at regular intervals to assess effect of implantation.

2.5.4 Tibial implantation

Cell suspensions of K-HOS and KRIB in sterile PBS at either (2 X 10^3 , 5 X 10^3 or 1 X 10^4 cells/20µl were used for the pilot study. All other *in vivo* experiments used 6 X $10^3/10\mu$ l of K-HOS cells for intra-tibial implantation. The mice were anaesthetised by subcutaneous (s.c.) injection with 80 mg ketamine/kg body weight and 10 mg

xylazine/kg body weight. The left tibia was wiped with 70% ethanol and using a 2 gassterilised Hamilton syringe, a 26 gauge needle was inserted through the tibial plateau with the knee flexed and the cells were injected (**Figure 2.1**). All animals were administered with PBS in the contralateral limb, as a control. After tibial implant mice were randomly assigned into groups. Mice were weighed regularly and radiographs were taken every two weeks to determine the extent of osteolysis. At sacrifice, all the major organs and both hind limbs were harvested in 10% buffered formalin for histological analysis.

2.5.5 Treatment

Recombinant human TRAIL, supplied by Immunex (Seattle, WA, USA) was used at 30mg/kg body weight, administered by intraperitoneal (i.p.) injection. Zoledronic acid (ZOL), supplied by Novartis Pharmaceuticals (Sydney, NSW, Australia), was dissolved in sterile water and prepared in sterile 1x PBS. Zoledronic acid at 100µg/kg body weight was administered by s.c. injection.

Protocol 1: In Vivo analysis of ZOL treatment.

Zoledronic acid was administered s.c. into female nude mice at weekly intervals starting one week after tibial implantation (Day 8, 15, 22, 29).

Protocol 2: *In Vivo* analysis of ZOL and TRAIL combination treatment.

Zoledronic acid was administered to mice at weekly intervals starting one week after tibial implantation (Day 8, 15, 22, 29). TRAIL was administered daily for 5 days, starting one week after tibial implantation and then once weekly (Day 8-12, 15, 22, 29).



Figure 2.1 Intra-tibial injection of human OS cells into a nude mouse. The left leg was held with knee flexed to expose the proximal tibia for needle insertion just below the anterior tubercle.

2.5.6 Histology

The tibias were fixed in 10% buffered formalin, followed by acid decalcification in 10% EDTA solution and 7% nitric acid at room temperature. Decalcification was confirmed by radiography before sectioning. Sections were paraffin embedded, sectioned longitudinally at 6 μ M and stained with haemotoxylin and eosin. Analysis was performed on a Nikon Eclipse TE300 inverted microscope (Nikon Corporation, Tokyo, Japan).

Histological sections were used to determine area of tumour burden using the *Scion* image analysis software. Sections were photographed and digital images were imported into the program. Using the program, the area of tumour burden was outlined and calculated as an area in mm² (*Scion* Image, Maryland, USA).

2.5.7 Radiography

Animals were anaesthetised by s.c. injection with 80 mg ketamine/kg body weight and 10 mg xylazine/kg body weight and were laid onto film with limbs spread for clear imaging. Refer to 2.3.4 for further details regarding equipment.

2.5.8 Micro-computer tomography analysis

Hind limbs required for micro-CT analysis were harvested in 100% ethanol. Both the right and left leg of each animal were mounted into the lid of the specimen tube containing 100% ethanol (to reduce dehydration). The specimen tube was securely placed into the SkyScan-1072 X-ray micro-CT Scanner and the specimen rotated 360° to ensure a clear field of view. The specimen was also elevated or lowered to further improve imaging results. The program was commenced with voltage set at 100 kV & 98 μ Amps, exposure time set at 1.7 sec, magnification set at 18 and beam hardening set to 100%. The TIFF files produced were converted into bmp files using the CONE-REC

software with Beam hardening set at 20%. Three dimensional (3D) reconstruction of images required the program 3D-creator with Step, Locality and Tolerance settings adjusted to 1 to give maximum image detail. Rotation of the images in any direction allowed the images obtained to be edited with the use of 3D-creator. CTan software was used for bone volume analysis of the scanned tibias. Using the two dimensional (2D) images obtained from the micro-CT scan the growth plate was identified and 150 sections starting from the growth plate/tibial interface and moving down the tibia, were selected. Histograms representing bone volume were generated and compared to the control tibia.

2.4.9 Statistical analysis

Fisher's exact test was used to assess the associations between treatment group and presence of osteolytic lesions and lung metastases. In each of these cases, the interaction between ZOL and TRAIL was assessed using the Breslow-Day test for homogeneity of odds ratios.

The 2-way Analysis of Covariance (ANCOVA) was used to assess the impact of ZOL and TRAIL and an interaction between the two on bone volume in the left leg where bone volume in the right leg was the covariate. Impact of ZOL and TRAIL, and an interaction between the two, on tumour area (where tumours were detected) was assessed using the 2-way Analysis of Variance (ANOVA).

CHAPTER THREE

ZOLEDRONIC ACID INDUCES APOPTOSIS IN

HUMAN OSTEOSARCOMA CELLS.

3.1 INTRODUCTION

Although known for their ability to inhibit osteoclast-mediated resorption, BPs, have also been identified as having direct anti-tumour cell activity. *In vitro* experiments have shown the potential for N-BPs to dose-dependently inhibit the proliferation and survival of tumour cells, including the induction of apoptosis in human myeloma [Senaratne and Colston 2002; Stearns and Wang 1996] breast [Fromigue *et al.* 2000; Sato *et al.* 1991] and prostate [Lee *et al.* 2001] cancer cells. The concentrations of N-BPs used to produce these cytostatic and apoptotic effects *in vitro* vary, depending on the potency of the N-BP and the cell line used [Fromigue *et al.* 2000].

Other reported effects of N-BPs include a reduction in tumour cell adhesion and invasion, necessary for effective metastases of tumour cells, as well as decreased angiogenesis, vital for the growth and survival of metastases [van der Pluijm *et al.* 1996; Wood *et al.* 2002]. These additional effects of N-BPs on tumour cells, encourages investigation of these agents for the preventative treatment of metastatic cancers.

ZOL is a new generation N-BP and is the most potent of the currently available BPs in terms of inhibition of bone resorption [Fleisch 2000]. It has been demonstrated in phase I trials of patients with bone metastases or multiple myeloma, that ZOL is safe and well tolerated at all dose levels tested [Berenson *et al.* 2002; Berenson *et al.* 2001]. Zoledronic acid is the first N-BP to show a reduction in skeletal complications including osteolytic lesions and thus significant clinical benefit in patients with bone metastases from various primary tumours [Berenson *et al.* 2001; Coleman and Seaman 2001; Rosen *et al.* 2004; Rosen *et al.* 2003; Rosen *et al.* 2004; Rogers *et al.* 2000].

The effect of N-BPs on OS cells has not been extensively studied, although recently the anti-proliferative effect of pamidronate in human and rat OS cells and alendronate in canine OS cells, *in vitro*, has been investigated [Sato *et al.* 1991; Stearns

and Wang 1996; Farese *et al.* 2004]. To date, there have been no reports of the effects of the potent bisphosphonate, ZOL on human OS cells, other than a paper based on results obtained as part of this thesis [Evdokiou *et al.* 2003]. This chapter describes investigation of the cytotoxic effects of ZOL on human OS cell lines and the mechanisms involved.

3.2 MATERIALS AND METHODS

3.2.1 Cells and cell culture

The osteosarcoma cell lines and the normal human osteoblast-like cells used in this chapter are listed in section 2.1.1. The methods for cell preparation and propagation are described in section 2.1.2 and 2.1.3.

3.2.2 Measuring cell viability

The method used to assess cell viability after zoledronic acid treatment is described in section 2.4.2. and 2.4.3.

3.2.3 Measuring caspase activity

The caspase inhibitor used is listed in section 2.2.4. Caspase activity protocols used in this chapter are described in section 2.4.4.

3.2.4 DAPI staining of nuclei

The method used to prepare and stain cells with DAPI is described in section 2.4.5.

3.2.5 Cell surface receptor analysis after ZOL treatment

The antibodies used in cell surface receptor analysis are listed in section 2.2.5. The method used is described in section 2.4.10.

3.2.6 Cell cycle analysis after ZOL treatment

The method used is described in section 2.4.11.

3.2.7 Statistical analysis

Microsoft excel was used to obtain mean standard deviations of experimental data. Data shown in each case are from a representative experiment, which was repeated at least three times.

3.3 **RESULTS**

3.3.1 ZOL induces dose-dependent apoptosis

The effects of ZOL on cell proliferation were tested against a panel of eight human OS cell lines (HOS, BTK-143, G-292, Saos2, SJSA-1, MG-63, K-HOS and KRIB). Cells were treated with increasing concentrations of ZOL at 0, 1, 10, 25, 50, 100 μ M. Treatment with ZOL for 72 hour dose-dependently resulted in fewer cells, compared with untreated control cultures, in all eight OS cell lines, as shown in **Figure 3.1**. ZOL was most potent in the HOS, BTK-143, K-HOS, G-292 and KRIB cells, with IC₅₀'s of 8 μ M, 10 μ M, 12 μ M, 15 μ M and 25 μ M, respectively. In contrast, there was a weaker response to ZOL in the SJSA-1, Saos2 and MG-63 cell lines, with IC₅₀'s of 35 μ M, 50 μ M and greater than 100 μ M, respectively. The efficacy of ZOL to induce apoptosis and thus reduce cell number, varied considerably between cell lines. The differential sensitivity to ZOL can be seen more clearly in **Figure 3.2**.

It is clearly evident that the decrease in cell number is due to apoptosis and not an increase in control cell number, due to the morphology of the cells after ZOL treatment, as shown in **Figure 3.3**. This will be discussed later.

It was previously suggested that a possible mechanism for the bisphosphonate effects on cell growth was the ability of these compounds to act as calcium chelating agents [Rodan 1997]. The possibility was investigated by the addition of equimolar concentrations of EDTA, with respect to the concentration of ZOL that reduced cell viability by 50%. We found no significant effect of EDTA treatment on cell viability in any of the cell lines tested, suggesting that calcium chelation was not responsible for the observed effects (data not shown).



Figure 3.1 The effect of ZOL on OS cell number. OS cells were seeded 1 x 10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of ZOL (1-100 μ M). Cell viability was determined using the crystal violet assay and is expressed as % of control, as described in the chapter 2. Data shown in each case are from a representative experiment repeated at least three times: points are means ±SD.



Figure 3.2 Differential sensitivity of OS cell lines. 1 x 10^4 cells per well were seeded in 96 well microtiter plates and allowed to adhere to the plate. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of ZOL (1-100 μ M). Cell viability was determined using the WST-1 cell proliferation reagent assay kit, as described in chapter 2. Data is shown as a percentage compared to control untreated cells and are from a representative experiment repeated at least three times: points are means ±SD.



Figure 3.3 Zoledronic acid treatment leads to apoptosis. Photographic image of HOS cells treated with (A) fresh media or (B) media containing $25\mu M$ ZOL for 48 hours. Prior to apoptosis the cells begin to detach, seen by the characteristic 'stretching' of the cells (B) instead of the common round compact form they take in normal control media (A).

3.3.2 ZOL induces time-dependent apoptosis

The effects of ZOL on cell proliferation were assessed in six OS cell lines by counting viable cells, as assessed by trypan blue exclusion, over a 96 hour period in control cultures or in cultures treated with 25 μ M ZOL as shown in **Figure 3.4**. The results demonstrate that, in the case of HOS, BTK-143 and G-292 cells, ZOL treatment resulted in a reduction in the number of cells, compared with the number of cells plated initially, indicating a time-dependent decrease in cell proliferation and survival. The effects of ZOL on cell viability were not immediate, in that there was a lag period of 48 to 72 hours before the onset of cell death. In the case of Saos2, MG-63 and SJSA-1 cell lines, the results suggest that ZOL at 25 μ M had no effect on cell viability but the rate of proliferation was significantly decreased.

3.3.3 ZOL treatment leads to increased caspase activity

It has been demonstrated in other cell types that bisphosphonate-induced apoptosis leads to activation of the caspase cascade [Fromigue *et al.* 2000; Coxon *et al.* 2000; Reszka *et al.* 1999; Sato *et al.* 1991]. To assess whether the observed ZOL-induced apoptosis in OS cells might also be mediated by caspase activation, caspase-3 activity in HOS, BTK-143, G-292, Saos2, SJSA-1 and MG-63 OS cells after ZOL treatment was measured, in relation to the effect of ZOL on cell number. Caspase activity increased in all cell lines, in a dose- and time-dependent manner, following treatment with ZOL (shown in **Figure 3.5** and **3.6**, respectively). HOS, BTK-143, and G-292 cells were significantly more sensitive to ZOL than MG-63, SJSA-1 and SaOS-2 cells, in terms of reduced cell viability, and this was reflected by the relative concentrations of ZOL required for caspase-3 induction in the cell lines (**Figure 3.5**).



* Refer to amendments

Figure 3.4 The effect of ZOL on OS cell number. Cells were incubated for the times indicated in the absence (\blacklozenge) or presence (\blacksquare) of 25 µM ZOL. The number of viable cells was quantitated using a heamocytometer and expressed as % of control. Data shown in each case are from representative experiments repeated at least three times: bars, ±SD.



Figure 3.5 Increasing doses of Zoledronic acid leads to increased caspase activity in OS cells. BTK-143, G-292, HOS, MG-63, SJSA-1 and SaOS-2 OS cells were treated with a fixed concentration of ZOL (25μ M) for 72 hours. Cell lysates were used to determine caspase-3-like activity, using the caspase-3 specific fluorogenic substrate, zDEVD-AFC, as described in the chapter 2. The data show that the increase in caspase 3-like activity is well correlated with the ZOL-dose-dependent decrease in cell number (Figure 3.1).



Figure 3.6 Zoledronic acid treatment leads to increased caspase activity over time in OS cells. BTK-143, G-292, HOS, MG-63, SJSA-1 and Saos2 OS cells were treated with a fixed concentration of ZOL (25 μ M) for 24, 48, 72 and 96 hours. Cell lysates were prepared and caspase-3-like activity was determined, using the caspase-3 specific fluorogenic substrate, zDEVD-AFC, as described in the chapter 2. The data show that the increase in caspase 3-like activity is well correlated with the ZOL-time-dependent decrease in cell number (Figure 3.4).
Moreover, caspase activity in response to 25 μ M ZOL was significantly increased above control levels by 72 hours in HOS, BTK-143 and G-292 cells. In contrast, there was only a minimal increase in caspase-3 activity in the MG-63, SJSA-1 and SaOS-2 cells after 72 hours of treatment (**Figure 3.6**).

3.3.4 Caspase inhibitors do not protect against ZOL-induced apoptosis

To investigate the role of caspase activation in the ZOL-mediated apoptosis of human OS cell lines, the effect of caspase inhibitors on the induction of apoptosis in the ZOL-sensitive HOS cells was examined. Cells were incubated for 72 hours with 25 μ M of ZOL, either alone or in the presence of the broad specificity caspase inhibitors, zVAD-fmk or BOC-D-fmk, or the specific caspase-3 specific inhibitor, zDEVD-fmk. Consistent with previous experiments, the HOS cell line showed a significant decrease in cell number in the presence of ZOL alone (seen in **Figure 3.7A**). However, none of the caspase inhibitors was able to prevent the ZOL-induced cell loss despite the complete and irreversible inhibition of caspase-3 activity achieved in their presence (**Figure 3.7B**).

Similarly, caspase inhibition failed to protect HOS cells from ZOL-induced morphological changes. Representative photographic images of the cells following DAPI staining, with the percentage of apoptotic cells in each treatment, is shown in **Figure 3.8**. The effect of caspase inhibition on ZOL-induced apoptosis was quantified by counting the number of pyknotic nuclei in the absence or presence of the general caspase inhibitor zVAD-FMK (**Figure 3.8A**). The result was 3 apoptotic cells in the control untreated cells and 71 and 80 apoptotic cells in the treatment media containing

Figure 3.7 Caspase inhibitors do not protect against Zoledronic acid induced apoptosis. (A) HOS cells were treated for 72 hours with 25 μ M ZOL alone, or were co-incubated with the broad specificity caspase inhibitors z-VAD-fmk (50 μ M) or Boc-D-fmk (50 μ M), or the caspase-3 specific inhibitor z-DEVD-fmk (50 μ M). To investigate possible toxic effects of the inhibitors, cells were also treated with each inhibitor alone. Cell viability determined using the WST-1 cell proliferation reagent assay kit. The absorbance was measured using an ELISA reader at 420-480 nm and cell viability expressed as % of control. Data points show means of quadruplicate results from a representative experiment, repeated at least twice; *bars* ±SD. (B) Cell lysates were used to determine caspase-3-like activity, using the caspase-3 specific fluorogenic substrate, zDEVD-AFC.







Figure 3.8 *A. Top image:* Control, untreated HOS cells showing fluorescent nuclei homogeneously stained. *Middle image*: Cells were treated for 72 hours with 25 μ M ZOL. The DAPI nuclear fluorescence stain reveals changes in the nuclei consistent with induction of apoptosis. *Bottom image:* Cells were treated for 72 hours with 25 μ M ZOL in combination with the general caspase inhibitor z-VAD-FMK (50 μ M). Nuclear apoptotic bodies containing DNA condensations were seen, as for the ZOL-treated cells in the absence of z-VAD-FMK. The percentage of apoptotic cells from each treatment is indicated (mean ± SEM from three random fields of 200 cells each).

B. HOS cells were treated for 72 hours with 25 μ M ZOL, either alone or in combination with the general caspase inhibitor z-VAD-FMK (50 μ M). Genomic DNA was isolated and subjected to agarose gel electrophoresis. Intranucleosomal genomic DNA fragmentation, characteristic of apoptosis, was observed in the ZOL alone and in the ZOL + Z-VAD-FMK treated cells.

25μM ZOL and 25μM ZOL with 50μM zVAD-FMK respectively. In addition, caspase inhibition did not prevent ZOL-induced DNA fragmentation, as assessed by agarose gel electrophoresis (**Figure 3.8B**). These results with OS cell lines therefore differ from findings reported previously for breast cancer cells [Fromigue *et al.* 2000] and osteoclasts [Reszka *et al.* 1999], showing that bisphosphonate-induced apoptosis in these other cell types could be suppressed by the co-addition of the broad specificity caspase inhibitor z-VAD-fmk. Our results suggest that ZOL-induced apoptosis of OS cell lines is independent of caspase

3.3.5 Geranylgeraniol protects against ZOL-induced apoptosis

The N-BPs, including ZOL, inhibit the mevalonate pathway by inhibiting geranylgeranylation of small GTP-binding proteins such as Ras, Rho, Rac and Rab [Luckman *et al.* 1998; Coxon *et al.* 2000; Dunford *et al.* 2001; Fisher *et al.* 2000; Rodan 1998; Rogers *et al.* 2000]. These proteins are important for many cellular processes, including proliferation, cytoskeletal organization and intracellular signalling. To examine the involvement of the mevalonate pathway on the ZOL-mediated apoptosis of human OS cells, we assessed the ability of geranylgeraniol (GGO), an intermediate of the mevalonate pathway, to protect HOS and BTK-143 cells from ZOL-induced apoptosis. As seen in **Figure 3.9**, the addition of GGO protected these cells, in a dose dependent manner, from ZOL-induced apoptosis. For BTK-143 cells, complete protection was achieved with 50µM GGO, whereas 200µM GGO was required to achieve the same level of protection in the HOS cells. Concentrations of up to 500µM GGO, when used alone, had no effect on cell proliferation. Geranylgeraniol was able to protect all human OS cell lines from ZOL-induced apoptosis. Using a standard amount



Figure 3.9 Geranylgeraniol dose dependently abrogates the effect of ZOL in OS cells. HOS and BTK-143 cells were treated for 72 hours with 25 μ M ZOL alone (\blacksquare), or were co-incubated with ZOL and increasing concentrations of geranylgeraniol (\blacksquare), as indicated. Cell viability is represented as a percentage of control untreated cells. Data points show means of triplicate results from a representative experiment, repeated at least twice; *bars* ±SD.

of 300µM GGO, and 25µM ZOL, **Figure 3.10** shows protection from cell death in four cell lines; HOS, BTK-143, K-HOS and KRIB cells.

In addition, a dose-dependent increase of GGO added to BTK-143 and HOS cells, along with 25µM ZOL, inhibited an increase in caspase activation, as shown in **Figure 3.11**. These results are consistent with the importance of the mevalonate pathway as an intracellular target for the effects of ZOL on cell proliferation.

We compared the effects of ZOL on cell viability with its ability to inhibit protein prenylation. Lysates at each dose of treatment were analysed by western blotting for the presence of the non-prenylated form of the small GTPase, Rap1A. Our results show that there was a clear correlation between ZOL dose on cell viability and the ability to inhibit protein prenylation. For example, in the HOS cells, ZOL at a dose of 5 μ M, which reduced cell viability by approximately 50% (see dose response curves, **Figure 3.1**), also effectively inhibited Rap 1A prenylation. By contrast, the same concentration of ZOL, which was less potent at reducing cell viability of MG-63 cells (~5%) failed to inhibit Rap 1A prenylation (**Figure 3.12**). In these cells, the presence of unprenylated Rap1 A was not obvious until the concentration of ZOL reached 10 μ M. The levels of total Rap 1A did not change with treatment

3.3.6 ZOL treatment leads to anoikis

As seen in **Figure 3.3**, the induction of cell death in OS cells by ZOL resembled "anoikis', a special mode of apoptosis that occurs when adherent cells detach or lose the particular attachment contacts with the extracellular matrix that confer survival signals to the cells. To investigate whether anoikis may be the mechanism for induction of cell death in OS cells, experiments were carried out in which adhesion of HOS cells by their



Figure 3.10 Geranylgeraniol protects against ZOL-induced apoptosis in OS cell lines. The co-addition of 300 μ M of GGO to HOS, BTK-143, K-HOS and KRIB cells was able to protect against the apoptotic effects of ZOL (25 μ M) when compared to cells treated with ZOL alone for 72 hours. *Columns*, Cell viability as a percentage of control untreated cells. Data points show means of triplicate results from a representative experiment, repeated at least twice; *bars* ±SD.



Figure 3.11 Geranylgeraniol (GGO) inhibits ZOL mediated increases in caspase activation. HOS and BTK-143 cells were treated with increasing concentrations of GGO (0, 1, 10, 25, 50, 100 μ M) either with (\blacksquare) or without (\bullet) 25 μ M ZOL, for 72 hours. Cell lysates were used to determine caspase-3-like activity, using the caspase-3 specific fluorogenic substrate, zDEVD-AFC. Data points show means of quadruplicate results from a representative experiment, repeated at least twice; *bars* ±SD.



Figure 3.12 ZOL inhibits protein prenylation. HOS and MG-63 OS cells were seeded in 25cm^2 flask and incubated for 72 hours with media containing increasing concentrations of ZOL (1-50 μ M) for 48 hours. Cell lysates were analysed on western blots for the presence of the unprenylated form of Rap 1A and total Rap 1A.

culture was prevented, by using flasks coated with polyHEMA. The non-ionic nature of polyHEMA prevents both matrix deposition and subsequent cell attachment. We found that the number of dead cells in the non-anchored cell population increased progressively with time from 15% at 8 hours to more than 35% at 24 hours, compared to anchored cells (see Figure 3.13A). In addition, the non-anchored HOS cells demonstrated a 3-fold increase in the level of caspase-3-like activity at 24 hours as shown in Figure 3.13B and displayed morphological changes characteristic of apoptosis, as seen in the ZOL treated cells (see Figure 3.13C).

3.3.7 ZOL induces S-phase arrest of the cell cycle

We next performed flow cytometric analysis of DNA content to identify cell cycle perturbations following treatment with ZOL over a 96 hour period. In the MG-63 cells, in which ZOL was cytostatic and did not cause a loss of cell viability, there was an accumulation of cells in S-phase (**Figure 3.14**). The effects of ZOL on cell cycle parameters in MG-63 cells were not obvious at 24 hours. However, by 48 hours, ZOL caused an approximate 2-fold increase in the number of cells arrested in S-phase (55% vs 25%), concomitant with a reduction of cells in G_0/G_1 and G_2/M phases. There were no cells in the apoptotic sub- G_0/G_1 peak at any time following treatment with ZOL in these cells, consistent with the results showing no loss of viability and confirming the absence of ZOL-induced apoptosis in MG-63 cells. In contrast, ZOL treatment resulted in S-phase arrest in the HOS cells, exerting its effects as early as 24 hours, with the number of cells in S-phase increasing from 25% in control untreated cells to 55% in the corresponding ZOL treated cells. At 48 hours the number of cells in S phase in the ZOL treated cultures increased to 69% of the total population. The proportion of hypodiploid



Figure 3.13 HOS cells were plated on polyHEMA-coated or control tissue culture wells and cultured for the indicated times. (A) Cell viability was assessed by trypan blue dye exclusion and the number of dead cells at each time point was expressed as a percentage of the total cells. (B) Cells grown on polyHEMA (non-anchored) for 24 hours demonstrated a three fold increase in caspase-3-like activity compared with the control (anchored) cells and (C) displayed morphological changes characteristic of apoptosis, as assessed by DAPI staining of their nuclei. Data points show means of triplicate results from a representative experiment, repeated at least twice; *bars* \pm SD.



Figure 3.14 Zoledronic acid induces S-phase arrest of the cell cycle. HOS and BTK-143 cells were incubated for 24, 48, 72 and 96 hours in the absence (control) or presence of 25μ M ZOL. At each time point, cells were harvested, fixed and stained with propidium iodide. The positions on the histograms of the hypodiploid Apo (sub-G₀/G₁), G₀/G₁, S, G2/M peaks and the percentage of cells in each of the cycle phases, in a representative experiment, are indicated.

cells found in the apoptotic sub- G_0/G_1 peak, consistent with the onset of apoptosis, increased progressively to 6%, 42% and 53% at 48, 72 and 96 hours, respectively, as shown in **Figure 3.14**.

3.4 DISCUSSION

In this study we have shown that ZOL treatment of OS cells induced a dose- and time-dependent decrease in proliferation in all of the cell lines tested, although it was variably effective in reducing cell number in the different cell lines. The antiproliferative effect of ZOL in the SaOS-2, SJSA-1 and MG-63 cell lines was the result of inhibition of cell cycle progression associated with a time-dependent increase in the proportion of cells arrested in S-phase. In contrast, the decrease in number of the HOS, BTK-143, G-292, K-HOS and KRIB cells was the result of increased cell death. Flow cytometric analysis of these cells demonstrated a time-related increase in the proportion of cells in the apoptotic sub Go/G1 peak of the cell cycle. However, the increase in apoptotic cells was preceded by an increase in the proportion of cells arrested in Sphase. Other N-BPs have been shown to also cause S-phase arrest [Iguchi et al. 2003; Forsea et al. 2004]. The reasons for the observed differences in activity and potency of ZOL in the different cell lines are unclear but may reflect differences in bioavailability, cellular uptake or intracellular effects of ZOL in different cell types. Nonetheless, our results indicate that the underlying mechanisms of the ZOL effect on cell proliferation involve the inhibition of cell cycle progression, mainly due to S-phase arrest, and the induction of apoptosis.

In those OS cell lines, in which ZOL decreased cell viability, the effects were not immediate and were seen after a lag period of 48 to 72 hours. This is consistent with previous published studies showing similar delayed effects of BPs in other cancer cells, as well as osteoclasts [Ito *et al.* 1999; Fromigue *et al.* 2000; Senaratne and Colston 2002]. As suggested previously [Rodan 1997], this lag period may represent the time required for the turnover of prenylated proteins already present in the cells, before the inhibition of prenylation of newly synthesised proteins by BPs, in this case ZOL, takes

effect. The concentrations, at which ZOL was effective in our experimental model (1-50 µM), are unlikely to be reached in the circulation using current in vivo treatment regimens. The maximum concentration of ZOL in human plasma after a 4 mg dose is only 2 µM [Rogers et al. 2000]. However, it has been suggested that because of the high affinity of BPs for bone mineral, the local concentration of BPs in the bone microenvironment will be much higher than serum levels [Sato et al. 1991; Sahni et al. 1993]. It is generally accepted that, in vivo, it is the osteoclasts that are most likely to be exposed to high levels of BP during active resorption of bone. In the absence of any convincing evidence to suggest that BPs can exert direct effects on cancer cells themselves within the bone microenvironment, we can only speculate that the close proximity of tumour cells and actively resorbing osteoclasts within osteolytic lesions will result in the sufficient release of the BP required to induce tumour cell apoptosis. It is therefore crucial to test the effectiveness of BPs against OS in vivo. This will be discussed further in chapter 6 and 7. Whether the effects of ZOL on growth and survival of OS cells in bone occurs through its direct actions or via secondary effects mediated by osteoclast inhibition remains to be determined. The positive effects of inhibiting osteoclast activity may include preventing the release of factors which inhibit osteoblast differentiation and/or promote tumour cell growth from the osteoclasts themselves and the bone mineral and ultimately lead to the inhibition of bone loss, which may be adequate treatment in itself if direct effects are unattainable in vivo (Figure 3.15).

Treatment of the ZOL-sensitive OS cells, for example HOS cells, with ZOL caused significant cell detachment, as viewed by light microscopy. In addition, staining of nuclei with the nuclear fluorescent stain DAPI demonstrated morphological changes characteristic of apoptosis, including chromatin condensation, nuclear fragmentation and the formation of dense rounded apoptotic bodies. Induction of intranucleosomal



Figure 3.15 Inhibition of bone resorption by N-BPs. N-BPs act *via* multiple mechanisms to reduce osteoclast number and activity and thus inhibit bone resorption. This can occur directly by inducing loss of function and apoptosis, or indirectly by stimulating osteoblast proliferation and influencing changes in osteoblastic gene expression and the secretion of soluble factors involved in regulating osteoclast differentiation and activity.

DNA fragmentation, as assessed by agarose gel electrophoresis, confirmed the induction of apoptosis by ZOL. We investigated the induction of apoptosis of OS cells by ZOL, by monitoring the activation of caspase-3. We found that caspase activation following treatment with ZOL paralleled the decreased cell viability in a dose- and time-dependent manner. Thus, ZOL induced the largest increase, and earliest activation, of caspase-3 in the HOS, BTK-143 cells and G-292, which were the cell lines that were most sensitive to the effects of ZOL on cell survival. In contrast, the increase in caspase activity in the ZOL-resistant SaOS-2, SJSA-1 and MG-63 cells was relatively small and delayed. The general specificity caspase inhibitors, Boc-D-fmk and zVAD-fmk, and the specific caspase-3 inhibitor z-DEVD-fmk, blocked the ZOL-induced activation of caspases but did not prevent cell detachment or cell death. Our results therefore differ from findings previously reported for breast cancer cells [Fromigue et al. 2000] or [Reszka et al. 1999] showing that BP-induced apoptosis in these other cell types could be suppressed by the co-addition of the broad specificity caspase inhibitor z-VAD-fmk. It is more likely that cell detachment is the critical event in the induction of cell death by ZOL, with the activation of caspases a consequence of loss of critical cell attachments, rather than a primary cause of cell death. The induction of cell death in OS cells by ZOL therefore resembled "anoikis" [Frisch and Francis 1994], a special mode of apoptosis that occurs when adherent cells detach or lose the particular attachment contacts with the extracellular matrix that confer survival signals to the cells (Figure 3.16). Evidence supporting anoikis as the means of inducing cell death in OS cells was their increased apoptosis in the absence of cell-matrix contacts by plating on polyHEMA. These data are consistent with a ZOL-specific induction of cell apoptosis that involves cell detachment as a prerequisite (anoikis), and in which caspase activation occurs secondarily and is redundant as a mediator of cell death.



Figure 3.16 ZOL inhibits protein prenylation. N-BPs such as ZOL inhibit an enzyme of the mevalonate pathway, preventing downstream processes including farnesylation and geranylgeranylation, required for post-translational modifications of GTPase proteins. These GTPases are involved in numerous roles vital for cell survival.

Consistent with several previous reports in osteoclasts [Coxon *et al.* 2000; Reszka *et al.* 1999], and in other cancer cell types [Shipman *et al.* 1997], we have shown that the effects of ZOL on cell proliferation of OS cells are mediated mainly through the inhibition of the mevalonate pathway. These effects of ZOL were reversed in the presence of geranylgeraniol (GGO), an intermediate of the mevalonate pathway (**Figure 3.17**). Therefore, the effects of ZOL on cell detachment are likely to be mediated by changes in cytoskeletal integrity brought about by the disruption of cell-cell and cell-extracellular matrix interactions. Further studies investigating the prenylation of small GTPases that regulate these processes will be required to delineate further the molecular mechanisms involved.



Figure 3.17 The mevalonate pathway. N-BPs inhibit the enzyme as indicated above, preventing downstream processes including farnesylation and geranylgeranylation. Geranylgeraniol (GGO) acts as an intermediate for geranylgeranylation in the presence of N-BPs.

CHAPTER FOUR

ZOLEDRONIC ACID SENSITISES HUMAN OSTEOSARCOMA CELLS TO TRAIL INDUCED APOPTOSIS:

Are dual treatments more effective?

4.1 INTRODUCTION

The chief objective of anticancer strategies is the selective induction of apoptosis in cancer cells but not normal cells. Investigating combinations of agents with synergistic or additive activity is appealing because this approach potentially enables the use of lower drug doses and therefore reduced toxic side effects.

BPs and TRAIL induce apoptosis in cancer cells, at least in vitro. Cytostatic and cytotoxic effects due to TRAIL have been reported in transformed cell lines including colon, lung, breast, central nervous system, kidney, brain and skin [Ashkenazi and Dixit 1999; Pitti et al. 1996; Wiley et al. 1995]. Sensitivity to TRAIL differs in various cell lines and the reason for this is unknown, although sensitivity appears not to depend solely on TRAIL receptor expression. TRAIL was only weakly apoptotic in a panel of human OS cell lines [Evdokiou et al. 2002]. However, treatment of TRAIL-resistant cell lines with chemotherapeutic agents resulted in their sensitization to TRAIL, suggesting that TRAIL-induced apoptosis can be enhanced [Griffith et al. 1998]. The mode of action of sensitizing agents may be via the stimulation of either or both the extrinsic or intrinsic apoptotic signalling pathways. Both in vivo and in vitro studies have confirmed the effect of chemotherapeutic agents to enhance the antitumor activity of TRAIL in many cell lines, including human OS [Griffith et al. 1998; Thomas and Hersey 1998; Gliniak and Le 1999; Ashkenazi et al. 1999; Keane et al. 1999; Mizutani et al. 2001; Gibson et al. 2000; Clayer et al. 2001; Evdokiou et al. 2001]. Conversely, TRAIL does not seem to affect normal cells in vivo, unlike its TNF-ligand superfamily counterparts, Fas and TNF- α which are cytotoxic, making TRAIL a potential anticancer agent [French and Tschopp 1999; Ashkenazi et al. 1999].

There is a pressing need to specifically target cancer cells within the bone. The highly selective localisation and retention of BPs in bone is the basis for its use in skeletal disorders. The most potent BP currently available is ZOL and its use in animal

models ensures that the highest possible doses can be achieved in vivo [Green et al. 1994]. In vitro experiments have shown that ZOL induces apoptosis in a variety of cancer cell types including breast, prostate, myeloma, colon and OS cell lines [Fromigue et al. 2000; Jagdev et al. 2001; Evdokiou et al. 2003; Lee et al. 2001]. It has been reported that the apoptotic effect of ZOL is far greater when combined with some chemotherapeutic drugs such as paclitaxel and dexamethasone than with either agent alone [Jagdev et al. 2001; Tassone et al. 2000]. Studies have also demonstrated that the effect of ZOL was not enhanced when combined with chemotherapeutic drugs such as etoposide and doxorubicin. This may be due to the particular cisplatin. chemotherapeutic agent and cell types examined [Evdokiou et al. 2002]. Many chemotherapeutic agents are cytotoxic mainly to actively cycling cells. Some agents are phase-specific; that is, they are cytotoxic to cells in a particular phase of the cell cycle and others are capable of exerting cytotoxicity at any phase of cell cycle [Chabner and Longo 1996]. In this case it is the depletion of prenylated proteins that leads to the apoptotic efforts of ZOL, thus hindering this process by halting cell cycling may not provide a combined apoptotic effect. For this reason it may be more efficacious to combine ZOL with agents that act by different mechanisms.

The work described in this chapter investigated the combination treatment of TRAIL, which acts *via* the TRAIL receptors to activate caspase-mediated apoptosis, and ZOL, which acts to inhibit the mevalonate pathway, in a panel of human OS cell lines.

4.2 MATERIALS AND METHODS

4.2.1 Cells and cell culture

The osteosarcoma cell lines and the osteoblast-like cells from donors of normal human bone used in this chapter are listed in section 2.1.1. The methods for cell preparation and propagation are described in section 2.1.2 and 2.1.3.

4.2.2 Measuring cell viability of ZOL and TRAIL treated cells

The method used to assess cell viability after zoledronic acid and/or TRAIL treatment is described in section 2.4.2. and 2.4.3.

4.2.3 Measuring caspase activity

The caspase inhibitors used are listed in section 2.2.4. Caspase activity protocols used in this chapter are described in section 2.4.4.

4.2.4 Cell attachment dependent survival using polyHema

The preparation of polyHema plates and the method used to assess cell attachment dependent survival is described in section 2.4.8.

4.2.5 Effect of geranylgeraniol on ZOL and TRAIL combination treatment in OS Cells

The method used to assess cell viability after treatment is described in section 2.4.2. and 2.4.3.

4.2.6 Western blot analysis

The antibodies used in the western blots are listed in section 2.2.5. The method used is described in section 2.4.6. and 2.4.9.

4.2.7 Cell surface receptor analysis after treatment

The antibodies used in cell surface receptor analysis are listed in section 2.2.5. The method used is described in section 2.4.10.

4.2.8 Cell cycle analysis after treatment

The method used is described in section 2.4.11.

4.2.9 Statistical analysis

Microsoft excel was used to obtain mean standard deviations of experimental data. Data shown in each case are from a representative experiment, which was repeated at least three times.

4.3 **RESULTS**

4.3.1 ZOL augments TRAIL-induced apoptosis of human OS cells

It has been reported previously that ZOL can exert cytotoxic or cytostatic effects on OS cells [Evdokiou et al. 2003], and that, while most OS are poorly sensitive to TRAIL as a single agent, relevant chemotherapeutic agents often used in the clinic for OS treatment were shown to be highly effective in sensitising these cells to the apoptotic effects of TRAIL [Evdokiou et al. 2002]. To investigate the effect of ZOL on TRAIL-induced cytotoxicity, a panel of OS cells (HOS, BTK-143, G-292, Saos2, SJSA-1, MG-63, K-HOS and KRIB) were exposed to increasing concentrations of ZOL in the absence or presence of TRAIL, as shown in Figure 4.1. The apoptotic effects of ZOL alone varied between cancer cell lines; sensitive cell lines were HOS, BTK-143, K-HOS, G-292 and KRIB and resistant cell lines were MG-63, SJSA-1 and Saos2. However, the concurrent treatment of ZOL plus TRAIL for 72 hours induced more apoptosis than single agent alone, in all OS cell lines to varying degrees. Combination treatment significantly augmented the apoptotic effect seen with ZOL alone, with as little as 10 μ M ZOL in HOS, BTK-143 and G-292 and 25 μ M ZOL in K-HOS resulting in 30%, 30%, 40% and 65% greater cell death, respectively, in these cell lines compared with ZOL treatment alone. Saos2, SJSA-1, MG-63, and KRIB required 10, 25, 25 and 25 µM of ZOL to induce 13%, 18%, 20% and 15% greater cell death, respectively, in combination with TRAIL. The cell lines that were highly sensitive to ZOL as a single agent gave the highest potentiated apoptotic effect when combined with TRAIL (Figure 4.2). This suggests that the role of ZOL is important for the augmented apoptotic effect of the combination treatment, however, the differences in sensitivity highlight the complexity of the mechanisms involved.



Figure 4.1 Zoledronic Acid augments TRAIL-induced apoptosis in OS cells. Cells were seeded at 1 x 10⁴ cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of ZOL (1-100 μ M), either with (\blacksquare) or without (\blacksquare) 100ng/ml TRAIL. Cell viability was determined using the crystal violet assay (as described in chapter 2) and is expressed as % of control. Data shown in each case are from a representative experiment which was repeated at least three times: points are means ±SD.



Figure 4.2 Differential response of OS cells to combined treatment with ZOL and TRAIL. 1 x 10^4 cells per well were seeded in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing 100ng/ml TRAIL and increasing concentrations of ZOL (1-100 μ M). Cell viability was determined using the WST-1 cell proliferation reagent assay kit, as described in chapter 2. Data is expressed as a % of control cells and are from a representative experiment which was repeated at least three times. Error bars have been omitted for better clarity.

4.3.2 The augmented apoptotic effect of ZOL and TRAIL is timedependent

Experiments were designed to monitor any significant changes in the time required for the onset of apoptosis in the combination of ZOL and TRAIL treated OS cells compared with cells treated with ZOL alone (as shown in **Figure 4.3**). This suggests that the time required for ZOL to exert its effects is critical to the observed synergistic activity with TRAIL co-treatment. From **Figure 4.3**, augmentation is clearly evident for BTK-143 cells and moderately evident for HOS and G-292 cells. It is not known what differences exist between cell lines, to cause differential sensitivity to ZOL, however if these could be defined, treatments could be adjusted based on different cell characteristics of OS cells.

Morphological changes characteristic of apoptosis, including chromatin condensation, nuclear fragmentation and the formation of dense rounded apoptotic bodies, were clearly evident in a high percentage of cells from all cell lines, following treatment for 72 hours with ZOL and TRAIL (**Figure 4.4**). Analysis by agarose gel electrophoresis showed that combination treatment of HOS cells also resulted in an increase in the level of intranucleosomal genomic DNA fragmentation (**Figure 4.5A**), a key feature of apoptosis that arises from activation of endogenous endonucleases. Western Blot analysis detected PARP cleavage, an early specific marker of apoptosis. At 48 hours, PARP cleavage was more apparent in the combination treated cells than in cells treated with either agent alone (**Figure 4.5B**).



Figure 4.3 The augmented apoptotic effect of Zoledronic acid and TRAIL is time-dependent. Cells were incubated for the times indicated with media alone (|), media containing 100ng/ml TRAIL (\diamond), 25 µM ZOL (\blacksquare) and combination 100ng/ml TRAIL and 25 µM ZOL (\blacktriangle). The number of viable cells was quantitated, using a heamocytometer. Data shown in each case are from representative experiments repeated at least three times: bars, ±SD.



Control



TRAIL



Figure 4.4 Morphological characteristics of apoptosis in HOS cells treated with 25μ M ZOL and/or 100ng/ml TRAIL, as assessed by DAPI staining of their nuclei.



Figure 4.5 HOS cells were incubated for 72 hours with 25 μ M ZOL either alone or in combination with 100ng/ml TRAIL. *A*. Genomic DNA was isolated and subjected to agarose gel electrophoresis. Intranucleosomal genomic DNA fragmentation was observed in the ZOL alone and in the ZOL + TRAIL treated cells. *B*. Cell lysates were analysed by western blot for the presence of PARP cleavage, a common characteristic of apoptosis. Actin was used as a control to assess overall level of protein in lysates.

4.3.3 ZOL-induced apoptosis in OS cells is associated with caspase activation

Treatment of OS cells with ZOL leads to the activation of the caspase cascade, although the addition of caspase inhibitors failed to prevent ZOL-induced cell loss [Evdokiou *et al.* 2003]. Treatment of HOS and Saos2 cells with ZOL and TRAIL resulted in greater activation of caspase-3 in the combination treated cells than the ZOL alone, regardless of the dose of ZOL (**Figure 4.6A**). In HOS cells the level of caspase-3 activity increased as early as 24 hours and remained higher at all time points in the combination treatment, when compared to the single agent treated cells (**Figure 4.6B**). In Saos2 cells the level of caspase-3 activity in the combination treated cells only exceeded that in ZOL treated cells at 72 hours and only marginally (**Figure 4.6B**). As HOS cells are more sensitive to ZOL than Saos2 cells it is perhaps not surprising that these results are also reflected in the combination treatments.

The addition of the caspase inhibitor z-VAD-fink failed to protect BTK-143R and K-HOS cells from ZOL-induced cell death as shown in **Figure 4.7**. Consistent with previous results, caspase inhibition did protect the cells from the augmented apoptotic effect, resulting in cell loss equal to the level induced by ZOL alone. This suggests that ZOL may sensitise OS cells to the effects of TRAIL *via* the activation of the caspase cascade and that the addition of the caspase inhibitor z-VAD-fink protects against this sensitisation by irreversibly inhibiting caspase-3 activity.

4.3.4 Geranylgeraniol prevents sensitisation to TRAIL by ZOL

The addition of geranylgeraniol (GGO) with ZOL treatment alone was able to protect BTK-143 and K-HOS cells from ZOL-induced apoptosis (Figure 4.8). Cells



Figure 4.6 Caspase activation in OS cells treated with ZOL and TRAIL is dose and time dependent. *A.* HOS and Saos2 OS cells were incubated for 72 hours with increasing concentrations of ZOL (1-50 μ M) and 100ng/ml TRAIL, either alone (unfilled bars) or in combination (filled bars). *B.* HOS and Saos2 cells were either untreated (**1**) or treated with a fixed concentration of ZOL (25 μ M) (**A**), TRAIL (100ng/ml) (**0**) or combination (**x**) for 24, 48, or 72hours. Cell lysates were collected and used to determine caspase-3-like activity, using the caspase-3 specific fluorogenic substrate, zDEVD-AFC, as described in chapter 2. Data are from representative experiments, which were repeated at least three times.



Figure 4.7 Caspase inhibition prevents augmentation of cell death by TRAIL in combination treatment with ZOL. BTK-143 and K-HOS cells were incubated for 72 hours with 25 μ M ZOL, 100ng/ml TRAIL or the broad specificity caspase inhibitors z-VAD-fmk (50 μ M), either alone or in combination. Cell viability was determined using the crystal violet assay. The absorbance was measured using an ELISA reader at 420-480 nm and cell viability is expressed as % of control. Data points show means of quadruplicate results from a representative experiment, which was repeated at least twice; *bars* ±SD.


Figure 4.8 Geranylgeraniol (GGO) prevents sensitisation to TRAIL in combination treatment with ZOL. The co-addition of 300 μ M of GGO to BTK-143 and K-HOS cells was able to protect against ZOL (25 μ M) sensitisation of the OS cells to the apoptotic effects of TRAIL (100ng/ml). *Columns*, Cell viability is expressed as a percentage of control untreated cells. Data points show means of triplicate results from a representative experiment, which was repeated at least twice; *bars* ±SD.

treated with ZOL, TRAIL and GGO exhibited cell death similar to that seen with TRAIL alone. GGO alone, or in combination with TRAIL, had no effect on cell number in either cell line. These results support the significance of inhibition of the mevalonate pathway by ZOL, as a factor in the sensitisation of OS cells to the effects of TRAIL.

Western blot analysis was employed to measure inhibition of protein prenylation using lysates from cells treated with ZOL alone, TRAIL alone, or ZOL and TRAIL together. The non-prenylated form of Rap 1A was detected specifically in ZOL treated cells, confirming the role of ZOL in inhibiting protein prenylation (**Figure 4.9**). Total Rap 1A protein levels did not change in any of the treatments. Western Blot analysis of Bcl-2 and Bax resulted in no changes between cells treated with ZOL and TRAIL both alone and in combination (**Figure 4.9**).

4.3.5 Loss of cell anchorage sensitises cells to TRAIL-induced apoptosis

As stated in chapter 3, the addition of ZOL to OS cells leads to cell detachment prior to apoptosis by a process resembling anoikis [Evdokiou *et al.* 2003; Frisch and Francis 1994]. A comparison of anchored versus non- anchored cells, either untreated or treated with TRAIL, revealed a large proportion of cell death of 40% at 8 hours and up to 75% at 12 hours in the TRAIL treated non- anchored cells, compared to 20% cell death in the untreated anchored cells (**Figure 4.10A**). The morphological changes were extremely apparent between the anchored and non- anchored cells at both time points. Almost no apoptotic cells were identified in the anchored cells with TRAIL treatment. The non-anchored cells showed typical apoptotic characteristics including cell shrinkage, cell surface blebbing and fragmentation (**Figure 4.10B**). These results suggest that cell detachment is an important factor in sensitising the cells to TRAIL.

114



Figure 4.9 Western Blot analysis of HOS cells treated with ZOL and TRAIL either alone or in combination. HOS cells were seeded in 25 cm^2 flask and incubated for 48 hours with media containing 25 μ M ZOL, 100ng/ml TRAIL or combination of both. Cell lysates were analysed by western blot using antibodies for the unprenylated form of Rap 1A, total Rap 1A, Bcl-2, Bax and actin (see chapter 2 for details).



Figure 4.10 *A.* Loss of cell anchorage sensitises HOS cells to TRAIL-induced apoptosis. HOS cells were added to standard culture flasks (\blacksquare anchored) or polyHEMA-coated flasks (non-anchored) either with (\blacksquare) or without (\blacksquare) TRAIL (100ng/ml). The number of dead cells was counted at 8, 12 and 24 hours and and expressed as percent of total cells. Data points show means of triplicate results from a representative experiment, which was repeated at least twice; *bars* ±SD.



Figure 4.10 *B*. Comparison photographic images of HOS cells from anchored (plastic flasks) versus non-anchored (polyHEMA-coated flasks). (A) anchored cells treated with 100ng/ml TRAIL for 8 hours (B) anchored cells treated with 100ng/ml TRAIL for 12 hours; (C) non-anchored cells treated with 100ng/ml TRAIL for 8 hours; (D) non-anchored cells treated with with 100ng/ml TRAIL for 12 hours.

4.3.6 Cell cycle analysis of ZOL treated OS cells leads to S-phase arrest whereas combination with TRAIL leads to apoptosis

Cell cycle perturbations were characterised in HOS and BTK-143 cells following co-treatment with ZOL and TRAIL, using flow cytometric analysis. TRAIL treatment alone had no effect on the cell cycle, even after 4 days of treatment (Figure 4.11). As described previously, ZOL leads to a cytostatic response in OS cells, with an accumulation of cells in S-phase, as early as 24-48 hours [Evdokiou et al. 2003]. In the present experiments, the proportion of HOS cells in S-phase decreased progressively with ZOL and TRAIL treatment, from 55, 38, 23 and 10%, compared with ZOL alone with 55, 69, 40 and 20% at 24, 48, 72 and 96 hours, respectively (Figure 4.11). The decrease in S phase correlated with an increase in the apoptotic sub- G_0/G_1 peak to 5, 44, 65 and 83% at 24, 48, 72 and 96 hours, respectively. Similar trends were observed in the BTK-143 cell line. By 72 hours ZOL treatment, the proportion of BTK-143 cells in the S-phase peaked at 38% compared with 26% in control cells. The proportion of BTK-143 cells in S-phase after treatment with ZOL and TRAIL decreased progressively from 30, 24, 16 and 10% at 24, 48, 72 and 96 hours, respectively (Figure 4.11), correlating with an increase in the apoptotic sub- G_0/G_1 peak to 12, 27, 59 and 66% at 24, 48, 72 and 96 hours, respectively. These trends were also identified in the G-292, and Saos2, but not the MG-63 and SJSA-1 cell lines, at least at these concentrations of agents (data not shown).



Figure 4.11 ZOL leads to S-phase arrest in OS cells whereas combination with TRAIL leads to apoptosis. HOS and BTK-143 cells were incubated with fresh media containing TRAIL (100ng/ml) in the absence (control) or presence of 25 μ M ZOL for 24, 48, 72 and 96 hours. At each time point, cells were harvested, fixed and stained with propidium iodide for cell cycle analysis using the flow cytometer. The positions on the histograms of the hypodiploid Apo (sub-G₀/G₁), G₀/G1, S, G2/M peaks and the percentage of cells in each of the cycle phases, in a representative experiment, are indicated.

4.3.7 Normal human bone donor cells are not sensitised to TRAILinduced apoptosis by ZOL

NHB cells obtained from patients at the time of joint replacement surgery were used to assess the effect of ZOL and TRAIL combination treatment. NHB cells did not exhibit augmented apoptotic effects with co-treatment of ZOL and TRAIL, even at the highest concentrations of ZOL (100μ M), (**Figure 4.12**). Although there was a reduction in the overall survival of cells at the higher concentrations, cell survival still exceeded that seen in any of the human OS cell lines. Apart from the dose experiments, all other experiments conducted throughout chapter 3 and 4 used ZOL at a concentration of 25µM, at which no augmentation was observed in any of the NHB cell donors tested.

4.3.8 ZOL upregulates TRAIL cell surface death receptors

In HOS cells it was clearly evident that ZOL treatment lead to the upregulation of both DR4 and DR5 death receptors, while no changes were observed in the expression of DcR1, DcR2 and OPG decoy receptors (**Figure 4.13**). Consistent with the lack of effect of ZOL and TRAIL on NHB viability, these treatments did not affect expression of death and decoy receptors on NHB cells (**Figure 4.13**). These data suggest that ZOL may act to sensitise OS cells to TRAIL *via* upregulation of the death receptors.



Figure 4.12 Normal human osteoblast-like cells derived from two different donors, were seeded into plates and allowed to adhere before treatment for 72 hours with increasing doses of ZOL (\blacksquare) or a fixed concentration of 100ng/ml TRAIL with increasing ZOL doses (\square). Cell viability was determined using the crystal violet assay, as described in the chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.



Figure 4.13 Zoledronic acid upregulates TRAIL death receptors on the cell surface of OS cells. HOS and NHB cells were left untreated or treated with 25 μ M ZOL for 48 hours. Monoclonal antibodies specific for human TRAIL receptors, DR4, DR5, DcR1, DcR2 were used to analyse cell surface receptor expression in intact cells by FACS. Filled histograms represent Zol-treated cells.

4.4 **DISCUSSION**

In this study the combination treatment with ZOL and TRAIL led to an augmented apoptotic effect in OS cell lines but not in normal human bone cells (NHB). Combination treatment with ZOL and TRAIL resulted in varied degrees of augmented cell death, which reflected the sensitivity of cells to ZOL treatment alone. While combination treatment led to a greater apoptotic effect than either agent alone, the duration of this effect was not notably swifter than ZOL alone, suggesting the depletion of geranylgeranylated proteins as a crucial factor in sensitisation to TRAIL. Flow cytometric analysis of cells given the combination treatment demonstrated a timerelated decrease in the proportion of cells in the apoptotic sub-G0/G1 peak of the cell cycle. Unlike treatment with ZOL alone, combination treatment did not cause S-phase arrest, but instead led to direct apoptosis. It remains unclear why there is variation in the response to ZOL between cell lines but may be due to differences in bioavailability, cellular uptake (although the mode of transport has yet to be identified) and intracellular differences such as altered FPP synthase expression [Green 2003; Salomo et al. 2003]. Nonetheless, although these differences exist, it is evident that some cell lines, which are normally resistant to TRAIL induced cell death, are strongly sensitised by the coaddition of ZOL.

Morphological changes characteristic of apoptosis, including chromatin condensation, nuclear fragmentation and the formation of dense rounded apoptotic bodies, were clearly evident in OS cells stained with the nuclear fluorescent stain, DAPI, following ZOL and TRAIL treatment. Apoptosis, confirmed by agarose gel electrophoresis, showed that combination treatment also resulted in an increase in the level of intranucleosomal genomic DNA fragmentation, a key feature of apoptosis that arises from activation of endogenous endonucleases. PARP cleavage was detected as an early specific marker of apoptosis. Further investigation of caspase activation revealed a

117

higher level of caspase activity in the combination treated cells than cells treated with ZOL alone, regardless of ZOL concentration and cell type. This was not surprising as TRAIL acts via the extrinsic and intrinsic pathway to directly induce apoptosis via caspase activation. The amount of caspase activity between the single and dual treatments was greater in HOS than Saos2 cells but this may reflect the sensitivity differences to ZOL, which were observed between the two cell lines. This may also be the reason for the early onset of caspase activation in the combination treated cells seen at 48 hours in HOS cells, compared to 72 hours in Saos2 cells. The addition of the caspase inhibitor zVAD-fmk failed to protect the cells from ZOL-induced cell loss, as described previously [Evdokiou et al. 2003]. The presence caspase inhibitor prevented the augmented apoptotic effect in cells treated with the combination of ZOL and TRAIL and the degree of cell death was equal to that observed with ZOL treatment alone. The prevention of cell attachment and the subsequent exposure to TRAIL resulted in considerably more cell death than attached cells exposed to TRAIL. This suggests that cell detachment, is a potential mechanism for the sensitisation of OS cell lines by ZOL and supports the importance of prenylation of proteins required for cell attachment and integrity [Zhang and Casey 1996; Ridley and Hall 1992].

The use of intermediates of the mevalonate pathway such as geranylgeraniol (GGO) to reverse the effects of ZOL was shown previously [Reszka *et al.* 1999] and when administered to cells with combination treatment it prevented the augmented apoptotic effect and prevented ZOL-induced cell loss, again confirming that ZOL sensitises cells to the effects of TRAIL. This was further confirmed by the presence of unprenylated Rap 1A protein found only in ZOL treated cell lines. No changes were observed in the levels of Bcl-2 and Bax protein between treatments of HOS cells. This conflicts previous reports of down-regulation of the Bcl-2 protein with ZOL treatment [Oades *et al.* 2003; Senaratne *et al.* 2000]. Changes in other Bcl-2 family proteins may

be involved and further investigation is needed to better understand the involvement of the intrinsic pathway during ZOL-induced apoptosis. Other changes observed in ZOL treated cells was the upregulation of the cell surface receptors DR4 and DR5, commonly referred to as death receptors, due to their ability to elicit apoptosis *via* caspase activation. No change in receptor expression was observed in normal cells. It is possible that a depletion of prenylated proteins would lead to cell stress and upregulation of death receptors at the cell surface, which may be one mechanism, by which ZOL sensitises OS cells to TRAIL.

The use of ZOL in the treatment of OS has exciting potential however it may be unrealistic to believe that high enough concentrations can be achieved *in vivo*, despite suggestions that there is an elevated localised concentration in the bone microenvironment due to the high affinity of ZOL for bone, compared to serum levels. The use of combination treatment with synergistic or additive activity however, reduces the concentration of ZOL required to achieve significant apoptosis, improving the chances of attaining sufficient concentration levels *in vivo* with reduced toxic side effects. [Sato *et al.* 1991].

CHAPTER FIVE

OTHER INHIBITORS OF THE MEVALONATE PATHWAY: STATINS AND PRENYL TRANSFERASE INHIBITORS INDUCE APOPTOSIS IN OSTEOSARCOMA CELLS- PRELIMINARY STUDIES. Are there effective alternatives to Zoledronic Acid?

5.1 INTRODUCTION

BPs have a high affinity for the hydroxyapatite mineral in bone and are selectively taken up and adsorbed to mineral surfaces at sites of increased bone turnover where they exert their effects. The statins and prenyl transferase inhibitors (PTIs), like the N-BPs, act by targeting an enzyme of the mevalonate pathway. However, unlike N-BPs, which preferentially localise to the bone, both the statins and PTIs are systemically distributed *in vivo*. This makes them very appealing as alternative agents to ZOL not only for tumours localised to the bone, but also for soft tissue tumours.

Much like N-BPs, agents such as statins and prenyl transferase inhibitors (PTIs), also hinder the progress of the mevalonate pathway by inhibiting key enzymes. Statins, principally used for the treatment of hypercholesterolemia, inhibit 3-hydroxy-3-methylglutaryl coenzyme A reductase, which is required for cholesterol biosynthesis [Goldstein and Brown 1990]. The PTIs inhibit the enzymes farnesyl transferase and geranylgeranyl transferase and thus prevent farnesylation and geranylgeranylation, respectively. Both the statins and the PTIs have similar effects as N-BPs, such as anti-proliferation and apoptosis in human cancer cells [Newman *et al.* 1994; Wong *et al.* 2002; Ashar *et al.* 2001; Bolick *et al.* 2003; Morgan *et al.* 2003; van de Donk *et al.* 2004].

N-BPs, statins and PTI have all been shown to exhibit anti-cancer properties that are amplified in combination with chemotherapeutic agents [Sun *et al.* 1999; Feleszko *et al.* 2002; Feleszko *et al.* 1999; Jagdev *et al.* 2001; Tassone *et al.* 2003]. The high affinity of ZOL for bone makes it potentially useful for the treatment of OS because its bioavailability may be greater in proximity to bone than in other tissues, whereas the use of statins and PTIs may be more useful in OS metastases and other soft tissue sarcomas. More importantly, like ZOL, clinical trials using statins and FTIs have shown these to be well tolerated and safe, with little overt toxicity [Thibault *et al.* 1996; Larner

121

et al. 1998; Brunner et al. 2003; Karp et al. 2001; Gotlib et al. 2002]. These characteristics make them worth considering for cancer therapy.

In this chapter the effect of statins and PTIs in combination with TRAIL was investigated. Such an outcome would bring the hope of achieving substantial apoptotic response both in this tumour type and soft tissue tumours, without the toxic side effects of current therapies.

5.2.1 MATERIALS AND METHODS

5.2.1 Cells and cell culture

The osteosarcoma cell lines and the normal human bone donors used in this chapter are listed in section 2.1.1. The methods for cell preparation and propagation are described in section 2.1.2. and 2.1.3.

5.2.2 Measuring cell viability

The method used to assess cell viability is described in section 2.4.2. and 2.4.3.

5.2.3 Measuring caspase activity

The caspase inhibitor used is listed in section 2.2.4. Caspase activity protocols used in this chapter are described in section 2.4.4.

5.2.4 Western blot analysis

The antibodies used in the western blots are listed in section 2.2.5. The method used is described in section 2.4.6. and 2.4.9.

5.2.5 Statistical analysis

Microsoft excel was used to obtain mean standard deviations of experimental data. Data shown in each case are from a representative experiment, which was repeated at least three times.

5.3 **RESULTS**

5.3.1 Statins induce dose-dependent apoptosis

The cytotoxic effects of the statins, Lovostatin (LOV) and Mevastatin (MEV), on cell proliferation were tested against a panel of six human OS cell lines (HOS, BTK-143, G-292, Saos2, SJSA-1, MG-63). Cells were treated for 72 hours with increasing concentrations of LOV or MEV at 0, 1, 10, 25, 50 μ M. Treatment with LOV or MEV dose-dependently resulted in fewer cells, compared with untreated control cultures, in all six OS cell lines, (**Figure 5.1** and **5.2**). LOV was most potent in the G-292, BTK-143, Saos2 and HOS cells, with IC₅₀'s of 6 μ M, 7 μ M, 15 μ M, and 18 μ M, respectively. In contrast, there was a weaker response to LOV in the MG-63 and SJSA-1 cell lines, with IC₅₀'s of 22 μ M and 52 μ M, respectively. The sensitivity to LOV was reflected in the MEV treatments, with the most sensitive cell lines being G-292, BTK-143, Saos2 and HOS cells, with IC₅₀'s of 8 μ M, 10 μ M, 20 μ M, and 32 μ M respectively. The more resistant cell lines were MG-63 and SJSA-1, with IC₅₀'s of 55 μ M and 100 μ M, respectively.

As observed with ZOL in chapter 3 and 4, differential sensitivity between cell lines was also observed with statin treatment. Furthermore, the same relative order of sensitivity was preserved for the two types of agents. Comparison of **Figure 3.1** (showing ZOL dose-response) and **Figure 5.1** and **5.2** (statin dose-response) shows a similar pattern of responsiveness between treatments of the same cell line.



Figure 5.1 The effect of Lovostatin on OS cells. Cells were seeded at 1 x 10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of LOV (1-100 μ M). Cell viability was determined using crystal violet staining, as described in the chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.



Figure 5.2 The effect of Mevastatin on OS cells. Cells were seeded at 1 x 10⁴ cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of MEV (1-100 μ M). Cell viability was determined using crystal violet staining, as described in chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.

5.3.2 Statins augment TRAIL-induced apoptosis in a dose-dependent manner

Experiments were performed to determine whether the augmented apoptotic response observed previously with combination of ZOL and TRAIL treatment (chapter 4), could also be seen with TRAIL in combination with the statins. HOS, BTK-143, G-292, Saos2, SJSA-1 and MG-63 OS cells were exposed to increasing concentrations of LOV or MEV (1-50 μ M) in the absence or presence of TRAIL. As described for ZOL in chapter 4, the concurrent treatment of LOV or MEV plus TRAIL for 72 hours induced more apoptosis than single drug alone, in all OS cell lines, although to varying degrees (**Figure 5.3** and **5.4**).

Lovostatin and TRAIL combination treatment reduced the IC₅₀ in all the OS cell lines when compared to Lovastatin alone, as shown in **Table 5.1**. In the G-292, BTK-143, Saos2, HOS, MG-63 and SJSA-1 cell lines respectively, up to 8, 19, 12, 25, 18 and 15% greater cell death respectively, was observed. Likewise MEV and TRAIL also reduced the in all the cell lines with G-292, BTK-143, Saos2, HOS, MG-63 and SJSA-1 cell lines, resulting in up to 14, 22, 26, 40, 15 and 8% increased cell death respectively (**Table 5.1**).

Cell morphology prior to cell death in the statin treated cells resembled that seen in ZOL treated cells, with loss of cell attachment a common factor. The time taken for apoptosis to occur was also similar, with most cell death occurring at 48-72 hours (data not shown).



Figure 5.3 The effects of Lovostatin and TRAIL combination treatment. OS cells were seeded at 1×10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of LOV 1-100µM, with ($_{\odot}$) and without ($_{\odot}$) 100ng/ml TRAIL. Cell viability was determined using crystal violet staining, as described in the chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.



Figure 5.4 The effect of Mevastatin and TRAIL combination treatment in OS cells. Cells were seeded at 1×10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of MEV 1-100µM, (\blacksquare) with and without (\bullet) 100ng/ml TRAIL. Cell viability was determined using crystal violet staining, as described in t.he chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.

	IC50 MEV	IC50 MEV+T	% Increase in cell death	IC50 LOV	IC50 LOV+T	% Increase in cell death
BTK-143	10	5	22% at 10μM	7	4	19% at 1μM
G-292	8	5	14% at 10μM	6	4	8 % at 1µM
ноѕ	32	16	40% at 25µM	18	13	25% at 10μM
MG-63	55	25	15% at 50μ Μ	22	16	18% at 25μM
Saos2	20	13	26% at 10μM	15	9	12% at 10μM
SJSA-1	100	90	8% at 25μM	51	41	15% at 25μM

Table 5.1 The IC₅₀ was determined for all OS cell lines either with Mevastatin (MEV) or Lovostatin (LOV) treatment alone or in combination with TRAIL. The maximum percent increase in cell death observed in the combination treatment is also shown for each cell line. IC₅₀ values are represented in μ M.

5.3.2 Prenyl transferase inhibitors induce dose-dependent apoptosis

To investigate the effect of PTIs on OS cells, HOS and BTK-143 cells were seeded into plates and allowed to adhere prior to the addition of fresh media containing increasing concentrations (1-50 μ M) of FTI and GGTI (see section 2.2.2 for details of PTIs used). These cells were shown to respond well to both ZOL and statins, and for this reason were chosen for the experiments described in this chapter. A dose-dependent decrease in cell viability was observed in both the FTI and GGTI treated cells, as shown in **Figures 5.5** and **5.6**. The IC₅₀'s of HOS and BTK-143 cells were 50 and 20 μ M, respectively, with FTI treatment, and 20 and 23 μ M, respectively, with GGTI treatment. The onset of cell death was similar to that observed with ZOL and statin treatment, as was the appearance of the cells prior to apoptosis.

5.3.4 Prenyl transferase inhibitors augment TRAIL-induced apoptosis in a dose-dependent manner

The addition of TRAIL to GGTI and FTI treated cells was investigated in HOS and BTK-143 cell lines and compared to single treatments to determine if combination treatment leads to an augmented apoptotic effect. Cells were treated for 72 hours with increasing concentrations of FTI and GGTI (1-50 μ M), in the presence or absence of TRAIL. Both FTI and GGTI in combination with TRAIL led to augmented cell death compared with single treatment alone, as shown in **Figures 5.7** and **5.8**. HOS cells showed up to 23% greater cell death in the combination treatments with TRAIL and both FTI and GGTI. An increase of 13% and 40%, respectively, greater cell death was observed in the BTK-143 cells in the combination treatments with FTI and GGTI.



Figure 5.5 The effect of Farnesyl transferase inhibitor on viability of OS cells. HOS and BTK-143 cells were seeded at 1×10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of FTI (1-50 μ M). Cell viability was determined using crystal violet staining, as described in the chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times. Results are expressed as % of control: points are means ±SD.



Figure 5.6 The effect of Geranylgeranyl transferase inhibitor on viability of OS cells. HOS and BTK-143 cells were seeded at 1×10^4 cells per well and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of GGTI (1-50µM). Cell viability was determined using crystal violet staining, as described in chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times. Results are expressed as % of control: points are means ±SD.



Figure 5.7 The effect of Farnesyl transferase inhibitor and TRAIL in combination. HOS and BTK-143 cells were seeded at 1×10^4 cells per well in 96 well microtiter plates and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of FTI 1-50µM, with (**■**) and without (**●**) 100ng/ml TRAIL. Cell viability was determined using crystal violet staining, as described in chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times. Results are expressed as % of control: points are means ±SD.



Figure 5.8 The effect of Geranylgeranyl transferase inhibitor and TRAIL combination on OS cells. HOS and BTK-143 cells were seeded at 1×10^4 cells per well and allowed to adhere. Cells were then incubated for 72 hours with fresh media containing increasing concentrations of GGTI 1-50µM, with (a) and without (b) 100ng/ml TRAIL. Cell viability was determined using crystal violet staining, as described in chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times. Results are expressed as % of control: points are means ±SD.

5.3.5 Normal human bone donor cells are not sensitised to TRAILinduced apoptosis by statins or prenyl transferase inhibitors

Experiments were performed to determine whether normal bone cells succumb to the effects of statins and PTIs either with or without TRAIL much like the OS cells, or whether they behave in a similar pattern to ZOL-treated NHB cells, that is, they are not sensitised to the combination treatments. Osteoblast-like cells derived from two donors were used to assess the effect of combination treatments of TRAIL with either statins or PTIs. The addition of either 25 μ M LOV, MEV, GGTI or FTI, with 100ng/ml TRAIL, to normal human bone (NHB) cells failed to exhibit augmented death in cells from either donor, for any of the combination treatments (shown in **Figure 5.9**). Therefore, NHB cells were not sensitised to TRAIL-induced apoptosis by statins or PTIs as were most of the OS cell lines. Although there was a slight reduction in the overall survival of cells, probably due to a small degree of inhibition to the mevalonate pathway, cell survival still exceeded that seen in any of the human OS cell lines.

5.3.6 Caspase inhibitors do not protect against apoptosis induced by inhibitors of the mevalonate pathway

The effect of the caspase inhibitor, z-VAD-fmk, on the induction of apoptosis in cells treated with a combination of statins or PTIs and TRAIL was examined. HOS cells were treated with the statins LOV or MEV, TRAIL and z-VAD-fmk, either alone or in combination. Addition of the caspase inhibitor z-VAD-fmk failed to protect these cells from statin-induced cell loss (**Figure 5.10A** and **5.10B**). However, the co-addition of the caspase inhibitor with TRAIL did prevent the augmented apoptotic effect, resulting in cell loss equal to the level induced by statin alone. Similar results were also observed



Figure 5.9 The inhibitors of the mevalonate pathway do not affect the viability of NHB cells. Cells from two donors (\blacksquare) and (\square)were seeded into plates and allowed to adhere before treatment for 72 hours with 25µM LOV, MEV, GGTI or FTI and a fixed concentration of 100ng/ml TRAIL. Cell viability was determined using the crystal violet assay and expressed as % of control, as described in chapter 2. Data shown in each case are from a representative experiment, which was repeated at least three times: points are means ±SD.



Figure 5.10 Caspase inhibitors do not protect against apoptosis induced by inhibitors of the mevalonate pathway. HOS cells were treated for 72 hours with (A) 10 μ M LOV, (B) 10 μ M MEV, (C) 10 μ M GGTI, or (D) 10 μ M FTI, 100ng/ml TRAIL, and the broad specificity caspase inhibitors z-VAD-fmk (50 μ M), either alone or in combination. Cell viability was determined using the crystal violet assay and cell viability is expressed as % of control. Data points show means of triplicate results from a representative experiment, which was repeated at least twice; *bars* ±SD.

using GGTI or FTI with TRAIL and z-VAD-fmk (**Figure 5.10C** and **5.10D**). This supports the previous suggestion that statins and PTIs, like ZOL, may sensitise OS cells to the effects of TRAIL *via* activation of the caspase cascade and that the addition of the caspase inhibitor z-VAD-fmk protects against this sensitisation by irreversibly inhibiting caspase-3 activity.

5.3.7 Geranylgeraniol is able to rescue cells from the effects of statins but not prenyl transferase inhibitors

Geranylgeraniol was added to combination treatments of statins or PTIs and TRAIL, to examine its effect on the induction of apoptosis. HOS cells were treated with GGO, TRAIL, or together with LOV, MEV, GGTI or FTI either alone or in combination. The addition of GGO to the statin treated cells was able to protect the cells from statin-induced apoptosis, as shown in **Figure 5.11A** and **5.11B**. Combination treatment with LOV, TRAIL and GGO led to a 60% increase in cell viability, when compared to the corresponding LOV and TRAIL treated cells. Similarly MEV, TRAIL and GGO treated cells showed 45% improvement when compared to MEV and TRAIL treated cells.

In contrast to the N-BPs and statins, GGO did not protect the cells from PTIinduced apoptosis. This was also observed in the combination of GGO, PTI, and TRAIL treated cells, with cell death equal to that observed with PTI and TRAIL alone (**Figure 5.11C** and **5.11D**). This suggests that GGO is unable to act as an intermediate of the mevalonate pathway and therefore unable to rescue the cells from apoptosis. This will be discussed further in section 5.4.

Western blot analysis revealed PARP cleavage in all combination IMP and TRAIL treated cell lysates (Figure 5.12). At 48 hours however, there was no PARP



Figure 5.11 Geranylgeraniol (GGO) is able to rescue cells from the effects of statins but not prenyl transferase inhibitors. HOS cells were treated for 72 hours with (A) 10 μ M LOV, (B) 10 μ M MEV, (C) 10 μ M GGTI, or (D) 10 μ M FTI, 100ng/ml TRAIL, and 300 μ M GGO, either alone or in combination. Cell viability was determined using the crystal violet assay and cell viability is expressed as % of control. Data points show means of triplicate results from a representative experiment, which was repeated at least twice; *bars* ±SD.

J



Figure 5.12 Western blot analysis of HOS cells after 48 hours treatment showing *(i)* increase in PARP cleavage when TRAIL (T) is combined with 25 μ M ZOL (Z) or LOV (L) or MEV (M). *(ii)* ZOL, LOV and MEV inhibit protein prenylation as assessed by the presence of unprenylated Rap 1A. These are all compared to control untreated cells (C). *(iii)* No significant changes in the levels of Bcl-2 and Bax are detected.

cleavage with IMP treatment only. Detection of the unprenylated form of RAP 1A was identified in all cell lysates given IMP treatment but not detected in control or TRAIL treated cells (**Figure 5.12**). As observed with ZOL treated cells, no significant changes in the levels of Bcl-2 or Bax were detected (**Figure 5.12**).
5.4 **DISCUSSION**

In this study, statins and PTIs were used, either alone or in combination with TRAIL, to assess their effectiveness as alternative inhibitors of the mevalonate pathway to ZOL. Like ZOL, both statins (MEV and LOV) and PTIs (FTI and GGTI) exhibited direct effects on proliferation and survival of OS cells *in vitro* in a dose-dependent manner, supporting previous reports in other tumour types [Dimitroulakos *et al.* 1999; Dimitroulakos *et al.* 2001; Newman *et al.* 1994; Rubins *et al.* 1998; Jones *et al.* 1994; Muller *et al.* 1998; Wong *et al.* 2002; Chan *et al.* 2003; Bolick *et al.* 2003; Barrington *et al.* 1998; Sun *et al.* 1998; Lobell *et al.* 2001].

The augmentation of TRAIL induced apoptosis was not restricted to ZOL but appears to be a general phenomenon of inhibitors of the mevalonate pathway, with statins and PTIs capable of augmenting TRAIL-induced apoptosis in a similar manner to that observed with ZOL. A similar pattern of reduced cell viability was reflected between ZOL, statin and PTI treatments of each cell line. Comparison of the IC₅₀'s between different treatments demonstrates modest augmentation in all cell lines (Table 5.2). Single treatment of the sensitive cell lines HOS, BTK-143 and G-292 with either IMP shows a similar pattern of sensitivity. The BTK-143 cell line is the most sensitive to all five agents with the lowest values seen with LOV, slightly better than ZOL and MEV. Combination treatment of all agents with TRAIL in the BTK-143 cells reduced the IC₅₀ value to between 4-11 μ M for all treatments. IMP treatments in HOS cells revealed ZOL to be more potent followed by the statins and the PTIs, However combination treatment reduced the IC_{50} values to between 5 and 16 μ M. In contrast the statins had a more potent effect in the G-292 cells than ZOL, however dual treatments with TRAIL reduced the IC50 values to between 4 and 8 µM. The SJSA-1, Saos2 and MG-63 cell lines, considered to be more resistant to ZOL, also required greater amounts of statin to achieve 50% cell death. However MG-63 and Saos2 cells were more

	IC ₅₀		IC ₅₀		IC ₅₀		IC ₅₀		IC ₅₀	
	Z	Z+T	М	M+T	L	L+T	F	F+T	G	G+T
BTK-143	10	5	10	5	7	4	21	11	25	7
HOS	8	5	32	16	18	13	50	10	19	13
G-292	15	8	8	5	6	4	-	-	21414	-
MG-63	>100	90	55	25	22	16	-		-	-
Saos2	50	27	20	13	15	9	-	-	-	-
SJSA-1	35	23	100	90	51	41	-	: ==	-	-
к-ноs	45	16	-	-	-	-	-	(#	-	-
KRIB	25	18	-	-	-	-	-	-	-	-

Table 5.2 Combined table of IC₅₀ values for each cell line. Treatments are listed as T (100ng/ml TRAIL), Z (25 μ M ZOL), M (25 μ M MEV), L (25 μ M LOV), F (25 μ M FTI) and G (25 μ M GGTI). IC₅₀ values are represented in μ M.

susceptible to statin than ZOL with much lower IC₅₀ values. This was not observed for the SJSA-1 cell line. All three cell lines resulted in reduced IC₅₀ values with the combination TRAIL treatments, with the Saos2 values within the same range as that seen in the sensitive cell lines. The possible reasons for the differences observed between agents may depend on various factors, such as cell type, which may lead to variations in the expression of proteins involved in apoptotic or survival pathways. Other factors include variations in the chemical structure of the agents, which can reflect potency, as well as differences in efficient uptake of the agent into the cell [Lennernas and Fager 1997]. Further analysis of a greater range of statins and PTIs may prove beneficial in identifying a more potent alternative to ZOL. The concentration of statins used in this study to induce apoptosis compare with achievable concentrations in vivo, especially when given in combination with TRAIL. Phase I clinical trials administering LOV at a dose of 25 mg/kg/day, led to plasma concentrations of up to 3.92 μ M, which corresponds to the dose range that can trigger apoptosis of sensitive tumour cells in vitro [Dimitroulakos et al. 2001; Dimitroulakos et al. 1999]. Since then phase II and III trials have used LOV at 40mg/kg/dose four times daily, increasing plasma concentrations further [Minden et al. 2000]. As there are many types of statins and PTIs available, further investigation using a greater number of each type of agent, may identify a more potent agent capable of inducing greater cell death. However, the degree of cell death induced by both the statins and PTIs chosen for this study is significant and warrants further investigation as potential anticancer agents.

An important goal of anticancer strategies is the selective induction of apoptosis in cancer cells but not normal cells, that is, to specifically inhibit malignant growth with minimal systemic toxicity. Both statin and PTI treatment of NHB cells resulted in reduced cell death than that observed in human OS cells, even with the co-addition of TRAIL, supporting previous results obtained with ZOL.

Caspase inhibitors failed to protect cells from statin and PTI induced apoptosis, suggesting that their mode of action mimics that of ZOL, that is, apoptosis does not rely on the activation of the caspase cascade directly. On the other hand, the addition of GGO, an intermediate of the mevalonate pathway, was able to rescue the cells treated with statins but not PTIs. The addition of GGO to cells treated with statins and TRAIL, protected them from the augmented apoptotic effect observed with statin and TRAIL treatment alone. Although protection was not equal to that of TRAIL treatment alone, cell viability in the GGO/statin/TRAIL treated cells was double that of the statin/TRAIL treated cells for both MEV and LOV. Combination of FTI or GGTI and TRAIL treated cells were not protected by GGO and cell death was equal to that observed in the dual combination treatments. This difference is due to the effect of GGO within the mevalonate pathway. As statins inhibit an enzyme required early in the mevalonate pathway, the addition of GGO rescues the cell as it is downstream, therefore acts as an intermediate (Figure 5.13). FTI and GGTI inhibit the enzymes responsible for prenylation itself and GGO is unable to rescue the cells at this point. Thus it is not surprising that this difference exists.

Consistent with the ZOL and TRAIL combination treated cells, HOS cells treated with both MEV and LOV in combination with TRAIL also exhibited PARP cleavage, confirming the increased apoptotic efficacy of the combination treatments. Similarly all cells treated with IMP displayed the presence of the unprenylated form of RAP 1A after western blotting, confirming that the mode of action of these agents is via inhibition of the mevalonate pathway.

It is possible that ZOL-induced anoikis, or prevention of attachment, regulates the levels of pro and anti-apoptotic factors below a threshold required for sensitisation of TRAIL-resistant OS cells to TRAIL-induced apoptosis. It was recently demonstrated that anoikis leads to the translocation of the pro-apoptotic protein, Bmf, to the



Figure 5.13 Geranylgeraniol (GGO) acts as an intermediate of the mevalonate pathway. Nitrogen-containing bisphosphonates (N-BPs), statins and Prenyl Transferase Inhibitors (PTIs) inhibit the enzymes indicated above, preventing downstream processes including farnesylation and geranylgeranylation., GGO allows geranylgeranylation to occur if enzymes upstream are inhibited.

mitochondria where it binds to, and antagonises the activity of, the pro-survival protein Bcl-2 [Puthalakath *et al.* 2001]. The extrinsic and intrinsic pathways communicate with each other, and death receptors can trigger activation of the *intrinsic* pathway by caspase 8 mediated cleavage of Bid, which interacts with other members of the Bcl-2 superfamily including Bax, leading to activation of caspase 9. No changes in Bcl-2 and Bax protein expression were observed for both ZOL and statin treated cells. This conflicts reports for ZOL and statins which have been shown to down-regulate Bcl-2 expression [Blanco-Colio *et al.* 2002; Wong *et al.* 2002; Senaratne *et al.* 2000; Oades *et al.* 2003]. In some cells the anti-apoptotic effects are partly attributed to the regulation of several inhibitors of apoptosis including c-IAP1, c-IAP2 and XIAP [Wang *et al.* 1998; Deveraux and Reed 1999]. A more exhaustive analysis is required to identify expression changes in proteins of the Bcl-2 superfamily as well as inhibitors of apoptosis, which may be involved [Wang *et al.* 1998; Deveraux and Reed 1999].

The use of statins and PTIs in *in vivo* models may prove more beneficial than ZOL, as they do not have an affinity for bone, thus their widespread systemic distribution may also be used for the treatment of other non-osseous malignancies including lung metastases, often associated with increased mortality in OS. Most importantly the lack of effect on normal bone cells by the IMP make them very desirable anti-cancer agents however further investigation is required to fully understand the mechanism of augmentation with TRAIL and why they are not uniformly effective in all cell lines. One way to overcome this problem may be in the form of a predictive test to determine the best course of therapy for OS patients.

CHAPTER SIX

ESTABLISHMENT OF IN VIVO OSTEOSARCOMA

MOUSE MODEL

6.1 INTRODUCTION

Thus far there have been no reports investigating the potential anti-tumour activity of TRAIL alone, or in combination with BPs, statins or chemotherapy, in bone tumours *in vivo*. The aim of the work described in this chapter and the next was to evaluate these potential anti-tumour agents in an *in vivo* model. It was also considered important to test the ability of these agents to suppress the metastatic spread of human OS.

To begin investigating these agents in an *in vivo* environment, an OS animal model was established. This model was based on a simple technique, by which human OS cells are transplanted orthotopically into the tibia of athymic mice, as described by Berlin and colleagues [Berlin *et al.* 1993]. Berlin used the v-Ki-ras-oncogene-transformed human osteosarcoma cell line (KRIB) to develop a spontaneous metastasis model, in which a small number of tumour cells are injected into the tibial bones of athymic mice. Four weeks after intra-tibial transplantation, the KRIB cell line was shown to develop bone tumours, which were radiographically and histologically similar to primary human osteosarcoma [Berlin *et al.* 1993]. Furthermore, as in the human disease, cells from these primary tumours efficiently colonize the lungs of athymic nude mice, resulting in reproducible and quantifiable pulmonary metastases, evident both upon gross inspection of the lungs and histologically 6 weeks after tumour inoculation.

A pilot study was set up to identify a suitable cell line and an appropriate cell number for use in animal experiments. As noted in earlier chapters, KRIB OS cells are refractory to the apoptotic effects of TRAIL as a single agent. In addition, the presence of ZOL *in vitro* has little effect in sensitising KRIB cells to TRAIL-induced apoptosis. Other human OS cell lines, such as K-HOS, are also unaffected to TRAIL treatment alone, however, are strongly sensitised to the effects of TRAIL with co-treatment of ZOL. For this reason K-HOS cells were also chosen as a comparison to KRIB cells, for the pilot study.

6.2 MATERIALS AND METHODS

6.2.1 Cells and cell culture

The osteosarcoma cell lines used in this chapter are listed in section 2.5.2. The methods for cell preparation and propagation are described in section 2.1.3 and 2.5.1. Cell suspensions of K-HOS and KRIB cells in sterile PBS, at 2 X 10^3 , 5 X 10^3 or 1 X 10^4 cells/20µl, were implanted into the proximal tibia. Mice were weighed twice weekly and progress of the tumour was monitored using radiography. At time of sacrifice, organs were collected for histology and tibiae were removed for micro-CT analysis.

6.2.2 Mouse specifications and maintenance

Animal specifications and housing are detailed in section 2.5.3.

6.2.3 Intra-tibial implantation: delivery and schedule

For detailed explanation of the intra-tibial implantation method please refer to section 2.5.4.

6.2.4 Histology and image analysis

Histological sections were prepared according to the protocol detailed in section 2.5.6. Radiography and micro-CT analysis were performed as listed in section 2.5.7 and 2.5.8 respectively.

6.3 **RESULTS**

6.3.1 Lytic lesions identified using radiograph analysis

The pilot study consisted of a total of 6 animals. Three animals were injected with varying concentrations of 2 X 10^3 , 5 X 10^3 or 1 X 10^4 cells/20µl using KRIB cells. The remaining animals were injected with K-HOS cells at the same concentrations. After a minimum lag time of 4 weeks from intra-tibial injection of both KRIB and K-HOS cells, macroscopic tumours were evident in the tibiae of athymic mice as shown in Figure 6.1. Assessment by radiography at 2 weekly intervals revealed the development of osteolytic lesions that also progressively increased with time. The animal injected with 1 X 10^4 cells/20µl of KRIB cells exhibited some evidence of osteolysis at 3 weeks while all KRIB-injected animals showed clear osteolytic lesions by 5 weeks. None of the animals inoculated with K-HOS cells showed any evidence of osteolysis at 3 weeks using radiography but osteolysis was evident at 5 weeks. Animals were sacrificed if they developed large tumours with associated soft-tissue swelling and had some difficulty with movement or when they reached 8 weeks post tibial-inoculation. All animals injected with KRIB OS cells were sacrificed by 6 weeks due to the significant degree of osteolysis observed by radiography, however only one of the K-HOS-injected animals was sacrificed prior to 8 weeks because it also exhibited significant osteolysis. Radiographs were taken on the day of sacrifice and compared (Figure 6.2 and 6.3). All 3 animals injected with KRIB OS cells demonstrated considerably more osteolysis than the animals injected with K-HOS OS cells. However, the animal injected with 5,000 K-HOS cells exhibited significant osteolysis and a decision was made to use K-HOS cells for the study of the anti-cancer potential of ZOL and TRAIL in vivo. As animals were sacrificed at varying time points depending on the degree of osteolysis, the cell number only gives an indication of the time required for lesions to occur.



Figure 6.1 Intra-tibial injection of human OS cells resulted in the development of tumours. Photos were taken 6 weeks post intra-tibial injection. Clearly defined tumours were evident in mice injected with either K-HOS or KRIB cells.



Figure 6.2 Radiographic analysis of mice tibiae from the K-HOS Pilot study. Mice were injected with (A) 2×10^3 , (B) 5×10^3 or (C) 1×10^4 cells/20µl into the left tibia for A and B and right tibia for C. Radiographs of both tibiae were taken at 4 weeks post inoculation.



Figure 6.3 Radiographic analysis of mice tibiae from the KRIB Pilot study. Mice were injected with (A) 2×10^3 , (B) 5×10^3 or (C) 1×10^4 cells/20µl into the left tibia for A and B and both right and left tibia for C. Radiographs of both tibiae were taken at 4 weeks post inoculation.

6.3.2 Improved imaging of lytic lesions using micro-CT analysis

Micro-computer tomography (μ CT) was performed on the tibias of these animals to generate images of superior quality as compared to the radiograph. For microtomographical reconstruction, transmission X-ray images are acquired from 200 rotation views over 180 degrees of rotation. Each rotation view contains 1,004 lines, which can produce a two-dimensional (2-D) section representing 17.36 microns, when the same line from each rotation view is collated. The use of sophisticated software allows further image reconstruction of these 2-D sections leading to the generation of a three dimensional (3-D) image, which is far superior in detail and provides more information to that obtained by standard radiography. The 3D images can be rotated in any direction offering improved analysis of osteolytic destruction than two-dimensional images. These images can also be manipulated so that the tibias can be dissected, to allow investigation of inner bone structure and the degree of osteolysis from within the tibia. In addition the use of μ CT and the associated computer programs, allows bone volume quantification to be assessed.

The μ CT images obtained from mice injected with either K-HOS (**Figure 6.4**) or KRIB (**Figure 6.5**) cells both exhibited bone destruction however the extent of osteolysis was more severe in the animal injected with KRIB cells. These are comparable with the radiographs taken at time of sacrifice, however, the poorly defined regions of osteolysis in the radiographs are clearly detailed in the μ CT, providing much greater insight into the destructive nature of the tumour. From the μ CT images it is evident that the tibias with the transplanted K-HOS cells are uneven in appearance as opposed to the smooth contour of the non-transplanted tibia (**Figure 6.4**).



Figure 6.4 Micro-Computer Tomographic analysis of the tibiae removed from a nude mouse. 5 X 10³ K-HOS cells were inoculated into the left proximal tibia and the animal was sacrificed 6 weeks later. *Top Image*: μ CT of both tibiae, *Bottom Image*: Sliced images of the left and right tibiae using μ CT software, reveals the extent of osteolytic damage in the left tibia.



Figure 6.5 Micro-Computer Tomographic analysis of the tibiae removed from a nude mouse. 2 X 10^3 KRIB cells were inoculated into the left proximal tibia and the animal was sacrificed 6 weeks later. μ CT analysis reveals the extent of osteolytic damage in the tibia.

6.3.3 Histological analysis

Following μ CT imaging, the tibias were decalcified and embedded in paraffin for sectioning and staining (refer to section 2.5.6 for further details). Both the K-HOS or KRIB cells reproducibly developed large lesions that invaded the marrow cavity and began to erode the cortical bone (**Figure 6.6**).

Histological analysis of kidney, liver, and spleen sections obtained from animals injected with either the K-HOS or KRIB cells revealed no metastatic lesions to these organs (data not shown). Multiple pulmonary macro- and micro-metastases were identified for both cell types in all animals (**Figure 6.7**).



Figure 6.6 Histological examination of intra-tibial OS. Tumours were identified within the bone marrow cavity after injection of either (A) KRIB and (B) K-HOS into tibiae of nude mice. Regions of tumour are identified with white arrows.



Figure 6.7 Histological examination of lung tissue from nude mice inoculated with human OS cells. Multiple micro-metastatic tumours were identified in the lungs of both (A) KRIB and (B) K-Hos inoculated mice. Regions of tumour are identified with white arrows.

6.4 **DISCUSSION**

The establishment of an in vivo mouse model of OS allows for the investigation of the efficacy of potential anti-cancer agents. Both K-HOS and KRIB OS cell lines were able to produce locally growing tumours in the immediate area surrounding the tibial site of injection. It is clearly evident that the animals inoculated with KRIB OS cells showed a higher degree of osteolysis when compared to animals inoculated with K-HOS OS cells. Histology also confirmed the presence of tumours within the marrow cavity and eroding into the cortical bone. Pulmonary metastases were identified in animals injected with either cell line, however metastases were not discovered in any other organs. Micro-CT analysis provided far greater image resolution than radiography and supported the radiographs taken at time of sacrifice. However unlike µCT imaging, the use of radiography allowed for the continual monitoring of the tumour area, with the first signs of osteolysis observed at week 3 in the mouse injected with the highest concentration of KRIB cells. One limitation of continual monitoring using radiography is that the tumour has to penetrate through the cortex before it is evident on a radiograph. Thus it is not a clear indication of early tumour development. Both cell lines however required at least 4 weeks to develop macroscopic tumours. A possible reason for the reduced amount of osteolysis in the animal injected with 1 X 10⁴ K-HOS cells compared to the animal with 5 X 10^3 K-HOS cells may be due to the delivery of the cells. This needs to be confirmed with a greater number of animals to test the accuracy of the delivery system. It is difficult to ensure that such a high number of cells in such a small volume are delivered accurately, therefore the number of cells delivered may vary. In addition, it is impossible to inject into exactly the same position of the tibia in every mouse, with variations in injection sites potentially providing differing environments for the cells. Finally, it is likely that cells are left in the surrounding soft tissue as the needle withdraws from the tibia, leading to extra-osseous growth of these

cells, which does not represent the normal growth and progression of OS. The tumour burden outside the tibia may be one reason for the uneven appearance of the bone surface observed in the K-HOS tumour-ridden tibiae but not within direct proximity to the injection site, indicating bone remodelling of the periosteal surface. It is possible that these tumours may have a mixed osteolytic and osteoblastic phenotype and the observed response may be due to the extra-osseous tumour stimulating bone remodelling in both the tibia and fibula.

As mentioned in previous chapters, the KRIB cell line does not show a significant augmented apoptotic effect when treated with both ZOL and TRAIL. Therefore the KRIB animal model of OS is unsuitable for the investigation of the efficacy of these agents *in vivo*. For this reason, the K-HOS animal model was chosen for future studies.

CHAPTER SEVEN

INVESTIGATION INTO THE EFFECT OF ZOLEDRONIC ACID AND TRAIL TREATMENT IN NUDE MICE:

Will it work *in vivo*?

7.1 INTRODUCTION

In vivo studies in numerous animal models of metastatic cancer have demonstrated a reduction in tumour-induced osteolysis with ZOL treatment [Dallas *et al.* 1999; Hall and Stoica 1994; Nemoto *et al.* 1990]. Anti-angiogenic properties of ZOL have also been reported, making this class of bisphosphonates attractive agents in the treatment of metastatic cancers [Croucher *et al.* 2003]. The use of ZOL in patients with bone metastases from various types of primary tumours has shown significant clinical benefits, perhaps superior to other BPs. To date however, there have been no reports of ZOL treatment of OS *in vivo*.

The ability of TRAIL to induce apoptotic cell death in a wide variety of tumour cell lines, but not normal cells *in vitro* or *in vivo*, makes it appealing for the treatment of cancer. More importantly, sensitisation of OS cells but not normal human bone cells to TRAIL induced-apoptosis by ZOL, delivers huge potential into anti-cancer therapy.

With the establishment of a mouse model of OS, the aim of the work described in this chapter was to investigate the effect of ZOL and TRAIL as single agents and in combination, in delivering an apoptotic stimulus to OS cells *in vivo*. Such an outcome would bring the hope of achieving substantial apoptotic response in this tumour type without the toxic side effects often seen with conventional and current therapies.

7.2 MATERIALS AND METHODS

7.2.1 Cells and cell culture

The OS cell line K-HOS used in this chapter is listed in section 2.5.2. The methods for cell preparation and propagation are described in section 2.1.3 and 2.5.1. Cell suspensions of K-HOS cells in sterile PBS, at 6 X 10^3 cells/10µl, were implanted into the tibiae of nude mice. Mice were weighed twice weekly and progress of the tumour growth and osteolytic activity was monitored using radiography. At time of sacrifice, organs were collected for histology and tibiae were removed for micro-CT analysis.

7.2.2 Mouse specifications and maintenance

Animal specifications and housing are detailed in section 2.5.3.

7.2.3 In vivo treatment: delivery and schedule

For detailed explanation of the intra-tibial implantation method refer to section 2.5.4. Treatment schedules for ZOL and TRAIL are listed in section 2.5.5.

7.2.4 Histology and image analysis

Histological sections were prepared according to the protocol detailed in section 2.5.6. Radiography and micro-CT analysis were performed as described in sections 2.5.8 and 2.5.7, respectively.

7.2.5 Statistical analysis

Data obtained for each treatment group regarding the presence of osteolytic lesions and lung metastases, was assessed using Fisher's exact test. In each of these cases, the interaction between ZOL and TRAIL was assessed using the Breslow-Day test for homogeneity of odds ratios.

Impact of ZOL and TRAIL and an interaction between the two on bone volume in the left leg was assessed *via* 2-way Analysis of Covariance (ANCOVA), where bone volume in the right leg was the covariate. Impact of ZOL and TRAIL, and an interaction between the two, on tumour area (where tumours were detected) was assessed using the 2-way Analysis of Variance (ANOVA).

7.3 **RESULTS**

7.3.1 ZOL protects the skeleton from tumour-induced osteolysis

K-HOS cells, at a concentration of $6 \ge 10^3/10\mu$ l (see section 2.5.4), were prepared for intra-tibial injection of 4 to 6 week old Balb/c Nu/Nu female mice. Tibiae of 39 animals were inoculated on Day 0 and animals were assigned randomly into 4 groups: (1) the control untreated group (9 mice), (2) the TRAIL treated group (9 mice), (3) the ZOL treated group (10 mice) and (4) the combination of ZOL and TRAIL group (11 mice). To monitor the progression of tumours, mice were weighed twice a week and radiographs were taken fortnightly while under anaesthetic.

Commencement of treatment was at Day 7 post cancer cell transplantation, to allow sufficient establishment of tumour cells within the bone environment. TRAIL was administered *via* intraperitoneal injection at a dose of 30 mg/kg/dose, once a day for 5 consecutive days, followed by once weekly injection at the same dose (Figure 7.1). ZOL was commenced on Day 7, and was given subcutaneously once weekly at a dose of $100 \text{ }\mu\text{g/kg}$ for each injection until sacrifice.

By week 4, several animals had developed large tumours that hindered normal movement of the inoculated leg and the mice were beginning to show signs of discomfort. It was recommended by the animal ethics committee that the experiment be terminated early, to prevent any suffering of the mice. For this reason the experiment was terminated at Day 35.

Radiographs of all the animals were taken on the day of sacrifice, prior to removal of the tibiae for μ CT and histological analysis. **Figure 7.2** shows radiographs, with clearly defined osteolytic lesions in 3 of the 9 control untreated animals. The TRAIL treated group resulted in 2 animals with noticeable lesions, based on the radiographs (**Figure 7.3**). **Figure 7.4** and **7.5** show radiographs taken from the ZOL and

TRAIL 30 mg/kg, (i.p.)





Figure 7.1 Dosing regimen for *in vivo* OS experiment. Mice were inoculated with K-HOS cells into the left proximal tibia at day 0. Treatments were commenced from day 7 onwards with TRAIL and/or ZOL. Mice were sacrificed at day 35 for further analysis. A suitable control group of mice were also included which received the intra-tibial injection but given no treatment.





Figure 7.2 Radiographs of both tibiae, in mice of the control untreated group. Radiographs were taken at Day 35. Evidence of osteolytic lesions are identified with white arrows.





Figure 7.3 Radiographs of both tibiae, in mice of the TRAIL treated group. Radiographs were taken at Day 35. Evidence of clearly defined osteolytic lesions are identified with white arrows.



d,



combination of ZOL and TRAIL treated groups, respectively. Both groups show no signs of bone lysis from the radiographs. The level of resolution of these images makes it difficult to identify small lesions with any certainty. Micro-CT analysis proved far more useful in identifying the extent of damage from the OS tumours, as well as the efficacy of the chosen treatments.

Using the SkyScan-1072 micro-CT, one pair of tibiae per run was scanned, with a total of 4 runs per day. Following scanning, the resulting TIFF files from one run (which totalled approximately 1 gigabyte of computer memory), were converted into BMP files, a process requiring between 6-8 hours. The files were then imported into the μ CT 3-D software and 3-D images generated. The 3-D software enables editing and rotation of the image, allowing modifications to be made and further investigation of the tibiae, such as slicing through the middle to reveal the inner bone cavity. Although the process of generating 3-D images using μ CT is a lengthy and laborious task, the quality of the images is considerably more informative than radiographs alone, which justifies the investment of time in this procedure.

The μ CT images generated for each group of animals were collated into treatment groups for comparative analyses. **Figure 7.6** shows a more comprehensive view of the effect of K-HOS cells in the tibia of untreated control mice when compared to the corresponding radiograph (**Figure 7.2**). Five out of 9 animals in this group exhibited obvious differences between the inoculated tibia and the control tibia. It is worth noting that, although the same 3 animals (#4, #8, #9) identified as having osteolysis in the radiographs show lytic lesions in the μ CT images, animals #1 and #5 also had small lytic lesions, which were not apparent on the radiographs. Animals #4 and #8 had both osteolytic and presumed 'osteoblastic' lesions, as discussed below. Animal #2 exhibited only the 'osteoblastic' phenotype. This feature would not have otherwise been identified, if μ CT analysis had not been conducted. The images from the untreated





Figure 7.6 Micro-CT generated images of tibiae in the control untreated group. Osteolytic lesions are indicated with red arrows. Osteoblastic lesions are indicated with blue arrows.

control group were sliced through the middle (Figure 7.7) and showed significant damage through the inner bone cavity, due to osteolytic lesions, in animals #1, #4, #5, #8 and # 9. Animal #9 also showed thickening of the bone in the areas of increased bone formation. For the TRAIL treated group, radiographs revealed osteolytic lesions in animal #3 and #7 (Figure 7.3). The lesions were confirmed using μ CT analysis, however another 3 animals (#1, #2 and #8) had the 'osteoblastic' phenotype in the inoculated tibia, as shown in Figure 7.8. Longitudinal tibial sections of μ CT images also confirmed this finding, however increased bone thickness was not evident in animal #8 as the outer bone surface is not visible in this photo (Figure 7.9).

None of the ZOL treated animals showed any signs of osteolysis, as judged from the radiographic images (**Figure 7.4**). This was reflected in the μ CT images, with 3 of the 10 animals showing the 'osteoblastic' phenotype, as depicted in **Figure 7.10** and **7.11**. Similarly, the radiographs from the combination of the ZOL and TRAIL treated animals showed no osteolytic lesions (**Figure 7.5**) and this was supported by the μ CT images; however, 7 of the 11 animals did show areas of increased bone formation, as seen in **Figure 7.12**, which was not always apparent in the sliced images (**Figure 7.13**).

Due to the absence of osteolytic lesions in both the ZOL and combination ZOL and TRAIL treatment groups, it is not possible to assess the homogeneity of odds ratios between ZOL and TRAIL, therefore it could not be tested whether the combined effect is statistically different from ZOL or TRAIL alone. However, the ZOL treated group is statistically significantly different to the control untreated group (p=0.01) in terms of the number of osteolytic lesions.

Of particular interest was the development of the 'osteoblastic' phenotype observed in 18 of the 39 tumour-injected animals. Figure 7.14 follows the process of μ CT analysis beginning with scanning of the tibia, generation of the 2-D images and then creation of a 3 dimensional image. Increased areas of bone density were evident in

Figure 7.7 Micro-CT generated images of tibiae sliced in half from all animals in the control untreated group.

Osteolytic lesions are indicated with red arrows. Osteoblastic lesions are indicated with blue arrows.

9



1. 16.5

, == y = ___




Figure 7.8 Micro-CT generated images of tibiae in the TRAIL treated group. Osteolytic lesions are indicated with red arrows. Osteoblastic lesions are indicated with blue arrows.

Figure 7.9 Micro-CT generated images of tibiae sliced in half, from all animals in the TRAIL group.

Osteolytic lesions are indicated with red arrows. Osteoblastic lesions are indicated with blue arrows.

 \widetilde{v}^{*}



 $\mathbf{a}^{(i)}$



Group 3- ZOL



an an thèin an an an

Group 3- ZOL









Figure 7.11 Micro-CT generated images of tibiae sliced in half, from all animals in the ZOL treated group. Osteoblastic lesions are indicated with blue arrows.





÷

Group 5- ZOL+TRAIL











Figure 7.13 Micro-CT generated images of tibiae sliced in half, from all animals in the ZOL and TRAIL treated group. Osteoblastic lesions are indicated with blue arrows.



Figure 7.14 Comparison images showing the characteristic 'osteoblastic' phenotype observed in the K-HOS treated tibia (left) but not the contralateral tibia (right). *A*. Image of the tibia as observed during μ CT analysis. *B*. 2D image of the tibia obtained during μ CT analysis. The image on the left shows areas of increased density surrounding the tibia, represented by darker flame-like protrusions which are absent in the non-injected right tibia *C*. 3D images generated from 2D data reveal a high degree of 'osteoblastic' activity in the tibia injected with K-HOS cells.

the 2-D images, with flame-like protrusions typical in all tibiae exhibiting the 'osteoblastic' phenotype. These protrusions only extend from the tibiae outwards and, rather than pockets of mineral deposits as a result of the tumour itself, appear to be regions of accelerated bone formation. Possible reasons for this are discussed later in the chapter. Due to the decalcification process prior to histology, the 'osteoblastic' phenotype was not clearly visible in many sections, however some sections of the inoculated tibiae show clear evidence of these mineral deposits, as shown in **Figure 7.15**.

7.3.2 ZOL increases bone volume

Closer examination of the radiographs demonstrated that in all ZOL treated animals, and not in the untreated animals, the bones were highly radio-dense, suggesting increased bone density. This effect of ZOL was more pronounced in areas of increased bone turnover such as the distal femurs and proximal tibiae and was not restricted to the tumour site since the contralateral tibiae also showed this effect (**Figure 7.16**). Longitudinal sections of μ CT images were compared between treatment groups. These images revealed that both groups treated with ZOL showed significant increases in trabecular density and cortical bone thickness, as shown in **Figure 7.16**).

The computer program CTan was used to analyse bone volume of the scanned tibiae. Using the 2-D images obtained from the μ CT scan, the growth plate was identified and 150 sections, starting from the growth plate/tibial interface and moving toward the distal tibia, were selected, with each section representing 17.36 microns. Graphs representing bone volume were generated and compared to the control tibia. Differences between the left and right tibia were noted across all treatment groups, as



Figure 7.15 The photograph shown is a haematoxylin and eosin stained section of a mouse tibiae at 35 days post inoculation with K-HOS cells. Black arrows indicate areas of uneven bone growth which are shown as protrusions in the μ CT images.



Figure 7.16 Radiographs of the inoculated tibiae at two week intervals. Photo A and B represent a control untreated animal; C and D represent a TRAIL treated animal; E and F represent a ZOL treated animal; and G and H represent a combination ZOL and TRAIL treated animal. White arrows indicate areas of osteolytic lesions; Red arrows indicated areas of increased radiodensity.



Figure 7.17 Micro-CT images of tibiae sliced through the middle from; *A*. control untreated animal and *B*. ZOL treated animal. Blue arrows indicate increases in trabecular density; Red arrows indicate increases in cortical bone thickness.

shown in **Figure 7.18**, with a slightly decreased average bone volume in the inoculated tibia of the control group and a slightly increased average bone volume in the inoculated tibia of the TRAIL treated group. However, it is important to note that there was a significant increase in bone volume between the ZOL treated groups and the non-ZOL treated groups. Bone volume increased by up to 2 mm³ with ZOL treatment. This correlates with the increased radio-density of the radiographs and the increased thickness of the trabecular and cortical bone observed with μ CT analysis. Marked differences in histology, between the ZOL treated animals were also apparent (**Figure 7.19**). The histological findings correlate well with the radiographs and the μ CT images.

There was a statistically significant increase in bone volume due to ZOL treatment (p=0.0414) compared to the non-ZOL treatment groups. There was no interaction between ZOL and TRAIL nor was there any significant effect of TRAIL on bone volume in tumour-bearing legs.

7.3.3 ZOL had no effect on tibial tumour burden but TRAIL reduced tumour burden

The tibiae were fixed in 10% buffered formalin, followed by acid decalcification at room temperature. Tibiae were paraffin embedded, sectioned longitudinally and stained with haemotoxylin and eosin prior to analysis on an inverted microscope. Analysis of the sections revealed that there were several animals that had no evidence of any tumour burden. A photograph was taken of the best representative section of the tibia for each animal and imported into the Scion program. The area of the tumour in mm^2 was determined for each animal and tabulated, as shown in T**able 7.1**. Animals #2, #4, #8 and #9 of the control untreated group, all had macroscopic tumours which is



Figure 7.18 Bone volume comparisons between the control untreated group, the TRAIL treated group, the ZOL treated group and the combination ZOL and TRAIL treatment group. The above graph shows the bone volume in the left tibiae and the right tibiae of each group based on analyses using the program CTan. Bone volume was measured from 150 sections of the proximal tibia starting from the growth plate. Data shown in each case is the average bone volume from all animals in that group: points are means \pm SD.



Figure 7.19 Photographs shown are representative of haematoxylin and eosin stained sections of mouse tibiae at 35 days post inoculation with K-HOS cells. Panel (A) and (B) represents an untreated control animal with normal trabecular density and cortical bone thickness. Panel (C) and (D) represents a ZOL treated animal with increased trabecular density and cortical bone in both the tibia and the femur.

Animal	UT	Т	Ζ	Z+T	
#1	0	22.21	29.04	8.16	
#2	26.34	4.58	28.96	9.26	
#3	0	5.58	21.2	16.77	
#4	11.58	0	0	3.65	
#5	0	0	0	0	
#6	0.1	0	0	0	
#7	0	18.97	0	7.94	
#8	19.71	3.76	0	0	
#9	14.81	0	0	8.33	
#10	-	-	0	31.21	
#11	2 .	-	-	0	

Table 7.1 Area of tumour in the inoculated tibia. Histological photographs were taken and tumour area measured in mm^2 . Animals showing no visible tumour were assigned a value of zero.

evident in the histology, resulting in large areas of tumour burden. Animal #6 had a very small area of tumour mass within the marrow cavity but showed no visible signs of tumour from either the radiographs or the µCT images. The average area of tumour in the affected animals of the control untreated group was 14.5 mm². In the TRAIL treated group, 5 out of 9 animals had tumours, based on histological analysis. Of these, only 3 out of the five were evident from the radiographs but all were identified from the μCT images. The average area of tumour in the affected animals of the TRAIL treated group was 11.02 mm². It is clearly evident from the radiograph (Figure 7.4) that animals #1, #2 and #3 of the ZOL treated group had visible tumours and this is reflected in the tumour area calculated from the histological sections. In the combination of ZOL and TRAIL treated group, 7 out of 11 animals developed quantifiable tumours, with an average area of 12.2 mm². However, tumour growth was solely confined to regions outside the bone for all animals treated with ZOL. The average tumour area for each group was calculated, excluding animals with no tumour cells present, and is shown in Figure 7.20. From the graph, it can be seen that ZOL treated animals had the greatest tumour area, with TRAIL treated animals showing the least tumour area. There was a statistically significant reduction in tumour area in the treatment groups given TRAIL (P=0.0481), when compared to non-TRAIL treated groups.

7.3.4 ZOL and TRAIL have no effect on lung metastases

Upon sacrifice of the animals, the lungs were removed and prepared for histology. Approximately 3 sections at 3 levels were obtained for each animal and viewed under the microscope for the detection of lung metastases. The number of animals identified with lung metastases were, 2, 3, 3, and 3 for the untreated control group, TRAIL treated group, ZOL treated group and combination ZOL and TRAIL treated group,



Figure 7.20 Average tumour area per group. Histological sections were photographed and used to measure tumour area in mm^2 . Animals with no tumour cells present were eliminated from the calculation. Data shown in each case are from a representative section of each animal: points are means \pm SD.

respectively. All mice showing lung metastases also exhibited osteoblastic lesions but only 1 mouse showed both osteolytic and osteoblastic lesions (**Table 7.2**). Mice presenting with only osteolytic lesions did not possess lung metastases.

There was no statistically significant interaction between ZOL and TRAIL, and neither ZOL nor TRAIL significantly impacted independently on the presence of lung metastases.

	UT		Т		Z		Z+T					
Animal	Osteolytic Lesions	Osteoblastic Lesions	Lung Metästases	Osteolytic Lesions	Osteoblastic Lesions	Lung Metastases	Osteolytic Lesions	Osteoblastic Lesions	Lung Metastases	Osteolytic Lesions	Osteoblastic Lesions	Lung Metastases
#1	+	-	-	-	+	+	3 7	+	+		+	+
#2	_	+	+	-	+	+	1. 1.	+	+		+	+
#3	-			+	+	1944		+	+	-	+	+
#4	+	+	+	-	-	-	-	-	-	-	+	-
#5	+	-	9 111 11	-	-	(-)	-		-	-	-	H
#6		-	-	-		2 — 2		-		-		
#7	-	-		+	+	+	-	-	-	::	+	-
#8	+	+	_	-	+	-		,)		-	+	-
#9	+	-	-	-	÷	-	-	-	-		+	-
#10								-	-	-		-
#11										m	-	-

Table 7.2 Frequency of osteolytic, osteoblastic lesions and lung metastases. Animals left untreated (UT) and treated with T (TRAIL) and/or Z (ZOL) are assigned + for the presence of, and - for absence of lesions or lung metastases. Osteolytic and osteoblastic lesions were identified using micro-CT analysis and lung metastases were identified using histological sections.

7.4 **DISCUSSION**

This is the first study to investigate the effect of ZOL in an animal model of human OS. K-HOS cells were inoculated into the left proximal tibia of nude mice. When left untreated, these developed large lesions that invaded the marrow cavity and began to erode the cortical bone. Treatment with 30mg/kg/dose of TRAIL for 5 consecutive days followed by once weekly injections, had little effect, and large osteolytic lesions similar to control animals were seen. In mouse models, TRAIL has demonstrated remarkable efficacy against tumour xenografts of colon carcinoma [LeBlanc et al. 2002; Ashkenazi et al. 1999] breast carcinoma [Walczak et al. 1999] and glioma [Pollack et al. 2001; Fulda et al. 2002]. In this current study, it was not surprising that osteolytic lesions developed despite TRAIL treatment, as TRAIL had no effect on K-HOS cells in vitro. In contrast, all animals treated with 100 µg/kg/dose of ZOL, once weekly for four weeks, showed remarkable conservation of the tibiae with no radiographic or µCT evidence of bone destruction. In addition, no evidence of tumour growth within the bone was identified upon histological examination. This data supports previously published reports of the protection of bone by ZOL from cancerinduced osteolysis [Alvarez et al. 2003; Hiraga et al. 2004; Lee et al. 2002; Croucher et al. 2003; Green 2003; Green 2002; Peyruchaud et al. 2001]. As both the ZOL treated and combination of ZOL and TRAIL treated animals showed no osteolysis, it could not be determined whether the combination treatment was any more beneficial than ZOL alone. A dose response study with differing ZOL concentrations would be required to investigate this matter further.

It has been reported that at an active bone resorption site the concentration of BPs is highest and equates to the concentration required to induce apoptosis of cancer cells *in vitro* [Sato *et al.* 1991]. This concentration of BPs is not achievable in visceral organs *in vivo* [Fleisch 2000]. As OS-induced lesions are areas of high osteoclastic activity, it

may be possible that ZOL is released in the direct vicinity of the surrounding cancer cells. It is not known whether ZOL directly affects cancer cells *in vivo* or whether it causes changes to the bone environment that indirectly impact these cells. Studies demonstrating increased apoptosis of tumour cells in osteolytic lesions in mice treated with ZOL and other BPs have been demonstrated but this is not conclusive proof of a direct anti-tumour effect [Hiraga *et al.* 2001; Yaccoby *et al.* 2002]. To reproduce the augmented apoptotic effects observed with the combination treatment of ZOL and TRAIL *in vitro*, it is therefore essential that the tumour cells be directly exposed to ZOL *in vivo*.

Comparisons of radiographs between treatment groups revealed highly radiodense areas within the bone of the ZOL treated animals, suggesting increased bone density. The 100 µg/kg/dose of ZOL used is approximately equivalent to the human 4 mg dose, which is given once a month to tumour patients and not every week, as was administered to the mice [Berenson et al. 2001; Berenson et al. 2002]. This corresponds to a 4-fold higher dose and this was perhaps manifested in the development of dense bone, especially at the ends of the growing long bones. The use of μCT 3-D reconstructions and histological sections confirmed that ZOL treated animals showed significant increases in cortical bone thickness and trabecular density, when compared to the untreated animals. This observation is commonly found in BP treated animals [Sohara et al. 2003; Corey et al. 2003]. The bone volume, as measured by µCT, significantly increased in ZOL treated animals and was more pronounced in areas of increased bone turnover, such as the distal femurs and proximal tibiae, but was not restricted to the tumour site since the contralateral tibiae also showed this effect. This correlated with the radio-dense radiographs and the histology. Statistical analyses confirmed a significant increase of bone volume in ZOL treated animals compared to untreated. Clinically, patients are given BPs primarily to prevent further bone loss,

rather than increasing bone density [Fleisch 2000], although some increase in bone mass appears to occur with this class of anti-resorptive [Reid *et al.* 2002; Orwoll *et al.* 2000; Schnitzer *et al.* 2000]. However, it is difficult to compare events in young growing animals with those in older and osteoporotic human individuals.

The average area of tumour growth was lowest in the TRAIL treated and the combination of ZOL and TRAIL treated groups than the untreated and ZOL treated groups, suggesting that TRAIL may have an effect on the extra-osseous tumour. Although TRAIL alone had no effect on K-HOS cells *in vitro*, it is plausible that there may be an effect *in vivo* due to other factors. Statistical analysis based only on the animals with detectable tumour cells, showed a significant decrease in tumour area in TRAIL treated animals. The average tumour area per group was also calculated using these animals. The rationale for this was to exclude animals, in which there may have been technical errors upon intra-tibial injection, resulting in the absence of a lesion. However, the statistical significance did not hold true upon inclusion of all animals per group. Further experiments are required to support or reject the current data.

Histological examination of lung tissue revealed no difference between the treatment groups with regard to metastatic lesions. The only correlation made between the mice with metastases was that they all had 'osteoblastic' lesions. It is possible that the large non-osseous tumours often associated with these lesions had a larger, more complex vasculature and were therefore more likely to allow transport of cancer cells to the lungs, where they were able to establish and grow. Although ZOL protects the skeleton from cancer-induced cell death, it does not prevent lung metastases, which are often associated with increased mortality. This may indicate that there are no direct effects of ZOL *in vivo* under these conditions and that the observed effects on tumours in bone are related to inhibition of osteolysis. Alternatively, the lack of inhibitory effects on lung metastases may be due to poor bioavailability of ZOL, because it

155

. . .

accumulates in bone with limited access to soft tissues. Reports of the effect of N-BPs on visceral metastases have been conflicting, with trends towards either a decrease [Alvarez et al. 2003; Nobuyuki et al. 2001; Sasaki et al. 1998] or an increase in visceral metastases [Stearns and Wang 1996; Sasaki et al. 1998; Sasaki et al. 1995; Cruz et al. 2001]. The use of ZOL in conjunction with current chemotherapy would provide treatment for both OS-induced osteolysis and lung metastases. The efficacy of ZOL combined with anti-cancer agents in vitro has been well documented [Matsumoto et al. 2005; Vogt et al. 2004; Hiraga et al. 2003; Tassone et al. 2000; Jagdev et al. 2001]. Most clinical studies have been performed using BPs in combination with conventional anti-cancer agents [Body 2003; Body et al. 1998; Diel et al. 1998; Coleman and Seaman 2001; Berenson et al. 2001; Rogers et al. 2000; Rosen et al. 2003; Rosen et al. 2004; Rosen et al. 2003]. These clinical studies have all focused on metastases to the bone rather than OS, so there is a need to investigate the protective effect of BPs, specifically ZOL, in combination with agents capable of targeting both osseous and non-osseous tumours. Alternatively, ZOL could be used in combination with other IMP such as the statins as another way of targeting both osseous and non-osseous tumours.

Micro-CT analysis proved to be the most informative tool in assessing the effect of ZOL on human OS cells *in vivo*. It provided an extremely detailed analysis of the inoculated tibia, which could be further manipulated for other purposes, such as dissecting, rotating and bone volume analyses. Radiographs clearly showed relatively large osteolytic lesions in 3 animals of the control untreated group, however, the improved sensitivity of μ CT allowed smaller lesions to be recognized, identifying a total of 5 animals with lesions. Micro-CT confirmed the absence of osteolysis in the tibiae of all animals treated with ZOL. In addition, μ CT identified the presence of an 'osteoblastic' phenotype, in the form of small protrusions extending from the bone surface in over half the animals, which could not be discerned in the radiographs, or in

most of the histological sections due to the decalcification process. One possibility is that K-HOS cells may possess a mixed osteolytic and osteoblastic phenotype. ZOL has been shown to exhibit inhibitory effects on both osteoblastic and osteolytic lesions and is the only BPs to show a direct reduction in skeletal related events in the treatment of osteolytic and osteoblastic metastatic bone disease [Body 2003; Corey et al. 2003; Lipton et al. 2002]. A contradictory report has suggested that ZOL is effective in limiting osteolytic lesions in vivo but not osteoblastic lesions associated with prostate tumour cells, suggesting that osteoclast activity may not be critical for the development of osteoblastic lesions [Lee et al. 2002]. Another possibility for the observed 'osteoblastic' phenotype is that OS produce bone morphogenetic proteins (BMPs), which are potent inducers of osteoblasts and are able to induce ectopic bone formation in vivo [Yoshikawa et al. 2004; Raval et al. 1996; Anderson et al. 1992]. One report described this as "sunburst" spiculation of the periosteum as observed by radiography [Yoshikawa et al. 1985]. Almost all animals exhibiting the 'osteoblastic' phenotype had large macroscopic tumours. It is possible that cells, which were left in the surrounding soft tissue as the needle withdrew from the tibia, led to extra-osseous tumour growth, and does not represent typical development in vivo. As the areas of increased bone formation include the fibula and regions well below the inoculation site, it is likely that the extra osseous tumour growth is responsible for this phenomenon and is not representative of normal tumour progression. The observed phenotype may partly be a consequence of the method of delivery of cells to the tibia and this may need to be revised. Regardless of this observation, the lack of any osteolytic lesions in the ZOL treated animals is very exciting and may represent changes in the bone environment that make it less amenable to tumour growth. This may be the reason for the increased tumour area observed in ZOL treated animals, forcing the cells to establish outside the bone.

One limitation of this study is the lack of osteolysis observed in some of the control untreated animals. There could be numerous reasons for this observation and the premature termination of this experiment due to animal ethics issues also remains a possible reason for the incomplete collection of adequate data. This was the first time that the animal ethics committee were confronted with this procedure and it proved to be problematic for them. If the experiment had been taken to full term, it is possible that more lesions may have developed and that greater variations may have been observed between the treatment groups. This however remains to be seen and further experiments are required with the addition of analgesics, to reduce pain of the animals and extend the length of the experiment. Other reasons for the lack of osteolysis include, variations between tibial injections including inadequate delivery of cells and variations in injection sites potentially providing differing environments for the cells. The volume of inoculum was reduced from 20µl in the pilot study to 10µl in the current study. This was to improve the likelihood of complete delivery of entire volume into the tibia and to reduce variability between injections. Experience and practice should help reduce the large variations between animals. The use of mice at 4 weeks of age may improve the delivery technique, as the bones are much softer and easier to inject into.

Although there is extensive *in vivo* preclinical evidence that BPs, and in particular ZOL, have anti-tumour activity and can reduce skeletal and perhaps also extraskeletal tumour burden in a variety of tumour types, no such data exist for OS. These preliminary animal studies on human OS supplement the current literature by showing that ZOL treatment protects the skeleton from OS-induced osteolysis.

CHAPTER EIGHT

GENERAL DISCUSSION

8.1 GENERAL DISCUSSION

The data reported in this thesis are part of an ongoing investigation into new and effective agents for the treatment of OS, which is an aggressive and destructive malignancy. Individuals with OS, and particularly older patients continue to have a poor prognosis despite considerable improvement in the treatment regimes. New alternative approaches for the treatment of OS are essential for improvement in patient survival. Such agents include TRAIL and inhibitors of the mevalonate pathway including BPs, statins and PTIs.

Bisphosphonates have the ability to inhibit osteoclast-mediated resorption, and have also been identified as having direct anti-tumour activity. Reports of the more potent N-BPs to dose-dependently inhibit the proliferation and survival of tumour cells have been made in numerous tumour cell types [Senaratne and Colston 2002; Stearns and Wang 1996; Fromigue et al. 2000; Sato et al. 1991; Lee et al. 2001]. The most potent of the currently available N-BPs, ZOL, was investigated for effects on human OS cells. ZOL treatment induced a dose- and time-dependent decrease in proliferation and survival in all of the OS cell lines tested, although it was variably effective in reducing cell number in the different cell lines. Variations in the cytostatic and apoptotic effects of N-BPs in vitro have been reported and most probably depend on the potency of the N-BP and the cell line used [Fromigue et al. 2000]. ZOL acts by inhibiting a key enzyme of the mevalonate pathway leading to the cell's inability to post-translationally modify (or prenylate) proteins, such as the GTPases, which are required for essential processes of survival. [Coxon et al. 2000; Dunford et al. 2001; Luckman et al. 1998; Fisher et al. 2000; Rodan 1998; Rogers et al. 2000]. Treatment of OS cells with ZOL in vitro led to loss of attachment followed by apoptosis and was concomitant with an increase in caspase activity, most likely due to a secondary effect since caspase inhibition did not protect these cells from the ZOL-mediated effects. Alternatively,

addition of geranylgeraniol (GGO), was able to protect these cells by acting as an intermediate of the mevalonate pathway, allowing geranylgeranylation to proceed and thus cell survival.

Other effects of ZOL on human OS cells include S-phase arrest of the cell cycle and an increased expression of cell surface TRAIL death receptors. The increase of death receptors but not the decoy receptors at the cell surface may explain the observed augmented apoptotic effect of ZOL and TRAIL in most cell lines, however previous reports have not found a correlation between receptor expression and TRAIL sensitivity [Hersey and Zhang 2001; Ozoren and El-Deiry 2003]. Treatment of OS cells with ZOL and TRAIL together led to an augmented apoptotic effect, which was observed in all cell lines to some extent, correlating to the sensitivity of the cell lines to ZOL alone. More importantly, the cell lines that were resistant to TRAIL treatment alone were strongly sensitised to TRAIL-induced apoptosis by the co-addition of ZOL. As was noted with ZOL treatment, OS cells given the combination treatment showed higher levels of caspase activity. The co-addition of caspase inhibitor to ZOL and TRAIL combination treatments led to an apoptotic effect equal to that seen with ZOL alone. On the other hand, addition of GGO to the combination treatment prevented sensitisation to TRAIL-induced apoptosis.

Normal human osteoblast-like (NHB) cells did not exhibit such an augmented apoptotic effect with co-treatment of ZOL and TRAIL. Similarly, there was no upregulation of the TRAIL death receptors at the cell surface as was observed in the OS cells, which may explain the lack of augmentation with combination treatment. This is very encouraging as the combination of ZOL and TRAIL is much more effective at inducing apoptosis in cancer cells than normal cells, which could reduce toxicity of *in vivo* treatment, compared with that associated with currently used chemotherapeutic regimes.

Other inhibitors of the mevalonate pathway (IMP), such as statins, FTIs and GGTIs were also employed to investigate their effectiveness as alternative agents for the treatment of OS. As was observed with ZOL, treatment of OS cells with statins, FTI and GGTI also led to dose-dependent apoptosis that was augmented with the co-addition of TRAIL. Resembling the ZOL effects on NHB cells, these other IMP all failed to exhibit an augmented apoptotic effect in NHB cells using combination treatment with TRAIL. This highlights the importance of the mevalonate pathway inhibitors as anti-cancer agents.

Although *in vitro* experiments are necessary for the initial investigation of potential therapeutic agents, these experiments are invalid if the agents are not able to reach their destination at the concentration required to generate the desired effect without associated toxicity. Fortunately, BPs and statins have been used for many years for the treatment of various diseases and FTIs are currently being assessed in clinical trials [Goldstein and Brown 1990; Body 2000; Brunner *et al.* 2003]. Although GGTIs have not reached clinical trials yet, there is encouraging pre-clinical data and further investigation is warranted. This provides a sound basis for the use of these agents for *in vivo* analysis. It is not known whether tumour cells are directly exposed to ZOL *in vivo*, due to its high affinity for the bone. As the *in vitro* data described in this thesis suggests, ZOL availability to cancer cells is vital for sensitisation to TRAIL-induced apoptosis. Therefore, the use of statins, FTIs and GGTIs as alternatives to ZOL, with potentially improved bioavailability, may offer advantages. In addition, these agents may also be used for the treatment of other non-osseous cancers.

An OS mouse model involving the injection of human cells into the tibia was established in a pilot study and the most suitable conditions were identified for further investigations. The K-HOS cell line was chosen for the *in vivo* approach due to the ability of these cells to establish in the marrow cavity and cause osteolytic lesions in

mice. Also the synergistic apoptotic activity of ZOL and TRAIL co-treatment *in vitro* has been demonstrated, an effect not seen in KRIB cells. The animals were inoculated into the left proximal tibia and treatments were commenced one week post cancer cell transplantation.

Using radiography, micro computed tomography (μ CT) and histology, inoculated tibiae were assessed from animals of the control untreated, TRAIL-treated, ZOL-treated and combination ZOL and TRAIL-treated groups. All 21 animals given ZOL, showed remarkable conservation of the tibiae with no radiographic evidence of bone destruction. The radiographs demonstrated that in all ZOL treated animals, and not in the untreated animals, the bones were highly radio-dense, suggesting increased bone density. Longitudinal sections of μ CT 3-D reconstructions and histological sections of the corresponding tibiae clearly showed significant increases in cortical thickness and trabecular density in the ZOL treated animals when compared to the untreated animals. This was supported with significant increases in bone volume using the 2-D images obtained by μ CT analysis.

Using histological sections, tumour area was calculated and comparisons between affected animals were made. A significant reduction in tumour area was observed in animals treated with TRAIL, but no effect was seen with ZOL treatment. As the number of animals per group was small, the experiments need to be repeated to place any confidence in the results. The number of animals identified with lung metastases revealed no differences between groups.

From the μ CT data it was noted that many animals exhibited 'osteoblastic'-like lesions, which were not apparent from the radiographs. There are many possible reasons for the observed effect, including the release of factors by the surrounding soft-tissue tumour, almost always associated with this phenomena, leading to the stimulation of bone growth.

The development of very large macroscopic tumours which hindered movement and caused pain in a few animals, led to the premature termination of the experiment. However at this time point some animals did not exhibit any signs of osteolysis, possibly due to insufficient time for tumour development to occur. Alternatively, technical difficulties may also be the reason for the variations observed.

The *in vivo* results support current reports that ZOL protects the skeleton from OS-induced osteolysis but fail to exhibit any positive interaction between ZOL and TRAIL combination treatment *in vivo*. The direct effect of ZOL on cancer cells *in vivo* is not known, but may not be adequate to sensitise the cells to TRAIL-induced apoptosis. Further investigation is required to determine whether ZOL exposure to OS cells is adequate to exhibit an augmented apoptotic effect when combined with TRAIL.

The use of TRAIL in combination with inhibitors of the mevalonate pathway offers an exciting therapeutic potential. However, in order to realise this potential, it is important to proceed systematically through cell biology studies, preclinical studies before embarking on clinical trials. Ongoing cell studies are required to further understand the complex biology of TRAIL and its interaction with molecules such as ZOL, statins and PTIs that will enable more informed treatment, particularly of TRAIL-resistant tumours. Further *in vivo* experiments will prove extremely useful for the study of human OS growth and metastasis and will enable a thorough *in vivo* evaluation of these agents, which have been shown to have anti-proliferative and cytotoxic effects *in vitro*. This information will have important implications for future therapy of OS.

8.2 FUTURE DIRECTIONS

This thesis has provided data to support the anti-cancer activity of ZOL, statins, FTIs and GGTIs alone, and in combination with TRAIL, in human OS cells. Some of the mechanisms of action have been explored but many questions remain unanswered. Why are there differences in the sensitivity of OS cells to IMP-induced apoptosis? The reasons for the observed differences in activity and potency of IMP in the different cell lines are unclear but may reflect differences in bioavailability, cellular uptake or intracellular effects of IMP in different cell types. It is important to identify the reasons for these differences so that their effectiveness in inducing apoptosis can be improved. Due to the high degree in variability of OS cells with regards to sensitivity to these treatments, it would be beneficial to the patients to use biopsy sample to test the responsiveness of the tumour cells to these agents *in vitro*. Thus identifying the best course of treatment for each individual patient and improving prognosis.

Further investigation into the molecular mechanisms of action of TRAIL and IMP will assist in their exploitation and possibly enable identification of alternative agents to work in combination with the IMP. Which Bcl-2 family members are involved? Which GTPase proteins are critical for cell attachment? How exactly do IMP sensitise the cell to TRAIL-induced apoptosis? Which caspase independent pathways are involved? Despite the many questions, one important thing remains, that IMP in combination with TRAIL are very effective in inducing apoptosis in some human OS cells but not normal cells, providing the possibility of a new cancer treatment without the associated toxicity. This combination of agents must first be assessed in pre-clinical models.

Although the *in vivo* experiments conducted were able to show no tumour growth or bone destruction in all the animals treated with ZOL, it could not be determined whether the combination treatment was any more beneficial than ZOL alone. Direct investigation of the degree of exposure of ZOL to OS cells *in vivo* may help determine

whether this combination could cause a similar effect in vivo to that observed in vitro. Detection of unprenylated protein in OS cells in treated animals might provide some insight. If this proves difficult, a dose response study with differing ZOL concentrations and TRAIL would be required to investigate this matter further. Prior to any further in vivo experiments, several factors need to be addressed. The technique of intra-tibial injections needs to be refined to improve delivery and reduce variation between animals. The early onset of macroscopic tumours in some individuals must be considered. The use of analgesics will need to be administered to allow the experiment to be taken to full term. Alternatively, the use of bioluminescence equipment will allow better evaluation of the effects of these agents over time. More importantly, it will help identify and monitor the effect of these agents on well-established tumours, so that termination of the experiment will give the most meaningful results. If bioavailability remains a problem with ZOL treatment, the use of other agents, such as statins and PTIs, are alternative possibilities. These agents may act to target OS cells and other nonosseous cancer cells in vivo. The use of IMP in combination with other clinically relevant anticancer drugs may also prove extremely useful for the study of human OS growth and metastasis.

Taken together the use of IMP and TRAIL as alternative anti-tumour agents for the treatment of human OS has considerable potential and may lead to a more effective therapeutic regime that has the added advantage of being non-toxic and safe.
BIBLIOGRAPHY

Adami, S. (1997). Bisphosphonates in prostate carcinoma. Cancer Cell 80:1674-9.

Adams, J. M. and Cory, S. (1998). The Bcl-2 protein family: arbiters of cell survival. *Science* 281:1322-26.

Adjei, A. A., Davis, J. N., Bruzek, L. M., Erlichman, C. and Kaufmann, S. H. (2001). Synergy of the protein farnesyltransferase inhibitor SCH66336 and cisplatin in human cancer cell lines. *Clin Cancer Res* 7:1438-45.

Adrain, C., Slee, E. A., Harte, M. T. and Martin, S. J. (1999). Regulation of apoptotic protease activating factor-1 oligomerization and apoptosis by the WD-40 repeat region. *J Biol Chem* 274:20855-60.

Agarwal, B., Bhendwal, S., Halmos, B., Moss, S. F., Ramey, W. G. and Holt, P. R. (1999). Lovastatin augments apoptosis induced by chemotherapeutic agents in colon cancer cells. *Clin Cancer Res* 5:2223-9.

Agarwal, B., Rao, C. V., Bhendwal, S., Ramey, W. R., Shirin, H., Reddy, B. S. and Holt, P. R. (1999). Lovastatin augments sulindac-induced apoptosis in colon cancer cells and potentiates chemopreventive effects of sulindac. *Gastroenterology* 117:838-47.

Ahmad, M. and Shi, Y. (2000). TRAIL-induced apoptosis of thyroid cancer cells: potential for therapeutic intervention. *Oncogene* 19:3363-71.

Ali, S. M., Esteva, F. J., Hortobagyi, G., Harvey, H., Seaman, J., Knight, R., Costa, L. and Lipton, A. (2001). Safety and efficacy of bisphosphonates beyond 24 months in cancer patients. *J Clin Oncol* 19:3434-7.

Alnemri, E. S., Livingston, D. J., Nicholson, D. W., Salvesen, G., Thornberry, N. A., Wong, W. W. and Yuan, J. (1996). Human ICE/CED-3 protease nomenclature. *Cell* 87:171.

Alonso, D. F., Farina, H. G., Skilton, G., Gabri, M. R., De Lorenzo, M. S. and Gomez, D. E. (1998). Reduction of mouse mammary tumor formation and metastasis by lovastatin, an inhibitor of the mevalonate pathway of cholesterol synthesis. *Breast Cancer Res Treat* 50:83-93.

Alsina, M., Overton, R., Belle, N. and al., e. (2002). Farnesyl transferase inhibitor R115777 is well tolerated, induces stabilization of disease and inhibits farnesylation and oncogenic/tumor survival pathways in patients with advanced multiple myeloma. *Proc Am Assoc Cancer Res* 43:1000.

Alvarez, E., Westmore, M., Galvin, R. J., Clapp, C. L., Considine, E. L., Smith, S. J., Keyes, K., Iversen, P. W., Delafuente, D. M., Sulaimon, S., Zambrano, C., Ma, L., Sato, M., Martin, T. J., Teicher, B. A. and Galbreath, E. J. (2003). Properties of bisphosphonates in the 13762 rat mammary carcinoma model of tumor-induced bone resorption. *Clin Cancer Res* 9:5705-13.

Andela, V. B., Rosenblatt, J. D., Schwarz, E. M., Puzas, E. J., O'Keefe, R. J. and Rosier, R. N. (2002). Synergism of aminobisphosphonates and farnesyl transferase inhibitors on tumor metastasis. *Clin Orthop* 397:228-39.

Anderson, H. C., Sugamoto, K., Morris, D. C., Hsu, H. H. and Hunt, T. (1992). Bone inducing agent (BIA) from cultured human Saos-2 osteosarcoma cells. *Bone Miner* 16:49-62.

Antonsson, B. (2001). Bax and other pro-apoptotic Bcl-2 family "killer proteins" and their victim, the mitochondrion. *Cell Tissue Res* 306:347-61.

Ashar, H. R., James, L., Gray, K., Carr, D., McGuirk, M., Maxwell, E., Black, S., Armstrong, L., Doll, R. J., Taveras, A. G., Bishop, W. R. and Kirschmeier, P. (2001).

The farnesyl transferase inhibitor SCH 66336 induces a $G(2) \rightarrow M$ or G(1) pause in sensitive human tumor cell lines. *Exp Cell Res* 262:17-27.

Ashkenazi, A. and Dixit, V. M. (1999). Apoptosis control by death and decoy receptors. *Curr Opin Cell Biol* 11:255-60.

Ashkenazi, A., Pai, R. C., Fong, S., Leung, S., Lawrence, D. A., Marsters, S. A., Blackie, C., Chang, L., McMurtrey, A. E., Hebert, A., DeForge, L., Koumenis, I. L., Lewis, D., Harris, L., Bussiere, J., Koeppen, H., Shahrokh, Z. and Schwall, R. H. (1999). Safety and antitumor activity of recombinant soluble Apo2 ligand. *J Clin Invest* 104:155-62.

Ashkenazi, A. (2002). Targeting death and decoy receptors of the tumour-necrosis factor superfamily. *Nat Rev Cancer* 2:420-30.

Atkins, G. J., Bouralexis, S., Evdokiou, A., Hay, S., Labrinidis, A., Zannettino, A. C., Haynes, D. R. and Findlay, D. M. (2002). Human osteoblasts are resistant to Apo2L/TRAIL-mediated apoptosis. *Bone* 31:448-56.

Banner, D. W., D'Arcy, A., Janes, W., Gentz, R., Schoenfeld, H. J., Broger, C., Loetscher, H. and Lesslauer, W. (1993). Crystal structure of the soluble human 55 kd TNF receptor-human TNF beta complex: implications for TNF receptor activation. *Cell* 73:431-45.

Barberis, L., Wary, K. K. and Fiucci, G. (2000). Distinct roles of the adaptor protein Shc and focal adhesion kinase in integrin signalling to ERK. *J Biol Chem* 275:36532-40.

Bardeleben, R. V., Dunkern, T., Kaina, B. and Fritz, G. (2002). The HMG-CoA reductase inhibitor lovastatin protects cells from the antineoplastic drugs doxorubicin and etoposide. *Int J Mol Med* 10:473-9.

Bare, S., Kimmel, D., Binkley, N., Schaffer, V. and Green, J. (1997). Zoledronate (CGP 42'446) suppresses turnover without affecting mineralization in cancellous bone of the ovariectomized nonhuman primate. *J Bone Miner Res* 12:S473-82.

Barrington, R. E., Subler, M. A., Rands, E., Omer, C. A., Miller, P. J., Hundley, J. E., Koester, S. K., Troyer, D. A., Bearss, D. J., Conner, M. W., Gibbs, J. B., Hamilton, K., Koblan, K. S., Mosser, S. D., O'Neill, T. J., Schaber, M. D., Senderak, E. T., Windle, J. J., Oliff, A. and Kohl, N. E. (1998). A farnesyltransferase inhibitor induces tumor regression in transgenic mice harboring multiple oncogenic mutations by mediating alterations in both cell cycle control and apoptosis. *Mol Cell Biol* 18:85-92.

Bauer, D. C. (2003). HMG CoA reductase inhibitors and the skeleton: a comprehensive review. *Osteoporos Int* 14:273-82.

Baylin, S. and Bestor, T. H. (2002). Altered methylation patterns in cancer cell genomes: cause or consequence? *Cancer Cell* 1:299-305.

Bellosta, S., Ferri, N., Bernini, F., Paoletti, R. and Corsini, A. (2000). Non-lipid-related effects of statins. *Ann Med* 32:164-76.

Berenbaum, M. C. (1977). Synergy, additivism and antagonism in immunosuppression. *Clin Exp Immunol* 28:1-18.

Berenson, J. R., Rosen, L. S., Howell, A., Porter, L., Coleman, R. E., Morley, W., Dreicer, R., Kuross, S. A., Lipton, A. and Seaman, J. J. (2001). Zoledronic acid reduces skeletal-related events in patients with osteolytic metastases. *Cancer* 91:1191-200.

Berenson, J. R., Hillner, B. E., Kyle, R. A., Anderson, K., Lipton, A., Yee, G. C. and Biermann, J. S. (2002). American Society of Clinical Oncology Bisphosphonates Expert Panel. American Society of Clinical Oncology clinical practice guidelines: the role of bisphosphonates in multiple myeloma. *J Clin Oncol* 20:3719-36.

Beresford, P. J., Zhang, D., Oh, D. Y., Fan, Z., Greer, E. L., Russo, M. L., Jaju, M. and Lieberman, J. (2001). Granzyme A activates an endoplasmic reticulum-associated

caspase-independent nuclease to induce single-stranded DNA nicks. J Biol Chem 276:43285-93.

Bergeron, L., Perez, D., MacDonald, G., Shi, L., Sun, Y., Jurisicova, A., Varmuza, S., Latham, K. E., Flaws, J. A., Salter, J. C., Hara, H., Moskowitz, M. A., Li, E., Greenberg, A., Tilly, J. L. and Yuan, J. (1998). Defects in regulation of apoptosis in caspase-2 deficient mice. *Genes Dev* 12:1304-14.

Berlin, O., Samid, D., Donthineni-Rao, R., Akeson, W., Amiel, D. and Woods, V. L. (1993). Development of a novel spontaneous metastasis model of human osteosarcoma translanted orthotopically into bone of athymic mice. *Cancer Res* 53:4890-4895.

Bernhard, E. J., Kao, G., Cox, A. D., Sebti, S. M., Hamilton, A. D., Muschel, R. J. and McKenna, W. G. (1996). The farnesyltransferase inhibitor FTI-277 radiosensitizes H-ras-transformed rat embryo fibroblasts. *Cancer Res* 56:1727-30.

Bernhard, E. J., McKenna, W. G., Hamilton, A. D., Sebti, S. M., Qian, Y., Wu, J. M. and Muschel, R. J. (1998). Inhibiting Ras prenylation increases the radiosensitivity of human tumor cell lines with activating mutations of ras oncogenes. *Cancer Res* 58:1754-61.

Bettaieb, A., Dubrez-Daloz, L., Launay, S., Plenchette, S., Rebe, C., Cathelin, S. and Solary, E. (2003). Bcl-2 proteins: targets and tools for chemosensitisation of tumor cells. *Curr Med Chem Anti-Canc Agents* 3:307-18.

Binkley, N., Kimmel, D., Bruner, J., Haffa, A., Davidowitz, B., Meng, C., Schaffer, V. and Green, J. (1998). Zoledronate prevents the development of absolute osteopenia following ovariectomy in adult rhesus monkeys. *J Bone Miner Res* 13:1775-82.

Bjerre, L. M. and LeLorier, J. (2001). Do statins cause cancer? A meta-analysis of large randomized clinical trials. *Am J Med* 110:716-23.

Blanco-Colio, L. M., Villa, A., Ortego, M., Hernandez-Presa, M. A., Pascual, A., Plaza, J. J. and Egido, J. (2002). 3-Hydroxy-3-methyl-glutaryl coenzyme A reductase

inhibitors, atorvastatin and simvastatin, induce apoptosis of vascular smooth muscle cells by downregulation of Bcl-2 expression and Rho A prenylation. *Atherosclerosis* 161:17-26.

Blink, E., Maianski, N. A., Alnemri, E. S., Zervos, A. S., Roos, D. and Kuijpers, T. W. (2004). Intramitochondrial serine protease activity of Omi/HtrA2 is required for caspase-independent cell death of human neutrophils. *Cell Death Differ* In Press.

Bodmer, J. L., Holler, N., Reynard, S., Vinciguerra, P., Schneider, P., Juo, P., Blenis, J. and Tschopp, J. (2000). TRAIL receptor-2 signals apoptosis through FADD and caspase-8. *Nat Cell Biol* 2:241-43.

Bodmer, J. L., Meier, P., Tschopp, J. and Schneider, P. (2000). Cysteine 230 is essential for the structure and activity of the cytotoxic ligand TRAIL. *J Biol Chem* 275:20632-7.

Bodmer, J. L., Schneider, P. and Tschopp, J. (2002). The molecular architecture of the TNF superfamily. *Trends Biochem Sci* 27:19-26.

Body, J. J., Bartl, R., Burckhardt, P., Delmas, P. D., Diel, I. J., Fleisch, H., Kanis, J. A., Kyle, R. A., Mundy, G. R., Paterson, A. H. and Rubens, R. D. (1998). Current use of bisphosphonates in oncology. *J Clin Oncol* 16:3890-9.

Body, J. J. (2000). Current and future directions in medical therapy: hypercalcemia. *Cancer Cell* 88:3054-8.

Body, J. J. (2003). Rationale for the use of bisphosphonates in osteoblastic and osteolytic bone lesions. *Breast Cancer Res* 12:S37-44.

Boissier, S., Ferreras, M., Peyruchaud, O., Magnetto, S., Ebetino, F. H., Colombel, M., Delmas, P., Delaisse, J. M. and Clezardin, P. (2000). Bisphosphonates inhibit breast and prostate carcinoma cell invasion, an early event in the formation of bone metastases. *Cancer Res* 60:2949-54.

Boldin, M. P., Goncharov, T. M., Goltsev, Y. V. and Wallach, D. (1996). Involvement of MACH, a novel MORT1/FADD-interacting protease, in Fas/APO-1 and TNF receptor-induced cell death. *Cell* 85:803-15.

Bolick, S. C., Landowski, T. H., Boulware, D., Oshiro, M. M., Ohkanda, J., Hamilton, A. D., Sebti, S. M. and Dalton, W. S. (2003). The farnesyl transferase inhibitor, FTI-277, inhibits growth and induces apoptosis in drug-resistant myeloma tumor cells. *Leukemia* 17:451-7.

Bonavida, B., Ng, C. P., Jazirehi, A., Schiller, G. and Mizutani, Y. (1999). Selectivity of TRAIL-mediated apoptosis of cancer cells and synergy with drugs: the trail to non-toxic cancer therapeutics (review). *Int J Oncol* 15:793-802.

Borner, C., Martinou, I., Mattmann, C., Irmler, M., Schaerer, E., Martinou, J. C. and Tschopp, J. (1994). The protein Bcl- 2α does not require membrane attachment, but two conserved domains to suppress apoptosis. *J Cell Biol* 126:1059-68.

Borner, C. (2003). The Bcl-2 protein family: sensors and checkpoints for life-or-death decisions. *Mol Immun* 39:615-47.

Bouralexis, S., Clayer, M., Atkins, G. J., Labrinidis, A., Hay, S., Graves, S., Findlay, D. M. and Evdokiou, A. (2004). Sensitivity of fresh isolates of soft tissue sarcoma, osteosarcoma and giant cell tumour cells to Apo2L/TRAIL and doxorubicin. *Int J Oncol* 24:1263-70.

Bretz, J. D., Rymaszewski, M., Arscott, P. L., Myc, A., Ain, K. B., Thompson, N. W. and Baker, J. R. J. (1999). TRAIL death pathway expression and induction in thyroid follicular cells. *J Biol Chem* 274:23627-32.

Britten, C. D., Rowinsky, E. K., Soignet, S., Patnaik, A., Yao, S. L., Deutsch, P., Lee, Y., Lobell, R. B., Mazina, K. E., McCreery, H., Pezzuli, S. and Spriggs, D. (2001). A phase I and pharmacological study of the farnesyl protein transferase inhibitor L-778,123 in patients with solid malignancies. *Clin Cancer Res* 7:3894-903.

173

10 10

Broitman, S. A., Wilkinson, J. t., Cerda, S. and Branch, S. K. (1996). Effects of monoterpenes and mevinolin on murine colon tumor CT-26 in vitro and its hepatic "metastases" *in vivo*. Adv Exp Med Biol:111-30.

Brown, J. M. and Wouters, B. G. (1999). Apoptosis, p53, and tumor cell sensitivity to anticancer agents. *Cancer Res* 59:1391-9.

Browning, J. L., Ngam-ek, A., Lawton, P., DeMarinis, J., Tizard, R., Chow, E. P., Hession, C., O'Brine-Greco, B., Foley, S. F. and Ware, C. F. (1993). Lymphotoxin beta, a novel member of the TNF family that forms a heteromeric complex with lymphotoxin on the cell surface. *Cell* 72:847-56.

Brunner, T. B., Hahn, S. M., Gupta, A. K., Muschel, R. J., McKenna, W. G. and Bernhard, E. J. (2003). Farnesyltransferase inhibitors: an overview of the results of preclinical and clinical investigations. *Cancer Res* 63:5656-68.

Bruno, B., Rotta, M., Giaccone, L., Massaia, M., Bertola, A., Palumbo, A. and Boccadoro, M. (2004). New drugs for treatment of multiple myeloma. *Lancet Oncol* 5:430-42.

Buckley, C. D., Pilling, D., Henriquez, N. V., Parsonage, G., Threlfall, K., Scheel-Toellner, D., Simmons, D. L., Akbar, A. N., Lord, J. M. and Salmon, M. (1999). RGD peptides induce apoptosis by direct caspase-3 activation. *Nature* 397:534-39.

Budihardjo, I. I., Poirier, G. G. and Kaufmann, S. H. (1998). Apparent cleavage of poly(ADP-ribose) polymerase in non-apoptotic mouse LTA cells; An artifact of cross reactive secondary antibody. *Mol Cell Biochem* 178:245-49.

Burns, T. F. and El-Deiry, W. S. (2001). Identification of inhibitors of TRAIL-induced death (ITIDs) in the TRAIL-sensitive colon carcinoma cell line SW480 using a genetic approach. *J Biol Chem* 276:37879-86.

Campanacci, M. (1999). Bone and Soft Tissue Tumors. Padova, Piccin Nuova Libraria.

Carboni, J. M., Yan, N., Cox, A. D., Bustelo, X., Graham, S. M., Lynch, M. J., Weinmann, R., Seizinger, B. R., Der, C. J. and Barbacid, M. (1995). Farnesyltransferase inhibitors are inhibitors of Ras but not R-Ras2/TC21, transformation. *Oncogene* 10:1905-13.

Cardone, M. H., Salvesen, G. S., Widmann, C., Johson, G. and Frisch, S. M. (1997). The regulation of anoikis: MEKK-1 activation requires cleavage by caspases. *Cell* 90:315-23.

Casey, P. J. (1992). Biochemistry of protein prenylation. J Lipid Res 33:1731-40.

Chabner, B. A. and Longo, D. L. (1996). <u>Cancer Chemotherapy and Biotherapy:</u> Principles and Practice. Philadelphia, Lippincott.

Chai, J., Shiozaki, E., Srinivasula, S. M., Wu, Q., Datta, P., Alnemri, E. S., Shi, Y. and Dataa, P. (2001). Structural basis of caspase-7 inhibition by XIAP. *Cell* 104:769-80.

Chan, H. S., Grogan, T. M., Haddad, G., DeBoer, G. and Ling, V. (1997). Pglycoprotein expression: critical determinant in the response to osteosarcoma chemotherapy. *J Natl Cancer Inst* 89:1706-15.

Chan, K. A., Andrade, S. E., Boles, M., Buist, D. S., Chase, G. A., Donahue, J. G., Goodman, M. J., Gurwitz, J. H., LaCroix, A. Z. and Platt, R. (2000). Inhibitors of hydroxymethylglutaryl-coenzyme A reductase and risk of fracture among older women. *Lancet Oncol* 355:2185-8.

Chan, K. K., Oza, A. M. and Siu, L. L. (2003). The statins as anticancer agents. *Clin Cancer Res* 9:10-9.

Chang, D. W., Xing, Z., Pan, Y., Algeciras-Schimnich, A., Barnhart, B. C., Yaish-Ohad, S., Peter, M. E. and Yang, X. (2002). c-FLIP(L) is a dual function regulator for caspase-8 activation and CD95-mediated apoptosis. *EMBO J* 21:3704-14.

Chaudhary, P. M., Eby, M., Jasmin, A., Bookwalter, A., Murray, J. and Hood, L. (1997). Death receptor 5, a new member of the TNFR family, and DR4 induce FADD-dependent apoptosis and activate the NF-kappaB pathway. *Immunity* 7:821-30.

Cheng, E. H., Kirsch, D. G., Clem, R. J., Ravi, R., Kastan, M. B., Bedi, A., Ueno, K. and Hardwick, J. M. (1997). Conversion of Bcl-2 to a Bax-like death effector by caspases. *Science* 278:1966-68.

Chinnaiyan, A. M., O'Rourke, K., Tewari, M. and Dixit, V. M. (1995). FADD, a novel death domain-containing protein, interacts with the death domain of Fas and initiates apoptosis. *Cell* 81:505-12.

Chinnaiyan, A. M., Prasad, U., Shankar, S., Hamstra, D. A., Shanaiah, M., Chenevert, T. L., Ross, B. D. and Rehemtulla, A. (2000). Combined effect of tumor necrosis factorrelated apoptosis-inducing ligand and ionizing radiation in breast cancer therapy. *Proc Natl Acad Sci U S A*:1754-9.

Chirgwin, J. M. and Guise, T. A. (2000). Molecular mechanisms of tumor-bone interactions in osteolytic metastases. *Crit Rev Eukaryot Gene Expr* 10:159-78.

Clayer, M., Bouralexis, S., Evdokiou, A., Hay, S., Atkins, G. J. and Findlay, D. M. (2001). Enhanced apoptosis of soft tissue sarcoma cells with chemotherapy: a potential new approach using TRAIL. *J Orth Surg* 9:41-44.

Clohisy, D. R., O' Keefe, P. F. and Ramnaraine, M. L. (2001). Pamidronate decreases tumor-induced osteoclastogenesis in osteopetrotic mice. *J Orthop Res* 19:554-8.

Cloughesy, T. F., Kuhn, J., Wen, P. and al., e. (2002). Phase II trial of R115777 (Zarnestra) in patients with recurrent glioma not taking enzyme inducing antiepileptic drugs (EIAED). *Proc Am Soc Clin Oncol* 21:80a.

Clutterbuck, R. D., Millar, B. C., Powles, R. L., Newman, A., Catovsky, D., Jarman, M. and Millar, J. L. (1998). Inhibitory effect of simvastatin on the proliferation of human

myeloid leukaemia cells in severe combined immunodeficient (SCID) mice. Br J Haematol 102:522-7.

Cohen, G. M., Sun, X. M., Snowden, R. T., Dinsdale, D. and Skilleter, D. N. (1992). Key morphological features of apoptosis may occur in the absence of internucleosomal DNA fragmentation. *Biochem J* 286:331-34.

Cohen, S. J., Ho, L., Ranganathan, S. and al., e. (2002). Phase II and pharmacodynamic trial of farnesyl transferase inhibitor R115777 as initial therapy in patients with metastatic pancreatic adenocarcinoma. *Proc Am Soc Clin Oncol* 21:137a.

Coleman, R. E. and Seaman, J. J. (2001). The role of zoledronic acid in cancer: clinical studies in the treatment and prevention of bone metastases. *Semin Oncol* 28:11-6.

Coniglio, S. J., Jou, T. S. and Symons, M. (2001). Rac1 protects epithelial cells against anoikis. *J Biol Chem* 276:28113-20.

Corey, E., Brown, L. G., Quinn, J. E., Poot, M., Roudier, M. P., Higano, C. S. and Vessella, R. L. (2003). Zoledronic acid exhibits inhibitory effects on osteoblastic and osteolytic metastases of prostate cancer,. *Clin Cancer Res*:295-306.

Cox, A. D. and Der, C. J. (1997). Farnesyltransferase inhibitors and cancer treatment: targeting simply Ras? *Biochim Biophys Acta* 1333:F51-71.

Coxon, F. P., Helfrich, M. H., Van't Hof, R., Sebti, S., Ralston, S. H., Hamilton, A. and Rogers, M. J. (2000). Protein geranylgeranylation is required for osteoclast formation, function, and survival: inhibition by bisphosphonates and GGTI-298. *J Bone Miner Res* 15:1467-76.

Coxon, F. P. and Rogers, M. J. (2003). The role of the prenylated small GTP-binding proteins in the regulation of osteoclast function. *Calcif Tissue Int* 72:80-84.

Crick, D. C., Andres, D. A., Danesi, R., Macchia, M. and Waechter, C. J. (1998). Geranylgeraniol overcomes the block of cell proliferation by lovastatin in C6 glioma cells. *J Neurochem* 70:2397-405.

Croucher, P. I., De Hendrik, R., Perry, M. J., Hijzen, A., Shipman, C. M., Lippitt, J., Green, J., Van Marck, E., Van Camp, B. and Vanderkerken, K. (2003). Zoledronic acid treatment of 5T2MM-bearing mice inhibits the development of myeloma bone disease: evidence for decreased osteolysis, tumor burden and angiogenesis, and increased survival. *J Bone Miner Res* 18:482-92.

Cruz, J. C., Alsina, M., Craig, F., Yoneda, T., Anderson, J. L., Dallas, M. and Roodman,G. D. (2001). Ibandronate decreases bone disease development and osteoclast stimulatory activity in an in vivo model of human myeloma. *Exp Hematol* 29:441-7.

Cuello, M., Ettenberg, S. A., Nau, M. M. and Lipkowitz, S. (2001). Synergistic induction of apoptosis by the combination of trail and chemotherapy in chemoresistant ovarian cancer cells. *Gynecol Oncol*:380-90.

Cunningham, D., de Gramont, A., Scheitauer, W., Smakal, M., Humblet, Y., Kurteva, G., Iveson, T., Andre, T., Dostalova, J., Illes, A., Jia, X. and Palmer, P. (2002). Randomized double-blind placebo-controlled trial of the farnesyl transferase inhibitor R115777 (Zarnestra) in advanced refractory colorectal cancer. *Proc Am Soc Clin Oncol* 21:502.

Dallas, S. L., Garrett, I. R., Oyajobi, B. O., Dallas, M. R., Boyce, B. F., Bauss, F., Radl, J. and Mundy, G. R. (1999). Ibandronate reduces osteolytic lesions but not tumor burden in a murine model of myeloma bone disease. *Blood* 93:1697-706.

D'Amours, D., Desnoyers, S., D'Silva, I. and Poirier, G. G. (1999). Poly(ADPribosylation) reaction in the regulation of nuclear functions polymerase. *Biochem J* 342:249-68.

Dancey, J. and Sausville, E. A. (2003). Issues and progress with protein kinase inhibitors for cancer treatment. *Nat Rev Drug Discov*:296-313.

Degli-Esposti, M. (1999). To die or not to die- the quest of the TRAIL receptors. *J of Leuk Biol* 65:535-42.

Degli-Esposti, M. A., Dougall, W. C., Smolak, P. J., Waugh, J., Smith, C. A. and Goodwin, R. G. (1997). The novel receptor TRAIL-R4 induces NF-kB and protects against TRAIL-mediated apoptosis, yet retains an incomplete death domain. *Immunity* 7:813-20.

Degli-Esposti, M. A., Smolak, P. J., Walczak, H., Waugh, J., Huang, C. P., DuBose, R. F., Goodwin, R. G. and Smith, C. A. (1997). Cloning and characterization of TRAIL-R3, a novel member of the emerging TRAIL receptor family. *J Exp Med* 186:1165-70.

Dejosez, M., Ramp, U., Mahotka, C., Krieg, A., Walczak, H., Gabbert, H. E. and Gerharz, C. D. (2000). Sensitivity to TRAIL/APO-2L-mediated apoptosis in human renal cell carcinomas and its enhancement by topotecan. *Cell Death Differ* 7:1127-36.

Deng, Y., Lin, Y. and Wu, X. (2002). TRAIL-induced apoptosis requires Baxdependent mitochondrial release of Smac/DIABLO. *Genes Dev* 16:33-45.

Denoyelle, C., Vasse, M., Korner, M., Mishal, Z., Ganne, F., Vannier, J. P., Soria, J. and Soria, C. (2001). Cerivastatin, an inhibitor of HMG-CoA reductase, inhibits the signaling pathways involved in the invasiveness and metastatic properties of highly invasive breast cancer cell lines: an *in vitro* study. *Carcinogenesis* 22:1139-48.

Desai, A. A., Innocenti, F. and Ratain, M. J. (2003). Pharmacogenomics: road to anticancer therapeutics nirvana? *Oncogene* 22:6621-28.

Deveraux, Q. L., Takahashi, R., Salvesen, G. S. and Reed, J. C. (1997). X-linked IAP is a direct inhibitor of cell-death proteases. *Nature* 388:300-303.

Deveraux, Q. L. and Reed, J. C. (1999). IAP family protein-suppressors of apoptosis. *Genes Dev* 13:239-52.

Di Paolo, A., Danesi, R., Caputo, S., Macchia, M., Lastella, M., Boggi, U., Mosca, F., Marchetti, A. and Del Tacca, M. (2001). Inhibition of protein farnesylation enhances the chemotherapeutic efficacy of the novel geranylgeranyltransferase inhibitor BAL9611 in human colon cancer cells. *Br J Cancer* 84:1535-43.

DiCaprio, M. R. and Friedlaender, G. E. (2003). Malignant bone tumors: limb sparing versus amputation. *Journal of American Academy of Orthopaedic Surgery* 11(1):25-37.

Diel, I. J., Solomayer, E. F., Costa, S. D., Gollan, C., Goerner, R., Wallwiener, D., Kaufmann, M. and Bastert, G. (1998). Reduction in new metastases in breast cancer with adjuvant clodronate treatment. *N Engl J Med* 339:357-63.

Dimitroulakos, J., Nohynek, D., Backway, K. L., Hedley, D. W., Yeger, H., Freedman, M. H., Minden, M. D. and Penn, L. Z. (1999). Increased sensitivity of acute myeloid leukemias to lovastatin-induced apoptosis: A potential therapeutic approach. *Blood* 93:1308-18.

Dimitroulakos, J., Ye, L. Y., Benzaquen, M., Moore, M. J., Kamel-Reid, S., Freedman, M. H., Yeger, H. and Penn, L. Z. (2001). Differential sensitivity of various pediatric cancers and squamous cell carcinomas to lovostatin-induced apoptosis: therapeutic implications. *Clin Cancer Res* 7:158-67.

Dinsmore, C. J., Williams, T. M., O'Neill, T. J., Liu, D., Rands, E., Culberson, J. C., Lobell, R. B., Koblan, K. S., Kohl, N. E., Gibbs, J. B., Oliff, A. I., Graham, S. L. and Hartman, G. D. (1999). Imidazole-containing diarylether and diarylsulfone inhibitors of farnesyl-protein transferase. *Bioorg Med Chem Lett* 9:3301-6.

Dinsmore, C. J., Zartman, C. B., Bergman, J. M., Abrams, M. T., Buser, C. A., Culberson, J. C., Davide, J. P., Ellis-Hutchings, M., Fernandes, C., Graham, S. L., Hartman, G. D., Huber, H. E., Lobell, R. B., Mosser, S. D., Robinson, R. G. and Williams, T. M. (2004). Macrocyclic piperazinones as potent dual inhibitors of farnesyltransferase and geranylgeranyltransferase-I. *Bioorg Med Chem Lett* 14:639-43.

Doney, M. K. and Vilke, G. M. (2004). Large osteosarcoma not apparent on conventional radiography. *Visual Diagnosis in Emergency Medicine* 26(3):351-352.

Dousset, M. and Buthiau, D. (1998). CT and MRI in oncology. New York, Springer.

Du, C., Fang, M., Li, Y., Li, L. and Wang, X. (2000). Smac, a mitochondrial protein that promotes cytochrome c-dependent caspase activation by eliminating IAP inhibition. *Cell* 102:33-42.

Dunford, J. E., Thompson, K., Coxon, F. P., Luckman, S. P., Hahn, F. M., Poulter, C. D., Ebetino, F. H. and Rogers, M. J. (2001). Structure-activity relationships for inhibition of farnesyl diphosphate synthase *in vitro* and inhibition of bone resorption *in vivo* by nitrogen-containing bisphosphonates. *J Pharmacol Exp Ther* 296:235-42.

Earnshaw, W. C., Martins, L. M. and Kaufmann, S. H. (1999). Mammalian caspases: structure, activation, substrates, and functions during apoptosis. *Annu Rev Biochem* 68:383-424.

Ebetino, F. H., Francis, M. D., Rogers, M. J. and Russell, R. G. G. (1998). Etidronate. Mechanisms of action of etidronate and other bisphosphonates. *Rev Contemp Pharm* 9:233-43.

Edwards, P. A. and Ericsson, J. (1998). Signaling molecules derived from the cholesterol biosynthetic pathway: mechanisms of action and possible roles in human disease. *Curr Opin Lipidol* 9:433-40.

Eggert, A., Grotzer, M. A., Zuzak, T. J., Wiewrodt, B. R., Ho, R., Ikegaki, N. and Brodeur, G. M. (2001). Resistance to tumor necrosis factor-related apoptosis-inducing ligand (TRAIL)-induced apoptosis in neuroblastoma cells correlates with a loss of caspase-8 expression. *Cancer Res* 61:1314-9.

El-Deiry, W. S. (2001). Insights into cancer therapeutic design based on p53 and TRAIL receptor signaling. *Cell Death Differ* 8:1066-75.

Ellis, H. M. and Horvitz, H. R. (1986). Genetic control of programmed cell death in the nematode *C. elegans. Cell* 44:817-29.

Emery, J. G., McDonnell, P., Burke, M. B., Deen, K. C., Lyn, S., Silverman, C., Dul, E., Appelbaum, E. R., Eichman, C., DiPrinzio, R., Dodds, R. A., James, I. E., Rosenberg, M., Lee, J. C. and Young, P. R. (1998). Osteoprotegerin is a receptor for the cytotoxic ligand TRAIL. *J Biol Chem* 273:14363-67.

End, D. W., Smets, G., Todd, A. V., Applegate, T. L., Fuery, C. J., Angibaud, P., Venet, M., Sanz, G., Poignet, H., Skrzat, S., Devine, A., Wouters, W. and Bowden, C. (2001). Characterization of the antitumor effects of the selective farnesyl protein transferase inhibitor R115777 *in vivo* and *in vitro*. *Cancer Res* 61:131-7.

Evan, G. and Littlewood, T. (1998). A matter of life and death. Science 281:1317-22.

Evdokiou, A., Atkins, G. J., Bouralexis, S., Hay, S., Raggatt, L. J., Cowled, P. A., Graves, S. E., Clayer, M. and Findlay, D. M. (2001). Expression of alternatively-spliced MDM2 transcripts in giant cell tumours of bone. *Int J Oncol* 19(3):625-32.

Evdokiou, A., Bouralexis, S., Atkins, G. J., Chai, F., Hay, S., Clayer, M. and Findlay, D. M. (2002). Chemotherapeutic agents sensitize osteogenic sarcoma cells, but not normal human bone cells, to Apo2L/TRAIL-induced apoptosis. *Int J Cancer* 99:491-504.

Evdokiou, A., Labrinidis, A., Bouralexis, S., Hay, S. and Findlay, D. M. (2003). Induction of cell death of human osteogenic sarcoma cells by zoledronic acid resembles anoikis. *Bone* 33:216-28.

Exner, G. U., Von Hochstetter, A. R., Augustiny, N. and Von Schulthess, G. (1990). Magnetic resonance imaging in malignant bone tumors. *International orthopaedics* 14:49-55. Fanger, N. A., Maliszewski, C. R., Schooley, K. and Griffith, T. S. (1999). Human dendritic cells mediate cellular apoptosis via tumour necrosis factor-related apoptosis-inducing ligand (TRAIL). *J Exp Med* 190:1155-64.

Farese, J. P., Ashton, J. and Milner R, A. L., Van Gilder J. (2004). The effect of the bisphosphonate alendronate on viability of canine osteosarcoma cells in vitro. *In Vitro Cell Dev Biol Anim* 40:113-7.

Farrow, S. N. and Brown, R. (1996). New members of the Bcl-2 family and their protein partners. *Curr Opin Genet Dev* 6:45-49.

Feleszko, W., Lasek, W., Golab, J. and Jakobisiak, M. (1995). Synergistic antitumor activity of tumor necrosis factor-alpha and lovastatin against MmB16 melanoma in mice. *Neoplasma* 42:69-74.

Feleszko, W., Zagozdzon, R., Golab, J. and Jakobisiak, M. R. (1998). Potentiated antitumour effects of cisplatin and lovastatin against MmB16 melanoma in mice. *Eur J Cancer* 34:406-11.

Feleszko, W., Balkowiec, E. Z., Sieberth, E., Marczak, M., Dabrowska, A., Giermasz, A., Czajka, A. and Jakobisiak, M. (1999). Lovastatin and tumor necrosis factor-alpha exhibit potentiated antitumor effects against Ha-ras-transformed murine tumor via inhibition of tumor-induced angiogenesis. *Int J Cancer* 81:560-7.

Feleszko, W., Mlynarczuk, I., Balkowiec-Iskra, E. Z., Czajka, A., Switaj, T., Stoklosa, T., Giermasz, A. and Jakobisiak, M. (2000). Lovastatin potentiates antitumor activity and attenuates cardiotoxicity of doxorubicin in three tumor models in mice. *Clin Cancer Res* 6:2044-52.

Fisher, J. E., Rogers, M. J., Halasy, J. M., Luckman, S. P., Hughes, D. E., Masarachia, P. J., Wesolowski, G., Russell, R. G., Rodan, G. A. and Reszka, A. A. (1999). Alendronate mechanism of action: geranylgeraniol, an intermediate in the mevalonate pathway, prevents inhibition of osteoclast formation, bone resorption, and kinase activation in vitro. *Proc Natl Acad Sci US A* 96:133-8.

Fisher, J. E., Rodan, G. A. and Reszka, A. A. (2000). *In vivo* effects of bisphosphonates on the osteoclast mevalonate pathway. *Endocrin* 141:4793-96.

FitzGerald, T. J., Daugherty, C., Kase, K., Rothstein, L. A., McKenna, M. and Greenberger, J. S. (1985). Activated human N-ras oncogene enhances x-irradiation repair of mammalian cells in vitro less effectively at low dose rate. Implications for increased therapeutic ratio of low dose rate irradiation. *Am J Clin Oncol* 8:517-22.

Fleisch, H., Russell, R. G. G., Bisaz, S., Casey, P. and Muhlbauer, R. (1968). The influence of pyrophosphate analogues (diphosphates) on the precipitation and dissolution of calcium phosphate *in vitro* and *in vivo*. *Calcif Tissue Res* 2(Suppl):10-10a.

Fleisch, H. (2000). <u>Bisphosphonates in bone disease</u>. From the laboratory to the patient. San Diego, Academic Press.

Forsea, A. M., Muller, C., Riebeling, C., Orfanos, C. E. and Geilen, C. C. (2004). Nitrogen-containing bisphosphonates inhibit cell cycle progression in human melanoma cells. *Br J Cancer* 91:803-10.

Fraser, A. and James, C. (1998). Fermenting debate: Do yeast undergo apoptosis? *Trends Cell Biol* 8:219-21.

French, L. E. and Tschopp, J. (1999). The TRAIL to selective tumor death. *Nat Med* 5:146-7.

Frisch, S. M. and Francis, H. (1994). Disruption of epithelial cell-matrix interactions induces apoptosis. *Journal of Cell Biology* 124:619-26.

Fromigue, O., Lagneaux, L. and Body, J. J. (2000). Bisphosphonates induce breast cancer cell death *in vitro*. *J Bone Miner Res* 15:2211-21.

Fromigue, O. and Body, J. J. (2002). Bisphosphonates influence the proliferation and the maturation of normal human osteoblasts. *J Endocrinol Invest* 25:539-46.

Fulda, S., Kufer, M. U., Meyer, E., van Valen, F., Dockhorn-Dworniczak, B. and Debatin, K. M. (2001). Sensitization for death receptor- or drug-induced apoptosis by re-expression of caspase-8 through demethylation or gene transfer. *Oncogene* 20:5865-77.

Fulda, S., Wick, W., Weller, M. and Debatin, K. M. (2002). Smac agonists sensitize for Apo2L/TRAIL- or anticancer drug-induced apoptosis and induce regression of malignant glioma in vivo. *Nat Med* 8:808-15.

Garcia, A. M., Rowell, C., Ackermann, K., Kowalczyk, J. J. and Lewis, M. D. (1993). Peptidomimetic inhibitors of Ras farnesylation and function in whole cells. *J Biol Chem* 268:18415-8.

Garcia-Moreno, C., Serrano, S., Nacher, M., Farre, M., Diez, A., Marinoso, M. L., Carbonell, J., Mellibovsky, L., Nogues, X., Ballester, J. and Aubia, J. (1998). Effect of alendronate on cultured normal human osteoblasts. *Bone* 22:233-9.

Gaur, U. and Aggarwal, B. B. (2003). Regulation of proliferation, survival and apoptosis by members of the TNF superfamily. *Biochem Pharmacol* 66:1403-8.

Gazitt, Y. (1999). TRAIL is a potent inducer of apoptosis in myeloma cells derived from multiple myeloma patients and is not cytotoxic to hematopoietic stem cells. *Leukemia* 13:1817-24.

Gibbs, J. B. and Oliff, A. (1997). The potential of farnesyltransferase inhibitors as cancer chemotherapeutics. *Annu Rev Pharmacol Toxicol* 37:143-66.

Gibson, S. B., Oyer, R., Spalding, A. C., Anderson, S. M. and Johnson, G. L. (2000). Increased expression of death receptors 4 and 5 synergizes the apoptosis response to combined treatment with etoposide and TRAIL. *Mol Cell Biol* 20(1):205-12.

Giuliani, N., Pedrazzoni, M., Negri, G., Passeri, G., Impicciatore, M. and Girasole, G. (1998). Bisphosphonates stimulate formation of osteoblast precursors and mineralized nodules in murine and human bone marrow cultures *in vitro* and promote early osteoblastogenesis in young and aged mice *in vivo*. *Bone* 22:455-61.

Giuliani, N., Pedrazzoni, M., Passeri, G. and Girasole, G. (1998). Bisphosphonates inhibit IL-6 production by human osteoblast-like cells. *Scand J Rheumatol*:38-41.

Gliniak, B. and Le, T. (1999). Tumor necrosis factor-related apoptosis-inducing ligand's antitumor activity in vivo is enhanced by the chemotherapeutic agent CPT-11. *Cancer Res* 59:6153-8.

Goldstein, J. L. and Brown, M. S. (1990). Regulation of the mevalonate pathway. *Nature* 343:425-30.

Goltzman, D. (2001). Osteolysis and cancer. J Clin Invest 107:1219-20.

Gong, B. and Almasan, A. (2000). Apo2 ligand/TNF-related apoptosis-inducing ligand and death receptor 5 mediate the apoptotic signaling induced by ionizing radiation in leukemic cells. *Cancer Res* 60:5754-60.

Gotlib, J., Dugan, K., U., K. and al., e. (2002). PhaseI/II study of farnesyl transferase inhibitor, R115777 (Zarnestra) in patients with myeloproliferative disorders (MPDs): Preliminary results. *Proc Natl Acad Sci* 21:4a.

Gray, C. W., Ward, R. V., Karran, E., Turconi, S., Rowles, A., Viglienghi, D., Southan,
C., Barton, A., Fantom, K. G., West, A., Savopoulos, J., Hassan, N. J., Clinkenbeard,
H., Hanning, C., Amegadzie, B., Davis, J. B., Dingwall, C., Livi, G. P. and Creasy, C.
L. (2000). Characterization of human HtrA2, a novel serine protease involved in the mammalian cellular stress response. *Eur J Biochem* 267:5699-710.

Green, D. and Reed, J. C. (1998). Mitochondria and apoptosis. Science 281:1309-12.

Green, D. R. (1998). Apoptotic pathways: the road to ruin. Cell 94:695-98.

Green, D. R. (2000). Apoptotic pathways: paper wraps stone blunts scissors. *Cell* 102:1-4.

Green, J. (2003). <u>Effects of bisphosphonates on tumor cells.</u> IVth International Conference on Cancer-Induced Bone Diseases, San Antonio, TX, USA.

Green, J. R., Muller, K. and Jaeggi, K. A. (1994). Preclinical pharmacology of CGP 42,446 a new, potent, heterocyclic bisphosphonate compound. *J Bone Miner Res* 9:745-51.

Green, J. R. (2002). Bisphosphonates in cancer therapy. Curr Opin Oncol:609-15.

Green, J. R. (2003). Antitumor effects of bisphosphonates. Cancer 97:840-7.

Griffith, T. S., Chin, W. A., Jackson, G. C., Lynch, D. H. and Kubin, M. Z. (1998). Intracellular regulation of TRAIL-induced apoptosis in human melanoma cells. *J Immunol* 161:2833-40.

Griffith, T. S., Wiley, S. R., Kubin, M. Z., Sedger, L. M., Maliszewski, C. R. and Fanger, N. A. (1999). Monocyte-mediated tumoricidal activity via the tumour necrosis factor-related cytokine, TRAIL. *J Exp Med* 189:1343-54.

Griffith, T. S., Anderson, R. D., Davidson, B. L., Williams, R. D. and Ratliff, T. L. (2000). Adenoviral-mediated transfer of the TNF-related apoptosis-inducing ligand/Apo-2 ligand gene induces tumor cell apoptosis. *J Immunol* 165:2886-94.

Griffith, T. S. and Broghammer, E. L. (2001). Suppression of tumor growth following intralesional therapy with TRAIL recombinant adenovirus. *Mol Ther* 4:257-66.

Grimaud, E., Redini, F. and Heymann, D. (2001). Osteoprotegerin: a new therapeutic agent for the treatment of bone disease. *Drug Discov Today* 6:1241-1242.

Gronthos, S., Zannettino, A. C., Graves, S. E., Ohta, S., Hay, S. J. and Simmons, P. J. (1999). Differential cell surface expression of the STRO-1 and alkaline phosphatase antigens on discrete developmental stages in primary cultures of human bone cells. *J Bone Miner Res* 14:47-56.

Gruss, H. J. and Dower, S. K. (1995). The TNF ligand superfamily and its relevance for human diseases. *Cytokines Mol Ther* 1:75-105.

Guan, B., Yue, P., Clayman, G. L. and Sun, S. Y. (2001). Evidence that the death receptor DR4 is a DNA damage-inducible, p53-regulated gene. *J Cell Physiol* 188:98-105.

Gura, T. (1997). How TRAIL kills cancer cells, but not normal cells. Science 277:768.

Haas, N., Peereboom, D., Ranganathan, S. and al., e. (2002). Phase II trial of R115777, an inhibitor of farnesyltransferase, in patients with hormone refractory prostate cancer. *Proc Am Soc Clin Oncol* 21:181a.

Hakem, R., Hakem, A., Duncan, G. S., Henderson, J. T., Woo, M., Soengas, M. S., Elia, A., de la Pompa, J. L., Kagi, D., Khoo, W., Potter, J., Yoshida, R., Kaufman, S. A., Lowe, S. W., Penninger, J. M. and Mak, T. W. (1998). Differential requirement for caspase 9 in apoptotic pathways *in vivo*. *Cell* 94:339-52.

Halaas, O. R., Vik, R., Ashkenazi, A. and Espevik, T. (2000). Lipopolysaccharide induces expression of APO2 ligand/TRAIL in human monocytes and macrophages. *Scan J Immunol* 51:244-50.

Hall, D. G. and Stoica, G. (1994). Effect of the bisphosphonate risedronate on bone metastases in a rat mammary adenocarcinoma model system. *J Bone Miner Res* 9:221-30.

Hanada, M., Aime-Sempe, C., Sato, T. and Reed, J. C. (1995). Structure-function analysis of Bcl-2 protein: identification of conserved domains important for

homodimerization with Bcl-2 and heterodimerization with Bax. J Biol Chem 270:11962-69.

Harmon, T. P. and Norton, K. S. (1966). Osteogenic sarcoma in four siblings. *Journal* of Bone and Joint Surgery 48:493-498.

Hawk, M. A., Cesen, K. T., Siglin, J. C., Stoner, G. D. and Ruch, R. J. (1996). Inhibition of lung tumor cell growth in vitro and mouse lung tumor formation by lovastatin. *Cancer Lett* 109:217-22.

Hayakawa, Y., Smyth, M. J., Kayagaki, N., Kakuta, S., Iwakura, Y., Yagita, H., Okumura, K. and Takeda, K. (2001). Involvement of tumour necrosis factor-related apoptosis-inducing ligand in surveillance of tumour metastasis by liver natural killer cells. *J Exp Med* 7:94-100.

Hebert, P. R., Gaziano, J. M., Chan, K. S. and Hennekens, C. H. (1997). Cholesterol lowering with statin drugs, risk of stroke, and total mortality. An overview of randomized trials. *JAMA* 278:313-21.

Hedgecock, E. M., Sulston, J. E. and Thomson, J. N. (1983). Mutations affecting programmed cell deaths in the nematode *Caenorhabditis elegans*. *Science* 220:1277-79.

Hegde, R., Srinivasula, S. M., Zhang, Z., Wassell, R., Mukattash, R., Cilenti, L., DuBois, G., Lazebnik, Y., Zervos, A. S., Fernandes-Alnemri, T. and Alnemri, E. S. (2002). Identification of Omi/HtrA2 as a mitochondrial apoptotic serine protease that disrupts inhibitor of apoptosis protein-caspase interaction. *J Biol Chem* 277:432-8.

Heikkila, P., Teronen, O., Moilanen, M., Konttinen, Y. T., Hanemaaijer, R., Laitinen, M., Maisi, P., van der Pluijm, G., Bartlett, J. D., Salo, T. and Sorsa, T. (2002). Bisphosphonates inhibit stromelysin-1 (MMP-3), matrix metalloelastase (MMP-12), collagenase-3 (MMP-13) and enamelysin (MMP-20), but not urokinase-type plasminogen activator, and diminish invasion and migration of human malignant and endothelial cell lines. *Anticancer Drugs* 13:245-54.

.

Hengartner, M. O., Ellis, H. M. and Horvitz, H. R. (1992). *Caenorhabditis elegans* gene *ced-9* protects cells from programmed cell death. *Nature* 356:494-99.

Hersey, P. and Zhang, X. D. (2001). How melanoma cells evade trail-induced apoptosis. *Nat Rev Cancer* 1:142-50.

Heymach, J. V., Johnson, D. H., Khuri, F. R., Safran, H., Schlabach, L. L., Yunus, F., DeVore, R. F. r., De Porre, P. M., Richards, H. M., Jia, X., Zhang, S. and Johnson, B. E. (2004). Phase II study of the farnesyl transferase inhibitor R115777 in patients with sensitive relapse small-cell lung cancer. *Ann Oncol* 15:1187-93.

Hiraga, T., Williams, P. J., Mundy, G. R. and Yoneda, T. (2001). The bisphosphonate ibandronate promotes apoptosis in MDA-MB-231 human breast cancer cells in bone metastases. *Cancer Res* 61:4418-24.

Hiraga, T., Ueda, A., Tamura, D., Hata, K., Ikeda, F., Williams, P. J. and Yoneda, T. (2003). Effects of oral UFT combined with or without zoledronic acid on bone metastasis in the 4T1/luc mouse breast cancer. *Int J Cancer* 106:973-9.

Hiraga, T., Williams, P. J., Ueda, A., Tamura, D. and Yoneda, T. (2004). Zoledronic acid inhibits visceral metastases in the 4T1/luc mouse breast cancer model. *Clin Cancer Res* 10:4559-67.

Holcik, M. and Korneluk, R. G. (2001). XIAP, the guardian angel. Nat Rev Mol Cell Biol 7:550-556.

Holen, I., Croucher, P. I., Hamdy, F. C. and Eaton, C. L. (2002). Osteoprotegerin (OPG) is a survival factor for human prostate cancer cells. *Cancer Res* 62:1619-23.

Holstein, S. A. and Hohl, R. J. (2001). Interaction of cytosine arabinoside and lovastatin in human leukemia cells. *Leuk Res* 25:651-60.

Hsu, H., Xiong, J. and Goeddel, D. V. (1995). The TNF receptor 1-associated protein TRADD signals cell death and NF-kappa B activation. *Cell* 81:495-504.

Huang, D. C. and Strasser, A. (2000). BH3-proteins- essential initiators of apoptotic cell death. *Cell* 103:839-42.

Huang, Y., Park, Y. C., Rich, R. L., Segal, D., Myszka, D. G. and Wu, H. (2001). Structural basis of caspase inhibition by XIAP: differential roles of the linker versus the BIR domain. *Cell* 104:781-90.

Hughes, D. E., Wright, K. R., Uy, H. L., Sasaki, A., Yoneda, T., Roodman, G. D., Mundy, G. R. and Boyce, B. F. (1995). Bisphosphonates promote apoptosis in murine osteoclasts *in vitro* and *in vivo*. *J Bone Miner Res* 10:1478-87.

Huvos, A. G. (1991). Bone tumors: diagnosis, treatment and prognosis. Philadelphia, W.B. Saunders.

Hymowitz, S. G., Christinger, H. W., Fuh, G., Ultsch, M., O'Connell, M., Kelley, R. F., Ashkenazi, A. and de Vos, A. M. (1999). Triggering cell death: the crystal structure of Apo2L/TRAIL in a complex with death receptor 5. *Mol Cell* 4:563-71.

Idriss, H. T. and Naismith, J. H. (2000). TNF alpha and the TNF receptor superfamily: structure-function relationship(s). *Microsc Res Tech* 50:184-95.

Igney, F. H. and Krammer, P. H. (2002). Immune escape of tumors: apoptosis resistance and tumor counterattack. *J Leukoc Biol* 71:907-20.

Igney, F. H. and Krammer, P. H. (2002). Death and anti-death: tumour resistance to apoptosis. *Nat Rev Cancer* 2:277-88.

Iguchi, T., Miyakawa, Y., Yamamoto, K., Kizaki, M. and Ikeda, Y. (2003). Nitrogencontaining bisphosphonates induce S-phase cell cycle arrest and apoptosis of myeloma cells by activating MAPK pathway and inhibiting mevalonate pathway. *Cell Signal* 15:719-27. Im, G. I., Qureshi, S. A., Kenney, J., Rubash, H. E. and Shanbhag, A. S. (2004). Osteoblast proliferation and maturation by bisphosphonates. *Biomaterials* 25:4105-15.

Inano, H., Suzuki, K., Onoda, M. and Wakabayashi, K. (1997). Anti-carcinogenic activity of simvastatin during the promotion phase of radiation-induced mammary tumorigenesis of rats. *Carcinogenesis* 18:1723-7.

Irmler, M., Thome, M., Hahne, M., Schneider, P., Hofmann, K., Steiner, V., Bodmer, J. L., Schroter, M., Burns, K., Mattmann, C., Rimoldi, D., French, L. E. and Tschopp, J. (1997). Inhibition of death receptor signals by cellular FLIP. *Nature* 388:190-95.

Istvan, E. S., Palnitkar, M., Buchanan, S. K. and Deisenhofer, J. (2000). Crystal structure of the catalytic portion of human HMG-CoA reductase: insights into regulation of activity and catalysis. *EMBO J* 19:819-30.

Istvan, E. S. and Deisenhofer, J. (2001). Structural mechanism for statin inhibition of HMG-CoA reductase. *Science* 292:1160-4.

Ito, M., Amizuka, N., Nakajima, T. and Ozawa, H. (1999). Ultrastructural and cytochemical studies on cell death of osteoclasts induced by bisphosphonate treatment. *Bone* 25:447-52.

Izumo, N., Fujita, T., Nakamuta, H. and Koida, M. (2001). Lipophilic statins can be osteogenic by promoting osteoblastic calcification in a Cbfa1- and BMP-2-independent manner. *Methods Find Exp Clin Pharmacol* 23:389-94.

Jacobson, M. D., Weil, M. and Raff, M. C. (1997). Programmed cell death in animal development. *Cell* 88:347-54.

Jagdev, S. P., Coleman, R. E., Shipman, C. M., Rostami, H. A. and Croucher, P. I. (2001). The bisphosphonate, zoledronic acid, induces apoptosis of breast cancer cells: evidence for synergy with paclitaxel. *Br J Cancer* 84:1126-34.

James, G. L., Goldstein, J. L. and Brown, M. S. (1995). Polylysine and CVIM sequences of K-RasB dictate specificity of prenylation and confer resistance to benzodiazepine peptidomimetic *in vitro*. *J Biol Chem* 270:6221-6.

Jani, J. P., Specht, S., Stemmler, N., Blanock, K., Singh, S. V., Gupta, V. and Katoh, A. (1993). Metastasis of B16F10 mouse melanoma inhibited by lovastatin, an inhibitor of cholesterol biosynthesis. *Invasion Metastasis* 13:314-24.

Jazirehi, A. R., Ng, C. P., Gan, X. H., Schiller, G. and Bonavida, B. (2001). Adriamycin sensitizes the adriamycin-resistant 8226/Dox40 human multiple myeloma cells to Apo2L/tumor necrosis factor-related apoptosis-inducing ligand-mediated (TRAIL) apoptosis. *Clin Cancer Res* 7:3874-83.

Jeggo, P. A. (1998). DNA repair: PARP- Another guardian angel? Current Biol 8:49-51.

Jiang, X. and Wang, X. (2000). Cytochrome c promotes caspase-9 activation by inducing nucleotide binding to Apaf-1. *J Biol Chem* 275:31199-203.

Jiang, Y., Woronicz, J. D., Liu, W. and Goeddel, D. V. (1999). Prevention of constitutive TNF receptor 1 signaling by silencer of death domains. *Science* 283:543-6.

Jin, Z., Dicker, D. T. and El-Deiry, W. S. (2002). Enhanced sensitivity of G1 arrested human cancer cells suggests a novel therapeutic strategy using a combination of simvastatin and TRAIL. *Cell Cycle* 1:82-9.

Jo, M., Kim, T. H., Seol, D. W., Esplen, J. E., Dorko, K., Billiar, T. R. and Strom, S. C. (2000). Apoptosis induced in normal human hepatocytes by tumor necrosis factor-related apoptosis-inducing ligand. *Nat Med* 6:564-7.

Johnsen, A. C., Haux, J., Steinkjer, B., Nonstad, U., Egeberg, K., Sundan, A., Ashkenazi, A. and Espevik, T. (1999). Regulation of APO-2 ligand/TRAIL expression in NK cells: involvement in NK cell-mediated cytotoxicity. *Cytokines Mol Ther* 11:664-72.

Johnson, D. E. (2000). Noncaspase proteases in apoptosis. Leukemia 14:1695-1703.

Johnston, S. R. D., Hickish, T., Houston, S. and al., e. (2002). Efficacy and tolerability of two dosing regimens of R115777 (Zarnestra), a farnesyl transferase inhibitor, in patients with advanced breast cancer. *Proc Am Soc Clin Oncol* 21:35a.

Johnstone, R. W., Ruefli, A. A. and Lowe, S. W. (2002). Apoptosis: a link between cancer genetics and chemotherapy. *Cell* 108:153-64.

Jones, K. D., Couldwell, W. T., Hinton, D. R., Su, Y., He, S., Anker, L. and Law, R. E. (1994). Lovastatin induces growth inhibition and apoptosis in human malignant glioma cells. *Biochem Biophys Res Commun* 205:1681-7.

Jones, R. A., Johnson, V. L., Hinton, R. H., Poirier, G. G., Chow, S. C. and Kass, G. E. N. (1999). Liver poly(ADP-ribose) polymerase is resistant to cleavage by caspases. *Biochem Biophys Res Commun* 256:436-41.

Joza, N., Susin, S. A., Daugas, E., Stanford, W. L., Cho, S. K., Li, C. Y., Sasaki, T., Elia, A. J., Cheng, H. Y., Ravagnan, L., Ferri, K. F., Zamzami, N., Wakeham, A., Hakem, R., Yoshida, H., Kong, Y. Y., Mak, T. W., Zuniga-Pflucker, J. C., Kroemer, G. and Penninger, J. M. (2001). Essential role of the mitochondrial apoptosis-inducing factor in programmed cell death. *Nature* 410:549-54.

Juin, P. and Evan, G. (2000). Caspase 8: the killer you can't live without. *Nat Med* 6:498-500.

Jung, A., Bisaz, S. and Fleisch, H. (1973). The binding of pyrophosphate and two diphosphonates by hydroxyapatite crystals. *Calcif Tissue Res* 11:269-80.

Juo, P., Kuo, C. J., Yuan, J. and Blenis, J. (1998). Essential requirement for caspase-8/FLICE in the initiation of the Fas-induced apoptotic cascade. *Curr Biol* 8:1001-08. 10 TO 10

Jurgensmeier, J. M., Xie, Z., Deveraux, Q. L., Ellerby, D., Bredeson, D. and Reed, J. C. (1998). Bax directly induces release of cytochrome c from isolated mitochondria. *Proc Natl Acad Sci* 95:4997-5002.

Kabakoff, B. D., Doyle, J. W. and Kandutsch, A. A. (1990). Relationships among dolichyl phosphate, glycoprotein synthesis, and cell culture growth. *Arch Biochem Biophys* 276:382-9.

Kagawa, S., He, C., Gu, J., Koch, P., Rha, S. J., Roth, J. A., Curley, S. A., Stephens, L. C. and Fang, B. (2001). Antitumor activity and bystander effects of the tumor necrosis factor-related apoptosis-inducing ligand (TRAIL) gene. *Cancer Res* 61:3330-8.

Kang, S. J., Wang, S., Hara, H., Peterson, E. P., Namura, S., Amin-Hanjiani, S., Huang, Z., Srinivasan, A., Tomaselli, K. J., Thornberry, N. A., Moskowitz, M. A. and Yuan, Y. (2000). Dual role of caspase-11 in mediating caspase activation of caspase-1 and caspase-3 under pathological conditions. *J Cell Biol* 149:613-622.

Karp, J. E., Lancet, J. E., Kaufmann, S. H., End, D. W., Wright, J. J., Bol, K., Horak, I., Tidwell, M. L., Liesveld, J., Kottke, T. J., Ange, D., Buddharaju, L., Gojo, I., Highsmith, W. E., Belly, R. T., Hohl, R. J., Rybak, M. E., Thibault, A. and Rosenblatt, J. (2001). Clinical and biologic activity of the farnesyltransferase inhibitor R115777 in adults with refractory and relapsed acute leukemias: a phase 1 clinical-laboratory correlative trial. *Blood* 97:3361-9.

Kashii, Y., Giorda, R., Herberman, R. B., Whiteside, T. L. and Vujanovic, N. L. (1999). Constitute expression and role of the TNF family ligands in apoptotic killing of tumour cells by human NK cells. *J Immunol* 163:5358-66.

Kataoka, T. B. R., Holler, N., Thome, M., Martinon, F., Irmler, M., Burns, K., Hahne, M., Kennedy, N., Kovacsovics, M. and Tschopp, J. (2000). The caspase-8 inhibitor FLIP promotes activation of NF-kappaB and erk signalling pathways. *Curr Biol* 10:640-48.

Kaufmann, S. H., Desnoyers, S., Ottaviano, Y., Davidson, N. E. and Poirier, G. G. (1993). Specific proteolytic cleavage of poly(ADP-ribose) polymerase: an early marker of chemotherapy-induced apoptosis. *Cancer Res* 53:3976-85.

Kawata, S., Kakimoto, H., Ishiguro, H., Yamasaki, E., Inui, Y. and Matsuzawa, Y. (1992). Effect of pravastatin, a potent 3-hydroxy-3-methylglutaryl-coenzyme A reductase inhibitor, on survival of AH130 hepatoma-bearing rats. *Jpn J Cancer Res* 83:1120-3.

Kawata, S., Yamasaki, E., Nagase, T., Inui, Y., Ito, N., Matsuda, Y., Inada, M., Tamura, S., Noda, S., Imai, Y. and Matsuzawa, Y. (2001). Effect of pravastatin on survival in patients with advanced hepatocellular carcinoma. A randomized controlled trial. *Br J Cancer* 84:886-91.

Kayagaki, N., Yamaguchi, N., Nakayama, M., Eto, H., Okumura, K. and Yagita, H. (1999). Type I interferons (IFNs) regulate tumour necrosis factor related apoptosisinducing ligand (TRAIL) expression on human T cells: a novel mechanism for the antitumour effects of type I IFNs. *J Exp Med* 189:1451-60.

Kayagaki, N., Yamaguchi, N., Nakayama, M., Takeda, K., Tsutsui, H., Okumura, H., Nakanishi, K., Okumura, K. and Yagita, H. (1999). Expression and function of the TNF-related apoptosis-inducing ligand on murine activated NK cells. *J Immunol* 163:1906-13.

Keane, M. M., Ettenberg, S. A., Nau, M. M., Russell, E. K. and Lipkowitz, S. (1999). Chemotherapy augments TRAIL-induced apoptosis in breast cell lines. *Cancer Res* 59:734-41.

Keating, A. (2002). Chronic myeloid leukemia: current therapies and the potential role of farnesyltransferase inhibitors. *Semin Hematol* 39:11-7.

Kelekar, A. and Thompson, C. B. (1998). Bcl-2 family proteins: the role of the BH3 domain in apoptosis. *Trends Cell Biol* 8:324-30.

Kelley, S. K., Harris, L. A., Xie, D., Deforge, L., Totpal, K., Bussiere, J. and Fox, J. A. (2001). Preclinical studies to predict the disposition of Apo2L/tumor necrosis factorrelated apoptosis-inducing ligand in humans: characterization of *in vivo* efficacy, pharmacokinetics, and safety. *J Pharmacol Exp Ther* 299:31-8.

Kerr, J. F. K. and Harmon, B. V. (1991). Definition and incidence of apoptosis: A historical perspective. <u>Apoptosis: The molecular basis of cell death.</u> Tomei, L. D. and Cope, F. O. Cold Spring Harbour, NY, Cold Spring Harbour Laboratory Press.

Kerr, J. F. R., Wyllie, A. H. and Currie, A. H. (1972). Apoptosis, a basic biological phenomenon with wider applications in tissue kinetics. *Br J Cancer* 26:239-45.

Keyomarsi, K., Sandoval, L., Band, V. and Pardee, A. B. (1991). Synchronization of tumor and normal cells from G1 to multiple cell cycles by lovastatin. *Cancer Res* 51:3602-9.

Khokher, M. A. and Dandona, P. (1989). Diphosphonates inhibit human osteoblast secretion and proliferation. *Metabolism* 38:184-7.

Khwaja, A. and Downward, J. (1997). Lack of correlation between activation of Jun-NH2-terminal kinase and induction of apoptosis after detachment of epithelial cells. *J Cell Biol* 139:1017-23.

Kikuchi, T., Nagata, Y. and Abe, T. (1997). *In vitro* and *in vivo* antiproliferative effects of simvastatin, an HMG-CoA reductase inhibitor, on human glioma cells. *J Neurooncol* 34:233-9.

Kim, I. K., Chung, C. W., Woo, H. N., Hong, G. S., Nagata, S. and Jung, Y. (2000). Reconstitution of caspase-8 sensitises JB6 cells to TRAIL. *Biochem and Biophys Res Comm* 277:311-16.

Kim, W. S., Kim, M. M., Choi, H. J., Yoon, S. S., Lee, M. H., Park, K., Park, C. H. and Kang, W. K. (2001). Phase II study of high-dose lovastatin in patients with advanced gastric adenocarcinoma. *Invest New Drugs* 19:81-3.

Kimura, S., Kuroda, J., Segawa, H., Sato, K., Nogawa, M., Yuasa, T., Ottmann, O. G. and Maekawa, T. (2004). Antiproliferative efficacy of the third-generation bisphosphonate, zoledronic acid, combined with other anticancer drugs in leukemic cell lines. *Int J Hematol* 79:37-43.

Kischkel, F. C., Lawrence, D. A., Tinel, A., LeBlanc, H., Virmani, A., Schow, P., Gazdar, A., Blenis, J., Arnott, D. and Ashkenazi, A. (2001). Death receptor recruitment of endogenous caspase-10 and apoptosis initiation in the absence of caspase-8. *J Biol Chem* 276:46639-46.

Kohl, N. E., Omer, C. A., Conner, M. W., Anthony, N. J., Davide, J. P., deSolms, S. J., Giuliani, E. A., Gomez, R. P., Graham, S. L. and Hamilton, K. (1995). Inhibition of farnesyltransferase induces regression of mammary and salivary carcinomas in ras transgenic mice. *Nat Med* 1:792-7.

Kostenuik, P. J., Orr, F. W., Suyama, K. and Singh, G. (1993). Increased growth rate and tumor burden of spontaneously metastatic Walker 256 cancer cells in the skeleton of bisphosphonate-treated rats. *Cancer Res* 53:5452-7.

Kothakota, S., Azuma, T., Reinhard, C., Klippel, A., Tang, J., Chu, K., McGarry, T. J., Kirschner, M. W., Koths, K., Kwiatkowski, D. J. and Williams, L. T. (1997). Caspase-3 generated fragment of gelsolin: effector of morphological change in apoptosis. *Science* 278:294-98.

Kuida, K., Lippke, J. A., Ku, G., Harding, M. W., Livingston, D. J., Su, M. S. and Flavell, R. A. (1995). Altered cytokine export and apoptosis in mice deficient in interleukin-1β converting enzyme. *Science* 267:2000-03.

Kuida, K., Haydar, T. F., Kuan, C. Y., Gu, Y., Taya, C., Karasuyama, H., Su, M. S., Rakic, P. and Flavell, R. A. (1998). Reduced apoptosis and cytochrome *c*-mediated caspase activation in mice lacking caspase 9. *Cell* 94:325-37.

Kumta, S. M., Cheng, J. C., Li, C. K., Griffith, J. F., Chow, L. T. and Quintos, A. D. (2002). Scope and limitations of limb-sparing surgery in childhood sarcomas. *J Pediatr Orthop* 22(2):244-48.

Kuroda, J., Kimura, S., Segawa, H., Kobayashi, Y., Yoshikawa, T., Urasaki, Y., Ueda, T., Enjo, F., Tokuda, H., Ottmann, O. G. and Maekawa, T. (2003). The third-generation bisphosphonate zoledronate synergistically augments the anti-Ph+ leukemia activity of imatinib mesylate. *Blood* 102:2229-35.

Kurzrock, R., Cortes, J. and Ryback, M. E. (2001). Phase II study of a farnesyl transferase inhibitor, R115777, in patients with myelodysplastic syndrome. *Blood* 98:848.

Kurzrock, R., Sebti, S. M., M., K. H. and al., e. (2001). Phase I study of a farnesyl transferase inhibitor, R115777, in patients with myelodysplastic syndrome. *Blood* 98:623a.

Kusama, T., Mukai, M., Iwasaki, T., Tatsuta, M., Matsumoto, Y., Akedo, H. and Nakamura, H. (2001). Inhibition of epidermal growth factor-induced RhoA translocation and invasion of human pancreatic cancer cells by 3-hydroxy-3-methylglutaryl-coenzyme a reductase inhibitors. *Cancer Res* 61:4885-91.

Lacey, D. L., Timms, E., Tan, H. L., Kelley, M. J., Dunstan, C. R., Burgess, T., Elliott, R., Colombero, A., Elliott, G., Scully, S., Hsu, H., Sullivan, J., Hawkins, N., Davy, E., Capparelli, C., Eli, A., Qian, Y. X., Kaufman, S., Sarosi, I., Shalhoub, V., Senaldi, G., Guo, J., Delaney, J. and Boyle, W. J. (1998). Osteoprotegerin ligand is a cytokine that regulates osteoclast differentiation and activation. *Cell* 93:165-76.

Lacour, S., Hammann, A., Wotawa, A., Corcos, L., Solary, E. and Dimanche-Boitrel, M. T. (2001). Anticancer agents sensitize tumor cells to tumor necrosis factor-related apoptosis-inducing ligand-mediated caspase-8 activation and apoptosis. *Cancer Res* 61:1645-51.

Lantry, L. E., Zhang, Z., Yao, R., Crist, K. A., Wang, Y., Ohkanda, J., Hamilton, A. D., Sebti, S. M., Lubet, R. A. and You, M. (2000). Effect of farnesyltransferase inhibitor FTI-276 on established lung adenomas from A/J mice induced by 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone. *Carcinogenesis* 21:113-6.

Larner, J., Jane, J., Laws, E., Packer, R., Myers, C. and Shaffrey, M. (1998). A phase I-II trial of lovastatin for anaplastic astrocytoma and glioblastoma multiforme. *Am J Clin Oncol* 21:579-83.

Lawrence, D., Shahrokh, Z., Marsters, S., Achilles, K., Shih, D., Mounho, B., Hillan, K., Totpal, K., DeForge, L., Schow, P., Hooley, J., Sherwood, S., Pai, R., Leung, S., Khan, L., Gliniak, B., Bussiere, J., Smith, C. A., Strom, S. S., Kelley, S., Fox, J. A., Thomas, D. and Ashkenazi, A. (2001). Differential hepatocyte toxicity of recombinant Apo2L/TRAIL versions. *Nat Med* 7:383-5.

LeBlanc, A. C. (2003). Natural cellular inhibitors of caspases. *Prog Neuropsychopharmacol Biol Psychiatry* 27:215-29.

LeBlanc, H., Lawrence, D., Varfolomeev, E., Totpal, K., Morlan, J., Schow, P., Fong, S., Schwall, R., Sinicropi, D. and Ashkenazi, A. (2002). Tumor-cell resistance to death receptor--induced apoptosis through mutational inactivation of the proapoptotic Bcl-2 homolog Bax. *Nat Med* 8:274-81.

Lee, J., Hampl, M., Albert, P. and Fine, H. A. (2002). Antitumor activity and prolonged expression from a TRAIL-expressing adenoviral vector. *Neoplasia* 4:312-23.

Lee, M. V., Fong, E. M., Singer, F. R. and Guenette, R. S. (2001). Bisphosphonate treatment inhibits the growth of prostate cancer cells. *Cancer Res* 61:2602-8.

Lee, Y. P., Schwarz, E. M., Davies, M., Jo, M., Gates, J., Zhang, X., Wu, J. and Lieberman, J. R. (2002). Use of zoledronate to treat osteoblastic versus osteolytic lesions in a severe-combined-immunodeficient mouse model. *Cancer Res* 62:5564-70.

Lennernas, H. and Fager, G. (1997). Pharmacodynamics and pharmacokinetics of the HMG-CoA reductase inhibitors. Similarities and differences. *Clin Pharmacokinet* 32:403-25.

Letai, A., Bassik, M. C., Walensky, L. D., Sorcinelli, M. D., Weiler, S. and Korsmeyer, S. J. (2002). Distinct BH3 domains either sensitize or activate mitochondrial apoptosis, serving as prototype cancer therapeutics. *Cancer Cell* 2:183-92.

Leverkus, M., Sprick, M. R., Wachter, T., Mengling, T., Baumann, B., Serfling, E., Brocker, E. B., Goebeler, M., Neumann, M. and Walczak, H. (2003). Proteasome inhibition results in TRAIL sensitization of primary keratinocytes by removing the resistance-mediating block of effector caspase maturation. *Mol Cell Biol* 23:777-90.

Levine, A. J. (1997). p53, the cellular gatekeeper for growth and division. *Cell* 88:323-31.

Li, H., Zhu, H., Xu, C. J. and Yuan, J. (1998). Cleavage of Bid by caspase-8 mediates the mitochondrial damage in the Fas pathway of apoptosis. *Cell* 94:491-501.

Li, L. Y., Luo, X. and Wang, X. (2001). Endonuclease G is an apoptotic DNase when released from mitochondria. *Nature* 412:95-9.

Li, P., Nijhawan, I., Budihardjo, S., Srinivasula, M., Ahmad, E., Alnemri, E. S. and Wang, X. (1997). Cytochrome *c* and dATP-dependent formation of Apaf-1/Caspase-9 complex initiates an apoptotic protease cascade. *Cell* 91:479-89.

Link, M. P., Goorin, A. M., Miser, A. W., Green, A. A., Pratt, C. B., Belasco, J. B., Pritchard, J., Malpas, J. S., Baker, A. R. and Kirkpatrick, J. A. (1986). The effect of adjuvant chemotherapy on relapse-free survival in patients with osteosarcoma of the extremity. *New England Journal of Medicine* 314:1600-06.

Lipton, A. (2000). Bisphosphonates and breast carcinoma. Cancer Suppl 88:3033-37.

Lipton, A., Small, E., Saad, F., Gleason, D., Gordon, D., Smith, M., Rosen, L., Kowalski, M. O., Reitsma, D. and Seaman, J. (2002). The new bisphosphonate, Zometa (zoledronic acid), decreases skeletal complications in both osteolytic and osteoblastic lesions: a comparison to pamidronate. *Cancer Invest* 20:45-54.

Liu, X. and Zou, H. (1997). DFF, a heterodimeric protein that functions downstream of caspase-3 to trigger DNA fragmentation during apoptosis. *Cell* 89:175-84.

Lobell, R. B., Omer, C. A., Abrams, M. T., Bhimnathwala, H. G., Brucker, M. J., Buser,
C. A., Davide, J. P., deSolms, S. J., Dinsmore, C. J., Ellis-Hutchings, M. S., Kral, A.
M., Liu, D., Lumma, W. C., Machotka, S. V., Rands, E., Williams, T. M., Graham, S.
L., Hartman, G. D., Oliff, A. I., Heimbrook, D. C. and Kohl, N. E. (2001). Evaluation of
farnesyl:protein transferase and geranylgeranyl:protein transferase inhibitor
combinations in preclinical models. *Cancer Res* 61:8758-68.

Lockshin, R. and Williams, C. (1965). Programmed cell death. II. Endocrine presentation of the breakdown of the intersegmental muscles of silkworms. *Insect Physiology* 11:803-09.

Locksley, R. M., Killeen, N. and Lenardo, M. J. (2001). The TNF and TNF receptor superfamilies: integrating mammalian biology. *Cell* 104:487-501.

Looney, R. J., Boyd, A., Totterman, S., Seo, G. S., Tamez-Pena, J., Campbell, D., Novotny, L., Olcott, C., Martell, J., Hayes, F. A., O'Keefe, R. J. and Schwarz, E. M. (2002). Volumetric computerized tomography as a measurement of periprosthetic acetabular osteolysis and its correlation with wear. *Arthritis Res* 4:59-63.

Lorenzo, H. K., Susin, S. A., Penninger, J. M. and Kroemer, G. (1999). Apoptosisinducing factor (AIF): a phylogenetically old, caspase-independent effector of cell death. *Cell Death and Differ* 6:516-24.

Luckman, S. P., Hughes, D. E., Coxon, F. P., Russell, R. G. G. and Rogers, M. J. (1998). Nitrogen-containing bisphosphonates inhibit the mevalonate pathway and
prevent posttranslational prenylation of GTP-binding proteins, including Ras. J Bone Miner Res 13:581-89.

Luo, X., Budihardjo, I., Zou, H., Slaughter, C. and Wang, X. (1998). Bid, a Bcl2 interacting protein, mediates cytochrome c release from mitochondria in response to activation of cell surface death receptors. *Cell* 94:481-90.

Lutcke, A., Jansson, S., Parton, R. G., Chavrier, P., Valencia, A., Huber, L. A., Lehtonen, E. and Zerial, M. (1993). Rab17, a novel small GTPase, is specific for epithelial cells and is induced during cell polarization. *J Cell Biol* 121:553-64.

Lutter, M., Fang, M., Luo, X., Nishijima, M., Xie, X. and Wang, X. (2000). Cardiolipin provides specificity for targeting of tBid to mitochondria. *Nat Cell Biol* 2:754-61.

MacKeigan, J. P., Collins, T. S. and Ting, J. P. (2000). MEK inhibition enhances paclitaxel-induced tumor apoptosis. *J Biol Chem* 275:38953-6.

Magnetto, S., Boissier, S., Delmas, P. D. and Clezardin, P. (1999). Additive antitumour activities of taxoids in combination with the bisphosphonate ibandronate against invasion and adhesion of human breast carcinoma cells to bone. *Int J Cancer* 83:263-69.

Mallett, S. and Barclay, A. N. (1991). A new superfamily of cell surface proteins related to the nerve growth factor receptor. *Immunol Today* 12:220-3.

Maltese, W. A., Defendini, R., Green, R. A., Sheridan, K. M. and Donley, D. K. (1985). Suppression of murine neuroblastoma growth *in vivo* by mevinolin, a competitive inhibitor of 3-hydroxy-3-methylglutaryl-coenzyme A reductase. *J Clin Invest* 76:1748-54.

Mancini, M., Nicholson, D. W., Roy, S., Thornberry, N. A., Peterson, E. P., Casciola-Rosen, L. A. and Rosen, A. (1998). The caspase-3 precursor has a cytosolic and mitochondrial distribution: implications for apoptotic signaling. *J Cell Biol* 140:1485-95.

Mandruzzato, S., Brasseur, F., Andry, G., Boon, T. and van der Bruggen, P. A. (1997). CASP-8 mutation recognized by cytolytic T lymphocytes on a human head and neck carcinoma. *J Exp Med* 186:785-93.

Mankin, H. J., Lange, T. A. and Spanier, S. S. (1982). The hazards of biopsy in patients with malignant primary bone and soft tissue tumors. *Journal of Bone and Joint Surgery* 64:1121.

Manolagas, S. C. and Jilka, R. L. (1995). Bone marrow, cytokines and bone remodelling. *N Engl J Med* 332:305-11.

Mariani, S. M. and Krammer, P. H. (1998). Differential regulation of TRAIL and CD95 ligand in transformed cells of the T and B lymphocyte lineage. *Eur J Immunol* 28:973-82.

Marsters, S. A., Sheridan, J. P., Pitti, P. M., Huang, A., Skubatch, M., Baldwin, D., Yuan, J., Gurney, A., Goddard, A. D., Godaowski, P. and Ashkenazi, A. (1997). A novel receptor for ApoL/TRAIL contains a truncated death domain. *Curr Biol* 7:1003-6.

Martin, T. J. and Moseley, J. M. (2000). Mechanisms in the skeletal complications of breast cancer. *Endocrin Relat Cancer* 7:271-84.

Masarachia, P., Weinreb, M., Balena, R. and Rodan, G. A. (1996). Comparison of the distribution of 3H-alendronate and 3H-etidronate in rat and mouse bones. *Bone* 19:281-290.

Matar, P., Rozados, V. R., Roggero, E. A. and Scharovsky, O. G. (1998). Lovastatin inhibits tumor growth and metastasis development of a rat fibrosarcoma. *Cancer Biother Radiopharm* 13:387-93.

Matar, P., Rozados, V. R., Binda, M. M., Roggero, E. A., Bonfil, R. D. and Scharovsky, O. G. (1999). Inhibitory effect of Lovastatin on spontaneous metastases derived from a rat lymphoma. *Clin Exp Metastasis* 17:19-25.

Mathai, J. P., Germain, M., Marcellus, R. C. and Shore, G. C. (2002). Induction and endoplasmic reticulum location of BIK/NBK in response to apoptotic signaling by E1A and p53. *Oncogene* 21:2534-44.

Mathiasen, M. S. and Jaatela, M. (2002). Triggering caspase-independent cell death to combat cancer. *Trends Mol Med* 8:212-20.

Matsumoto, S., Kimura, S., Segawa, H., Kuroda, J., Yuasa, T., Sato, K., Nogawa, M., Tanaka, F., Maekawa, T. and Wada, H. (2005). Efficacy of the third-generation bisphosphonate, zoledronic acid alone and combined with anti-cancer agents against small cell lung cancer cell lines. *Lung Cancer* 47:31-9.

McFall, A., Ulku, A., Lambert, Q. T., Kusa, A., Rogers-Graham, K. and Der, C. J. (2001). Oncogenic ras blocks anoikis activation of a novel effector pathway independent of phosphatidylinositol 3-kinase. *Mol Cell Biol* 21:5488-99.

Medema, J. P., Scaffidi, C., Kischkel, F. C., Shevchenko, A., Mann, M., Krammer, P. H. and Peter, M. E. (1997). FLICE is activated by association with the CD95 deathinducing signalling complex (DISC). *EMBO J* 16:2794-804.

Medina, V., Edmonds, B., Young, G. P., James, R., Appleton, S. and Zalewski, P. D. (1997). Induction of caspase-3 protease activity and apoptosis by butyrate and trichostatin A (inhibitors of histone deacetylase): dependence on protein synthesis and synergy with a mitochondrial/cytochrome c-dependent pathway. *Cancer Res* 57:3697-707.

Meier, C. R., Schlienger, R. G., Kraenzlin, M. E., Schlegel, B. and Jick, H. (2000). HMG-CoA reductase inhibitors and the risk of fractures. *JAMA* 283:3205-10.

Michigami, T., Hiraga, T., Williams, P. J., Niewolna, M., Nishimura, R., Mundy, G. R. and Yoneda, T. (2002). The effect of the bisphosphonate ibandronate on breast cancer metastasis to visceral organs. *Breast Cancer Res Treat* 75:249-58.

Miller, A. C., Kariko, K., Myers, C. E., Clark, E. P. and Samid, D. (1993). Increased radioresistance of EJras-transformed human osteosarcoma cells and its modulation by lovastatin, an inhibitor of p21ras isoprenylation. *Int J Cancer* 53:302-7.

Milner, A. E., Palmer, D. H., Hodgkin, E. A., Eliopoulos, A. G., Knox, P. G., Poole, C. J., Kerr, D. J. and Young, L. S. (2002). Induction of apoptosis by chemotherapeutic drugs: the role of FADD in activation of caspase-8 and synergy with death receptor ligands in ovarian carcinoma cells. *Cell Death Differ* 9:287-300.

Minden, M. D., Dimitroulakos, J., Nohynek, D. and Penn, L. Z. (2000). Lovastatin induced control of blast cell growth in an elderly patient with acute myeloblastic leukemia. *Leuk Lymphoma* 40:659-62.

Miramar, M. D., Costantini, P., Ravagnan, L., Saraiva, L. M., Haouzi, D., Brothers, G., Penninger, J. M., Peleato, M. L., Kroemer, G. and Susin, S. A. (2001). NADH oxidase activity of mitochondrial apoptosis-inducing factor. *J Biol Chem* 276:16391-8.

Mitsiades, C. S., Treon, S. P., Mitsiades, N., Shima, Y., Richardson, P., Schlossman, R., Hideshima, T. and Anderson, K. C. (2001). TRAIL/Apo2L ligand selectively induces apoptosis and overcomes drug resistance in multiple myeloma: therapeutic applications. *Blood* 98:795-804.

Mitsiades, N., Poulaki, V., Mitsiades, C. and Tsokos, M. (2001). Ewing's sarcoma family tumors are sensitive to tumor necrosis factor-related apoptosis-inducing ligand and express death receptor 4 and death receptor 5. *Cancer Res* 61:2704-12.

Mizutani, Y., Nakao, M., Ogawa, O., Yoshida, O., Bonavida, B. and Miki, T. (2001). Enhanced sensitivity of bladder cancer cells to tumor necrosis factor related apoptosis inducing ligand mediated apoptosis by cisplatin and carboplatin. *J Urol* 165:263-70.

Moasser, M. M., Sepp-Lorenzino, L., Kohl, N. E., Oliff, A., Balog, A., Su, D. S., Danishefsky, S. J. and Rosen, N. (1998). Farnesyl transferase inhibitors cause enhanced mitotic sensitivity to taxol and epothilones. *Proc Natl Acad Sci U S A* 95:1369-74.

Morgan, M. A., Wegner, J., Aydilek, E., Ganser, A. and Reuter, C. W. (2003). Synergistic cytotoxic effects in myeloid leukemia cells upon cotreatment with farnesyltransferase and geranylgeranyl transferase-I inhibitors. *Leukemia* 17:1508-20.

Muller, C., Bockhorn, A. G., Klusmeier, S., Kiehl, M., Roeder, C., Kalthoff, H. and Koch, O. M. (1998). Lovastatin inhibits proliferation of pancreatic cancer cell lines with mutant as well as with wild-type K-ras oncogene but has different effects on protein phosphorylation and induction of apoptosis. *Int J Oncol* 12:717-23.

Mundy, G., Garrett, R., Harris, S., Chan, J., Chen, D., Rossini, G., Boyce, B., Zhao, M. and Gutierrez, G. (1999). Stimulation of bone formation in vitro and in rodents by statins. *Science* 286:1946-9.

Murphey, M. D., wan Jaovisidha, S., Temple, H. T., Gannon, F. H., Jelinek, J. S. and Malawer, M. M. (2003). Telangiectatic osteosarcoma: radiologic-pathologic comparison. *Radiology* 229(2):545-53.

Nagane, M., Pan, G., Weddle, J. J., Dixit, V. M., Cavenee, W. K. and Huang, H. J. (2000). Increased death receptor 5 expression by chemotherapeutic agents in human gliomas causes synergistic cytotoxicity with tumor necrosis factor-related apoptosis-inducing ligand *in vitro* and *in vivo*. *Cancer Res* 60:847-53.

Nagane, M., Huang, H. J. and Cavenee, W. K. (2001). The potential of TRAIL for cancer chemotherapy. *Apoptosis* 6:191-7.

Nagasu, T., Yoshimatsu, K., Rowell, C., Lewis, M. D. and Garcia, A. M. (1995). Inhibition of human tumor xenograft growth by treatment with the farnesyl transferase inhibitor B956. *Cancer Res* 55:5310-4.

Nagata, S. (1997). Apoptosis by death factor. Cell 88:355-65.

Nakagawa, T. and Yuan, J. (2000). Cross-talk between two cysteine protease families. Activation of caspase-12 by calpain in apoptosis. *J Cell Biol* 150:887-94.

Nakano, K. and Vousden, K. H. (2001). PUMA, a novel proapoptotic gene, is induced by p53. *Mol Cell* 7:683-94.

Narisawa, T., Fukaura, Y., Tanida, N., Hasebe, M., Ito, M. and Aizawa, R. (1996). Chemopreventive efficacy of low dose of pravastatin, an HMG-CoA reductase inhibitor, on 1,2-dimethylhydrazine-induced colon carcinogenesis in ICR mice. *Tohoku J Exp Med* 180:131-8.

Narisawa, T., Morotomi, M., Fukaura, Y., Hasebe, M., Ito, M. and Aizawa, R. (1996). Chemoprevention by pravastatin, a 3-hydroxy-3-methylglutaryl-coenzyme A reductase inhibitor, of N-methyl-N-nitrosourea-induced colon carcinogenesis in F344 rats. *Jpn J Cancer Res* 87:798-804.

Nemoto, R., Kanoh, S., Koiso, K. and Harada, M. (1988). Establishment of a model to evaluate inhibition of bone resorption induced by human prostate cancer cells in nude mice. *J Urol* 140:875-9.

Nemoto, R., Sato, S., Nishijima, Y., Miyakawa, I., Koiso, K. and Harada, M. (1990). Effects of a new bisphosphonate (AHBuBP) on osteolysis induced by human prostate cancer cells in nude mice. *J Urol* 144:770-4.

Nemoto, R., Satou, S., Miyagawa, I. and Koiso, K. (1991). Inhibition by a new bisphosphonate (AHBuBP) of bone resorption induced by the MBT-2 tumor of mice. *Cancer* 67:643-8.

Nemoto, R., Satou, S., Mochizuki, T. and Okabe, K. (1992). Response of MBT-2 bladder carcinoma-induced osteolysis to various agents. *Cancer Cell* 69:2316-21.

Nemoto, R., Nishijima, Y., Uchida, K. and Koiso, K. (1993). Inhibition by a new bisphosphonate (YM175) of bone resorption induced by the MBT-2 tumour of mice. *Br J Cancer* 67:893-7.

Ness, G. C., Zhao, Z. and Lopez, D. (1996). Inhibitors of cholesterol biosynthesis increase hepatic low-density lipoprotein receptor protein degradation. *Arch Biochem Biophys* 325:242-8.

Newman, A., Clutterback, R. D., Powles, R. L. and Millar, J. L. (1994). Selective inhibition of primary acute myeloid leukaemia cell growth by simvastatin. *Leukemia* 8:2023-29.

Newman, A., Clutterbuck, R. D., Powles, R. L., Catovsky, D. and Millar, J. L. (1997). A comparison of the effect of the 3-hydroxy-3-methylglutaryl coenzyme A (HMG-CoA) reductase inhibitors simvastatin, lovastatin and pravastatin on leukaemic and normal bone marrow progenitors. *Leuk Lymphoma* 24:533-7.

Ng, C. P., Zisman, A. and Bonavida, B. (2002). Synergy is achieved by complementation with Apo2L/TRAIL and actinomycin D in Apo2L/TRAIL-mediated apoptosis of prostate cancer cells: role of XIAP in resistance. *Prostate* 53:286-99.

Nicholson, D. W. and Thornberry, N. A. (1997). Caspases: killer proteases. *Trends Biochem Sci* 22:299-306.

Nicholson, D. W. (1999). Caspase structure, proteolytic substrates and function during apoptotic cell death. *Cell Death and Differ* 6:1028-42.

Nobuyuki, H., Hiraga, T. and Williams, P. J. (2001). The bisphosphonate zoledronic acid inhibits metastases to bone and liver with suppression of osteopontin production in mouse mammary tumor. *J Bone Miner Res* 16:S191.

Norris, J. S., Hyer, M. L., Voelkel-Johnson, C., Lowe, S. L., Rubinchik, S. and Dong, J. Y. (2001). The use of Fas Ligand, TRAIL and Bax in gene therapy of prostate cancer. *Curr Gene Ther* 1:123-36.

Oades, G. M., Senaratne, S. G., Clarke, I. A., Kirby, R. S. and Colston, K. W. (2003). Nitrogen containing bisphosphonates induce apoptosis and inhibit the mevalonate

pathway, impairing Ras membrane localization in prostate cancer cells. *J Urol* 170:246-52.

Ochi, K., Katagiri, T., Nagayama, S., Tsunoda, T., Myoui, A., Naka, N., Araki, N., Kudawara, I., Ieguchi, M., Toyana, Y., Toguchida, J., Yoshikawa, H. and Nakamura, Y. (2004). Prediction of response to neoadjuvant chemotherapy for osteosarcoma by gene expression profiles. *International Journal of Oncology* 24(3):647-55.

Oda, E., Ohki, R., Murasawa, H., Nemoto, J., Shibue, T., Yamashita, T., Tokino, T., Taniguchi, T. and Tanaka, N. (2000). Noxa, a BH3-only member of the Bcl-2 family and candidate mediator of p53-induced apoptosis. *Science* 288:1053-8.

Orwoll, E., Ettinger, M., Weiss, S., Miller, P., Kendler, D., Graham, J., Adami, S., Weber, K., Lorenc, R., Pietschmann, P., Vandormael, K. and Lombardi, A. (2000). Alendronate for the treatment of osteoporosis in men. *N Engl J Med* 343:604-10.

Oxlund, H., Dalstra, M. and Andreassen, T. T. (2001). Statin given perorally to adult rats increases cancellous bone mass and compressive strength. *Calcif Tissue Int* 69:299-304.

Ozoren, N., Kim, K., Burns, T. F., Dicker, D. T., Moscioni, A. D. and El-Deiry, W. S. (2000). The caspase 9 inhibitor Z-LEHD-FMK protects human liver cells while permitting death of cancer cells exposed to tumor necrosis factor-related apoptosis-inducing ligand. *Cancer Res* 60:6259-65.

Ozoren, N. and El-Deiry, W. S. (2002). Defining characteristics of Types I and II apoptotic cells in response to TRAIL. *Neoplasia* 4:551-7.

Ozoren, N. and El-Deiry, W. S. (2003). Cell surface Death Receptor signaling in normal and cancer cells. *Semin Cancer Biol* 13:135-47.

Pan, B., Farrugia, A. N., To, L. B., Findlay, D. M., Green, J., Lynch, K. and Zannettino, A. C. (2004). The nitrogen-containing bisphosphonate, zoledronic acid, influences

RANKL expression in human osteoblast-like cells by activating TNF-alpha converting enzyme (TACE). *J Bone Miner Res* 19:147-54.

Pan, G., Ni, J., Wei, Y. F., Yu, G. I., Gentz, R. and Dixit, V. M. (1997). An antagonist decoy receptor and a death domain-containing receptor for TRAIL. *Science* 277:815-18.

Pan, G., O'Rourke, K., Chinnaiyan, A. M., Gentz, R., Ebner, R., Ni, J. and Dixit, V. M. (1997). The receptor for the cytotoxic receptor TRAIL. *Science* 276:111-13.

Pan, G., O'Rourke, K. and Dixit, V. M. (1998). Caspase-9, Bcl-XL, and Apaf-1 form a ternary complex. *J Biol Chem* 273:5841-5.

Pasco, J. A., Kotowicz, M. A., Henry, M. J., Sanders, K. M. and Nicholson, G. C. (2002). Statin use, bone mineral density, and fracture risk: Geelong Osteoporosis Study. *Arch Intern Med* 162:537-40.

Pedersen, T. R., Wilhelmsen, L., Faergeman, O., Strandberg, T. E., Thorgeirsson, G., Troedsson, L., Kristianson, J., Berg, K., Cook, T. J., Haghfelt, T., Kjekshus, J., Miettinen, T., Olsson, A. G., Pyorala, K. and Wedel, H. (2000). Follow-up study of patients randomized in the Scandinavian simvastatin survival study (4S) of cholesterol lowering. *Am J Cardiol* 86:257-62.

Peyruchaud, O., Winding, B., Pecheur, I., Serre, C. M., Delmas, P. and Clezardin, P. (2001). Early detection of bone metastases in a murine model using fluorescent human breast cancer cells: application to the use of the bisphosphonate zoledronic acid in the treatment of osteolytic lesions. *J Bone Miner Res* 16:2027-34.

Phillips, T. A., Ni, J., Pan, G., Ruben, S. M., Wei, Y. F. and Hunt, J. S. (1999). TRAIL (Apo-2L) and TRAIL receptors in human placentas: implications for immune privilege. *J Immunol* 162:6053-59.

Pitti, R. M., Marsters, S. A., Rupper, S., Donahue, C. J., Moore, A. and Ashkenazi, A. (1996). Induction of apoptosis by Apo-2 ligand, a new member of the tumour necrosis factor cytokine family. *J Biol Chem* 271:12687-90.

Pollack, I. F., Erff, M. and Ashkenazi, A. (2001). Direct stimulation of apoptotic signaling by soluble Apo2l/tumor necrosis factor-related apoptosis-inducing ligand leads to selective killing of glioma cells. *Clin Cancer Res* 7:1362-9.

Pollard, M. and Luckert, P. H. (1985). Effects of dichloromethylene diphosphonate on the osteolytic and osteoplastic effects of rat prostate adenocarcinoma cells. *J Natl Cancer Inst* 75:949-54.

Pollard, M. and Luckert, P. H. (1986). The beneficial effects of diphosphonate and piroxicam on the osteolytic and metastatic spread of rat prostate carcinoma cells. *Prostate* 8:81-6.

Pringle, J. A. S. (1999). Bone-forming neoplasms arising within bone. <u>Pathology of bone and joint neoplasms</u>. Helliwell, T. R. Philadelphia, Saunders.

Provisor, A. J., Ettinger, L. J., Nachman, J. B., Krailo, M. D., Makley, J. T., Yunis, E. J., Huvos, A. G., Betcher, D. L., Baum, E. S., Kisket, C. T. and Miser, J. S. (1997). Treatment of nonmetastatic osteosarcoma of the extremity with preoperative and postoperative chemotherapy: a report from the Children's Cancer Group. *Journal of Clinical Oncology* 15:76-84.

Puri, L., Wixson, R. L., Stern, S. H., Kohli, J., Hendrix, R. W. and Stulberg, S. D. (2002). Use of helical computed tomography for the assessment of acetabular osteolysis after total hip arthroplasty. *J Bone Joint Surg Am* 84-A:609-14.

Puthalakath, H., Villunger, A., O'Reilly, L. A., Beaumont, J. G., Coultas, L., Cheney, R. E., Huang, D. C. and Strasser, A. (2001). Bmf: a proapoptotic BH3-only protein regulated by interaction with the myosin V actin motor complex, activated by anoikis. *Science* 293:1829-32.

Puthalakath, H. and Strasser, A. (2002). Keeping killers on a tight leash: transcriptional and post-translational control of the pro-apoptotic activity of BH3-only proteins. *Cell Death and Differ* 9:505-12.

212

10.2

Qin, J. Z., Chaturvedi, V., Bonish, B. and Nickoloff, B. J. (2001). Avoiding premature apoptosis of normal epidermal cells. *Nat Med* 7:385-386.

Rao, L., Perez, D. and White, E. (1996). Lamin proteolysis facilitates nuclear event during apoptosis. *J Cell Biol* 135:1441-45.

Raval, P., Hsu, H. H. T., Schneider, D. J., Sarras, M. P. J., Masuhara, K., Bonewald, L.
F. and Anderson, H. C. (1996). Expression of bone morphogenetic proteins by osteoinductive and non-osteoinductive human osteosarcoma cells. *J Dent Res* 75:1518-23.

Reed, J. C. and Bischoff, J. R. (2000). BIRinging chromosomes through cell divisionand survivin' the experience. *Cell* 102:545-48.

Refaat, Y., Gunnoe, J., Hornicek, F. J. and Mankin, H. J. (2002). Comparison of quality of life after amputation or limb salvage. *Clinical Orthopaedics* 397:298-305.

Reid, I. R., Brown, J. P., Burckhardt, P., Horowitz, Z., Richardson, P., Trechsel, U., Widmer, A., Devogelaer, J. P., Kaufman, J. M., Jaeger, P., Body, J. J., Brandi, M. L., Broell, J., Di Micco, R., Genazzani, A. R., Felsenberg, D., Happ, J., Hooper, M. J., Ittner, J., Leb, G., Mallmin, H., Murray, T., Ortolani, S., Rubinacci, A., Saaf, M., Samsioe, G., Verbruggen, L. and Meunier, P. J. (2002). Intravenous zoledronic acid in postmenopausal women with low bone mineral density. *N Engl J Med* 346:653-61.

Reinholz, G. G., Getz, B., Pederson, L., Sanders, E. S., Subramaniam, M., Ingle, J. N. and Spelsberg, T. C. (2000). Bisphosphonates directly regulate cell proliferation, differentiation, and gene expression in human osteoblasts. *Cancer Res* 60:6001-7.

Reszka, A. A., Halasy-Nagy, J. M., Masarachia, P. J. and Rodan, G. A. (1999). Bisphosphonates act directly on the osteoclast to induce caspase cleavage of mst1 kinase during apoptosis. A link between inhibition of the mevalonate pathway and regulation of an apoptosis-promoting kinase. *J Biol Chem* 274:34967-73.

ces the
s. *Cell*). The
ffling.
c). The
b. *Br J*c). W.,
c) of

Ridley, A. J. and Hall, A. (1992). The small GTP-binding protein Rho regulates the assembly of focal adhesions and actin stress fibers in response to growth factors. *Cell* 70:389-99.

Ridley, A. J., Paterson, H. F., Johston, C. L., Diekmann, D. and Hall, A. (1992). The small GTP-binding protein Rac regulates growth factor-induced membrane ruffling. *Cell* 70:401-10.

Riebeling, C., Forsea, A. M., Raisova, M., Orfanos, C. E. and Geilen, C. C. (2002). The bisphosphonate pamidronate induces apoptosis in human melanoma cells *in vitro*. *Br J Cancer* 87:366-71.

Riedl, S. J., Renatus, M., Schwarzenbacher, R., Zhou, Q., Sun, C., Fesik, S. W., Liddington, R. C. and Salvesen, G. S. (2001). Structural basis for the inhibition of caspase-3 by XIAP. *Cell* 104:791-800.

Rieger, J., Naumann, U., Glaser, T., Ashkenazi, A. and Weller, M. (1998). APO2 ligand: a novel lethal weapon against malignant glioma? *FEBS Lett* 427:124-8.

Roberg, K., Johansson, U. and Ollinger, K. (1999). Lysosomal release of cathepsin D precedes relocation of cytochrome c and loss of mitochondrial transmembrane potential during apoptosis induced by oxidative stress. *Free Radic Biol Med* 27:1228-37.

Rodan, G. A. (1996). Coupling of bone resorption and formation during bone remodeling. <u>Osteoporosis</u>. Marcus, R., Feldman, D. and Kelsey, J. San Diego, Academic Press: 290-99.

Rodan, G. A. (1997). Bone mass homeostasis and bisphosphonate action. Bone 20:1-4.

Rodan, G. A. (1998). Mechanisms of action of bisphosphonates. *Annu Rev Pharmacol Toxicol* 38:375-88.

Rogers, M. J., Gordon, S., Benford, H. L., Coxon, F. P., Luckman, S. P., Monkkonen, J. and Frith, J. C. (2000). Cellular and molecular mechanisms of action of bisphosphonates. *Cancer Cell* 88:2961-78.

Rohn, T. A., Wagenknecht, B., Roth, W., Naumann, U., Gulbins, E., Krammer, P. H., Walczak, H. and Weller, M. (2001). CCNU-dependent potentiation of TRAIL/Apo2Linduced apoptosis in human glioma cells is p53-independent but may involve enhanced cytochrome *c* release. *Oncogene* 20:4128-37.

Rosen, L. S., Gordon, D., Kaminski, M., Howell, A., Belch, A., Mackey, J., Apffelstaedt, J., Hussein, M. A., Coleman, R. E., Reitsma, D. J., Chen, B. L. and Seaman, J. J. (2003). Long-term efficacy and safety of zoledronic acid compared with pamidronate disodium in the treatment of skeletal complications in patients with advanced multiple myeloma or breast carcinoma: a randomized, double-blind, multicenter, comparative trial. *Cancer Cell* 98:1735-44.

Rosen, L. S., Gordon, D., Tchekmedyian, S., Yanagihara, R., Hirsh, V., Krzakowski, M., Pawlicki, M., de Souza, P., Zheng, M., Urbanowitz, G., Reitsma, D. and Seaman, J. J. (2003). Zoledronic acid versus placebo in the treatment of skeletal metastases in patients with lung cancer and other solid tumors: a phase III, double-blind, randomized trial--the Zoledronic Acid Lung Cancer and Other Solid Tumors Study Group. *J Clin Oncol* 21:3150-7.

Rosen, L. S., Gordon, D., Tchekmedyian, N. S., Yanagihara, R., Hirsh, V., Krzakowski, M., Pawlicki, M., De Souza, P., Zheng, M., Urbanowitz, G., Reitsma, D. and Seaman, J. (2004). Long-term efficacy and safety of zoledronic acid in the treatment of skeletal metastases in patients with nonsmall cell lung carcinoma and other solid tumors: a randomized, Phase III, double-blind, placebo-controlled trial. *Cancer* 100:2613-21.

Rosen, L. S., Gordon, D. H., Dugan, W. J., Major, P., Eisenberg, P. D., Provencher, L., Kaminski, M., Simeone, J., Seaman, J., Chen, B. L. and Coleman, R. E. (2004). Zoledronic acid is superior to pamidronate for the treatment of bone metastases in breast carcinoma patients with at least one osteolytic lesion. *Cancer* 100:36-43.

Rosenson, R. S. and Tangney, C. C. (1998). Antiatherothrombotic properties of statins: implications for cardiovascular event reduction. *JAMA* 279:1643-50.

Rosenson, R. S., Tangney, C. C. and Casey, L. C. (1999). Inhibition of proinflammatory cytokine production by pravastatin. *Lancet Oncol* 353:983-4.

Roth, W., Isenmann, S., Naumann, U., Kugler, S., Bahr, M., Dichgans, J., Ashkenazi, A. and Weller, M. (1999). Locoregional Apo2L/TRAIL eradicates intracranial human malignant glioma xenografts in athymic mice in the absence of neurotoxicity. *Biochem Biophys Res Commun* 265:479-83.

Rougraff, B. T., Simon, M. A., Kneisl, J. S., Greenberg, D. B. and Mankin, H. J. (1994). Limb salvage compared with amputation for osteosarcoma of the distal end of the femur. A long term oncological, functional and quality-of-life study. *Journal of Bone and Joint Surgery America* 76(5):649-56.

Rowell, C. A., Kowalczyk, J. J., Lewis, M. D. and Garcia, A. M. (1997). Direct demonstration of geranylgeranylation and farnesylation of Ki-Ras *in vivo*. *J Biol Chem* 272:14093-7.

Roy, S. and Nicholson, D. W. (2000). Cross-talk in cell death signaling. J Exp Med 192:21-6.

Rubins, J. B., Greatens, T., Kratzke, R. A., Tan, A. T., Polunovsky, V. A. and Bitterman, P. (1998). Lovastatin induces apoptosis in malignant mesothelioma cells. *Am J Respir Crit Care Med* 157:1616-22.

Ruoslahti, E. and Reed, J. C. (1994). Anchorage dependence, integrins and apoptosis. *Cell* 77:477-78.

Rytomaa, M., Lehmann, K. and Downward, J. (2000). Matrix detachment induces caspase dependent cytochrome *c* release from mitochondria: Inhibition by PKB/Akt but not Raf signalling. *Oncogene* 19:4461-68.

Saad, F., Gleason, D. M., Murray, R., Tchekmedyian, S., Venner, P., Lacombe, L., Chin, J. L., Vinholes, J. J., Goas, J. A. and Chen, B. (2002). A randomized, placebocontrolled trial of zoledronic acid in patients with hormone-refractory metastatic prostate carcinoma. *J Natl Cancer Inst* 94:1458-68.

Sadowski-Debbing, K., Coy, J. F., Mier, W., Hug, H. and Los, M. (2002). Caspasestheir role in apoptosis and other physiological processes as revealed by knock-out studies. *Arch Immunol Ther Exp (Warsz)* 50:19-34.

Saelens, X., Festjens, N., Walle, L. V., van Gurp, M., van Loo, G. and Vandenabeele, P. (2004). Toxic proteins released from mitochondria in cell death. *Oncogene* 23:2861-74.

Sahni, M., Guenther, H. L., Fleisch, H., Collin, P. and Martin, T. J. (1993). Bisphosphonates act on rat bone resorption through the mediation of osteoblasts. *J Clin Invest.* 91:2004-11.

Sakahira, H., Enari, M. and Nagata, S. (1998). Cleavage of CAD inhibitor in CAD activation and DNA degradation during apoptosis. *Nature* 391(96-99).

Salomo, M., Jurlander, J., Nielsen, L. B. and Gimsing, P. (2003). How myeloma cells escape bisphosphonate-mediated killing: development of specific resistance with preserved sensitivity to conventional chemotherapeutics. *Br J Haematol* 122:202-10.

Salvesen, G. S. and Duckett, C. S. (2002). IAP proteins: blocking the road to deaths door. *Nat Rev Mol Cell Biol* 3:401-10.

Sasaki, A., Boyce, B. F., Story, B., Wright, K. R., Chapman, M., Boyce, R., Mundy, G. R. and Yoneda, T. (1995). Bisphosphonate risedronate reduces metastatic human breast cancer burden in bone in nude mice. *Cancer Res* 55:3551-7.

Sasaki, A., Kitamura, K., Alcalde, R. E., Tanaka, T., Suzuki, A., Etoh, Y. and Matsumura, T. (1998). Effect of a newly developed bisphosphonate, YH529, on osteolytic bone metastases in nude mice. *Int J Cancer* 77:279-85.

Sato, M., Grasser, W., Endo, N., Akins, R., Simmons, H., Thompson, D. D., Golub, E. and Rodan, G. A. (1991). Bisphosphonate action. Alendronate localization in rat bone and effects on osteoclast ultrastructure. *J Clin Invest* 88:2095-105.

Sawada, K., Morishige, K., Tahara, M., Kawagishi, R., Ikebuchi, Y., Tasaka, K. and Murata, Y. (2002). Alendronate inhibits lysophosphatidic acid-induced migration of human ovarian cancer cells by attenuating the activation of rho. *Cancer Res* 62:6015-20.

Sayers, T. J., Brooks, A. D., Koh, C. Y., Ma, W., Seki, N., Raziuddin, A., Blazar, B. R., Zhang, X., Elliott, P. J. and Murphy, W. J. (2003). The proteasome inhibitor PS-341 sensitizes neoplastic cells to TRAIL-mediated apoptosis by reducing levels of c-FLIP. *Blood* 102:303-10.

Scaffidi, C., Schmitz, I., Zha, J., Korsmeyer, J. K., Krammer, P. H. and Peter, M. E. (1999). Differential modulation of apoptosis sensitivity in CD95 type I and type II cells. *J Biol Chem* 274:22532-38.

Schnitzer, T., Bone, H. G., Crepaldi, G., Adami, S., McClung, M., Kiel, D., Felsenberg, D., Recker, R. R., Tonino, R. P., Roux, C., Pinchera, A., Foldes, A. J., Greenspan, S. L., Levine, M. A., Emkey, R., Santora, A. C. n., Kaur, A., Thompson, D. E., Yates, J. and Orloff, J. J. (2000). Therapeutic equivalence of alendronate 70 mg once-weekly and alendronate 10 mg daily in the treatment of osteoporosis. Alendronate Once-Weekly Study Group. *Aging (Milano)* 12:1-12.

Schulze-Osthoff, K., Ferrari, D., Los, M., Wesselborg, S. and Peter, M. E. (1998). Apoptosis signaling by death receptors. *Eur J Biochem* 254:439-59.

Schwarz, E. M., Campbell, D., Totterman, S., Boyd, A., O'Keefe, R. J. and Looney, R. J. (2003). Use of volumetric computerized tomography as a primary outcome measure to evaluate drug efficacy in the prevention of peri-prosthetic osteolysis: a 1-year clinical pilot of etanercept vs. placebo. *J Orthop Res* 21:1049-55.

Screaton, G. and Xu, X. N. (2000). T cell life and death signalling via TNF-receptor family members. *Curr Opin Immunol* 12:316-22.

Searle, J., Lawson, T. A., Abbott, P. J., Harmon, B. and Kerr, J. F. K. (1975). An electron-microscope study of the mode of cell death induced by cancer therapeutic agents in populations of proliferating normal and neoplastic cells. *J Path* 116:129-38.

Sebti, S. M. and Hamilton, A. D. (1997). Inhibition of Ras prenylation: a novel approach to cancer chemotherapy. *Pharmacol Ther* 74:103-14.

Sebti, S. M. and Hamilton, A. D. (2000). Farnesyltransferase and geranylgeranyltransferase I inhibitors and cancer therapy: lessons from mechanism and bench-to-bedside translational studies. *Oncogene* 19:6584-93.

Sebti, S. M. and Hamilton, A. D. (2000). Farnesyltransferase and geranylgeranyltransferase I inhibitors in cancer therapy: important mechanistic and bench to bedside issues. *Expert Opin Investig Drugs* 9:2767-82.

Sedger, L. M., Shows, D. M., Blanto, R. A., Peschon, J. J., Goodwin, R. G., Cosman, D. and Wiley, S. R. (1999). IFN-gamma mediates a novel antiviral activity through dynamic modulation of TRAIL and TRAIL receptor expression. *J Immunol* 163:920-26.

Senaratne, S. G., Pirianov, G., Mansi, J. L., Arnett, T. R. and Colston, K. W. (2000). Bisphosphonates induce apoptosis in human breast cancer cell lines. *Br J Cancer* 82:1459-68.

Senaratne, S. G. and Colston, K. W. (2002). Direct effects of bisphosphonates on breast cancer cells. *Breast Cancer Res* 4:18-23.

Seol, D. W. and Billiar, T. R. (1999). A caspase-9 variant missing the catalytic site is an endogenous inhibitor of apoptosis. *J Biol Chem* 274:2072-76.

Sepp-Lorenzino, L., Ma, Z., Rands, E., Kohl, N. E., Gibbs, J. B., Oliff, A. and Rosen, N. (1995). A peptidomimetic inhibitor of farnesyl:protein transferase blocks the anchorage-dependent and -independent growth of human tumor cell lines. *Cancer Res* 55:5302-9.

Sezer, O., Jakob, C., Zavrski, I., Heider, U., Fleissner, C. and Freund, M. (2003). Bisphosphonates in early multiple myeloma. *Eur J Haematol* 71:231-2.

Sharma, S., Kemeny, N., Kelsen, D. P., Ilson, D., O'Reilly, E., Zaknoen, S., Baum, C., Statkevich, P., Hollywood, E., Zhu, Y. and Saltz, L. B. (2002). A phase II trial of farnesyl protein transferase inhibitor SCH 66336, given by twice-daily oral administration, in patients with metastatic colorectal cancer refractory to 5-fluorouracil and irinotecan. *Ann Oncol* 13:1067-71.

Sheikh, M. S., Burns, T. F., Huang, Y., Wu, G. S., Amundson, S., Brooks, K. S., Fornace, A. J. J. and El-Deiry, W. S. (1998). p53-dependent and -independent regulation of the death receptor KILLER/DR5 gene expression in response to genotoxic stress and tumor necrosis factor alpha. *Cancer Res* 58:1593-8.

Sheridan, J. P., Marsters, S. A., Pitti, R. M., Gurney, A., Skubatch, M., Baldwin, D., Ramakrishnan, L., Gray, C. L., Baker, K., Wood, W. I., Goddard, A. D., Godowski, P. and Ashkenazi, A. (1997). Control of TRAIL-induced apoptosis by a family of signaling and decoy receptors. *Science* 277:818-21.

Shevrin, D. H., Gorny, K. I., Rosol, T. J. and Kukreja, S. C. (1991). Effect of etidronate disodium on the development of bone lesions in an animal model of bone metastasis using the human prostate cancer cell line PC-3. *Prostate* 19:149-54.

Shi, B., Yaremko, B., Hajian, G., Terracina, G., Bishop, W. R., Liu, M. and Nielsen, L. L. (2000). The farnesyl protein transferase inhibitor SCH66336 synergizes with taxanes in vitro and enhances their antitumor activity *in vivo*. *Cancer Chemother Pharmacol* 46:387-93.

Shinoda, H., Adamek, G., Felix, R., Fleisch, H., Schenk, R. and Hagan, P. (1983). Structure-activity relationships of various bisphosphonates. *Calcif Tissue Int* 35:87-99. Shipman, C. M., Rogers, M. J., Apperley, J. F., Russell, R. G. and Croucher, P. I. (1997). Bisphosphonates induce apoptosis in human myeloma cell lines: a novel anti-tumour activity. *Br J Haematol* 98:665-72.

Shipman, C. M., Croucher, P. I., Russell, R. G., Helfrich, M. H. and Rogers, M. J. (1998). The bisphosphonate incadronate (YM175) causes apoptosis of human myeloma cells in vitro by inhibiting the mevalonate pathway. *Cancer Res* 58:5294-7.

Shipman, C. M., Rogers, M. J., Apperley, J. F., Graham, R. and Croucher, P. I. (1998). Anti-tumour activity of bisphosphonates in human myeloma cells. *Leuk and Lymph* 32:129-38.

Shipman, C. M., Rogers, M. J., Vanderkerken, K., Van Camp, B., Graham, R., Russell, G. and Croucher, P. I. (2000). Bisphosphonates--mechanisms of action in multiple myeloma. *Acta Oncol* 39:829-35.

Simonet, W. S., Lacey, D. L., Dunstan, C. R., Kelley, M., Chang, M. S., Luthy, R., Nguyen, H. Q., Wooden, S., Bennett, L., Boone, T., Shimamoto, G., DeRose, M., Elliott, R., Colombero, A. and al., e. (1997). Osteoprotegerin: a novel secreted protein involved in the regulation of bone density. *Cell* 89:309-19.

Slee, E. A., Adrain, C. and Martin, S. J. (1999). Serial killers: ordering caspase activation events in apoptosis. *Cell Death Differ* 6:1067-74.

Smith, C. A., Farrah, T. and Goodwin, R. G. (1994). The TNF receptor superfamily of cellular and viral proteins: activation, costimulation, and death. *Cell* 76:959-62.

Sohara, Y., Shimada, H., Scadeng, M., Pollack, H., Yamada, S., Ye, W., Reynolds, C. P. and A., D. Y. (2003). Lytic bone lesions in human neuroblastoma xenograft involve osteoclast recruitment and are inhibited by bisphosphonate. *Cancer Res* 63:3026-31.

Solit, D. B., Basso, A. D., Olshen, A. B., Scher, H. I. and Rosen, N. (2003). Inhibition of heat shock protein 90 function down-regulates Akt kinase and sensitizes tumors to Taxol. *Cancer Res* 63:2139-44.

Soma, M. R., Pagliarini, P., Butti, G., Paoletti, R., Paoletti, P. and Fumagalli, R. (1992). Simvastatin, an inhibitor of cholesterol biosynthesis, shows a synergistic effect with N,N'-bis(2-chloroethyl)-N-nitrosourea and beta-interferon on human glioma cells. *Cancer Res* 52:4348-55.

Song, K., Chen, Y., Goke, R., Wilmen, A., Seidel, C., Goke, A., Hilliard, B. and Chen, Y. (2000). Tumour necrosis factor-related apoptosis inducing ligand (TRAIL) is an inhibitor of autoimmune inflammation and cell cycle progression. *J Exp Med* 191:1095-104.

Song, Q., Kuang, Y., Dixit, V. M. and Vincenz, C. (1999). Boo, a novel negative regulator of cell death, interacts with Apaf-1. *EMBO J* 18:167-78.

Sonnemann, J., Eckervogt, V., Truckenbrod, B., Boos, J., Winkelmann, W. and van Valen, F. (2003). The bisphosphonate pamidronate is a potent inhibitor of Ewing's sarcoma cell growth *in vitro*. *Anticancer Drugs* 14:767-71.

Sprick, M. R., Rieser, E., Stahl, H., Grosse-Wilde, A., Weigand, M. A. and Walczak, H. (2002). Caspase-10 is recruited to and activated at the native TRAIL and CD95 deathinducing signalling complexes in a FADD-dependent manner but can not functionally substitute caspase-8. *EMBO J* 21:4520-30.

Srinivasula, S., Ahmad, M., Fernandes-Alnemri, T. and Alnemri, E. (1998). Autoactivation of procaspse-9 by apaf-1-mediated oligomerization. *Mol Cell* 1:949-57.

Srinivasula, S. M., Hegde, R., Saleh, A., Datta, P., Shiozaki, E., Chai, J., Lee, R. A., Robbins, P. D., Fernandes-Alnemri, T., Shi, Y. and Alnemri, E. S. (2001). A conserved XIAP-interaction motif in caspase-9 and Smac/DIABLO regulates caspase activity and apoptosis. *Nature* 410:112-6.

Srinivasula, S. M., Datta, P., Kobayashi, M., Wu, J. W., Fujioka, M., Hegde, R., Zhang, Z., Mukattash, R., Fernandes-Alnemri, T., Shi, Y., Jaynes, J. B. and Alnemri, E. S.

(2002). sickle, a novel Drosophila death gene in the reaper/hid/grim region, encodes an IAP-inhibitory protein. *Curr Biol* 12:125-30.

Srinivasula, S. M., Gupta, S., Datta, P., Zhang, Z., Hegde, R., Cheong, N., Fernandes-Alnemri, T. and Alnemri, E. S. (2003). Inhibitor of apoptosis proteins are substrates for the mitochondrial serine protease Omi/HtrA2. *J Biol Chem* 278:31469-72.

Stamenkov, R., Howie, D., Taylor, J., Findlay, D., McGee, M., Kourlis, G., Carbone, A. and Burwell, M. (2003). Measurement of bone defects adjacent to acetabular components of hip replacement. *Clin Orthop* 412:117-24.

Stearns, M. E. and Wang, M. (1996). Effects of alendronate and taxol on PC-3 ML cell bone metastases in SCID mice. *Invasion Metastasis* 16:116-31.

Strasser, A., O'Connor, L. and Dixit, D. M. (2000). Apoptosis signalling. Annu Rev Biochem 69:217-45.

Sugiyama, M., Kodama, T., Konishi, K., Abe, K., Asami, S. and Oikawa, S. (2000). Compactin and simvastatin, but not pravastatin, induce bone morphogenetic protein-2 in human osteosarcoma cells. *Biochem Biophys Res Commun* 271:688-92.

Sulston, J. E. and Horvitz, H. R. (1977). Post-embryonic cell lineages of the nematode, *Caenorhabditis elegans. Developmental Biology* 56:110-56.

Sun, J., Qian, Y., Hamilton, A. D. and Sebti, S. M. (1998). Both farnesyltransferase and geranylgeranyltransferase I inhibitors are required for inhibition of oncogenic K-Ras prenylation but each alone is sufficient to suppress human tumor growth in nude mouse xenografts. *Oncogene* 16:1467-73.

Sun, J., Blaskovich, M. A., Knowles, D., Qian, Y., Ohkanda, J., Bailey, R. D., Hamilton, A. D. and Sebti, S. M. (1999). Antitumor efficacy of a novel class of non-thiol-containing peptidomimetic inhibitors of farnesyltransferase and geranylgeranyltransferase I: combination therapy with the cytotoxic agents cisplatin, Taxol, and gemcitabine. *Cancer Res* 59:4919-26.

Sun, J., Qian, Y., Chen, Z., Marfurt, J., Hamilton, A. D. and Sebti, S. M. (1999). The geranylgeranyltransferase I inhibitor GGTI-298 induces hypophosphorylation of retinoblastoma and partner switching of cyclin-dependent kinase inhibitors. A potential mechanism for GGTI-298 antitumor activity. *J Biol Chem* 274:6930-4.

Sun, S. Y., Yue, P., Hong, W. K. and Lotan, R. (2000). Augmentation of tumor necrosis factor-related apoptosis-inducing ligand (TRAIL)-induced apoptosis by the synthetic retinoid 6-[3-(1-adamantyl)-4-hydroxyphenyl]-2-naphthalene carboxylic acid (CD437) through up-regulation of TRAIL receptors in human lung cancer cells. *Cancer Res* 60:7149-55.

Susin, S. A., Lorenzo, H. K., Zamzami, N., Marzo, I., Brenner, C., Larochette, N., Prevost, M. C., Alzari, P. M. and Kroemer, G. (1999). Mitochondrial release of caspase-2 and -9 during the apoptotic process. *J Exp Med* 189:381-94.

Susin, S. A., Lorenzo, H. K., Zamzami, N., Marzo, I., Snow, B. E., Brothers, G. M., Mangion, J., Jacotot, E., Costantini, P., Loeffler, M., Larochette, N., Goodlett, D. R., Aebersold, R., Siderovski, D. P., Penninger, J. M. and Kroemer, G. (1999). Molecular characterization of mitochondrial apoptosis-inducing factor. *Nature* 397:441-6.

Suzuki, Y., Imai, Y., Nakayama, H., Takahashi, K., Takio, K. and Takahashi, R. (2001). A serine protease, HtrA2, is released from the mitochondria and interacts with XIAP, inducing cell death. *Mol Cell* 8:613-21.

Takai, Y., Sasaki, A. and Matozaki, T. (2001). Small GTP-binding proteins. *Physiol Rev* 81:153-208.

Talanian, R. V., Yang, X., Turbov, J., Seth, P., Ghayur, T., Casiano, C. A., Orth, K. and Froelich, C. J. (1997). Granule-mediated killing: pathways for granzyme B-initiated apoptosis. *J Exp Med* 186:1323-31.

Tassone, P., Forciniti, S., Galea, E., Morrone, G., Turco, M. C., Martinell, V., Tagliaferri, P. and Venuta, S. (2000). Growth inhibition and synergistic induction of

apoptosis by zoledronate and dexamethasone in human myeloma cell lines. *Leukemia* 14:841-44.

Tassone, P., Tagliaferri, P., Viscomi, C., Palmieri, C., Caraglia, M., D'Alessandro, A., Galea, E., Goel, A., Abbruzzese, A., Boland, C. R. and Venuta, S. (2003). Zoledronic acid induces antiproliferative and apoptotic effects in human pancreatic cancer cells *in vitro*. *Br J Cancer* 88:1971-8.

Taube, T., Elomaa, I., Blomqvist, C., Beneton, M. N. and Kanis, J. A. (1994). Histomorphometric evidence for osteoclast-mediated bone resorption in metastatic breast cancer. *Bone* 15:161-6.

Teitz, T., Wei, T., Valentine, M. B., Vanin, E. F., Grenet, J., Valentine, V. A., Behm, F. G., Look, A. T., Lahti, J. M. and Kidd, V. J. (2000). Caspase 8 is deleted or silenced preferentially in childhood neuroblastomas with amplification of MYCN. *Nat Med* 6:529-35.

Teronen, O., Heikkila, P., Konttinen, Y. T., Laitinen, M., Salo, T., Hanemaaijer, R., Teronen, A., Maisi, P. and Sorsa, T. (1999). MMP inhibition and downregulation by bisphosphonates. *Ann N Y Acad Sci* 878:453-65.

Thibault, A., Samid, D., Tompkins, A. C., Figg, W. D., Cooper, M. R., Hohl, R. J., Trepel, J., Liang, B., Patronas, N., Venzon, D. J., Reed, E. and Myers, C. E. (1996). Phase I study of lovastatin, an inhibitor of the mevalonate pathway, in patients with cancer. *Clin Cancer Res* 2:483-91.

Thomas, W. D. and Hersey, P. (1998). TNF related apoptosis-inducing ligand (TRAIL) induces apoptosis in Fas ligand-resistant melanoma cells and mediates CD4 T cell killing of target cells. *J Immunol* 161:3513-18.

Thornberry, N. A., Rano, T. A., Peterson, E. P., Rasper, D. M., Timkey, T., Garcia-Calvo, M., Houtzager, V. M., Nordstrom, P. A., Roy, S., Vaillancourt, J. P., Chapman, K. T. and Nicholson, D. W. (1997). A combinatorial approach defines specificities of members of the caspase family and granzyme B. Functional relationships established for key mediators of apoptosis. *J Biol Chem* 272:17907-911.

Thornberry, N. A. and Lazebnik, Y. (1998). Caspases; enemies within. *Science* 281:1312-16.

Tokuda, H., Kozawa, O., Harada, A. and Uematsu, T. (1998). Tiludronate inhibits interleukin-6 synthesis in osteoblasts: inhibition of phospholipase D activation in MC3T3-E1 cells. *J Cell Biochem* 69:252-9.

Trapani, J. A., Davis, J., Sutton, V. R. and Smyth, M. J. (2000). Proapoptotic functions of cytotoxic lymphocyte granule constituents in vitro and in vivo. *Curr Opin Immunol* 12:323-9.

Trump, B. F., Goldblatt, P. J. and Stowell, R. E. (1965). Studies of mouse liver necrosis *in vitro*. Ultrastructural and cytochemical alternations in hepatic parenchymal cell nuclei. *Lab Invest* 14:1969-99.

Truneh, A., Sharma, S., Silverman, C., Khandekar, S., Reddy, M. P., Deen, K. C., McLaughlin, M. M., Srinivasula, S. M., Livi, G. P., Marshall, L. A., Alnemri, E. S., Williams, W. V. and Doyle, M. L. (2000). Temperature sensitive differential affinity of TRAIL for its receptors: DR5 is the highest affinity receptor. *J Biol Chem* 275:23319-25.

Tschopp, J., Irmler, M. and Thome, M. (1998). Inhibition of fas death signals by FLIPs. *Curr Opin Immunol* 10:552-8.

Tsuchimoto, M., Azuma, Y., Higuchi, O., Sugimoto, I., Hirata, N., Kiyoki, M. and Yamamoto, I. (1994). Alendronate modulates osteogenesis of human osteoblastic cells *in vitro*. *Jpn J Pharmacol* 66:25-33.

Tsujimoto, Y., Louie, E., Bashir, M. M. and Croce, C. M. (1988). The reciprocal partners of both the t(14; 18) and the t(11; 14) translocations involved in B-cell neoplasms are rearranged by the same mechanism. *Oncogene* 2:347-51.

Tsujimoto, Y. (2003). Cell death regulation by the Bcl-2 protein family in the mitochondria. *J Cell Physiol* 195:158-67.

Unni, K. K. (1996). <u>Dahlin's Bone Tumors: general aspects and data on 11,087 cases.</u> Philadelphia, Lippincott-Raven.

Van Beek, E. R., Lowik, C. W. and Papapoulos, S. E. (2002). Bisphosphonates suppress bone resorption by a direct effect on early osteoclast precursors without affecting the osteoclastogenic capacity of osteogenic cells: the role of protein geranylgeranylation in the action of nitrogen-containing bisphosphonates on osteoclast precursors. *Bone* 30:64-70.

Van Cutsem, E., Karasek, P., Oettle, H., Vervenne, W. L., Szawlowski, A., Schoffski, P., Post, S., Neumann, H., Safran, H., Humblet, Y., van de Velde, H., Ma, Y. and Von Hoff, D. (2002). Phase III trial comparing gemcitabine + R115777 (Zarnestra) *versus* gemcitabine + placebo in advanced pancreatic cancer (PC). *Proc Am Soc Clin Oncol* 21:517.

van de Donk, N. W., Schotte, D., Kamphuis, M. M., van Marion, A. M., van Kessel, B., Bloem, A. C. and Lokhorst, H. M. (2003). Protein geranylgeranylation is critical for the regulation of survival and proliferation of lymphoma tumor cells. *Clin Cancer Res* 9:5735-48.

van der Pluijm, G., Vloedgraven, H., van Beek, E., van der Wee-Pals, L., Lowik, C. and Papapoulos, S. (1996). Bisphosphonates inhibit the adhesion of breast cancer cells to bone matrices *in vitro*. *J Clin Invest* 98:698-705.

Van Loo, G., Schotte, P., van Gurp, M., Demol, H., Hoorelbeke, B., Gevaert, K., Rodriguez, I., Ruiz-Carrillo, A., Vandekerckhove, J., Declercq, W., Beyaert, R. and Vandenabeele, P. (2001). Endonuclease G: a mitochondrial protein released in apoptosis and involved in caspase-independent DNA degradation. *Cell Death and Differ* 8:1136-42. van Loo, G., Demol, H., van Gurp, M., Hoorelbeke, B., Schotte, P., Beyaert, R., Zhivotovsky, B., Gevaert, K., Declercq, W., Vandekerckhove, J. and Vandenabeele, P. (2002). A matrix-assisted laser desorption ionization post-source decay (MALDI-PSD) analysis of proteins released from the isolated liver mitochondria treated with recombinant truncated Bid. *Cell Death and Differ* 9:301-08.

van Loo, G., Saelens, X., Matthijssens, F., Schotte, P., Beyaert, R., Declercq, W. and Vandenabeele, P. (2002). Caspases are not localized in mitochondria during life or death. *Cell Death Differ* 9:1207-11.

Van Valen, F., Fulda, S., Truckenbrod, B., Eckervogt, V., Sonnemann, J., Hillmann, A., Rodl, R., Hoffmann, C., Winkelmann, W., Schafer, L., Dockhorn-Dworniczak, B., Wessel, T., Boos, J., Debatin, K. M. and Jurgens, H. (2000). Apoptotic responsiveness of the Ewing's sarcoma family of tumours to tumour necrosis factor-related apoptosis-inducing ligand (TRAIL). *Int J Cancer* 88:252-9.

Van Valen, F., Fulda, S., Schafer, K. L., Truckenbrod, B., Hotfilder, M., Poremba, C., Debatin, K. M. and Winkelmann, W. (2003). Selective and nonselective toxicity of TRAIL/Apo2L combined with chemotherapy in human bone tumour cells vs. normal human cells. *Int J Cancer* 107:929-40.

Varfolomeev, E. E., Schuchmann, M., Luria, V., Chiannilkulchai, N., Beckmann, J. S., Mett, I. L., Rebrikov, D., Brodianski, V. M., Kemper, O. C., Kollet, O., Lapidot, T., Soffer, D., Sobe, T., Avraham, K. B., Goncharov, T., Holtmann, H., Lonai, P. and Wallach, D. (1998). Targeted disruption of the mouse Caspase 8 gene ablates cell death induction by the TNF receptors, Fas/Apo1, and DR3 and is lethal prenatally. *Immunity* 9:267-76.

Vaux, D. L., Cory, S. and Adams, J. M. (1988). Bcl-2 gene promotes haemopoietic cell survival and cooperates with c-myc to immortalize pre-B cells. *Nature* 355:440-42.

Vaux, D. L. and Silke, J. (2003). Mammalian mitochondrial IAP binding proteins. *Biochem Biophys Res Commun.* 304:499-504. Verhagen, A. M., Ekert, P. G., Pakusch, M., Silke, J., Connolly, L. M., Reid, G. E., Moritz, R. L., Simpson, R. J. and Vaux, D. L. (2000). Identification of DIABLO, a mammalian protein that promotes apoptosis by binding to and antagonizing IAP proteins. *Cell*:43-53.

Verhagen, A. M., Silke, J., Ekert, P. G., Pakusch, M., Kaufmann, H., Connolly, L. M., Day, C. L., Tikoo, A., Burke, R., Wrobel, C., Moritz, R. L., Simpson, R. J. and Vaux, D. L. (2002). HtrA2 promotes cell death through its serine protease activity and its ability to antagonize inhibitor of apoptosis proteins. *J Biol Chem* 277:445-54.

Vidalain, P. O., Azocar, O., Lamourille, B., Astier, A., Rabourdin-Combe, C. and Servet-Delprat, C. (2000). Measles virus induces functional TRAIL production by human dendritic cells. *J Virol* 74:556-59.

Viereck, V., Emons, G., Lauck, V., Frosch, K. H., Blaschke, S., Grundker, C. and Hofbauer, L. C. (2002). Bisphosphonates pamidronate and zoledronic acid stimulate osteoprotegerin production by primary human osteoblasts. *Biochem Biophys Res Commun* 291:680-6.

Vigushin, D. M., Mirsaidi, N., Brooke, G., Sun, C., Pace, P., Inman, L., Moody, C. J. and Coombes, R. C. (2004). Gliotoxin is a dual inhibitor of farnesyltransferase and geranylgeranyltransferase I with antitumor activity against breast cancer *in vivo*. *Med Oncol* 21:21-30.

Vitte, C., Fleisch, H. and Guenther, H. L. (1996). Bisphosphonates induce osteoblasts to secrete an inhibitor of osteoclast-mediated resorption. *Endocrinology* 137:2324-33.

Vogt, A., Sun, J., Qian, Y., Hamilton, A. D. and Sebti, S. M. (1997). The geranylgeranyltransferase-I inhibitor GGTI-298 arrests human tumor cells in G0/G1 and induces p21(WAF1/CIP1/SDI1) in a p53-independent manner. *J Biol Chem* 272:27224-9.

Vogt, U., Bielawski, K. P., Bosse, U. and Schlotter, C. M. (2004). Breast tumour growth inhibition in vitro through the combination of cyclophosphamide/metotrexate/5-

fluorouracil, epirubicin/cyclophosphamide, epirubicin/paclitaxel, and epirubicin/docetaxel with the bisphosphonates ibandronate and zoledronic acid. *Oncol Rep* 12:1109-14.

Walczak, H., Degli-Esposti, M., Johnson, R. S., Smolak, P. J., Waugh, J. Y., Boiani, N., Timour, M. S., Gerhart, M. J., Schooley, K. A., Smith, C. A., Goodwin, R. G. and Rauch, C. T. (1997). TRAIL-R2: a novel apoptosis-mediating receptor for TRAIL. *EMBO J* 16:5386-97.

Walczak, H., Miller, R. E., Ariail, K., Gliniak, B., Griffith, T. S., Kubin, M., Chin, W., Jones, J., Woodward, A., Le, T., Smith, C., Smolak, P., Goodwin, R. G., Rauch, C. T., Schuh, J. C. and Lynch, D. H. (1999). Tumoricidal activity of tumor necrosis factor-related apoptosis-inducing ligand *in vivo*. *Nat Med* 5:157-63.

Walczak, H. and Krammer, P. H. (2000). The CD95 (APO-1/Fas) and the TRAIL (APO-2L) apoptosis systems. *Exp Cell Res* 256:58-66.

Walker, N. P., Talanian, R. V., Brady, K. D., Dang, L. C., Bump, N. J., Ferenz, C. R., Franklin, S., Ghayur, T., Hackett, M. C. and Hammill, L. D. e. a. (1998). Crystal strusture of the cysteine protease interleukin-1 beta-converting enzyme: a (p20/p10)2 homodimer. *Cell* 78:243-52.

Walker, P. R., LeBlanc, J. and Sikorska, M. (1997). Evidence that DNA fragmentation in apoptosis is initiated and propagated by single strand breaks. *Cell Death and Differ* 4:506-15.

Wang, C. Y., Mayo, M. W., Korneluk, R. G., Goeddel, D. V. and Baldwin, A. S. J. (1998). NF-kappaB antiapoptosis: induction of TRAF1 and TRAF2 and c-IAP1 and c-IAP2 to suppress caspase-8 activation. *Science* 281:1680-3.

Wang, J., Zheng, L., Lobito, A., Chan, F. K., Dale, J., Sneller, M., Yao, X., Puck, J. M., Straus, S. E. and Lenardo, M. J. (1999). Inherited human Caspase 10 mutations underlie defective lymphocyte and dendritic cell apoptosis in autoimmune lymphoproliferative syndrome type II. *Cell* 98:47-58.

Wang, J., Lobito, A. A., Shen, F., Hornung, F., Winoto, A. and Lenardo, M. J. (2000). Inhibition of Fas-mediated apoptosis by the B cell antigen receptor through c-FLIP. *Eur J Immunol* 30:155-63.

Wang, J., Chun, H. J., Wong, W., Spencer, D. M. and Lenardo, M. J. (2001). Caspase-10 is an initiator caspase in death receptor signaling. *Proc Natl Acad Sci U S A* 98:13884-8.

Wang, P. S., Solomon, D. H., Mogun, H. and Avorn, J. (2000). HMG-CoA reductase inhibitors and the risk of hip fractures in elderly patients. *JAMA* 283:3211-6.

Wang, S. and El-Deiry, W. S. (2003). TRAIL and apoptosis induction by TNF-family death receptors. *Oncogene* 22:8628-33.

Wang, W. Q., Miura, M., Bergeron, L., Zhu, H. and Yuan, J. (1995). Ich-1, an ICE/ced-3-related gene, encodes both positive and negative regulators of programmed cell death. *Cell* 78:739-50.

Wang, X. (2001). The expanding role of mitochondria in apoptosis. *Genes Dev* 15:2922-33.

Waterhouse, N. J., Finucane, D. M., Green, D. R., Elce, J. S., Kumar, S., Alnemri, E. S., Litwack, G., Khanna, K., Lavin, M. F. and Watters, D. J. (1998). Calpain activation is upstream of caspases in radiation-induced apoptosis. *Cell Death Differ* 5:1051-61.

Whyte, M. P. <u>Skeletal Neoplasms.</u> In: Favus MJ, ed. Primer on the metabolic bone diseases and disorders of skeletal metabolism. Philadelphia: Lippincott, Williams and Wilkins, 1999:397-405.

Widlak, P., Li, L. Y., Wang, X. and Garrard, W. T. (2001). Action of recombinant human apoptotic endonuclease G on naked DNA and chromatin substrates: cooperation with exonuclease and DNase. *J Biol Chem* 276:48404-09.

Wiley, S., Schooley, K., Smolak, P. J., Din, W. S., Huang, C. P., Nicholl, J. K., Sutherland, C. R. and Goodwin, R. G. (1995). Identification and characterization of a new member of the TNF family that induces apoptosis. *Immunity* 3:673-82.

Winquist, E., Moore, M. J., Chi, K. and al., e. (2001). A phase II study of a farnesyl transferase inhibitor, (SCH66336) in patients with unresectable or metastatic transitional cell carcinoma of the urothelial tract failing prior chemotherapy. *Proc Am Soc Clin Oncol* 20:197a.

Wong, W. W., Tan, M. M., Xia, Z., Dimitroulakos, J., Minden, M. D. and Penn, L. Z. (2001). Cerivastatin triggers tumor-specific apoptosis with higher efficacy than lovastatin. *Clin Cancer Res* 7:2067-75.

Wong, W. W., Dimitroulakos, J., Minden, M. D. and Penn, L. Z. (2002). HMG-CoA reductase inhibitors and the malignant cell: the statin family of drugs as triggers of tumor-specific apoptosis. *Leukemia* 16:508-19.

Wood, J., Bonjean, K., Ruetz, S., Bellahcene, A., Devy, L., Foidart, J. M., Castronovo, V. and Green, J. R. (2002). Novel antiangiogenic effects of the bisphosphonate compound zoledronic acid. *J Pharmacol Exp Ther* 302:1055-61.

Wu, G. S., Burns, T. F., McDonald, E. R. r., Jiang, W., Meng, R., Krantz, I. D., Kao, G., Gan, D. D., Zhou, J. Y., Muschel, R., Hamilton, S. R., Spinner, N. B., Markowitz, S., Wu, G. and el-Deiry, W. S. (1997). KILLER/DR5 is a DNA damage-inducible p53-regulated death receptor gene. *Nat Genet* 17:141-3.

Wu, X. X., Kakehi, Y., Mizutani, Y., Kamoto, T., Kinoshita, H., Isogawa, Y., Terachi, T. and Ogawa, O. (2002). Doxorubicin enhances TRAIL-induced apoptosis in prostate cancer. *Int J Oncol* 20:949-54.

Wu, X. X., Ogawa, O. and Kakehi, Y. (2004). TRAIL and chemotherapeutic drugs in cancer therapy. *Vitam Horm* 67:365-83.

Wyllie, A. H. (1980). Glucocorticoid-induced thymocyte apoptosis is associated with endogenous endonuclease activation. *Nature* 284:555-56.

Wyllie, A. H., Kerr, J. F. K. and Currie, A. H. (1981). Cell death: the significance of apoptosis. *Cytol* 68:251-305.

Wyllie, A. H. (1987). Apoptosis: cell death in tissue regulation. J Pathol 153:313-316.

Xia, Z., Tan, M. M., Wong, W. W., Dimitroulakos, J., Minden, M. D. and Penn, L. Z. (2001). Blocking protein geranylgeranylation is essential for lovastatin-induced apoptosis of human acute myeloid leukemia cells. *Leukemia* 15:1398-407.

Yaccoby, S., Pearse, R. N., Johnson, C. L., Barlogie, B., Choi, Y. and Epstein, J. (2002). Myeloma interacts with the bone marrow microenvironment to induce osteoclastogenesis and is dependent on osteoclast activity. *Br J Haematol* 116:278-90.

Yamanaka, T., Shiraki, K. and Sugimoto, K. (2000). Chemotherapeutic agents augment TRAIL-induced apoptosis in human hepatocellular carcinoma cell lines. *Hepatology* 32:482-90.

Yamashita, Y., Shimada, M., Tanaka, S., Okamamoto, M., Miyazaki, J. and Sugimachi, K. (2002). Electroporation-mediated tumor necrosis factor-related apoptosis-inducing ligand (TRAIL)/Apo2L gene therapy for hepatocellular carcinoma. *Hum Gene Ther* 13:275-86.

Yang, Q. H., Church-Hajduk, R., Ren, J., Newton, M. L. and Du, C. (2003). Omi/HtrA2 catalytic cleavage of inhibitor of apoptosis (IAP) irreversibly inactivates IAPs and facilitates caspase activity in apoptosis. *Genes Dev* 17:1487-96.

Yano, S., Zhang, H., Hanibuchi, M., Miki, T., Goto, H., Uehara, H. and Sone, S. (2003). Combined therapy with a new bisphosphonate, minodronate (YM529), and chemotherapy for multiple organ metastases of small cell lung cancer cells in severe combined immunodeficient mice. *Clin Cancer Res* 9:5380-5.

Ye, H., Cande, C., Stephanou, N. C., Jiang, S., Gurbuxani, S., Larochette, N., Daugas, E., Garrido, C., Kroemer, G. and Wu, H. (2002). DNA binding is required for the apoptogenic action of apoptosis inducing factor. *Nat Struct Biol* 9:680-4.

Yoneda, T., Michigami, T., Yi, B., Williams, P. J., Niewolna, M. and Hiraga, T. (1999). Use of bisphosphonates for the treatment of bone metastasis in experimental animal models. *Cancer Treat Rev* 25:293-9.

Yoneda, T., Michigami, T., Yi, B., Williams, P. J., Niewolna, M. and Hiraga, T. (2000). Actions of bisphosphonate on bone metastasis in animal models of breast carcinoma. *Cancer Cell* 88:2979-88.

Yoshikawa, H., Takaoka, K., Hamada, H. and Ono, K. (1985). Clinical significance of bone morphogenetic activity in osteosarcoma: a study of 20 cases. *Cancer* 56:1682-87.

Yoshikawa, H., Nakase, T., Myoui, A. and Uede, T. (2004). Bone morphogenetic proteins in bone tumours. *J of Ortho Sci* 9:334-40.

Yu, C., Wang, S., Dent, P. and Grant, S. (2001). Sequence-dependent potentiation of paclitaxel-mediated apoptosis in human leukemia cells by inhibitors of the mitogen-activated protein kinase kinase/mitogen-activated protein kinase pathway. *Mol Pharmacol* 60:143-54.

Yu, R., Mandlekar, S., Ruben, S., Ni, J. and Kong, A. N. (2000). Tumor necrosis factorrelated apoptosis-inducing ligand-mediated apoptosis in androgen-independent prostate cancer cells. *Cancer Res* 60:2384-9.

Yuan, J., Shaham, S., Ledoux, S., Ellis, H. M. and Horvitz, H. R. (1993). The *C. elegans* cell death gene *ced-3* encodes a protein similar to mammalian interleukin-1-beta-converting enzyme. *Cell* 75:641-52.

Zamai, L., Ahmad, M., Bennett, I. M., Azzoni, L., Alnemri, E. S. and Perussia, B. (1998). Natural killer (NK) cell-mediated cytotoxicity: differential use of TRAIL and Fas ligand by immature and mature primary human NK cell. *J Exp Med* 188:2375-80.

Zamzami, N. and Kroemer, G. (2003). Apoptosis: mitochondrial membrane permeabilization- the (w)hole story? *Curr Biol* 13:R71-R73.

Zhang, F. L. and Casey, P. J. (1996). Protein prenylation: molecular mechanisms and functional consequences. *Annu Rev Biochem* 65:241-69.

Zhang, X. D., Franco, A., Myers, K., Gray, C., Nguyen, T. and Hersey, P. (1999). Relation of TNF-related apoptosis-inducing ligand (TRAIL) receptor and FLICE-inhibitory protein expression to TRAIL-induced apoptosis of melanoma. *Cancer Res* 59:2747-53.

Zhao, H., Laitala-Leinonen, T., Parikka, V. and Vaananen, H. K. (2001). Downregulation of small GTPase Rab7 impairs osteoclast polarization and bone resorption. *J Biol Chem* 276:39295-302.

Zheng, T. S. and Flavell, R. A. (2000). Divinations and surprises: genetic analysis of caspase function in mice. *Exp Cell Res* 256:67-73.

Zhivotovsky, B., Samali, A., Gahm, A. and Orrenius, S. (1999). Caspases: their intracellular localization and translocation during apoptosis. *Cell Death Differ* 6:644-51.

Zhu, H., Zhang, L., Huang, X., Davis, J. J., Jacob, D. A., Teraishi, F., Chiao, P. and Fang, B. (2004). Overcoming acquired resistance to TRAIL by chemotherapeutic agents and calpain inhibitor I through distinct mechanisms. *Mol Ther* 9:666-73.

Zohn, I. M., Campbell, S. L., Khosravi-Far, R., Rossman, K. L. and Der, C. J. (1998). Rho family proteins and Ras transformation: the RHOad less traveled gets congested. *Oncogene* 17:1415-38.

Zou, H., Henzel, W. J., Liu, X., Lutschg, A. and Wang, X. (1997). Apaf-1, a human protein homologous to *C. elegans* CED-4, participates in cytochrome *c*-dependent activation of caspase-3. *Cell* 90:405-13.

Amendments

Introduction

•Pg 33: For more details of last paragraph refer to Hofbauer LC *et al.*, The OPG/RANKL/RANK system in metabolic bone diseases. *J Musculoskelet Neuronal Interact* 2004 4(3):268-75.

09PH

- Pg 47: For more details of paragraph 1 refer to Mundy GR. Mechanisms of bone metastasis. *Cancer* 1997 80(8S):1546-56, and Chirgwin JM. & Guise TA. Molecular mechanisms of tumor-bone interactions in osteolytic metastases. *Crit Rev Eukaryot Gene Expr* 2000 10(2):159-78.
- •Pg 51: First paragraph should read 'It has been proposed that these concentrations of BPs may be achievable in bone at an active resorption site although there is no conclusive evidence to prove this hypothesis [Sato *et al.*, 1991].'

Chapter 2

Section 2.4.12: The 2^{nd} sentence should read ' This program was used to determine standard deviation (SD) and *p* values, based on experiments that were performed in triplicate and repeated at least three times'.

Chapter 3

- ⁹Pg 95: The 2nd to last sentence of paragraph 1 should read 'the efficacy of ZOL to inhibit increases in cell number varied considerably between cell lines. The decrease in cell number could be due to inhibition of cell proliferation or to the induction of apoptosis'.
- Fig 3.4: See revised figure (next page).
- Fig 3.14: BTK-143 should be replaced with MG-63.
- Section 3.3.7: The 3rd should read 'However, by 72 hours, ZOL caused an increase in the number of cells arrested in S-phase (38% v 26%), concomitant with a reduction of cells in G₀/G₁ and G₂/M phases.' The 5th sentence should include the following amendment; '....S-phase increasing from 33%.....'
- Pg 99: Our results show a correlation between ZOL dose on cell viability and inhibition of protein prenylation in the more sensitive HOS cell line. A ZOL dose of 10μM, reduced cell viability by approx 60% (see dose curves, Figure 3.1) and also inhibited Rap 1A prenylation. By contrast, the same concentration of ZOL, which was less potent at reducing cell viability of MG-63 cells, still showed inhibition of prenylated Rap 1A. The presence of unprenylated protein may therefore not be a clear representation of cell viability.

Chapter 4

- Pg 112: Although there is some variability in the effect of ZOL on BTK-143 cells when compared to previous data, there is clear augmentation of apoptosis when TRAIL and ZOL are given in combination (as seen from multiple experiments).
- Fig 4.7: The columns representing Z+T for both cell lines had p values < 0.005 compared to both Z and T single treatments.
- Fig 4.8: The columns GGO+Z for both cell lines had p values < 0.001 compared to Z treatment alone.
- Fig 4.11: All references to BTK-143 should be replaced with the cell line MG-63.
- Section 4.3.6: All references to BTK-143 should be replaced with the cell line MG-63. The last line should read, ' These trends were also identified in the G-292, Saos2 and SJS-1 cell lines (data not shown).'

Chapter 5

- Fig 5.9: All t tests performed on data sets represented in this figure generated values of p > 0.05.
- Fig 5.11: t tests performed on (A) and (B) revealed p < 0.001 for L+ GGO+/-T and M+GGO+T and p < 0.003 for M+GGO. (C) and (D) had values of p > 0.05 for PTI+GGO+/-T.
- Table 5.1: Column should read 'increase in cell death as a % of control untreated cells'.

Chapter 7

Section 7.3.2: Add to the end of paragraph 1: 'It has been reported that an increase in bone mineral density adjacent to growth plates after bisphosphonate administration to growing animals is due to the inhibition of bone resorption and the retention of the primary spongiosa formed during growth (Schenk R. *et al.*, *Calc Tissue Res* 1973 11:196-214).'



Figure 3.4 The effect of ZOL on OS cell number. Cells were incubated for the times indicated in the absence (\rightarrow) or presence (\rightarrow) of 25 μ M ZOL. The number of viable cells was quantitated, based on trypan blue exclusion, using a heamocytometer and expressed as number of cells. Data shown in each case are from representative experiments repeated at least three times: bars, ±SD.